HADRON COLLIDER TESTS OF NEUTRINO MASS-GENERATING MECHANISMS

by

Richard Efrain Ruiz

B.S. in Mathematics, University of Chicago, 2010
B.A. (Honors) in Physics, University of Chicago, 2010
M.S. in Physics (Particle Theory and Collider Phenomenology), University of Wisconsin - Madison, 2012

Submitted to the Graduate Faculty of the Kenneth P. Dietrich School of Arts and Sciences in partial fulfillment of the requirements for the degree of **Doctor of Philosophy**

University of Pittsburgh

2015

UNIVERSITY OF PITTSBURGH

KENNETH P. DIETRICH SCHOOL OF ARTS AND SCIENCES

This dissertation was presented

by

Richard Efrain Ruiz

It was defended on

April 24, 2015

and approved by

Tao Han, Ph.D., Distinguished Professor, University of Pittsburgh

Ayres Freitas, Ph.D., Associate Professor, University of Pittsburgh

Richard Holman, Ph.D., Professor, Carnegie Mellon University Arthur Kosowsky, Ph.D., Professor, University of Pittsburgh Vladimir Savinov, Ph.D., Professor, University of Pittsburgh

Dissertation Director: Tao Han, Ph.D., Distinguished Professor, University of Pittsburgh

©

HADRON COLLIDER TESTS OF NEUTRINO MASS-GENERATING MECHANISMS

Richard Efrain Ruiz, PhD

University of Pittsburgh, 2015

The Standard Model of particle physics (SM) is presently the best description of nature at small distances and high energies. However, with tiny but nonzero neutrino masses, a Higgs boson mass unstable under radiative corrections, and little guidance on understanding the hierarchy of fermion masses, the SM remains an unsatisfactory description of nature. Well-motivated scenarios that resolve these issues exist but also predict extended gauge (e.g., Left-Right Symmetric Models), scalar (e.g., Supersymmetry), and/or fermion sectors (e.g., Seesaw Models). Hence, discovering such new states would have far-reaching implications.

After reviewing basic tenets of the SM and collider physics, several beyond the SM (BSM) scenarios that alleviate these shortcomings are investigated. Emphasis is placed on the production of a heavy Majorana neutrinos at hadron colliders in the context of low-energy, effective theories that simultaneously explain the origin of neutrino masses and their smallness compared to other elementary fermions, the so-called Seesaw Mechanisms. As probes of new physics, rare top quark decays to Higgs bosons in the context of the SM, the Types I and II Two Higgs Doublet Model (2HDM), and the semi-model independent framework of Effective Field Theory (EFT) have also been investigated. Observation prospects and discovery potentials of these models at current and future collider experiments are quantified.

Keywords: Hadron Colliders, Seesaw Mechanisms, Neutrino Physics, Collider Phenomenology.

TABLE OF CONTENTS

PR	EFA	CE	xiv
1.0	TH	E STANDARD MODEL OF PARTICLE PHYSICS	1
	1.1	Introduction	1
	1.2	Symmetries	1
	1.3	Continuous Symmetries I: Global Symmetries	2
		1.3.1 Spontaneously Broken Global Symmetries	3
	1.4	Continuous Symmetries II: Local Symmetries	6
		1.4.1 Quantum Electrodynamics	11
		1.4.2 The Higgs Mechanism: Spontaneously Broken Local Symmetries	13
	1.5	Introduction to the Standard Model of Particle Physics	17
	1.6	Gauge Sector of the Standard Model	18
	1.7	The Standard Model Higgs Sector and Electroweak Symmetry Breaking	21
		1.7.1 Electroweak Symmetry Breaking I: Massive Gauge Bosons	25
		1.7.2 Custodial Symmetry	29
	1.8	Fermion Sector of the Standard Model	32
		1.8.1 Fermion Content	32
		1.8.2 Fermion Lagrangian	34
		1.8.3 Electroweak Symmetry Breaking II: Fermion Masses	35
		1.8.4 Quark and Lepton Mass Mixing	36
	1.9	Beyond the Standard Model	39
2.0	PR	INCIPLES OF COLLIDER PHYSICS	40
	2.1	Introduction	40
	2.2	Helicity Amplitudes	40
		2.2.1 Spin One-Half Fermions	41

		2.2.1.1 Properties of Dirac Spinors			 42
		2.2.2 Spin-One Vector Bosons			 42
		2.2.3 Decay of Heavy Fermionic Top Quark Partner			 43
	2.3	Phase Space			 45
		2.3.1 One-Body Phase Space			 46
		2.3.2 Solid Angle in d Dimension $\ldots \ldots \ldots \ldots \ldots$			 47
		2.3.3 Two-Body Phase Space in d and 4 Dimensions			 48
		2.3.4 Three-Body Phase Space in d and 4 Dimensions			 49
		2.3.5 Phase Space Decomposition			 51
		2.3.6 One-Particle Phase Space Splitting			 53
	2.4	Partial Width			 53
		2.4.1 The Optical Theorem and Breit-Wigner Propagators			 54
		2.4.2 Narrow Width Approximation			 56
		2.4.3 Example: NWA Applied to Leptonic Decays of Top Qu	uarks		 57
	2.5	Partonic Level Cross Section			 61
		2.5.1 Example: Zh Production at Electron Colliders			 62
	2.6	Initial-State Photons from Electron- X Scattering			 64
		2.6.1 $e \rightarrow \gamma$ Splitting			 67
		2.6.2 Weizsäcker-Williams Approximation			 69
		2.6.3 Weak Boson Distribution Functions			 70
		2.6.4 Beyond Leading Logarithm			 71
		2.6.5 Elastic Photon PDF			 72
	2.7	Factorization Theorem, Parton Luminosities, and Hadronic G	Cross Sectio	n	 73
		2.7.1 Inelastic Photon PDF			 74
		2.7.2 Parton Luminosities			 76
		2.7.3 Parton-Vector Boson Luminosities			 77
		2.7.4 Vector Boson Scattering: Double Initial-State Parton S	Splitting		 77
	2.8	Statistics			 77
		2.8.1 Poisson Statistics			 77
		2.8.2 Gaussian Statistics			 78
3.0	HI	GGS BOSONS FROM THE TOP DECAYS			 79
	3.1	Introduction			 79

	3.2	Theor	retical F	Fra	ar	m	ev	vo	rk	2	•		•		•		•	•			•	•	•		•	•	•	•		•	•		•	• •		•	•	•			•	•	80	0
		3.2.1	The SM	M	a	ıs	a	n i	Ef	ffe	ec	ti	ive	e	F	iel	ld	Т	he	or	y			•										•		•					•		8	1
			3.2.1.1	E	E]	F	Т	\mathbf{fr}	ar	ne	ew	NC	or	k	a	nc	d I	pa	rai	m	ete	er	s					•		•	•			•		•				•	•	•	8	1
			3.2.1.2	E	E]	F	Т	С	or	ıst	tr	ca	in	nts	s		•	•										•		•				•		•				•	•	•	8	3
		3.2.2	Linear l	D)e	эр	en	ıd	en	ice	е	0	of]	E	F′	Т	0	pe	era	tc	ors	3.		•										•		•					•		84	4
			3.2.2.1	(C	\mathcal{O}_t	1	an	ıd	$\overline{\mathcal{C}}$	$\bar{\mathfrak{I}}_t$!1			•		•	•				•									•			•		•					•		8	5
			3.2.2.2	Ī	C	$\bar{\mathcal{I}}_t$	2						•		•		•	•				•									•			•		•					•	•	8	5
			3.2.2.3	Ī	Ċ	$\bar{\mathcal{I}}^{(}_{\varphi}$	p(1)						•		•		•	•				•									•			• •		•					•	•	80	6
		3.2.3	Type I	Т	Γv	wo	5]	Hi	gg	ζs	Γ	C	ou	ıb	le	et	Μ	00	lel			•												•		•					•		80	6
			3.2.3.1	Γ	Гj	УI	pe	Ι	2	H	D)N	Л	\mathbf{fr}	a	m	ew	voi	rk	aı	nd	ł)a:	ra	m	et	\mathbf{er}	5		•	•			•		• •				•	•		8'	7
			3.2.3.2	Γ	Гj	УĮ	pe	Ι	2	H	D)N	Л	С	o	\mathbf{ns}	str	aiı	nts	5.											•			•		•					•	•	88	8
		3.2.4	Type II	I	Т	w	70	Н	lig	gs	s	D)0	oul	bl	let	; N	Ло	de	l		•						•		•	•			•		• •				•	•	•	89	9
			3.2.4.1	Γ	Гj	УI	pe	Ľ	I 2	2H	łI	D.	Μ	I f	ra	an	ne	wc	ork	: 8	n	d	pa	ar	ar	ne	te	rs		•	•		•	•		• •				•	•		89	9
			3.2.4.2	Γ	Гj	УI	рe	Ľ	I 2	2H	łI	D.	M	[(Со	on	st	\mathbf{ra}	int	\mathbf{S}		•		•							•			•		• •				•	•		90	0
	3.3	Brand	ching Ra	ati	ic	\mathbf{OS}				•			•		•		•	•										•		•						•					•	•	9	1
		3.3.1	EFT B	R	t(1	t	\rightarrow	• V	Vi	bh	ı)		•	•	•		•	•		•		•								•	•		•	•		• •				•	•		92	2
		3.3.2	Type I	2	ΡH	ŦI	ΟN	Л	B	R	(t	t -	\rightarrow	, V	V	<i>b1</i>	H))		•	•	•			•	•					•		•	•		• •				•	•		90	б
			3.3.2.1	E	B]	R	.(t		>]	W	'b)h	ι, Ι	Η)	vs	s t	ar	$\mathbf{n} \boldsymbol{\beta}$	<i>, </i>	Δ	V			•	•					•		•	•		· •				•			90	б
			3.3.2.2	E	B]	R	(t	_	>]	W	'b)h	ı/.	H)	vs	s 1	n_h	n/E	I						•					•			•		. .				•			98	8
			3.3.2.3	E	B]	R	(t	_	+ ا	W	'b)A	4)	v	\mathbf{s}	tε	an	β,	<i>n</i>	1 _A	•					•				•	•			•		· •				•	•	•	99	9
		3.3.3	Type II	I 2	2]	Н	D	М	E	3R	₹((t	. —	\rightarrow	И	Vb	ЪH	()								•				•	•			•		•				•	•	•	10	1
			3.3.3.1	E	B]	R	(t	_	+ ا	W	'b	h	ι, Ι	H)	vs	s t	ar	$\mathbf{n} \boldsymbol{\beta}$, -	Δ	V				•				•	•			•		· •				•	•	•	10	1
			3.3.3.2	E	B]	R	(t	_	+ ا	W	'b	h	ι, Ι	H)	vs	5 r	n_{I}	·							•				•	•			•		· •				•	•	•	10	1
			3.3.3.3	E	B]	R	(t	_	+ ا	W	'b)A	4)	v	\mathbf{s}	tε	an	β,	<i>n</i>	ι_A	•					•				•	•			•		· •				•	•	•	10	2
	3.4	Obser	vation F	Pr	rc	DS	pe	ect	ts	at	t	С	Jo	olli	id	er	s	•										•		•				• •		•					•	•	10;	3
	3.5	Sumn	nary and	d	C	Co	one	clı	usi	io	n		•		•		•	•																• •		•					•	•	10;	3
4.0	INC	CLUS	IVE HI	E	A	41	V	Y	M	I A	1.	J	0	R	A	11	NA	4 1	N]	El	U'	Γ	R	IP	10	O	P	RO	DI	π	JC	CT	I	10	N	A'	Г	н	Α	D	R	0	N	
	CO	LLID	ERS .			•	•			•		•	•		•		•	•										•		•				•		•				•	•	•	10	5
	4.1	Intro	duction			•	•			•			•				•	•										•		•				• •		•					•	•	10	5
	4.2	Neutr	rino Mix	kin	ng	g	Fo	ori	ma	ali	is	m	a .				•											•								•					•		108	8
	4.3	Heavy	y $N \operatorname{Pro}$	od	lu	ict	tic	on	a	t	H	Ia	ıdı	.ro	on	L C	Co	llio	de	rs											•			•		•					•		108	8
		4.3.1	Constra	aiı	'n	its	3 C	m	Η	lea	av	vy	y]	Ne	eι	ıtı	rir	10	m	ix	in	g		•										•		•					•		11	1

		4.3.2 N Production via the Drell-Yan Process at NNLO $\ldots \ldots \ldots$
		4.3.3 Photon-Initiated Processes
		4.3.3.1 Elastic Scattering: Intact Final-State Nucleons
		4.3.3.2 Inelastic Scattering: Collinear Photons From Quarks
		4.3.3.3 Deeply Inelastic Scattering: High p_T Quark Jet
		4.3.3.4 Total Neutrino Production from γ -Initiated Processes
		4.3.4 Kinematic Features of N Production with Jets at 14 TeV $\ldots \ldots \ldots \ldots 122$
		4.3.5 Scale Dependence
	4.4	Heavy Neutrino Observability at Hadron Colliders
		4.4.1 Kinematic Features at 100 TeV
		4.4.2 Signal Definition and Event Selection: Same-Sign Leptons with Jets 128
		4.4.3 Background
		$4.4.3.1 \ t\bar{t} \dots \dots$
		4.4.3.2 Electron Charge Misidentification
		4.4.3.3 $W^{\pm}W^{\pm}$
		4.4.4 Discovery Potential at 100 TeV
		4.4.5 Updated Discovery Potential at 14 TeV LHC
	4.5	Summary
5.0	LE	PTON NUMBER VIOLATION AND W' CHIRAL COUPLINGS 147
	5.1	Introduction
	5.2	Theoretical Framework and Current Constraints
		5.2.1 Neutrino Mixing with W' Couplings $\ldots \ldots 149$
		5.2.1.1 Model-Independent W' Charged Current Couplings $\ldots \ldots \ldots 149$
		5.2.2 W' Chiral Coupling to Fermions
		5.2.3 Current Constraints on W'
		5.2.4 Current Constraints on N
	5.3	Derivation of Partonic Level Angular Distributions
		5.3.1 Summed Squared Matrix Element
		5.3.2 Phase Space Volume Element
		5.3.3 Partonic-Level Angular Distributions
		5.3.4 Partonic-Level Angular Distributions: L-Conserving Case
	5.4	W' and N Production and Decay at the LHC

	5.4.1 W' Production and Decay $\ldots \ldots \ldots$
	5.4.2 Heavy Neutrino Decay
5.5	Like-Sign Dilepton Signature
	5.5.1 Event Selection
	5.5.2 Characteristics of Kinematical Distributions
	5.5.3 Background Reduction and Statistical Significance
5.6	W' Chiral Couplings From Angular Correlations at the LHC $\ldots \ldots \ldots$
	5.6.1 W' Chiral Couplings To Leptons
	5.6.2 W' Chiral Couplings to Initial-State Quarks $\ldots \ldots \ldots$
5.7	Unlike-Sign Dilepton Angular Distributions
	5.7.1 W' Chiral Coupling from Angular Distributions
5.8	Summary
BIBLIC	DGRAPHY

LIST OF TABLES

1	Bosons of the Standard Model Before Electroweak Symmetry Breaking 20
2	Bosons of the Standard Model After Electroweak Symmetry Breaking
3	Matter Content of the Standard Model
4	Masses of the Elementary Standard Model Fermions
5	Components of CKM Matrix
6	Bounds on EFT couplings
7	Neutral Scalar Boson Couplings in the 2HDM(I) Relative to the SM Higgs Couplings 87
8	Neutral Scalar Boson Couplings in the $2HDM(II)$ Relative to the SM Higgs Couplings 90
9	$BR(t \rightarrow Wbh)$ for Benchmark Values of Anomalous $t\bar{t}h$ Couplings
10	$BR(t \rightarrow WbH)$ for Benchmark Values of Higgses in the 2HDM(I)
11	$BR(t \rightarrow WbH)$ for Benchmark Values of Higgses in the 2HDM(II)
12	Cross sections for $t\bar{t}$ and $t\bar{t} \to WW^*b\bar{b}h \to h(\ell^+\ell^{\prime-}\nu_\ell\overline{\nu}_{\ell'} + jj\ell^{\pm}\nu_\ell)$
13	LO and NNLO cross sections for $pp \to W^* \to \mu^{\pm} \nu$ at 14 and 100 TeV
14	Total cross sections of various $pp \to N\ell^{\pm}X$ channels for representative values of m_N . 120
15	Summary of scale dependence in $N\ell^{\pm}X$ production at 14 TeV and 100 TeV 125
16	Parton-level cuts on 100 TeV $\mu^{\pm}\mu^{\pm}jjX$ Analysis
17	Acceptance rates for $\mu^{\pm}\mu^{\pm}jjX$ at 100 TeV VLHC
18	Expected $\mu^\pm \mu^\pm j j X$ (bare) signal and SM background rates at 100 TeV VLHC $~$ 135
19	Same as Table 18 for $e^{\pm}\mu^{\pm}jjX$
20	Acceptance rates for SM $t\bar{t}$ at 100 TeV pp collider
21	Acceptance rates for SM $W^{\pm}W^{\pm}$ at 100 TeV pp collider
22	Sensitivity to the mixing parameter $S_{\ell\ell'}$ at the 14 TeV LHC and 100 TeV VLHC $$ 142
23	Sensitivity to heavy neutrino production in the $\mu\mu$ channel at 14 and 100 TeV 143
24	Parton-level cuts on 14 TeV $\mu^{\pm}\mu^{\pm}jjX$ Analysis

25	Same as Table 18 for 14 TeV LHC	143
26	Cross section for $pp \to W'_{L,R}^+ \to \mu^+ \mu^+ q \overline{q}' \dots \dots \dots \dots \dots \dots \dots \dots \dots$	169
27	Forward-backward asymmetry for $pp \to W'_{L,R}^+ \to \mu^+ \mu^+ q \overline{q}'$	182

LIST OF FIGURES

1	Interaction Feynman Rules for QED
2	Feynman vertex rules for $\phi^0 - \phi^0 - V - V$, $V = W^{\pm}$, Z, in the SM before EWSB. 25
3	Feynman rules for $h - h - V - V$, $h - V - V$, $V = W^{\pm}$, Z after EWSB 29
4	Diagrammatic representation of quasi-real, initial-state photon from $e\gamma$ splitting 65
5	Parton luminosity as a function of $\sqrt{\tau}$ at (a) 14 TeV and (b) 100 TeV
6	Feynman diagrams representing the leading transition $t \to H$
7	$BR(t \rightarrow Wbh)$ as a function of EFT coupling
8	The 2HDM(I) BR $(t \to WbH)$ as a function of $\tan \beta$ and $\Delta_V \ldots \ldots \ldots \ldots 95$
9	2HDM(I) BR($t \rightarrow WbH$) as a function of mass for a non-SM-like Higgs 97
10	2HDM(I) BR($t \to WbA$) as a function of $\tan \beta$ and $m_A \ldots \ldots \ldots \ldots \ldots \ldots $ 97
11	2HDM(II) BR $(t \to WbH)$ as a function of $\tan \beta$ and Δ_V
12	2HDM(I) BR($t \rightarrow WbH$) as a function of mass for a non-SM-like Higgs 100
13	2HDM(II) BR $(t \to WbA)$ as a function of $\tan \beta$ and m_A
14	Diagram representing resonant heavy Majorana neutrino production
15	Diagrammatic description of elastic & inelastic/deeply inelastic γp scattering 107
16	LHC and VLHC $N\ell^{\pm}$ cross section, divided by $ V_{\ell N} ^2$, at LO and NLO in QCD 110
17	Parton luminosities at 14 and 100 TeV
18	$N\ell X$ cross section, divided by $ V_{\ell N} ^2$
19	Feynman diagrams for photon-initiated process $q\gamma \to N\ell^{\pm}q'$
20	Feynman diagrams for the DIS process $q_1q_2 \rightarrow N\ell^{\pm}q'_1q'_2$
21	Total and differential cross section for $N\ell^{\pm}j$
22	Hadronic p_T and η differential distributions $\ldots \ldots \ldots$
23	Leptonic p_T and y/η differential distributions $\dots \dots \dots$
24	Cross section ratios relative to the default scale scheme

25	Kinematic distributions of $pp \to \ell^{\pm} \ell'^{\pm} jjX$ II
26	Kinematic distributions of $pp \to \ell^{\pm} \ell'^{\pm} jjX$ II
27	Missing transverse momentum in $pp \to \ell^{\pm} \ell^{\pm} j j X$
28	Reconstructed invariant mass of W and N
29	Total SM background versus m_N for (a) $\mu^{\pm}\mu^{\pm}$ and (b) $e^{\pm}\mu^{\pm}$ channels at 100 TeV 140
30	100 TeV discovery potential of N
31	At 14 TeV, (a) same as figure 30(a); (b) same as figure 30(c). $\dots \dots \dots$
32	Partonic-level process for heavy W'^+ production and decay into $\ell^{\pm}\ell^{\pm}q\overline{q'}$
33	The total decay width and branching fractions for W_R'
34	Heavy Majorana N total width and branching fractions $\hfill \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 165$
35	Partonic level process for a heavy W'^+ production and decay to like sign leptons \dots 167
36	Total cross section of $pp \to W'^+ \to \mu^+ \mu^+ W^-$ times $W^- \to q\bar{q}'$ branching ratio 168
37	Invariant mass distribution of $m_{\ell_1 j j}$ and $m_{\ell_2 j j}$
38	Transverse momentum distributions
39	Helicity and spin correlations in the chains $N_{L,R}\to \ell^+W^-\to \ell^+q\overline{q}'$ from W_L' decay $% M_L'$. 173
40	ΔR_{jj} distribution and $\Delta R_{\ell j}^{\min}$ distributions
41	Missing energy distribution for $pp \to W'_{L,R}^+ \to \mu^+ \mu^+ q \overline{q}'$ at the LHC
42	Integrated luminosity needed at 14 TeV LHC for discovery
43	Spin correlations for $q\bar{q}' \to W' \to N\ell^+$
44	Angular distribution of the charged lepton originating from neutrino decay 180
45	Spin correlations for neutrino production in the neutrino rest-frame
46	Spin correlations in the neutrino rest-frame
47	Φ distributions at the 14 TeV LHC for fully reconstructed events $\hdots \dots \dots \dots \dots 185$
48	Angular distributions of Oppose-Sign $\ell \ell j j$

PREFACE

In the process of writing this text, I came to the realization that the most important conclusions contained in this document will have already been published in refereed journals [1-3] by the time it has been defended. Experts and students have already the resources available to obtain copies of referenced works, the capabilities to work out (unpublished) intermediate results, and it becomes redundant to simply recompile everything into a new document. Therefore, very selfishly, I have decided to construct this dissertation for a single audience: myself; and for a very specific reason: to have a mostly complete and standalone set of notes from which, given the appropriate number of scratch paper and computational resources, Refs. [1-3] can be readily reproduced. While the contents of this book may not be sufficient to teach a graduate course on collider physics, I expect it to satisfy many of the requirements.

The text is laid out in the following fashion: In the first chapter, the Standard Model of Particle Physics (SM) is introduced. Fundamentals of collider physics is introduced after this. Three nontrivial extensions to the SM are then studied in great detail: (i) an additional scalar doublet gauged under the SM symmetries (Two Higgs Doublet Model); (ii) a right-handed neutrino with a Majorana mass (Seesaw Model Type I); and (iii) extended gauge sector with heavy Majorana neutrino (Seesaw Model Type I+II).

At this point, it is very much tradition to acknowledge and thank the many people who and institutions that have contributed to the completion of this thesis. This is a tradition I am very pleased to continue. First and foremost, in my completely biased and subjective opinion, I would like to thank Tao Han for his guidance, undervalued criticisms that I hope to one day appreciate, mentorship, support, and words of wisdom. To Efrain and Gilda, I am unsure how to appropriately acknowledge all that you have done. To write proportionally would fill libraries. Instead, I think acknowledging you two *inversely* to your importance is more appropriate, so to summarize: thanks.

There are many from around the world I wish to thank. To minimize omissions, I resort to the alphabet: Daniel Alva, Marguerite Braun, Helen Brown, Terry and Joey Carlos, Cindy Cercone,

Neil Christensen, Andrew Cox, Samuel Ducatman, Anna Elder and Kevin Sapp, Ayres Freitas, Athena Frost, Ayriole Frost, Caitlyn Hunter, Hiroyuki Ishida, LJ, Ali Kahler, Rachel Landau-Lazerus, Bryce Lanham, Ian Lewis, Zhen Liu, Amy Lowitz, Phil Luciano, Adriano Maron, Thomas McLaughlan, Ben Messerly, Danny and Debbie Molina, Jim Mueller, Shannon Patterson, Kevin Pedro, Zachary Pierpoint, Kara Ponder, Zhuoni Qian, my editors at Quantum Diaries, Angela Ruiz, Vladimir Savinov, Josh Sayre, Jared Schmitthenner, Ben Smart, Brittany Stalker, Brock Tweedie, Marina Tyquiengco, Xing Wang, Ying Wang, Cédric Weiland, Adam Weingarten, Kristen Welke, Daniel Wiegand, Krispi Williams, Ricky Williams, and to the many, many people on whose couches I have slept during my travels. Special appreciation and gratitude are owed to Zong-guo Si and Young-Kee Kim. Much of the research presented in this text was made possible through the generous assistance of the Graduate Dean's Office at the University of Pittsburgh, and the personal support of Kathleen Blee and Philippa Carter.

As of this writing, the relevant publications and working group reports can be found in Refs. [1–3] and Refs. [4,5]. Reproduction of Refs. [1–3] is authorized under license numbers 3610261016932 and 3610240521805 issued by Physical Review D, as well as 3604430001427 issued by the Journal of High Energy Physics. Apologies are owed for typographical errors, which, of course, are my responsibility. The realization (or lack thereof) of Seesaw Mechanisms in nature, on the other hand, is nature's responsibility.

Heavy Majorana neutrinos is a particularly favorite topic because it is an ideal, one-particle laboratory, similar to the top quark, where rich physics can be enjoyed and studied. Furthermore, the existence of tiny but nonzero neutrino masses indicate the existence of new particles and interactions not predicted by the Standard Model of Particle Physics. It goes without saying that the physics presented here is quite interesting.

> R. Ruiz Pittsburgh, Spring 2015

To Armando Pacheco and Michael Hattar.

1.0 THE STANDARD MODEL OF PARTICLE PHYSICS

1.1 INTRODUCTION

The Standard Model of Particle Physics (SM) is the quantum field theoretic model that, to date, best describes the interactions matter at small distances and large energies. Though incredibly successful, as we will discuss later, the SM does remain an incomplete description of nature. Before studying the particle content and forces of the SM, we begin by first considering what lies at the heart of the SM and physics in general: symmetries.

1.2 SYMMETRIES

A symmetry exists when a quantity, for example: linear momentum or electric charge, does not change (remains invariant) while a system undergoes a transformation, such as a spatial translation or a local U(1) phase shifts. Here we discuss continuous, both global and local, symmetries, emphasizing along the way the consequences of their spontaneous breakdown.

The spontaneous breakdown of a symmetry through the acquiring of a nonzero vacuum expectation value, or vev, by a scalar field is an interesting topic with subtle consequences. Typically in Quantum Field Theories (QFT), expectation values of fields, both bosonic and fermionic, are zero:

$$\langle 0|\phi(x), \ \psi(x), \ A^{\mu}(x)|0\rangle = 0.$$
 (1.1)

As fermions and vector bosons are nontrivial representations of the Lorentz group, they carry spinor and Lorentz indices. A nonzero vev for these fields would imply that the vacuum itself carries corresponding indices, indicating a preferred state, thereby breaking Lorentz invariance. Scalars, on the other hand, being trivial representations of the Lorentz group, whether elementary or composite, are allowed to form a condensate and acquire a nonzero vev without violating Lorentz invariance. However, whatever symmetries that are respected by scalars before acquiring a vev are not guaranteed to be preserved.

We now consider our first case study: continuous global symmetries.

1.3 CONTINUOUS SYMMETRIES I: GLOBAL SYMMETRIES

Global continuous symmetries are transformations that remain independent of spacetime coordinates. A familiar example intrinsic to all quantum mechanical processes is the invariance to an overall phase shift of the amplitude that leaves the physical probability density unchanged¹:

$$\mathcal{M} \to \mathcal{M}' = e^{-i\theta} \mathcal{M}, \quad \theta \in [0, 2\pi)$$
 (1.2)

$$|\mathcal{M}|^2 \to |\mathcal{M}'|^2 = |\mathcal{M}|^2 \tag{1.3}$$

Though moving the amplitude \mathcal{M} along the edge of a circle in the complex plane, such phase shifts are unobservable. Since the rotation holds for an arbitrary angle, it holds for all angles. The collection of all such transformation is the multiplicative group U(1):

$$U(1) = \{e^{i\theta} | \theta \in [0, 2\pi)\}$$
(1.4)

For infinitesimal rotations, we have

$$U(1) = 1 + i\theta \tag{1.5}$$

The spacetime independence of θ means that $\partial_{\mu} \exp[i\theta] = 0$. Hence, we say that physical probabilities derived from quantum mechanical amplitudes are symmetric (or invariant) under global U(1) symmetries (or transformations).

We study global symmetries by considering a Lagrangian density, or Lagrangian for short, at dimension-four consisting of both a Dirac fermion ψ and a complex scalar ϕ :

$$\mathcal{L} = \overline{\psi}i\partial \!\!\!/\psi + (\partial^{\mu}\phi^{*})\partial_{\mu}\phi - m_{\psi}\overline{\psi}\psi - m_{\phi}^{2}\phi^{*}\phi - \lambda(\phi^{*}\phi)^{2}.$$
(1.6)

Rotating ψ and ϕ under the same global U(1) transformations

$$\psi \to \psi' = U^{\dagger}\psi = e^{-i\theta}\psi, \qquad \phi \to \phi' = U^{\dagger}\phi = e^{-i\theta}\phi,$$
(1.7)

¹Throughout this text we employ *active* transformations $U^{-1} = e^{-i\theta}$, which differs from some texts, e.g., *Peskin & Schroeder* [6], which use *passive* transformations.

it is self-evident that Eq. (1.6) remains unchanged. Global symmetries, however, are not limited to simple Abelian transformations. Suppose that our ψ and ϕ fields were instead multiplets under a larger group, e.g., were in the fundamental representation of SU(n) or U(n):

$$\psi^T = (\psi_i \dots \psi_n), \quad \text{for} \quad SU(n) \text{ or } U(n)$$

$$(1.8)$$

$$\phi^T = (\phi_i \dots \phi_n), \quad \text{for} \quad SU(n) \text{ or } U(n). \tag{1.9}$$

The infinitesimal transformations now behave as

$$U^{\dagger} = e^{-i\theta} = 1 - i\theta \equiv 1 - i\theta^a T^a, \quad a = 1, \dots, n,$$
(1.10)

where T^a denotes the generators of SU(n) or U(n) and θ^a are the linearly independent, infinitesimal rotations in the space of our group's fundamental representation. As global (and local) transformations acting on scalars and fermions are unitary, i.e., $U^{-1} = U^{\dagger}$, we have

$$U^{-1} = e^{-i\theta^a T^a} = U^{\dagger} = \left(e^{i\theta^a T^a}\right)^{\dagger} = e^{-i\theta^{a*} T^{a\dagger}},\tag{1.11}$$

implying that the generator and its adjoint are related by the expression

$$\theta^a T^a = \theta^{a*} T^{a\dagger}. \tag{1.12}$$

However, generators of physical transformations are Hermitian, and so θ^a must be real:

$$0 = \theta^a T^a - \theta^{a*} T^{a\dagger} = T^a (\theta^a - \theta^{a*}) \implies \theta^a = \theta^{a*}.$$
 (1.13)

Despite the complication of non-Abelian groups, our Lagrangian remains unchanged under infinitesimal rotations

$$\mathcal{L} \to \mathcal{L}' = \overline{\psi'} i \partial \!\!\!/ \psi' + (\partial^{\mu} \phi'^{\dagger}) \partial_{\mu} \phi' - m_{\psi} \overline{\psi'} \psi' - m_{\phi}^2 \phi'^{\dagger} \phi' - \lambda (\phi'^{\dagger} \phi')^2$$
(1.14)
$$= \overline{\psi} U i \partial \!\!/ (U^{\dagger} \psi) + [\partial^{\mu} (\phi^{\dagger} U)] U^{\dagger} \partial_{\mu} \phi$$

$$= \overline{\psi}Ui\partial \hspace{-0.15cm}/ (U^{\dagger}\psi) + [\partial^{\mu}(\phi^{\dagger}U)]U^{\dagger}\partial_{\mu}\phi$$

$$- m_{\psi}\overline{\psi}UU^{\dagger}\psi' - m_{\phi}^{2}\phi^{\dagger}UU^{\dagger}\phi - \lambda(\phi^{\dagger}UU^{\dagger}\phi')^{2}$$
(1.15)

$$= \mathcal{L}. \tag{1.16}$$

Spontaneously Broken Global Symmetries 1.3.1

On our yellow brick road toward the emerald city of spontaneously broken gauge theories, we come across the related case of spontaneously broken global symmetries. Though sharing many mechanics with broken local transformations, the phenomenological outcomes radically differ. Consider the interacting theory of a complex scalar and a massive vector boson ρ

$$\mathcal{L} = (\partial^{\mu}\phi)^{*}\partial_{\mu}\phi - \frac{1}{4}\rho^{\mu\nu}\rho_{\mu\nu} - \frac{1}{2}M_{\rho}^{2}\rho_{\mu}\rho^{\mu} - m_{\phi}^{2}\phi^{*}\phi - \lambda(\phi^{*}\phi)^{2} - g_{\rho}\phi^{*}\phi\rho_{\mu}\rho^{\mu}, \qquad (1.17)$$

where the field strength $\rho_{\mu\nu}$ is

$$\rho_{\mu\nu} = \partial_{\mu}\rho_{\nu} - \partial_{\nu}\rho_{\mu}. \tag{1.18}$$

The theory is invariant under global U(1) transformations of ϕ . Inspecting the scalar potential

$$V(\phi^*\phi) = m_{\phi}^2(\phi^*\phi) + \lambda(\phi^*\phi)^2 + g_{\rho}\phi^*\phi\rho_{\mu}\rho^{\mu},$$
(1.19)

one sees that it is simply a quadratic function in $(\phi^*\phi)$ with coefficients m_{ϕ}^2 and λ . We ignore the contribution of ρ_{μ} as minima of vector fields must be zero to preserve Lorentz invariance. Potentials must also be bounded from below in order to bar tunneling to a state with infinite energy, so we require $\lambda > 0$. For $m_{\phi}^2 > 0$, the potential's minimum is zero at the origin

$$\left. \frac{\partial V}{\partial \phi} \right|_{\min} = \left. \frac{\partial V}{\partial (\phi^* \phi)} \frac{\partial (\phi^* \phi)}{\partial \phi} \right|_{\min}$$
(1.20)

$$= \left. \left(m_{\phi}^{2} + 2\lambda(\phi^{*}\phi) \right) \phi^{*} \right|_{\min} = 0 \implies \phi^{*}(x) \Big|_{\min} = 0 \tag{1.21}$$

$$\frac{\partial V}{\partial \phi^*} \bigg|_{\min} = \left(m_{\phi}^2 + 2\lambda(\phi^*\phi) \right) \phi \bigg|_{\min} = 0 \implies \phi(x) \bigg|_{\min} = 0 \quad (1.22)$$

However, curiously, when $m_{\phi}^2 < 0$, we discover a global minimum *away* from the origin

$$\frac{\partial V}{\partial \phi}\Big|_{\text{extrema}} = \left. \left(-|m_{\phi}^2| + 2\lambda(\phi^*\phi) \right) \phi^* \Big|_{\text{extrema}} \implies \phi^*(x) \Big|_{\text{extrema}} = 0, \ \sqrt{\frac{|m_{\phi}|^2}{2\lambda}} \quad (1.23)$$

$$\frac{\partial V}{\partial \phi^*} \bigg|_{\text{extrema}} = \left. \left(-|m_{\phi}^2| + 2\lambda(\phi^*\phi) \right) \phi \right|_{\text{extrema}} \implies \phi(x) \bigg|_{\text{extrema}} = 0, \ \sqrt{\frac{|m_{\phi}|^2}{2\lambda}} \qquad (1.24)$$

That is to say, the scalar ϕ possesses a nonzero vacuum expectation value (vev) given by

$$\langle \phi \rangle \equiv \frac{v}{\sqrt{2}} = \sqrt{\frac{|m_{\phi}^2|}{2\lambda}} > 0 \quad \Longrightarrow \quad v = \sqrt{2} \langle \phi \rangle = \sqrt{\frac{|m_{\phi}^2|}{\lambda}} > 0. \tag{1.25}$$

The factor of $\sqrt{2}$ accounts for the normalization of ϕ as it can be expanded in terms of its real and imaginary components, $\phi = (\Re(\phi) + i\Im(\phi))/\sqrt{2}$. Following this convention the kinetic term of ϕ results in properly normalized kinetic terms for $\Re(\phi)$ and $\Im(\phi)$.

We explore the consequences of quadratic (in $\phi^* \phi$) potential V and, effectively, the tachyonic mass m_{ϕ} by considering small perturbations of ϕ around v. We justify this by counting the degrees of freedom (states) in the theory before ϕ acquires a vev: two from the complex field ϕ and three from ρ (two transverse and one longitudinal polarization). Whether or not $\phi(x)$ is in a particular location, minimum or elsewhere, should not change the total number of physical states in the theory. So while their manifestations may depend on dynamics and momentum transfer scales, we expect to always have five physical states in our model. The seemingly missing degrees can be traced back to ϕ . As fields with zero vevs, e.g., ρ , are fluctuations about classical minima, we should expect to have fluctuations around $\langle \phi \rangle$. Therefore, expanding ϕ about its vev we have

$$\phi \to \phi \approx \frac{v+h+ia}{\sqrt{2}}, \quad \langle h \rangle = \langle a \rangle = 0,$$
 (1.26)

where h and a are real scalar fields with zero vevs. In passing, we note that the purely imaginary nature of ia implies that its interactions are odd under charge conjugation unlike h, which is C-even. Making the replacement, we see explicitly

$$\begin{aligned} \mathcal{L} \to \mathcal{L}' &= \frac{1}{2} \partial^{\mu} (v+h+ia)^* \partial_{\mu} (v+h+ia) + \frac{|m_{\phi}|^2}{2} (v+h+ia)^* (v+h+ia) - \frac{1}{2} M_{\rho}^2 \rho_{\mu} \rho^{\mu} \\ &- \frac{\lambda}{4} ((v+h+ia)^* (v+h+ia))^2 - g_{\rho} \rho_{\mu} \rho^{\mu} (v+h+ia)^* (v+h+ia) \end{aligned} \tag{1.27} \\ &= \frac{1}{2} \partial^{\mu} h \partial_{\mu} h + \frac{1}{2} \partial^{\mu} a \partial_{\mu} a + \frac{|m_{\phi}^2|}{2} (v^2 + 2vh + h^2 + a^2) - \frac{1}{2} M_{\rho}^2 \rho_{\mu} \rho^{\mu} \\ &- \frac{\lambda}{4} (v^4 + 4v^3h + 6v^2h^2 + 2v^2a^2 + 4vh^3 + h^4 + 2h^2a^2 + 4vha^2 + a^4) \\ &- \frac{g_{\rho}}{2} \rho_{\mu} \rho^{\mu} (v^2 + 2vh + h^2 + a^2) \end{aligned} \tag{1.28}$$

Regrouping terms in powers of h and a, we get

$$\mathcal{L} = \frac{1}{2} \partial^{\mu} h \partial_{\mu} h + \frac{1}{2} \partial^{\mu} a \partial_{\mu} a - \frac{1}{4} \rho^{\mu\nu} \rho_{\mu\nu}$$
(1.29)

$$-\frac{1}{2}\underbrace{(3\lambda v^{2} - |m_{\phi}^{2}|)}_{(h \text{ mass})^{2}}h^{2} - \frac{1}{2}\underbrace{(\lambda v^{2} - |m_{\phi}^{2}|)}_{\text{zero } a \text{ mass}}a^{2} - \frac{1}{2}\underbrace{(M_{\rho}^{2} + g_{\rho}v^{2})}_{(\rho \text{ mass shift})^{2}}\rho_{\mu}\rho^{\mu}$$
(1.30)

$$-\frac{\lambda}{2}h^{2}a^{2}\underbrace{-\lambda vh^{3}-\lambda vha^{3}}_{U(1)-\text{violating}} -\frac{\lambda}{4}h^{4} -\frac{\lambda}{4}a^{4}\underbrace{-g_{\rho}v\rho_{\mu}\rho^{\mu}h}_{U(1)-\text{violating}} -\frac{g_{\rho}}{2}\rho_{\mu}\rho^{\mu}h^{2} -\frac{g_{\rho}}{2}\rho_{\mu}\rho^{\mu}a^{2}$$
(1.31)

$$- \underbrace{v(\lambda v^{2} - |m_{\phi}^{2}|)}_{0^{2}}h + \underbrace{\frac{1}{2}\left(|m_{\phi}^{2}| - \frac{\lambda}{2}v^{2}\right)v^{2}}_{(\frac{1}{2}|m_{\phi}|v)^{2}}$$
(1.32)

As one may expect, expanding ϕ into real and imaginary components has the effect of making ϕ 's two degrees of freedom manifest in the form of two real states, h and a. It also gives rise to the four-point couplings h^4 , a^4 , $\rho^2 a^2$, etc. Setting v to zero has no impact on the existence of these couplings. Indeed, the original global U(1) symmetry is still respected by the vertices. A nonzero vev gives not-so-expected results. There are four immediate effects that merit our attention: (i) It endows the C-even component h with positive definite mass of larger in magnitude than m_{ϕ}

$$m_h = \sqrt{2\lambda}v = \sqrt{3\lambda v^2 - |m_{\phi}^2|} = \sqrt{2}|m_{\phi}|;$$
 (1.33)

(ii) renders massless the C-odd component of ϕ , a,

$$m_a = \sqrt{\lambda v^2 - |m_{\phi}^2|} = 0;$$
 (1.34)

(iii) it makes a positive definite shift to M_{ρ}

$$\tilde{M}_{\rho} = \sqrt{M_{\rho}^2 + g_{\rho} v^2} \ge M_{\rho}; \qquad (1.35)$$

(iv) and introduces global symmetry-violating, three-point interactions proportional to v

$$-\lambda vh^3 - \lambda vha^2 - g_\rho v\rho_\mu \rho^\mu h = -\sqrt{\frac{\lambda}{2}}m_h h^3 - \sqrt{\frac{\lambda}{2}}m_h ha^2 - \frac{g_\rho m_h}{\sqrt{2\lambda}}\rho_\mu \rho^\mu h.$$
(1.36)

The discrete charge conjugation symmetry protects (forbids) the Lagrangian from spontaneously generating interaction terms with odd powers of a. An imaginary vev, however, would generate such interactions. A generalization of this presentation to arbitrary global group symmetries with a countable number of group generators is known as **Goldstone's Theorem** [7]. It states that for each broken continuous global symmetry; equivalently, for each broken generator of a continuous global symmetry, a massless scalar appears. These massless scalars, such as a in our case, are called Nambu-Goldstone (NG) bosons. In the case of a spontaneously breakdown of an inexact global symmetry, the "inexactness" being controlled by some parameter M, the NG bosons acquire a mass proportional to ($M \times vev$) that vanishes in the limit that the global symmetry becomes exact. In that case, the NG bosons are called *pseudo-Nambu-Goldstone* (PNG) bosons, e.g., pions ($\pi^{0,\pm}$) in QCD after chiral symmetry breaking.

1.4 CONTINUOUS SYMMETRIES II: LOCAL SYMMETRIES

As the name suggests, local symmetries, also known as *gauge symmetries*, differ from global ones in that *continuous local symmetries* are infinitesimal transformations that are dependent on spacetime coordinates. In a sense, they are a generalization of global transformations. However, only derivatives acting on symmetry-respecting fields present a concern. Operators without derivatives but respect a global symmetry, e.g., $\phi^{\dagger}\phi$, also respect their local analogs

$$\phi^{\dagger}\phi \to \phi^{\prime\dagger}\phi^{\prime} = \phi^{\dagger}UU^{\dagger}\phi = \phi^{\dagger}\phi.$$
(1.37)

Thus our attention focuses on derivative operators, and in particular kinetic terms.

Lets consider a theory of a Dirac fermion ψ and a complex scalar ϕ

$$\mathcal{L} = \overline{\psi}i\partial\!\!\!/\psi + (\partial^{\mu}\phi^{\dagger})\partial_{\mu}\phi - m_{\psi}\overline{\psi}\psi - m_{\phi}^{2}\phi^{\dagger}\phi - \lambda(\phi^{\dagger}\phi)^{2}.$$
(1.38)

We assume ψ and ϕ satisfy some non-Abelian local symmetry but delay a discussion of the sign of λ and m_{ϕ} until the next section However, the scalar potential's resemblance to Eq. (1.19) is not coincidental. A global transformation on ψ (or ϕ) leaves kinetic terms unchanged since

$$\partial \!\!\!/ \psi \to \partial \!\!\!/ \psi' = \partial \!\!\!/ (e^{-i\theta}\psi) = \underbrace{(\partial \!\!\!/ e^{-i\theta})}_{0} \psi + e^{-i\theta} (\partial \!\!/ \psi) = e^{-i\theta} \partial \!\!\!/ \psi.$$
(1.39)

A spacetime-dependent symmetry rotation on the other hand, such as

$$\Theta(x) \equiv \Theta^a(x) T^a, \tag{1.40}$$

where T^a are the generators of the corresponding group and $\Theta^a(x)$ are spacetime-dependent rotations in the space of the group representation, leads to an additional term:

$$\partial_{\mu}\psi \to \partial_{\mu}\psi' = \partial_{\mu}(e^{-i\Theta(x)}\psi) = \underbrace{[-ie^{i\Theta(x)}\partial_{\mu}\Theta(x)]}_{\neq 0}\psi + e^{-i\Theta}(\partial_{\mu}\psi) \tag{1.41}$$

$$= e^{-i\Theta(x)} \left(\partial_{\mu} - i\partial_{\mu}\Theta(x)\right)\psi.$$
(1.42)

Suppressing Θ 's x^{μ} -dependence, the ψ and ϕ kinetic terms under local transformations are

$$\overline{\psi}i \ \partial \!\!\!/ \psi \to \overline{\psi}'i \ \partial \!\!\!/ \psi' = \overline{\psi}i \ \partial \!\!\!/ \psi + (\partial_\mu \Theta)\overline{\psi}\gamma^\mu\psi$$
$$(\partial_\mu \phi)^* \partial^\mu \phi \to (\partial_\mu \phi')^* \partial^\mu \phi' = \partial_\mu \phi^* \partial^\mu \phi + (\partial_\mu \Theta)(\partial^\mu \Theta)\phi^\dagger \phi - [i(\partial_\mu \phi^*)\phi(\partial^\mu \Theta) + \text{H.c}].$$

The existence of terms linear and quadratic in $\partial_{\mu}\Theta$ very much violate our notion of invariance under infinitesimal transformations. Therefore, if we must insist on such a symmetry, then additional terms must be introduced to cancel the $\partial_{\mu}\Theta$ terms. The Lorentz and group indices on $\partial_{\mu}\Theta =$ $\partial_{\mu}\Theta^{a}T^{a}$ provide us with much guidance.

As spin-zero and spin-half fields do not carry Lorentz four-vector indices, our symmetry-rescuing terms must come from modifications to our derivatives operators. This considerably constraints our options. A modification must then take the form

$$i\partial_{\mu} \to iD_{\mu} \stackrel{\text{preliminarily}}{=} i\partial_{\mu} - gA_{\mu}(x), \quad g > 0.$$
 (1.43)

where $A_{\mu}(x)$ is a quantum field. The resemblance of Eq. (1.43) to obtaining the Hamiltonian of a particle with electric charge e > 0 in classical electrodynamics using the substitution

$$p^{\mu} \to p^{\prime \mu} = p^{\mu} - eA^{\mu}(x),$$
 (1.44)

where $A^{\mu}(x^{\nu}) = (\Phi(x^{\nu}), \vec{A}(x^{\nu}))$ is the classical electrodynamic vector potential, is motivational. In the case of a local U(1) transformation with g = e, we can identify A_{μ} in Eq. (1.43) as the quantized version of A_{μ} in Eq. (1.44), familiarly known as the photon. For this reason, we take g > 0.

As the infinitesimal rotation Θ^a is actually a vector in the space spanned by the group generators T^a , it possesses as many independent components that spoil our symmetry as there are generators. For SU(N) and U(N) theories, respectively, there are

$$n = N^2 - 1$$
 and $n = N^2$ (1.45)

generators. So to systematically cancel these terms, we must introduce not just one A_{μ} but as many as there are Θ^a . To do this, A_{μ} , like Θ^a , must be a vector in the space spanned by T^a . Therefore, our derivative of Eq. (1.43) is more completely written as

$$i\partial_{\mu} \to iD_{\mu} = i\partial_{\mu} - gA_{\mu}, \quad A_{\mu} \equiv A^{a}_{\mu}T^{a}, \quad \operatorname{Tr}[T^{a}T^{b}] = \frac{1}{2}\delta^{ab}.$$
 (1.46)

The last equality represents the generator normalization convention we are adopting.

A local transformation on $D_{\mu}\psi$ is given by

$$D_{\mu}\psi \to D_{\mu}\psi' = (\partial_{\mu} + igA_{\mu})(e^{-i\Theta}\psi) = (\partial_{\mu} + igA^{a}_{\mu}T^{a})(\psi - i\Theta^{b}T^{b}\psi)$$
(1.47)

$$= \partial_{\mu}\psi - i\Theta^{b}T^{b}(\partial_{\mu}\psi) + igA^{a}_{\mu}T^{a}\psi - i(\partial_{\mu}\Theta^{b})T^{b}\psi + gA^{a}_{\mu}\Theta^{b}T^{a}T^{b}\psi.$$
(1.48)

Generically, the commutator for a non-Abelian field is expressed as

$$[A_{\mu}, A_{\nu}] \equiv [A^{b}_{\mu}T^{b}, A^{c}_{\nu}T^{c}] = A^{b}_{\mu}A^{c}_{\nu}[T^{b}, T^{c}] = if_{abc}A^{b}_{\mu}A^{c}_{\nu}T^{a}.$$
 (1.49)

where f_{abc} is the structure constant of the group. This allows us to rewrite Eq. (1.48) as

$$D_{\mu}\psi' = e^{-i\Theta}(\partial_{\mu}\psi) + igA^{a}_{\mu}T^{a}\psi - i(\partial_{\mu}\Theta^{b})T^{b}\psi + gA^{a}_{\mu}\Theta^{b}T^{b}T^{a}\psi + igf_{abc}A^{a}_{\mu}\Theta^{b}T^{c}\psi \quad (1.50)$$

$$= e^{-i\Theta}(\partial_{\mu}\psi) + gA^{a}_{\mu}\Theta^{b}T^{b} + ig\left(A^{a}_{\mu}T^{a} - \frac{1}{g}(\partial_{\mu}\Theta^{b})T^{b} + f_{abc}A^{a}_{\mu}\Theta^{b}T^{c}\right)\psi$$
(1.51)

As the group indices within the parentheses are not external, the labels a, b, c are dummy indices. Using the cyclic properties of f_{abc} , we rewrite the parenthetical term as

$$(\dots) = A^a_\mu T^a - \frac{1}{g} (\partial_\mu \Theta^a) T^a + f_{cba} A^c_\mu \Theta^b T^a = \left(A^a_\mu - \frac{1}{g} (\partial_\mu \Theta^a) - f_{abc} \Theta^b A^c_\mu \right) T^a.$$
(1.52)

Therefore, if A^a_μ also rotates under infinitesimal transformations such that

$$A^a_\mu \to A^{'a}_\mu = A^a_\mu + \frac{1}{g}(\partial_\mu \Theta^a) + f_{abc}\Theta^b A^c_\mu$$
(1.53)

then $D_{\mu}\psi$ transforms under local rotations (keeping terms at most linear in Θ) as

$$D'_{\mu}\psi' = e^{-i\Theta}(\partial_{\mu}\psi) + gA'^{a}_{\mu}\Theta^{b}T^{b}T^{a}\psi + ig\left(A'^{a}_{\mu} - \frac{1}{g}(\partial_{\mu}\Theta^{a}) - f_{abc}\Theta^{b}A^{c}_{\mu}\right)T^{a}\psi \qquad (1.54)$$

$$= e^{-i\Theta}(\partial_{\mu}\psi) + gA^{a}_{\mu}\Theta^{b}T^{b}T^{a}\psi + igA^{a}_{\mu}T^{a}\psi$$
(1.55)

$$= e^{-i\Theta}(\partial_{\mu}\psi) + ig(1-i\Theta^{b}T^{b})A^{a}_{\mu}T^{a}\psi$$
(1.56)

$$= e^{-i\Theta}(D_{\mu}\psi). \tag{1.57}$$

Our fermion and scalar kinetic terms now transform satisfactorily as

$$\overline{\psi}i \not\!\!\!D\psi \to \overline{\psi}'i \not\!\!\!D'\psi' = \overline{\psi} \left(e^{i\Theta} e^{-i\Theta} \right) i \not\!\!\!D\psi = \overline{\psi}i \not\!\!\!D\psi$$
(1.58)

$$(D_{\mu}\phi)^{\dagger}(D^{\mu}\phi) \to (D'_{\mu}\phi')^{\dagger}(D'^{\mu}\phi') = (D_{\mu}\phi)^{\dagger}(e^{i\Theta}e^{-i\Theta})(D^{\mu}\phi) = (D_{\mu}\phi)^{\dagger}(D^{\mu}\phi)$$
(1.59)

Thus, the replacement in our theory of derivatives ∂_{μ} with covariant derivatives D_{μ} :

$$\partial_{\mu} \to D_{\mu} \equiv \partial_{\mu} + ig A^a_{\mu} T^a, \qquad (1.60)$$

where ϕ , ψ , and A_{μ} transform under the local symmetry as

$$\psi \to \psi' = \psi e^{-i\Theta(x)} \tag{1.61}$$

$$\phi \to \phi' = \phi e^{-i\Theta(x)} \tag{1.62}$$

$$A^a_\mu \to A^{\prime a}_\mu = A^a_\mu + \frac{1}{g} (\partial_\mu \Theta^a), \quad \text{for Abelian symmetries}$$
(1.63)

$$A^a_\mu \to A^{'a}_\mu = A^a_\mu + \frac{1}{g}(\partial_\mu \Theta^a) + f_{abc}\Theta^b A^c_\mu$$
 for non-Abelian symmetries, (1.64)

renders the entire Lagrangian given by Eq. (1.38) invariant under local transformations. The gauge fields A^a_{μ} , as they carry Lorentz vector indices, and thus are in the vector boson representation of the Lorentz group. We refer to such objects that correspond to gauge transformations as *gauge bosons*. Colloquially, we also say that such a theory described

$$\mathcal{L} = \overline{\psi} i D \!\!\!/ \psi + (D^{\mu} \phi)^* D_{\mu} \phi - m_{\psi} \overline{\psi} \psi - m_{\phi}^2 \phi^* \phi - \lambda (\phi^* \phi)^2.$$
(1.65)

is gauge invariant. However, Eq. (1.65) is incomplete. As we have introduced the gauge field A_{μ} , or collection of gauge fields A^a_{μ} , we must also specify how it propagates through spacetime. In other words: its mass and kinetic terms.

Mass terms for vector bosons take the form

$$\mathcal{L}_{\text{Mass}} = \frac{1}{2} M_A^2 A_\mu^a A^{a\mu}.$$
 (1.66)

Under gauge transformations, however, we have

$$\frac{1}{2}M_A^2 A_\mu^a A^{a\mu} \rightarrow \frac{1}{2}M_A^2 A_\mu^{'a} A^{'a\mu}$$

$$= \frac{1}{2}M_A^2 \left(A_\mu^a + \frac{1}{g}(\partial_\mu \Theta^a) + f_{abc}\Theta^b A_\mu^c\right) \left(A^{\mu a} + \frac{1}{g}(\partial^\mu \Theta^a) + f_{ade}\Theta^d A^{\mu e}\right) (1.68)$$

Keeping terms that are no more than linear in Θ^a and permuting the indices of f, we obtain

$$\frac{1}{2}M_{A}^{2}A_{\mu}^{'a}A^{'a\mu} = \frac{1}{2}M_{A}^{2}\left[A_{\mu}^{a}A^{\mu a} + \frac{1}{g}(\partial_{\mu}\Theta^{a})A^{\mu a} + \frac{1}{g}(\partial^{\mu}\Theta^{a})A_{\mu}^{a}\right]$$

$$f_{abc}\Theta^{b}A_{\mu}^{c}A^{\mu a} + f_{cba}\Theta^{b}A_{\mu}^{c}A^{\mu a}\right]$$
(1.69)

$$= \frac{1}{2}M_A^2 \left[A_{\mu}^a A^{\mu a} + \frac{1}{g} (\partial_{\mu} \Theta^a) A^{\mu a} + \frac{1}{g} (\partial^{\mu} \Theta^a) A_{\mu}^a \right], \qquad (1.70)$$

which violates our gauge symmetry. Therefore, to preserve it, gauge bosons must be massless.

Kinetic terms for gauge bosons are constructed from the *field strength* tensor

$$A_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu} + ig[A_{\mu}, A_{\nu}]$$
(1.71)

$$\equiv A^a_{\mu\nu}T^a = \left(\partial_\mu A^a_\nu - \partial_\nu A^a_\mu - gf_{abc}A^b_\mu A^c_\nu\right)T^a, \qquad (1.72)$$

where we have used the commutator relationship of Eq. (1.49). Alternatively, we can construct the field strength tensor by evaluating the commutator of the covariant derivate:

$$A_{\mu\nu} = \frac{1}{ig} [D_{\mu}, D_{\nu}] = \frac{1}{ig} (\partial_{\mu} + igA_{\mu})(\partial_{\nu} + igA_{\nu}) - \frac{1}{ig} (\partial_{\nu} + igA_{\nu})(\partial_{\mu} + igA_{\mu})$$
(1.73)

$$= \frac{1}{ig}\partial_{\mu}\partial_{\nu} + (\partial_{\mu}A_{\nu}) + A_{\nu}\partial_{\mu} + A_{\mu}\partial_{\nu} + igA_{\mu}A_{\nu}$$

$$- \frac{1}{ig}\partial_{\nu}\partial_{\mu} - (\partial_{\nu}A_{\mu}) - A_{\mu}\partial_{\nu} - A_{\nu}\partial_{\mu} - igA_{\nu}A_{\mu}$$
(1.74)

$$= [(\partial_{\mu}A_{\nu}) - (\partial_{\nu}A_{\mu})] + ig(A_{\mu}A_{\nu} - A_{\nu}A_{\mu}).$$
(1.75)

In this notation, it is easy to see that non-Abelian field strengths transform under gauge a transformation U in the same manner as covariant derivatives do:

$$D_{\mu} \rightarrow D'_{\mu} = U^{\dagger} D_{\mu} U \qquad (1.76)$$

$$A_{\mu\nu} \rightarrow A'_{\mu\nu} = \frac{1}{ig} \left(U^{\dagger} D_{\mu} U U^{\dagger} D_{\nu} U - U^{\dagger} D_{\nu} U U^{\dagger} D_{\mu} U \right) = \frac{1}{ig} U^{\dagger} [D_{\mu}, D_{\nu}] U \qquad (1.77)$$

$$= U^{\dagger} A_{\mu\nu} U.$$

However, because of this property, we are not allowed to write a kinetic term simply as $A_{\mu\nu}A^{\mu\nu}$ if we require that it be independently gauge invariant from all other Lagrangian terms. We resolve this by taking only its diagonal elements, its trace. Physically, we can understand this as needing to match a gauge boson's gauge charge (group index) with itself to form a gauge singlet state. Subsequently,

$$Tr[A_{\mu\nu}A^{\mu\nu}] \to Tr[A'_{\mu\nu}A'^{\mu\nu}] = Tr[U^{\dagger}A_{\mu\nu}UU^{\dagger}A^{\mu\nu}U] = Tr[A_{\mu\nu}A^{\mu\nu}] = \frac{1}{2}A^{a}_{\mu\nu}A^{a\mu\nu}, \qquad (1.78)$$

there the factor of one-half came from the normalization of the generators, as given in Eq. (1.46). Scaling this trace by 1/2 is necessary for correct field normalization for identical particles. For Abelian gauge bosons, we have the additional property the field strengths are also individually gauge invariant:

$$A_{\mu\nu} \to A'_{\mu\nu} = \partial_{\mu}A'_{\nu} - \partial_{\nu}A'_{\mu} = \partial_{\mu}\left(A_{\nu} + \frac{1}{g}(\partial_{\nu}\Theta)\right) - \partial_{\nu}\left(A_{\mu} + \frac{1}{g}(\partial_{\mu}\Theta)\right)$$
(1.79)

$$= \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu} + \frac{1}{g}\left(\underbrace{\partial_{\mu}\partial_{\nu}\Theta - \partial_{\nu}\partial_{\mu}\Theta}_{0}\right) = A_{\mu\nu}.$$
(1.80)

Thus, Abelian kinetic terms do take the simple form $A_{\mu\nu}A^{\mu\nu}$. With this, we write

$$\mathcal{L} = \overline{\psi} i D \psi + (D^{\mu} \phi)^{\dagger} D_{\mu} \phi - m_{\psi} \overline{\psi} \psi - m_{\phi}^{2} \phi^{\dagger} \phi - \lambda (\phi^{\dagger} \phi)^{2} \underbrace{-\frac{1}{4} A_{\mu\nu} A^{\mu\nu}}_{\text{Abelian}} \underbrace{-\frac{1}{4} A_{\mu\nu}^{a} A^{a \ \mu\nu}}_{\text{non-Abelian}}$$
(1.81)

and our local gauge theory is now complete.

As a concrete example, we briefly consider a simple U(1) gauge theory: QED.

1.4.1 Quantum Electrodynamics

Quantum Electrodynamics (QED) is the theory of light at microscopic distances and, as the name suggests, is simply electrodynamics after second quantization, i.e., expansion of classical field in creation and annihilation operators. We begin by supposing a massive fermion ψ and complex scalar ϕ that transform according to a local U(1)_{EM} with generator \hat{Q}

$$\psi \to \psi' = \psi e^{-i\Theta(x)\hat{Q}}, \qquad \overline{\psi} \to \overline{\psi'} = \overline{\psi} e^{i\Theta(x)\hat{Q}}$$
(1.82)

$$\phi \to \phi' = \phi e^{-i\Theta(x)\hat{Q}}, \qquad \phi^* \to \phi'^* = \phi^* e^{i\Theta(x)\hat{Q}}.$$
 (1.83)

When \hat{Q} operates on ψ (ϕ) gives its electric charge q_{ψ} (q_{ϕ}) in units of e > 0. Following to our procedure above, our gauge invariant theory is given by the Lagrangian

$$\mathcal{L}_{\text{QED}} = \overline{\psi} i D \psi + (D^{\mu} \phi)^{\dagger} D_{\mu} \phi - \frac{1}{4} A_{\mu\nu} A^{\mu\nu} - m_{\psi} \overline{\psi} \psi - m_{\phi}^{2} \phi^{\dagger} \phi \qquad (1.84)$$



Figure 1: Interaction Feynman Rules for QED.

Expanding the covariant gauge derivatives gives one

$$\overline{\psi}i \not\!\!D\psi = \overline{\psi}i(\partial \!\!\!/ + ie \not\!\!A\hat{Q})\psi = \overline{\psi}i \not\!\!\partial\psi - eq_{\psi}\overline{\psi} \not\!\!A\psi$$
(1.85)

$$(D^{\mu}\phi)^{\dagger}D_{\mu}\phi = (\partial^{\mu}\phi + ieA^{\mu}\hat{Q}\phi)^{\dagger}(\partial_{\mu}\phi + ieA_{\mu}\hat{Q}\phi)$$
(1.86)

$$= (\partial^{\mu}\phi^{*})(\partial_{\mu}\phi) + e^{2}q_{\phi}^{2}A^{\mu}A_{\mu}\phi^{*}\phi + [ieq_{\phi}(\partial^{\mu}\phi^{*})A_{\mu}\phi + \text{H.c}],$$

Shuffling terms, we have

$$\mathcal{L}_{\text{QED}} = \mathcal{L}_{\text{Kinetic}} + \mathcal{L}_{\text{Mass}} + \mathcal{L}_{\text{Int.}}, \qquad (1.87)$$

$$\mathcal{L}_{\text{Kinetic}} = \overline{\psi} i \, \partial \!\!\!/ \psi + (\partial^{\mu} \phi^{*})(\partial_{\mu} \phi) - \frac{1}{4} A_{\mu\nu} A^{\mu\nu}$$
(1.88)

$$\mathcal{L}_{\text{Mass}} = -m_{\psi}\overline{\psi}\psi - m_{\phi}^{2}\phi^{\dagger}\phi \qquad (1.89)$$

$$\mathcal{L}_{\text{Int.}} = -eq_{\psi}\overline{\psi}\,\mathcal{A}\psi + e^2 q_{\phi}^2 g_{\mu\nu} A^{\mu} A^{\nu} \phi^* \phi + [ieq_{\phi}(\partial^{\mu}\phi^*)A_{\mu}\phi + \text{H.c}]$$
(1.90)

To efficiently obtain the *Feynman Rules* for QED from the interaction Lagrangian $\mathcal{L}_{\text{Int.}}$, we Fourier decompose the field operators ψ , ϕ , and A_{μ} under its Action. For example:

$$S_{\overline{\psi}\psi A} = i \int d^4x \mathcal{L}_{\overline{\psi}\psi A} = \int d^4x (-ieq_{\psi}) \overline{\psi} \,\mathcal{A}\psi$$

$$\propto \int d^4x \left[d^3p \right]^3 \sum_{\text{d.o.f.}} \left[a_s^{\dagger}(p') \overline{u}_s(p') e^{-ip' \cdot x} + b_s(p') \overline{v}_s(p') e^{ip' \cdot x} \right]$$

$$\times (-ieq_{\psi}\gamma^{\mu}) \left[a_{\lambda}(q) \varepsilon_{\lambda\mu}(q) e^{iq \cdot x} + a^{c\dagger}(q,\lambda) \varepsilon_{\mu}^{*}(q,\lambda) e^{-iq \cdot x} \right]$$

$$\times \left[a_s(p) u_s(p) e^{ip \cdot x} + b_s^{\dagger}(p) v_s(p) e^{-ip \cdot x} \right].$$
(1.91)
(1.91)
(1.91)
(1.92)

This represents every permutation of incoming/outgoing fermions/bosons that is allowed under

QFT and gauge invariance for the $\psi - \overline{\psi} - A$ coupling. The common factor is the term

$$V_{\overline{\psi}\psi A} \quad : \quad -ieq_{\psi}\gamma^{\mu}. \tag{1.93}$$

We may do the same for the $\phi - \phi^* - A - A$ interaction vertex. Keeping track of a multiplicity factor of 2 that originates from having identical bosons, i.e., A_{μ} , A_{ν} , we obtain

$$V_{\phi\phi AA}$$
 : $+2ie^2 q_{\phi}^2 g_{\mu\nu}.$ (1.94)

Last, we have the $\phi - \phi^* - A$ vertex. It is marginally more complicated since one must keep track of incoming/outgoing momenta. For incoming p and outgoing p', we have

$$S_{\phi\phi A} = i \int d^{4}x \mathcal{L}_{\phi\phi A} = \int d^{4}x (-eq_{\phi}) \left[(\partial^{\mu}\phi^{*}) A_{\mu}\phi - \phi^{*}A_{\mu}(\partial^{\mu}\phi) \right]$$
(1.95)

$$\propto \int d^{4}x \left[d^{3}p \right]^{3} \sum_{\text{d.o.f.}} \left[a^{\dagger}(p') \partial^{\mu}e^{-ip'\cdot x} + \underbrace{a^{c}(p')\partial^{\mu}e^{ip'\cdot x}}_{\text{Outgoing: } ip'^{\mu}} \right]$$
(1.95)

$$\times (-eq_{\phi}) \left[a_{\lambda}(q)\varepsilon_{\lambda\mu}(q)e^{iq\cdot x} + a^{c\dagger}(q,\lambda)\varepsilon^{*}_{\mu}(q,\lambda)e^{-iq\cdot x} \right]$$
(1.96)

$$\times \left[a(p)e^{ip\cdot x} + \underbrace{a^{c\dagger}(p)e^{-ip\cdot x}}_{\text{Incoming: } -ip^{\mu}} \right] - \cdots \text{(Incoming)}$$
(1.96)

Collecting terms gives us the coupling vertex

$$V_{\phi\phi A} : -ieq_{\phi}(p^{\mu} + p'^{\mu}) \quad \text{for incoming (outgoing) } p(p'). \tag{1.97}$$

We summarize these Feynman rules in Fig. 1.

1.4.2 The Higgs Mechanism: Spontaneously Broken Local Symmetries

At last, we turn to the topic of spontaneously broken gauge symmetries. Several of the intermediate steps here have been derived in Section 1.3.1, where spontaneously broken global symmetries are studied. We consider a model containing three complex scalars ϕ , Φ , and H. We let ϕ and Φ be respectively gauged under U(1)_A and U(1)_B, and normalize their couplings to unity. The field His charged under both U(1)_A and U(1)_B, also with unity charges. Notationally, it is often stated that under the gauge group

$$\mathcal{G} = \mathrm{U}(1)_A \times \mathrm{U}(1)_B \tag{1.98}$$

the fields ϕ , Φ , H are charged as follows:

 $\phi: (+1,0), \quad \Phi: (0,+1), \quad H: (+1,+1).$ (1.99)

Assuming that only H has a nonzero mass, our gauge invariant Lagrangian is

$$\mathcal{L} = (D^{\mu}\phi)^{\dagger}D_{\mu}\phi + (D^{\mu}\Phi)^{\dagger}D_{\mu}\Phi + (D^{\mu}H)^{\dagger}D_{\mu}H - \frac{1}{4}A_{\mu\nu}A^{\mu\nu} - \frac{1}{4}B_{\mu\nu}B^{\mu\nu} - V_1 - V_2, \qquad (1.100)$$

$$V_1 = m_H^2 H^* H + \lambda_{HH} (H^* H)^2 + \lambda_{H\phi} H^* H \phi^* \phi + \lambda_{H\Phi} H^* H \Phi^* \Phi, \qquad (1.101)$$

$$V_2 = \lambda_{\phi\phi}(\phi^*\phi)^2 + \lambda_{\Phi\Phi}(\Phi^*\Phi)^2 + \lambda_{\Phi\phi}\Phi^*\Phi\phi^*\phi, \qquad (1.102)$$

where $X_{\mu\nu}$ is the field strength of gauge boson X = A, B. The covariant derivatives are

$$D_{\mu}\phi = (\partial_{\mu} + ig_{A}A_{\mu}\hat{Q}_{A})\phi = (\partial_{\mu} + (+1)ig_{A}A_{\mu})\phi, \qquad (1.103)$$

$$D_{\mu}\Phi = (\partial_{\mu} + ig_{B}B_{\mu}\hat{Q}_{B})\Phi = (\partial_{\mu} + (+1)ig_{B}B_{\mu})\Phi, \qquad (1.104)$$

$$D_{\mu}H = (\partial_{\mu} + ig_A A_{\mu}\hat{Q}_A + ig_B B_{\mu}\hat{Q}_B)H = (\partial_{\mu} + (+1)ig_A A_{\mu} + (+1)ig_B B_{\mu})H, \quad (1.105)$$

where \hat{Q}_X denotes the charge generator of gauge interaction X. The potentials V_1 and V_2 have been written in such a way that strictly (H^*H) -dependent terms are contained in V_1 .

For the sake of avoiding a nonsensical theory, we require all four-point couplings λ_{XY} to be positive-definite. We require that neither ϕ nor Φ carry nonzero vevs,

$$\langle \phi \rangle = \langle \Phi \rangle = 0. \tag{1.106}$$

The quartic potential in (H^*H) , V_1 , gives rise to a nonzero vev for H if $m_H^2 < 0$. We may ignore ϕ and Φ in solving for the minimum of H since their vevs are (by hypothesis) zero. In this case, the extrema solutions of H are the same as those given in Eq. (1.24), and therefore H possesses a vev given by

$$v_H = \sqrt{2} \langle H \rangle = \sqrt{\frac{|m_H^2|}{\lambda_{HH}}}.$$
(1.107)

We now consider the covariant derivate acting on $\langle H \rangle$, which is the qualitatively new feature in spontaneously broken *local* symmetries. Setting H to its vev, we see for $D_{\mu}H$

$$D_{\mu}H = (\partial_{\mu} + ig_A A_{\mu} + ig_B B_{\mu}) \frac{v_H}{\sqrt{2}} = i \frac{v_H}{\sqrt{2}} (g_A A_{\mu} + g_B B_{\mu}).$$
(1.108)

Pairing it with its conjugate, we obtain

$$(D^{\mu}H)^{\dagger}(D_{\mu}H) = \frac{v_{H}^{2}}{2} \left(g_{A}A^{\mu} + g_{B}B^{\mu}\right) \left(g_{A}A_{\mu} + g_{B}B_{\mu}\right).$$
(1.109)

Without the loss of generality, we assume $g_A > g_B$ and define the quantities

$$\cos\theta_A \equiv \frac{g_A}{\sqrt{g_A^2 + g_B^2}}, \quad g_Z \equiv \sqrt{g_A^2 + g_B^2} = \frac{g_A}{\cos\theta_A}, \quad \text{and} \quad g_\gamma = g_A \sin\theta_A. \tag{1.110}$$

With these definitions, we may write Eq. (1.109) as

$$(D^{\mu}H)^{\dagger}(D_{\mu}H) = \frac{v_{H}^{2}}{2}g_{Z}^{2}\left(\cos\theta_{A}A_{\mu} + \sin\theta_{A}B_{\mu}\right)^{2}.$$
 (1.111)

However, one may recognize that the quantity in the parentheses is nothing more than a field redefinition in $g_A - g_B$ space given by the rotation

$$\begin{pmatrix} Z \\ \gamma \end{pmatrix} = \begin{pmatrix} \cos \theta_A & \sin \theta_A \\ -\sin \theta_A & \cos \theta_A \end{pmatrix} \begin{pmatrix} A \\ B \end{pmatrix}.$$
 (1.112)

The field γ is identified as the orthogonal state of Z as A, B are rotated by mixing angle θ_A . Expressing the kinetic term for H at its minimum in terms of this Z_{μ} vector boson gives us a remarkable result: a vector boson mass term.

$$(D^{\mu}H)^{\dagger}(D_{\mu}H) = \frac{v_{H}^{2}}{2}g_{Z}\underbrace{(\cos\theta_{A}A_{\mu} + \sin\theta_{A}B_{\mu})^{2}}_{Z^{\mu}Z_{\mu}} = \frac{M_{Z}}{2}Z_{\mu}Z^{\mu}, \qquad (1.113)$$

where the mass of the Z_{μ} boson is

$$M_Z = g_Z v_H. \tag{1.114}$$

As no such term for γ_{μ} materializes, it remains massless.

These results are notable because in Eq. (1.70) we showed that a gauge boson if forbidden to have mass as it would otherwise violate gauge invariance. However, as H was charged under the A and B gauge groups and has since acquired a vev, the vacuum too has acquired charges under A and B, breaking the local symmetry. The massless field γ , on the other hand, is free to make gauge transformations with respect to a new (Abelian) generator:

$$\hat{Q}_{\gamma} = \hat{Q}_A - \hat{Q}_B. \tag{1.115}$$

The fields ϕ and Φ , respectively, possess charges $q^{\phi}_{\gamma} = +1$ and $q^{\Phi}_{\gamma} = -1$ since

$$\hat{Q}_{\gamma}\phi = \hat{Q}_{A}\phi - \hat{Q}_{B}\phi = (+1)\phi - (0)\phi$$
 (1.116)

$$\hat{Q}_{\gamma}\Phi = \hat{Q}_{A}\Phi - \hat{Q}_{B}\Phi = (0)\Phi - (+1)\Phi.$$
 (1.117)

Their couplings to the Z and γ fields are discovered by applying the field rotation Eq. (1.112) to their covariant derivatives:

$$D_{\mu}\phi = (\partial_{\mu} + ig_A A_{\mu}\hat{Q}_A)\phi = \left[\partial_{\mu} + ig_A \left(\cos\theta_A Z_{\mu} - \sin\theta_A \gamma_{\mu}\right) \left(\hat{Q}_{\gamma} + \hat{Q}_B\right)\right]\phi \quad (1.118)$$

$$= \left(\partial_{\mu} + ig_Z \cos\theta_A^2 Z_{\mu} - ig_\gamma \gamma_{\mu}\right)\phi, \qquad (1.119)$$

$$D_{\mu}\Phi = (\partial_{\mu} + ig_{B}B_{\mu}\hat{Q}_{B})\Phi = \left[\partial_{\mu} + ig_{B}\left(\sin\theta_{A}Z_{\mu} + \cos\theta_{A}\gamma_{\mu}\right)\left(\hat{Q}_{A} - \hat{Q}_{\gamma}\right)\right]\Phi \quad (1.120)$$

$$= \left(\partial_{\mu} + ig_Z \sin \theta_A^2 Z_{\mu} + ig_\gamma \gamma_{\mu}\right) \Phi.$$
 (1.121)

Being a gauge interaction, γ couples to ϕ and Φ with equal strength and proportionally to their charge. Z_{μ} , however, couples non-universally: for $g_A > g_B$, ϕ interacts more strongly with Z_{μ} than Φ . This is partly due to Z_{μ} aligning more ($\theta_A > \pi/4$) with the gauge state A_{μ} than with B_{μ} , which is aligned more closely with γ . Altogether, this is the crux of spontaneously broken gauge symmetries: the generation of vector boson masses and the emergence of a "hidden" local symmetry. The hidden symmetry refers to H being charged under

$$U(1)_A \times U(1)_B,$$
 (1.122)

and so the associated gauge bosons are aware of its symmetry-breaking vev, but the subgroup

$$U(1)_{A-B} \subset U(1)_A \times U(1)_B,$$
 (1.123)

whose generator is given by $\hat{Q}_{\gamma} = \hat{Q}_A - \hat{Q}_B$, is unknown to *H* since it is *neutral* under this local symmetry. Being neutral, *H* is unable to charge the vacuum under this gauge group, and therefore it remains unbroken after the spontaneously breakdown of its component generators.

In analog to the global symmetry case, we now expand H about its minimum

$$H \approx \frac{v_H + h(x) + i\xi(x)}{\sqrt{2}}.$$
(1.124)

Writing this to lowest order in h and ξ , however, gives us

$$H = \frac{1}{\sqrt{2}} \left(v_H + h(x) \right) \left(1 + i \frac{\xi(x)}{v_H} \right) = \frac{1}{\sqrt{2}} \left(v_H + h(x) \right) e^{i \frac{\xi(x)}{v_H}},$$
(1.125)

which we recognize as a gauge transformation of the form

$$H \to H' = H e^{-i\Theta(x)}.\tag{1.126}$$

This also indicates that our gauge field transforms locally as

$$Z_{\mu} \to Z'_{\mu} = Z_{\mu} + \frac{1}{g_{\gamma}v}(\partial_{\mu}\xi). \qquad (1.127)$$

The field ξ is an unphysical degree of freedom that represents our ability to make gauge transformations, in contrast to the global symmetry case, where ξ was a real, massless scalar. The gauge choice of removing the unphysical fields by explicitly setting $\xi(x) = 0$ is known as the *unitary gauge*.

Continuing, the covariant derivate on H is now given by

$$D_{\mu}H = (\partial_{\mu} + ig_A A_{\mu} + ig_B B_{\mu}) \frac{1}{\sqrt{2}} (v_H + h) = \frac{1}{\sqrt{2}} \partial_{\mu}h + \frac{i}{\sqrt{2}} Z_{\mu} (M_Z + g_Z h). \quad (1.128)$$

Pairing $D_{\mu}H$ with its conjugate, we obtain

$$(D^{\mu}H)^{\dagger}(D_{\mu}H) = \frac{1}{2} \left[(\partial^{\mu}h) - iZ_{\mu}(M_{Z} + g_{Z}h) \right] \left[(\partial_{\mu}h) + iZ_{\mu}(M_{Z} + g_{Z}h) \right]$$
(1.129)

$$= \frac{1}{2} (\partial^{\mu} h) (\partial_{\mu} h) + \frac{1}{2} Z^{\mu} Z_{\mu} (M_Z + g_Z h)^2$$
(1.130)

$$= \frac{1}{2} (\partial^{\mu} h) (\partial_{\mu} h) + \frac{1}{2} M_Z^2 Z^{\mu} Z_{\mu} + g_Z M_Z Z^{\mu} Z_{\mu} h + \frac{g_Z^2}{2} Z^{\mu} Z_{\mu} h h, \quad (1.131)$$

which gives us three-point ZZh interactions proportional to M_Z and four-point ZZhh couplings that is suppressed by two powers of g_Z . For completeness, we turn to the potential V_1 . We observe the emergence of positive definite masses for h, ϕ and Φ as well as interactions that are linear and quadratic in h in the same manner that we witnessed for the global case:

$$V_{1} = m_{H}^{2}H^{*}H + \lambda_{HH}(H^{*}H)^{2} + \lambda_{H\phi}H^{*}H\phi^{*}\phi + \lambda_{H\Phi}H^{*}H\Phi^{*}\Phi, \qquad (1.132)$$

$$= \frac{m_{H}^{2}}{2}(v_{H}^{2} + h^{2} + 2v_{H}h) + \lambda_{HH}(v_{H}^{2} + h^{2} + 2v_{H}h)^{2}$$

$$+ \lambda_{H\phi}(v_{H}^{2} + h^{2} + 2v_{H}h)\phi^{*}\phi + \lambda_{H\Phi}(v_{H}^{2} + h^{2} + 2v_{H}h)\Phi^{*}\Phi \qquad (1.133)$$

$$= \frac{1}{2}\underbrace{(3\lambda_{HH}v_{H}^{2} - |m_{H}^{2}|)}_{(h \text{ mass})^{2}}hh + \frac{1}{2}\underbrace{(2\lambda_{H\phi}v_{H}^{2})}_{(\phi \text{ mass})^{2}}\phi\phi + \frac{1}{2}\underbrace{(2\lambda_{H\Phi}v_{H}^{2})}_{(\Phi \text{ mass})^{2}}\Phi\Phi$$

$$+ \text{ cubic interaction terms of the form } hhh, h\phi\phi, h\Phi\Phi,$$

+ quartic interaction terms of the form hhhh, $hh\phi\phi$, $hh\Phi\Phi$. (1.134)

This mechanism, proposed in 1964 first by Brout & Englert [8], Higgs [9, 10], and Guralnik, et. al. [11], later to be reviewed in Refs. [12, 13], is known as the *Brout-Englert-Higgs* (BEH) *Mechanism*, or more commonly, the the **Higgs Mechanism**. It is a subtle caveat of Goldstone's Theorem stating that for each broken continuous local symmetry, the gauge boson associated with the broken generator of the continuous local symmetry acquires a mass. The difference being that if the continuous symmetry is global (local), a massless scalar (massive vector boson) appears in the theory. With this framework in place, we now move onto our Standard Model adventure.

1.5 INTRODUCTION TO THE STANDARD MODEL OF PARTICLE PHYSICS

The Standard Model of particle physics, commonly denoted simply as SM, represents to-date our best understanding of matter and its interactions at energy scales on the order of 1 TeV and below. In terms of distance, this corresponding to scales as small as 10^{-19} meters. Though unsatisfactory,

for example its prediction of massless neutrinos, the impressive agreement between high precision predictions and experimental observations demonstrate that most any theory that supersedes it will contain the SM as its "low energy" effective field theory limit. In this section, we introduce the ingredients of the SM and derive some of its most fundamental properties. In doing so, we will be able to appreciate some of the more subtle aspects of SM extensions that alleviate its incompleteness.

Formally speaking, the SM contains a renormalizable Yang-Mills theory [14] together with of chiral spin-half fermions and spin-zero bosons (scalars) with varying charges under strongly coupled [15–19] and weakly coupled [20–22] gauge symmetries. Respecting both Abelian and non-Abelian transformations, the SM scalar sector breaks the weakly coupled gauge sector through the Higgs Mechanism. The remaining unbroken gauge symmetries, color (non-Abelian) and electromagnetism (Abelian), possess somewhat interesting dynamics and eventually give rise to atoms, which, to speak technically, are electronic bound states of light elementary fermions (electrons) and heavy hadronic bound states (nucleons). The applicability and utility of atoms are (presumably) familiar to the reader.

The dimension-four Lagrangian of the SM is given as

$$\mathcal{L}_{\rm SM} = \mathcal{L}_{\rm Gauge} + \mathcal{L}_{\rm Higgs} + \mathcal{L}_{\rm Fermion}, \qquad (1.135)$$

representing the gauge, scalar (or Higgs), and matter (or fermion) sectors of the models. We will now discuss each part in detail.

1.6 GAUGE SECTOR OF THE STANDARD MODEL

The gauge sector of the SM is categorized into two parts: (i) a strongly coupled (at low momentum transfers) but still asymptotically free (vanishing coupling at infinite momentum transfers) sector obeying an SU(3) symmetry, known as quantum chromodynamics (QCD); and (ii) a weakly coupled $SU(2) \times U(1)$ symmetry, known as the electroweak (EW) sector, that spontaneously breaks to a weakly coupled U(1) gauge symmetry.

The gauge field content of each symmetry constitute an adjoint representation of the group, meaning that there are as many gauge bosons of a local symmetry as there are generators of the group. For SU(N) theories, there are $N^2 - 1$ generators (bosons). Similarly for U(N) theories, there are N^2 generators (bosons). Thus, there is a total of 12 (8+3+1) massless spin-one (vector) bosons in the SM gauge group

$$\mathcal{G}_{\rm SM} = SU(3)_C \times SU(2)_L \times U(1)_Y. \tag{1.136}$$

The labels C, L, and Y denote color, left-handed weak isospin, and weak hypercharge, the names for the respectively conserved charges.

Defining indices a, b, c = 1, ..., 8 to denote color degrees of freedom and i, j, k = 1, ..., 3 to denote weak isospin degrees of freedom, and with a summation implied for repeated indices, the gauge sector Lagrangian is given by the vector boson kinetic terms

$$\mathcal{L}_{\text{Gauge}} = -\underbrace{\frac{1}{4} \operatorname{Tr}[G_{\mu\nu}G^{\mu\nu}]}_{\text{Color}} - \underbrace{\frac{1}{4} \operatorname{Tr}[W_{\mu\nu}W^{\mu\nu}]}_{\text{Weak Isospin}} - \underbrace{\frac{1}{4} F_{\mu\nu}F^{\mu\nu}}_{\text{Weak Hypercharge}}$$
(1.137)

$$= -\frac{1}{4}G^{a}_{\mu\nu}G^{a\ \mu\nu} - \frac{1}{4}W^{i}_{\mu\nu}W^{i\ \mu\nu} - \frac{1}{4}F_{\mu\nu}F^{\mu\nu}, \qquad (1.138)$$

For the SU(3) color gauge bosons G^a_{μ} , known as gluons, the field strength [See Eq. (1.75)] is

$$G^{a}_{\mu\nu} = \partial_{\mu}G^{a}_{\nu} - \partial_{\nu}G^{a}_{\mu} + ig_{s}[G_{\mu}, G_{\nu}]^{a}$$
(1.139)

$$= \partial_{\mu}G^{a}_{\nu} - \partial_{\nu}G^{a}_{\mu} - g_{s}f^{abc}G^{b}_{\mu}G^{c}_{\nu}.$$
(1.140)

A most striking feature of non-Abelian theories is the appearance of three-point $\partial G^a G^b G^c$ and four-point $G^b G^c G^d G^e$ interaction vertices among the gauge bosons. These self-interactions are proportional to both the coupling (quadratically in the four-gluon case) and the structure constant (product of structure constants in the four-gluon case). The reason for this self-coupling is due to the fact that the bosons in non-Abelian theories also carry gauge charges. In the gauge sector Lagrangian [Eq. (1.138)], this is why a trace over the non-Abelian generators to pair fields accordingly is required. For QCD and its SU(3)_C symmetry, its generators (in the fundamental representation) are proportional to the Gell-Man (or color) matrices λ^a . Explicitly, the generators are given by

$$\hat{T}^{a} = \frac{1}{2}\lambda^{a}, \qquad a = 1, \dots, 8, \qquad (1.141)$$

$$\lambda^{1} = \begin{pmatrix} 0 & 1 & 0 \\ 1 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad \lambda^{2} = \begin{pmatrix} 0 & -i & 0 \\ i & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad \lambda^{3} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & -1 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad \lambda^{4} = \begin{pmatrix} 0 & 0 & 1 \\ 0 & 0 & 0 \\ 1 & 0 & 0 \end{pmatrix},$$

$$\lambda^{5} = \begin{pmatrix} 0 & 0 & -i \\ 0 & 0 & 0 \\ -i & 0 & 0 \end{pmatrix}, \quad \lambda^{6} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & 1 \\ 0 & 1 & 0 \end{pmatrix}, \quad \lambda^{7} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -i \\ 0 & i & 0 \end{pmatrix}, \quad \lambda^{8} = \frac{1}{\sqrt{3}} \begin{pmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & -2 \end{pmatrix}.$$

Interaction	Symbol	Spin	$SU(3)_C \times SU(2)_L \times U(1)_Y$ Charge
Strong	G^a_μ	1	(3, 1, 0)
Weak	W^a_μ	1	(1, 2, 0)
Hypercharge	B_{μ}	1	(1, 1, 0)
Yukawa	Φ	0	(1, 2 , +1) (Complex)

Table 1: Bosons of the Standard Model Before Electroweak Symmetry Breaking

The nonzero, antisymmetric color structure constants are

$$f_{123} = 1$$
, $f_{458} = f_{678} = \frac{\sqrt{3}}{2}$, $f_{147} = f_{246} = f_{257} = f_{345} = f_{516} = f_{637} = \frac{1}{2}$. (1.142)

Turning to the SU(2) weak isospin gauge bosons W^i_{μ} , the field strength is

$$W^{i}_{\mu\nu} = \partial_{\mu}W^{i}_{\nu} - \partial_{\nu}W^{i}_{\mu} + ig_{W}[W_{\mu}, W_{\nu}]^{i}$$
(1.143)

$$= \partial_{\mu}W^{i}_{\nu} - \partial_{\nu}W^{i}_{\mu} - g_{W}\epsilon^{ijk}W^{j}_{\mu}W^{k}_{\nu}. \qquad (1.144)$$

As in the QCD case, we find three-point $\partial W^i W^j W^k$ and four-point $W^i W^j W^k W^j$ interaction vertices arising from the kinetic term. In the triplet (adjoint) representation of SU(2), the rotation matrices are equivalent to the SO(3) spatial rotations for angular momentum j = 1,

$$\hat{T}_{L}^{1} = \frac{1}{\sqrt{2}} \begin{pmatrix} 0 & 1 & 0 \\ 1 & 0 & 1 \\ 0 & 1 & 0 \end{pmatrix}, \quad \hat{T}_{L}^{2} = \frac{1}{\sqrt{2}} \begin{pmatrix} 0 & -i & 0 \\ i & 0 & -i \\ 0 & i & 0 \end{pmatrix}, \quad \hat{T}_{L}^{3} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & -1 \end{pmatrix}.$$
(1.145)

Immediately, we read off from \hat{T}_L^3 that the isospin charges of the weak bosons are

$$\hat{T}_{L}^{3}|W^{i}\rangle = \pm 1,0|W^{i}\rangle.$$
 (1.146)

In this relatively simple case, the $SU(2)_L$ generators (in the fundamental representation) are proportional to the Pauli spin matrices (hence the label "isospin"):

$$\hat{T}_{L}^{i} = \frac{\sigma^{i}}{2}, \qquad i = 1, \dots, 3, \quad \text{where}$$
 (1.147)

$$\sigma^{1} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \quad \sigma^{2} = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix}, \quad \sigma^{3} = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}.$$
 (1.148)
The structure constant is the antisymmetric, three-dimensional Levi-Civita tensor ϵ_{ijk}

$$[W_{\mu}, W_{\nu}] = W_{\mu}^{j} W_{\nu}^{k} [T^{j}, T^{k}] = W_{\mu}^{j} W_{\nu}^{k} \left[\frac{\sigma^{j}}{2}, \frac{\sigma^{k}}{2} \right] = W_{\mu}^{j} W_{\nu}^{k} \frac{i\sigma^{i}}{2} \epsilon_{ijk}.$$
 (1.149)

Lastly, for the U(1) weak hypercharge gauge boson B_{μ} , which carries zero hypercharge, we have

$$F_{\mu\nu} = \partial_{\mu}B_{\nu} - \partial_{\nu}B_{\mu}. \tag{1.150}$$

A special property of field strengths for Abelian gauge theories stems from its linear dependence on boson fields. Namely, that linear transformations acting on a gauge field also hold for its field strength. If we can express an Abelian gauge field by the following linear combination

$$B_{\mu} = \sum_{i} c_i A^i_{\mu}, \qquad (1.151)$$

then we have that the field strength obeys the analogous linear decomposition:

$$B_{\mu\nu} = \partial_{\mu}B_{\nu} - \partial_{\nu}B_{\mu} = \sum_{i} c_{i}\partial_{\mu}A^{i}_{\nu} - c_{i}\partial_{\nu}A^{i}_{\mu} = \sum_{i} c_{i}A^{i}_{\mu\nu}.$$
 (1.152)

Together, the isospin and hypercharge charge fields are the *unbroken electroweak* gauge bosons.

The gauge boson content of the SM is summarized in the first three rows of Table 1.

1.7 THE STANDARD MODEL HIGGS SECTOR AND ELECTROWEAK SYMMETRY BREAKING

The SM contains a single colorless, complex scalar field Φ that is gauged under the electroweak sector. Transforming as a doublet under SU(2)_L, i.e., under the fundamental representation, it possesses hypercharge Y = +1, mass μ , and is expressible as

$$\Phi = \frac{1}{\sqrt{2}} \begin{pmatrix} \phi^1 + i\phi^2 \\ \phi^0 + i\phi^3 \end{pmatrix} = \begin{pmatrix} \phi^+ \\ \frac{1}{\sqrt{2}}(\phi^0 + i\phi^3) \end{pmatrix}, \quad \phi^{\pm} \equiv \frac{1}{\sqrt{2}}(\phi^1 \pm i\phi^2).$$
(1.153)

From the requirements that it be charged under isospin (2 d.o.f.) and complex (×2 d.o.f.), Φ actually comprises four real scalar fields ϕ^0, \ldots, ϕ^4 . The fields ϕ^1, ϕ^2 , are written as a linear combination for reasons that will become clear shortly. It suffices for the moment to say that the two isospin components of Φ separately respect the a second U(1) gauge group with generator

$$\hat{Q}\Phi = \left(\hat{T}_{L}^{3} + \frac{1}{2}\hat{Y}\right)\Phi = \frac{1}{2}\left(\sigma^{3} + \mathbb{1}_{2}\right)\Phi = \begin{pmatrix} 1 & 0\\ 0 & 0 \end{pmatrix}\Phi,$$
(1.154)

indicating that ϕ^{\pm} have a charge $Q = \pm 1$ and ϕ^0, ϕ^1 are charge zero. In Eq. (1.154), we expanded \hat{T}_L^3 and \hat{Y} in SU(2)_L space, in which case $\hat{Y} = \mathbb{1}_2 Y_{\Phi} = (+1)\mathbb{1}_2$.

Ignoring fermions, the most general Lagrangian at dimension-four we can write for Φ is

$$\mathcal{L}_{\text{Higgs}} = \left(D^{\mu}\Phi\right)^{\dagger} D_{\mu}\Phi - V(\Phi), \quad V(\Phi) = -\mu^{2}\Phi^{\dagger}\Phi + \lambda(\Phi^{\dagger}\Phi)^{2}, \quad (1.155)$$

where the covariant derivative is given by

$$D_{\mu}\Phi = \left[\partial_{\mu} + ig_{W}\hat{T}_{L}^{i}W_{\mu}^{i} + i\frac{g_{Y}}{2}\hat{Y}B_{\mu}\right]\Phi$$
(1.156)

$$= \left[\partial_{\mu} + i\frac{g_W}{2}\sigma^i W^i_{\mu} + i\frac{g_Y}{2}B_{\mu}\right]\Phi, \qquad (1.157)$$

and $g_W(g_Y)$ denotes the weak isospin (hypercharge) coupling strength. \hat{Y} and g_Y are normalized such that a factor of 1/2 appears in the covariant derivative, but it is sometimes absorbed in to the definition of \hat{Y} . Though Eq. (1.156) appears harmless, its present form does not show its utility. Let us rewrite Eq. (1.156) by taking advantage of familiar results of SU(2) algebras. The raising and lowering ladder operators of SU(2) are canonically given by

$$\hat{T}_{L}^{\pm} = \hat{T}_{L}^{1} \pm i\hat{T}_{L}^{2} = \frac{1}{2}(\sigma^{1} \pm i\sigma^{2}) = \begin{cases} \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} & \text{for } + \\ \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} & \text{for } - \end{cases}$$
(1.158)

We now write the $SU(2)_L$ gauge fields in terms of raising and lowering operators

$$\hat{T}_{L}^{i}W_{\mu}^{i} = \frac{1}{2}(\hat{T}_{L}^{+} + \hat{T}_{L}^{-})W_{\mu}^{1} - \frac{i}{2}(\hat{T}_{L}^{+} - \hat{T}_{L}^{-})W_{\mu}^{2} + W_{\mu}^{3}\hat{T}_{L}^{3}$$
(1.159)

$$= \frac{1}{2}(W^{1}_{\mu} - iW^{2}_{\mu})\hat{T}^{+}_{L} + \frac{1}{2}(W^{1}_{\mu} + iW^{2}_{\mu})\hat{T}^{-}_{L} + W^{3}_{\mu}\hat{T}^{3}_{L}, \qquad (1.160)$$

and suggests the following linear field redefinitions

$$W_{\mu}^{\pm} = \frac{W_{\mu}^{1} \mp i W_{\mu}^{2}}{\sqrt{2}}.$$
(1.161)

In this form, the W^+, W^-, W^3 gauge bosons can be identified as increasing, decreasing, or leaving unchanged the weak isospin of a system that absorbs it. Alternatively, the three respectively lower, raise, or leave unchanged isospin when radiated. Equation (1.156) becomes

$$D_{\mu}\Phi = \left[\partial_{\mu} + \frac{ig_{W}}{\sqrt{2}}W_{\mu}^{+}\hat{T}_{L}^{+} + \frac{ig_{W}}{\sqrt{2}}W_{\mu}^{-}\hat{T}_{L}^{-} + ig_{W}W_{\mu}^{3}\hat{T}_{L}^{3} + i\frac{g_{Y}}{2}\hat{Y}B_{\mu}\right]\Phi.$$
 (1.162)

However, by decomposing \hat{T}_L^a in this manner, Eq. (1.162) suggests an additional action: a redefinition of W^3 and B. As W^3 and B share the same spacetime (massless, spin-1) and internal quantum numbers (zero isospin and hypercharge), they in principle can mix. This is very much like the gauge-mixing case we witnessed in Eq. (1.112).

To construct the appropriate field redefinitions, we first recognize that this will require at most a 2×2 matrix, where the level of mixing is controlled by an angle θ_W . Expanding \hat{T}_L^3 and \hat{Y} in SU(2)_L space gives us such an object:

$$g_W W^3_\mu \hat{T}^3_L + \frac{g_Y}{2} \hat{Y} B_\mu = \frac{1}{2} \begin{pmatrix} g_W W^3_\mu + g_Y B_\mu & 0\\ 0 & -g_W W^3_\mu + g_Y B_\mu \end{pmatrix}.$$
 (1.163)

Being diagonal, it is easy to read off the eigenstates, which we label A_{μ} and Z_{μ} ,

$$A_{\mu} \sim g_W W_{\mu}^3 + g_Y B_{\mu} \sim \sin \theta_W W_{\mu}^3 + \cos \theta_W B_{\mu}$$
(1.164)

$$Z_{\mu} \sim g_W W_{\mu}^3 - g_Y B_{\mu} \sim \cos \theta_W W_{\mu}^3 - \sin \theta_W B_{\mu}. \qquad (1.165)$$

The interaction states then relate to gauge states W^3_{μ} , B_{μ} by the $SU(2)_L - U(1)_Y$ rotation

$$\begin{pmatrix}
A_{\mu} \\
Z_{\mu}
\end{pmatrix} = \begin{pmatrix}
\cos \theta_{W} & \sin \theta_{W} \\
-\sin \theta_{W} & \cos \theta_{W}
\end{pmatrix} \begin{pmatrix}
B_{\mu} \\
W_{\mu}^{3} \\
W_{\mu}^{3}
\end{pmatrix}.$$
(1.166)

with coupling and mixing parameters defined by

$$\sin \theta_W \equiv \frac{g_Y}{\sqrt{g_W^2 + g_Y^2}} \quad \text{and} \quad g_Z \equiv \sqrt{g_W^2 + g_Y^2} = \frac{g}{\cos \theta_W}.$$
 (1.167)

Conventionally, we take the $g_Y \to 0$ limit to be the decoupling regime where the SU(2)_L and U(1)_Y gauge bosons do not mix. We now rewrite the last terms of $D_{\mu}\Phi$ in Eq. (1.162) as

$$g_{W}W_{\mu}^{3}\hat{T}_{L}^{3} + \frac{g_{Y}}{2}\hat{Y}B_{\mu} = g_{W}(\sin\theta_{W}A_{\mu} + \cos\theta_{W}Z_{\mu})\hat{T}_{L}^{3} + \frac{g_{Y}}{2}(\cos\theta_{W}A_{\mu} - \sin\theta_{W}Z_{\mu})\hat{Y}$$

$$= g_{W}\sin\theta_{W}\underbrace{(\hat{T}_{L}^{3} + \frac{1}{2}\hat{Y})}_{\hat{Q} \text{ of Eq. (1.154)}}A_{\mu} + \frac{g_{W}}{\cos\theta_{W}}(\cos^{2}\theta_{W}\hat{T}_{L}^{3} - \sin^{2}\theta_{W}\underbrace{\frac{1}{2}\hat{Y}}_{\hat{Q}-\hat{T}_{L}})Z_{\mu}$$

$$= e\hat{Q}A_{\mu} + g_{Z}(\hat{T}_{L}^{3} - \sin^{2}\theta_{W}\hat{Q})Z_{\mu}, \qquad (1.168)$$

where

$$e \equiv g_W \sin \theta_W. \tag{1.169}$$

Finally, we have

$$D_{\mu} = \left[\partial_{\mu} + \frac{ig_W}{\sqrt{2}}W_{\mu}^{+}\hat{T}_L^{+} + \frac{ig_W}{\sqrt{2}}W_{\mu}^{-}\hat{T}_L^{-} + ieA_{\mu}\hat{Q} + ig_Z Z_{\mu}(\hat{T}_L^3 - \sin^2\theta_W\hat{Q})\right].$$
 (1.170)

Equation (1.170) is a rather dense expression, so we take time and explore the consequences of our successive field definitions. The second and third terms, W^+ and W^- , being mixtures of pure $SU(2)_L$ fields that still carry $T_L^3 = \pm 1$ isospin, remain the gauge bosons of the local symmetry. As previously mentioned, they transmute lower and upper components of Φ into each other. With the ladder operators \hat{T}_L^{\pm} , it is clear that terms proportional to

$$W^+\phi^+$$
 or $W^-\phi^-$, (1.171)

which would violate weak isospin charge conservation, do not exist.

The third term, $A_{\mu}\hat{Q}$, is interesting because we will eventually identify these objects as the photon field and the electric charge generator of QED. The term is also interesting because only half the Higgs doublet is aware of its existence; see the discussion of "Hidden Local Symmetries" above Eq. (1.123). As the sum of the hypercharge and third weak isospin generators, the (electric) charges Q for the four ϕ^i fields are

$$Q_{\phi^{1,2}} = T_L^3_{\phi^{1,2}} + \frac{1}{2}Y_{\phi^{0,3}} = \frac{+1}{2} + \frac{1}{2} = 1, \qquad (1.172)$$

$$Q_{\phi^{0,3}} = T_L^3_{\phi^{0,3}} + \frac{1}{2}Y_{\phi^{0,3}} = \frac{-1}{2} + \frac{1}{2} = 0.$$
 (1.173)

The relevant portion of the Higgs doublet's covariant derivate then simplifies to

$$D_{\mu}\Phi \ni \left(\partial_{\mu} + ieA_{\mu}\hat{Q}\right)\Phi = \partial_{\mu}\Phi + ieA_{\mu}\begin{pmatrix} 1 & 0\\ 0 & 0 \end{pmatrix}\begin{pmatrix} \phi^{+}\\ \frac{1}{\sqrt{2}}(\phi^{0} + i\phi^{3}) \end{pmatrix}$$
(1.174)

$$\ni \quad (\partial_{\mu} + ieA_{\mu}) \phi^+, \tag{1.175}$$

which, as we studied in Section 1.4.1, is the covariant derivative for scalar QED. However, as W^{\pm} carry nonzero isospin but zero hypercharge, they too carry a net electric charge $Q_W = \pm 1$. These interaction terms are not present in $\mathcal{L}_{\text{Higgs}}$ because they emerge after applying the field redefinitions W^{\pm}_{μ} in Eq. (1.161) and Z_{μ}, A_{μ} in Eq. (1.166) to the W^a_{μ} and B_{μ} field strengths.

The last term of Eq. (1.170) is notable because its gives the appearance of predicting deviations from universal gauge couplings, even as the lower components of Φ have zero electromagnetic charge. Of course, as Z_{μ} is neither an isospin or hypercharge gauge boson, gauge coupling universality is not actually violated. It is enlightening to see the origin of slight coupling difference by considering the $\phi^0 \phi^0 VV$ coupling for $V = W^{\pm}, Z$. From the kinetic term in Eq. (1.155), we have

$$(D_{\mu}\Phi)^{\dagger} D^{\mu}\Phi \quad \ni \quad \frac{g_{W}^{2}}{2} W_{\mu}^{-} W^{\mu+} \Phi^{\dagger} \left(\hat{T}_{L}^{-} \hat{T}_{L}^{+}\right) \Phi + g_{Z}^{2} Z_{\mu} Z^{\mu} \Phi^{\dagger} \left(\hat{T}_{L}^{3} - \sin^{2}\theta_{W} \hat{Q}\right)^{2} \Phi \quad (1.176)$$

$$\ni \quad \frac{g_W^2}{2} \left(\frac{1}{\sqrt{2}}\right)^2 W_{\mu}^- W^{\mu+} \phi^0 \phi^0 + g_Z^2 \left(\frac{1}{2\sqrt{2}}\right)^2 Z_{\mu} Z^{\mu} \phi^0 \phi^0 \tag{1.177}$$



Figure 2: Feynman vertex rules for $\phi^0 - \phi^0 - V - V$, $V = W^{\pm}$, Z, in the SM before EWSB.

Accounting for the symmetry factors from identical pairs $\phi^0 \phi^0$ and ZZ, the Feynman vertex rules for the four-point interactions, and shown in Fig. 2, are

$$\phi^0 \phi^0 W^+ W^-$$
 : $ig\phi\phi WW = i\frac{g_W^2}{2}$ (1.178)

$$\phi^0 \phi^0 Z Z$$
 : $ig\phi\phi Z Z = i\frac{g_Z^2}{2} = i\frac{g_W^2}{2\cos^2\theta_W},$ (1.179)

and become identical in the decoupling limit of $g_Y \to 0$. At leading order, we therefore have

$$\lambda_{WZ} \stackrel{\text{Preliminarily}}{=} \frac{g_{\phi\phi WW}}{g_{\phi\phi ZZ} \cos^2 \theta_W} = \frac{g_W^2}{g_Z^2 \cos^2 \theta_W} = 1, \qquad (1.180)$$

which is actually quite stable under radiative corrections [23]. Equation (1.180) is very related to the notion of *custodial symmetry*, and will be visited shortly.

The scalar boson content of the SM is summarized in the last row of Table 1. We now turn our focus to the potential V in $\mathcal{L}_{\text{Higgs}}$ and the topic of *electroweak* (EW) symmetry breaking (EWSB).

1.7.1 Electroweak Symmetry Breaking I: Massive Gauge Bosons

From Section 1.4.2, we learned that if the Φ field mass μ^2 and the self-coupling λ are both positivedefinite, then its potential V,

$$V(\Phi) = -\mu^2 \Phi^{\dagger} \Phi + \lambda (\Phi^{\dagger} \Phi)^2, \qquad (1.181)$$

has a minimum at the origin and the Higgs field's ground state expectation value is zero. However, for $\mu^2 < 0$ and positive λ , the minimum is away from the origin, leading to a nonzero vev, triggering the Higgs Mechanism. The EW symmetries under which the Higgs transforms are broken spontaneously, and the associated EW gauge bosons of these now-broken symmetries generate masses proportional to the size of the vev, a process called *electroweak* (EW) symmetry breaking (EWSB). We now apply the Higgs Mechanism to the SM and denote the vev of Φ by

$$v \equiv \sqrt{2} \langle \Phi \rangle = \sqrt{\frac{|\mu^2|}{\lambda}}.$$
(1.182)

Empirically, v is measured from the muon lifetime [24]

$$\tau_{\mu} = \frac{1}{\Gamma_{\mu}} = \frac{192\pi^3}{G_F^2 m_{\mu}^2} \approx 2.2^{-6} s, \qquad (1.183)$$

where G_F is Fermi's constant and

$$\sqrt{\sqrt{2}G_F} = v \approx 246 \text{ GeV.}$$
(1.184)

Letting Φ settle at the minimum of its potential and take on the value of its vev, i.e.,

$$\Phi = \frac{1}{\sqrt{2}} \begin{pmatrix} 0\\ v \end{pmatrix}, \tag{1.185}$$

its covariant derivate is then

$$D_{\mu}\Phi = \left[\underbrace{\partial_{\mu}}_{\partial_{\mu}v=0} + \frac{ig_{W}}{\sqrt{2}}W_{\mu}^{+}\hat{T}_{L}^{+} + \frac{ig_{W}}{\sqrt{2}}\underbrace{W_{\mu}^{-}\hat{T}_{L}^{-}}_{\hat{T}_{L}^{-}\Phi=0} + ie\underline{A_{\mu}\hat{Q}}_{\hat{Q}\Phi=0} + ig_{Z}Z_{\mu}\left(\hat{T}_{L}^{3} - \sin^{2}\theta_{W}\hat{Q}\right)\right]\frac{v}{\sqrt{2}}\begin{pmatrix}0\\1\end{pmatrix}$$
(1.186)

$$= \frac{iv}{\sqrt{2}} \left[\frac{g_W}{\sqrt{2}} W^+_\mu \begin{pmatrix} 1\\0 \end{pmatrix} + \frac{g_Z}{2} Z_\mu \begin{pmatrix} 0\\1 \end{pmatrix} \right] = \frac{iv}{2} \begin{pmatrix} g_W W^+_\mu \\ -\frac{g_Z}{\sqrt{2}} Z_\mu \end{pmatrix}$$
(1.187)

The kinetic term of Φ at the bottom of the Higgs' potential then simplifies to

$$(D_{\mu}\Phi)^{\dagger}D^{\mu}\Phi = \frac{v^{2}}{2^{2}} \left(g_{W}W_{\mu}^{-} - \frac{-g_{Z}}{\sqrt{2}}Z_{\mu}\right) \left(\frac{g_{W}W_{\mu}^{+}}{-g_{Z}}\frac{-g_{Z}}{\sqrt{2}}Z_{\mu}\right)$$
(1.188)

$$= M_W^2 W_\mu^- W_\mu^+ + \frac{1}{2} M_Z^2 Z_\mu Z^\mu$$
 (1.189)

where we have define the mass terms

$$M_W^2 \equiv \frac{g_W^2 v^2}{4}$$
 and $M_Z^2 \equiv \frac{g_Z^2 v^2}{4} = \frac{(g_W^2 + g_Y^2)v^2}{4}$. (1.190)

The massive gauge bosons, W^{\pm} and Z, have been measured at many experiments since their first direct production at CERN's Super Proton Synchrotron by the UA1 and UA2 experiments in 1983. Presently, the world's best average for these masses are [25]

$$M_W^{\text{World Avg.}} = 80.385 \pm 0.015 \text{ GeV}$$
 and $M_Z^{\text{World Avg.}} = 91.1876 \pm 0.0021 \text{ GeV}.$ (1.191)

Name	Gauge Interaction	Symbol	Spin	Mass $[GeV]$ [25]	$SU(3)_C \times U(1)_{EM}$
Gluon	Strong	G^a_μ	1	0	(3,0)
W	Weak	W^{\pm}_{μ}	1	80.385 ± 0.015	$(1, \pm 1)$
Z	Weak	Z_{μ}	1	$91.1876 {\pm} 0.0021$	(1,0)
Photon	Electromagnetism	A_{μ}/γ	1	0	(1,0)
Higgs	Yukawa (Not Gauged)	h	0	125.09 ± 0.21	(1,0)

Table 2: Bosons of the Standard Model After Electroweak Symmetry Breaking

As tempted as we are to comment on the similarity of the masses, we continue with EWSB.

Counting degrees of freedom before EWSB, we had four fields from Φ , three SU(2)_L gauge fields, and one U(1)_Y gauge field. Since each (massless) gauge boson possess two transverse polarizations, this gives us 12 total degrees of freedom. Presently, we have recovered only nine from the massive W^{\pm} , Z bosons (two transverse and one longitudinal polarization). We saw in Section 1.4.2 that vector boson masses break gauge invariance, and thus M_W an M_Z in Eq. (1.189) ruin the generators

$$\hat{T}^{\pm}, \ \hat{T}_L^3 - \sin^2 \theta_W \hat{Q}.$$
 (1.192)

However, Φ does not carry a charge associated with generator

$$\hat{Q} = \hat{T}_L^3 + \frac{1}{2}\hat{Y},\tag{1.193}$$

and A_{μ} remains massless:

$$m_{\gamma} = 0 (1.194)$$

Two more physical degrees of freedom are thus recovered as transverse polarizations.

The last physical state comes from fluctuations of Φ around $\langle \Phi \rangle$. We define the field h with a vanishing vev such that

$$\Phi(x) \approx \frac{1}{\sqrt{2}} \begin{pmatrix} 0\\ v+h(x) \end{pmatrix}, \quad \langle h(x) \rangle = 0.$$
(1.195)

Recall from Eq. (1.125) that $\Im[\Phi(x)]$ around v is an unphysical field that represents the ability of

 A_{μ} to make a gauge transformation. The covariant derivate acting on Φ now takes the form

$$D_{\mu}\Phi = \left[\frac{1}{\sqrt{2}} \begin{pmatrix} 0\\ \partial_{\mu}h \end{pmatrix} + i\frac{g_{W}}{2}W_{\mu}^{+} \begin{pmatrix} v+h\\ 0 \end{pmatrix} + i\frac{g_{Z}}{2\sqrt{2}}Z_{\mu} \begin{pmatrix} 0\\ v+h \end{pmatrix}\right]$$
(1.196)
$$= \left(\frac{i\frac{g_{W}}{2}W_{\mu}^{+}(v+h)}{\frac{1}{\sqrt{2}}\partial_{\mu}h - i\frac{g_{Z}}{2\sqrt{2}}Z_{\mu}(v+h)}\right).$$
(1.197)

This gives rise to the three-point and four-point interaction terms

$$(D_{\mu}\Phi)^{\dagger}D^{\mu}\Phi = \left(\frac{-ig_{W}}{2}W_{\mu}^{-}(v+h), \frac{\partial_{\mu}h}{\sqrt{2}} + \frac{ig_{Z}}{\sqrt{2}}Z_{\mu}(v+h)\right) \left(\frac{ig_{W}}{2}W^{\mu+}(v+h) - \frac{\partial^{\mu}h}{\sqrt{2}} - \frac{ig_{Z}}{2\sqrt{2}}Z^{\mu}(v+h)\right)$$
(1.198)

$$= \frac{1}{2}\partial_{\mu}h\partial^{\mu}h + M_{W}^{2}W_{\mu}^{-}W^{\mu+} + M_{Z}^{2}Z_{\mu}Z^{\mu} + \underbrace{g_{W}M_{W}W_{\mu}^{-}W^{\mu+}h + g_{Z}M_{Z}Z_{\mu}Z^{\mu}h}_{hVV \text{ Coupling } \propto M_{V}}$$

$$+\underbrace{\frac{g_{W}^{2}}{4}W_{\mu}^{-}W^{\mu+}hh + \frac{g_{Z}^{2}}{8}Z_{\mu}Z^{\mu}hh}_{hhVV \text{ Coupling}}.$$
(1.199)

Expending Φ in the potential V, we obtain the mass and self-interaction terms for h:

$$V(\Phi) = -\mu^{2} \left(\frac{1}{\sqrt{2}}\right)^{2} \left(0, v+h\right) \left(\frac{0}{v+h}\right) + \lambda \left(\frac{1}{\sqrt{2}}\right)^{2} \left[\left(0, v+h\right) \left(\frac{0}{v+h}\right)\right]^{2} (1.200)$$
$$= \frac{1}{2} m_{H}^{2} h^{2} + \sqrt{\frac{\lambda}{2}} m_{H} h h h + h^{4} + \frac{\lambda}{4} h h h h + \frac{\lambda v^{4}}{2}, \qquad (1.201)$$

where the mass of the *Higgs bosons*, h, is given by

$$m_H \equiv v\sqrt{2\lambda} = \sqrt{2}|\mu| \tag{1.202}$$

Discovered only recently by the ATLAS and CMS experiments at the CERN's Large Hadron Collider [26,27], the discovery of h represent the completion of the SM as its last unknown parameter. Presently, the best combination measurement of the Higgs mass is [28]

$$m_H^{\text{ATLAS+CMS}} = 125.09 \pm 0.21 \text{ (stat.)} \pm 0.11 \text{(syst.)}.$$
 (1.203)

Direct measurements of the Higgs self-coupling have not been achieved at the time of this writing. Taking the central value of m_H , it is predicted to be

$$\lambda = \frac{m_H^2}{2v^2} = \frac{1}{\sqrt{2}} m_H^2 G_F \approx 0.129 \approx \frac{1}{8}.$$
 (1.204)



Figure 3: Feynman vertex rules for h - h - V - V (L) and h - V - V (C), $V = W^{\pm}$, Z, in the SM after EWSB. (R) W/Z mass.

The elementary boson content of the SM after EWSB is summarized in Table 2.

Before introducing the fermionic content of the SM, we return to the similarity of M_W and M_Z .

1.7.2 Custodial Symmetry

Recalling the definitions M_W and M_Z , we have

$$M_W^2 \equiv \frac{g_W^2 v^2}{4}$$
 (1.205)

$$M_Z^2 \equiv \frac{g_Z^2 v^2}{4} = \frac{(g_W^2 + g_Y^2) v^2}{4}, \qquad (1.206)$$

which means that the mass ratio of the two is a measure of how much, or how little, the isospin and hypercharge groups rotate into each other during EWSB:

$$\frac{M_W^2}{M_Z^2} = \frac{g_W^2}{g_W^2 + g_Y^2} = \cos\theta_W^2. \tag{1.207}$$

In the zero mixing limit, which arises from either $Y_{\Phi} = 0$ or a negligibly small g_Y , the W and Z bosons masses converge. From this, the observable ρ , also called the ρ -parameter, can be constructed. At tree-level in the SM, the ρ -parameter is defined to be

$$\rho \equiv \frac{M_W^2}{M_Z^2 \cos^2 \theta_W} = 1, \qquad (1.208)$$

changes very little under radiative corrections [23]. This stability is due to custodial symmetry. In the SM, the Higgs field obeys an approximate global $SU(2)_L \times SU(2)_R$ symmetry, and is exact in the zero hypercharge limit. After EWSB, the (approximate) left-right symmetry breaks down to an (approximate) vector symmetry, i.e.,

$$\operatorname{SU}(2)_L \times \operatorname{SU}(2)_R \to \operatorname{SU}(2)_V,$$
 (1.209)

thereby ensuring a near mass degeneracy among the gauge bosons.

However, it is straightforward to see how drastically ρ can change in the presence of new scalars participating in EWSB. Supposing it were the case that many complex scalars, all gauged under $SU(2)_L \times U(1)_Y$, acquire various vevs. For such a scalar Φ_i in weak isospin representation \hat{T}_i with weak isospin charge T_i , only its electrically neutral component, ϕ^0 can acquire a nonzero vev $v_i \equiv \langle \phi_i \rangle$ in order to preserve electromagnetism. As the electric charge is given by $Q_{\phi} = T_i^3 + Y_i/2$, it goes to show that $Y_i = -2T_i^3$ for each participating scalar, where T_i^3 is the isospin of the electrically neutral component ϕ_i^0 . For reference, in the SM, the Higgs field Φ is an $SU(2)_L$ double with isospin charge T = 1/2; its electrically neutral component ϕ^0 has isospin $T^3 = -1/2$ and vev

$$v = \sqrt{2} \langle \Phi \rangle = \langle \phi^0 \rangle, \tag{1.210}$$

where

$$\Phi = \begin{pmatrix} \phi^+ \\ \frac{1}{\sqrt{2}} \left(\phi^0 + i \phi^3 \right) \end{pmatrix}.$$
(1.211)

The covariant derivate acting on a generic Φ_i at its minimum is then

$$D_{\mu}\Phi_{i} \to D_{\mu}v_{i} = \left[\frac{ig_{W}}{\sqrt{2}}\left(W_{\mu}^{+}\hat{T}_{i}^{+} + W_{\mu}^{-}\hat{T}_{i}^{-}\right) + ig_{Z}Z_{\mu}\hat{T}_{i}^{3}\right]v_{i}, \qquad (1.212)$$

which implies a kinetic term

$$(D^{\mu}\Phi_{i})^{\dagger}(D_{\mu}\Phi_{i}) \to (D^{\mu}v_{i})^{\dagger}(D_{\mu}v_{i}) = \left[\frac{g_{W}^{2}}{2}W^{\mu-}W^{+}_{\mu}\left(\hat{T}_{i}^{-}\hat{T}_{i}^{+}+\hat{T}_{i}^{+}\hat{T}_{i}^{-}\right)+g_{Z}^{2}\left(\hat{T}_{i}^{3}\right)^{2}Z^{\mu}Z_{\mu}\right]v_{i}^{2}$$

Following the usual ladder operator algebra, we have

$$\hat{T}_i^{\pm} \hat{T}_i^{\mp} |v_i\rangle = \hat{T}_i^{\pm} \hat{T}_i^{\mp} |T_i, T_i^3\rangle$$
(1.213)

$$= \sqrt{T_i(T_i+1) - T_i^3(T_i^3 \mp 1)} \hat{T}_i^{\pm} |T_i, T_i^3 \mp 1\rangle$$
(1.214)

$$= \sqrt{T_i(T_i+1) - T_i^3(T_i^3 \mp 1)} \sqrt{T_i(T_i+1) - (T_i^3 \mp 1)T_i^3} |T_i, T_i^3\rangle$$
(1.215)

$$= \left[T_i(T_i+1) - T_i^3(T_i^3 \mp 1)\right] |T_i, T_i^3\rangle, \qquad (1.216)$$

indicating that for the W boson, we have

$$\left(\hat{T}_{i}^{-}\hat{T}_{i}^{+}+\hat{T}_{i}^{+}\hat{T}_{i}^{-}\right)v_{i}=2\left[T_{i}(T_{i}+1)-(T_{i}^{3})^{2}\right]v_{i}.$$
(1.217)

Similarly,

$$(\hat{T}_i^3)^2 v_i = (T_i^3)^2 v_i. \tag{1.218}$$

The kinetic term for ϕ now simplifies to

$$(D^{\mu}v_{i})^{\dagger}(D_{\mu}v_{i}) = \left[g_{W}^{2}v_{i}^{2}\left(T_{i}(T_{i}+1)-(T_{i}^{3})^{2}\right)W^{\mu-}W^{+}_{\mu}+g_{Z}^{2}v_{i}(T_{i}^{3})^{2}Z^{\mu}Z_{\mu}\right].$$
 (1.219)

Summing over all vev-acquiring fields Φ_i , the total kinetic term gives

$$\mathcal{L}_{\text{Kinetic}} = \sum_{i} (D^{\mu} \Phi_{i})^{\dagger} (D_{\mu} \Phi_{i}) \ni \sum_{i} (D^{\mu} \phi_{i}^{0})^{\dagger} (D_{\mu} \phi_{i}^{0}) \to \sum_{i} (D^{\mu} v_{i})^{\dagger} (D_{\mu} v_{i})$$
(1.220)

$$= \underbrace{\sum_{i} \left[g_{W}^{2} v_{i}^{2} (T_{i}(T_{i}+1) - (T_{i}^{3})^{2}) \right] W^{\mu -} W^{+}_{\mu} + \frac{1}{2} \underbrace{2 \sum_{i} \left[g_{Z}^{2} v_{i}^{2} (T_{i}^{3})^{2} \right] Z^{\mu} Z_{\mu}, \quad (1.221)}_{M_{W}^{2}}$$

giving us expressions for M_W and M_Z in terms of the various v_i and isospins

$$M_W^2 = \sum_i g_W^2 v_i^2 \left(T_i (T_i + 1) - (T_i^3)^2 \right), \qquad (1.222)$$

$$M_Z^2 = 2\sum_i g_Z^2 v_i^2 (T_i^3)^2. (1.223)$$

The ρ -parameter for an arbitrary number of Higgs doublets is then given by

$$\rho \equiv \frac{M_W^2}{M_Z^2 \cos^2 \theta_W} = \frac{\sum_i \left[g_W^2 v_i^2 (T_i (T_i + 1) - (T_i^3)^2) \right]}{2 \sum_i \left[g_Z^2 v_i (T_i^3)^2 \right] \cos^2 \theta_W}$$
(1.224)

$$= \frac{\sum_{i} v_{i}^{2} \left[T_{i}(T_{i}+1) - (T_{i}^{3})^{2} \right]}{2 \sum_{i} v_{i}^{2} (T_{i}^{3})^{2}}$$
(1.225)

As M_W , M_Z , and $\cos \theta_W$ in Eq. (1.208) can be measured independently, ρ represents a highprecision into the EW sector and the origin of EWSB. Accounting for smalls radiative corrections, labeled by $\hat{\rho}$, the best measurement for ρ is given by [25]

$$\rho_0 \equiv \frac{\rho}{\hat{\rho}} = \frac{M_W^2}{M_Z^2 \cos^2 \theta_W \hat{\rho}} = 1.00040 \pm 0.00024, \qquad (1.226)$$

and is consistent with the SM at 1.67σ . An proxy test of custodial symmetry is measuring the branching fraction ratios of Higgs boson decays to weak bosons.

As we saw in Eq. (1.180) as well as in Fig. 3, the three-point hVV and four-point hhVV couplings are proportional to the amount of mixing between the isospin and hypercharge bosons. In the vanishing g_Y limit, the two couplings for WW and ZZ become identical.

$$\lambda_{WZ} = \frac{\mathrm{BR}\left(H \to WW\right)}{\mathrm{BR}^{\mathrm{SM}}\left(H \to WW\right)} \times \frac{\mathrm{BR}^{\mathrm{SM}}\left(H \to ZZ\right)}{\mathrm{BR}\left(H \to ZZ\right)} \sim \frac{g_{hWW}^2}{g_{hWW}^2} \times \frac{g_{hZZ}^{2\,\mathrm{SM}}}{g_{hZZ}^2}$$
(1.227)

Measurements by ATLAS [29] and CMS [30] find it consistent with SM prediction of 1

$$\lambda_{WZ}^{\text{ATLAS}} = 0.81_{-0.15}^{+0.16}, \qquad (1.228)$$

$$\lambda_{WZ}^{\rm CMS} = 0.94_{-0.18}^{0.22}. \tag{1.229}$$

1.8 FERMION SECTOR OF THE STANDARD MODEL

The SM is a theory of massless, chiral fermion that are coupled through Yukawa interactions and interact via the exchange of gauge bosons. We now introduce the SM fermionic sector, their gauge and Yukawa interactions, and their spontaneous generation of mass.

1.8.1 Fermion Content

We first denote the LH (RH) components of a Dirac fermion ψ by

$$\psi_{L(R)} \equiv P_{L(R)}\psi \equiv \frac{1}{2}(1_4 \mp \gamma^5)\psi,$$
 (1.230)

where $P_{L(R)}$ is the chiral projection operator, and under charge conjugation one has

$$\psi_L^c \equiv (\psi^c)_L = (\psi_R)^c. \tag{1.231}$$

All known (anti)fermionic states that are gauged under a non-Abelian group are charged in a (anti)fundamental representation of the gauge group. The absoluteness of this statement is of much interest and speculation.

The fermionic content of the SM consists of the LH states

$$Q_L^{\alpha I} = \begin{pmatrix} u_L^{\alpha I} \\ d_L^{\alpha I} \end{pmatrix}, \quad L_L^I = \begin{pmatrix} \nu_L^I \\ e_L^I \end{pmatrix}, \quad (1.232)$$

and the RH states

$$u_R^{\alpha I}, \quad d_R^{\alpha I}, \quad e_R^I. \tag{1.233}$$

The LH objects are arranged to make manifest that they satisfy an SU(2) (weak isospin) gauge symmetry. The lowercase Greek index $\alpha = 1, \ldots, N_c = 3$ denote SU(3) (color) indices. The capital Roman index $I = 1, \ldots 3$ represent that there are three copies of these fields called *generations*, or sometimes *families*. The ordering is such that generation-*n* fields have smaller Yukawa couplings

Species	Symbol	$\mathrm{SU}(3)_C \times \mathrm{SU}(2)_L \times \mathrm{U}(1)_Y$ Rep.	$U(1)_{EM}$ Charge [Units of $e > 0$]
Quark	$Q_L = \begin{pmatrix} u_L \\ d_L \end{pmatrix}$	$(3,2,+rac{1}{3})$	$\begin{pmatrix} +2/3\\ -1/3 \end{pmatrix}$
Quark	u_R	$({f 3},{f 1},+{4\over3})$	+2/3
Quark	d_R	$({f 3},{f 1},-{2\over 3})$	-1/3
Lepton	$L_L = \begin{pmatrix} \nu_L \\ e_L \end{pmatrix}$	(1, 2, -1)	$\begin{pmatrix} 0\\ -1 \end{pmatrix}$
Lepton	e_R	(1, 2, -2)	-1

Table 3: Matter Content of the Standard Model

(masses) than generation-(n + 1) fields. Despite the wide body of literature, and despite its suggestive structure, presently there is no confirmed "theory of generations". Measurements of Higgs boson properties indicate that additional generations, if they exist and obtain their from the Higgs fields, must be very massive [31]. For a fixed generation, each of the seven fields posses a unique charge under $U(1)_Y$ hypercharge. However, for a fixed generation, the sum of all hypercharges is identically zero, thereby rendering it "anomaly free" [6]. As gauge quantum number assignments are independent of generation, this cancellation holds for each generation.

The SU(3)-colored fields Q_L , u_R , d_R are called *quarks*, and the SU(3)-neutral fermions L_L , e_R are the *leptons*. Leptons are further categorized into (electrically) *charged leptons*

$$e_L^I, \quad e_R^I, \tag{1.234}$$

and (electrically) neutral leptons or neutrinos

 $\nu_L^I.$ (1.235)

Though not used there, the notation

$$q_L^{\alpha iI}, \quad \ell_L^{iI}, \tag{1.236}$$

is very often found to denote LH quark and lepton doublets with $SU(2)_L$ index i = 1, 2.

The identity of each of the four LH fields in Eq. (1.232) is referred to as *flavor*. Accounting for three generations, there are 12 flavors in total. The RH analogs of Eq. (1.232), if they exist,

have the same flavor, e.g., $e_L^{I=1}$ is the LH electron and $e_R^{I=1}$ is the RH electron. When speaking of a particular particle species across generations, the qualifier *type* is used, e.g., a *u*-type or up-type quark represents

$$u_L^{\alpha,I}, \ u_R^{\alpha I} \quad \text{for} \quad I = 1, \dots, 3.$$
 (1.237)

The SM elementary fermion content is summarized in Table 3.

There are no RH neutrinos, N_R^I , in the SM.

1.8.2 Fermion Lagrangian

The last piece of the SM Lagrangian [Eq. (1.135)] is the fermionic contribution, given by

$$\mathcal{L}_{\text{Fermion}} = \mathcal{L}_{\text{Fermion Kin.}} - V_{\text{Yukawa}}$$
 (1.238)

where the kinetic term is given by

$$\mathcal{L}_{\text{Fermion Kin.}} = \overline{Q}_{L}^{\beta I} i \, \mathcal{D}^{\beta \alpha} Q_{L}^{\alpha I} + \overline{L}_{L}^{I} i \, \mathcal{D} L_{L}^{I} + \overline{u}_{R}^{\beta I} i \, \mathcal{D}^{\beta \alpha} u_{R}^{\alpha I} + \overline{d}_{R}^{\beta I} i \, \mathcal{D}^{\beta \alpha} d_{R}^{\alpha I} + \overline{e}_{R}^{I} i \, \mathcal{D} e_{R}^{I}$$
(1.239)

and Yukawa potential by

$$V_{\text{Yukawa}} = y_u^{JI} \overline{Q}_L^{\alpha J} \tilde{\Phi} u_R^{\alpha I} + y_d^{JI} \overline{Q}_L^{\alpha J} \Phi d_R^{\alpha I} + y_e^{JI} \overline{L}_L^J \tilde{\Phi} e_R^I + \text{H.c.}$$
(1.240)

We unpack Eq. (1.238) by first listing the covariant derivatives explicitly using Eq. (1.170):

$$\begin{split} \mathcal{P}^{\beta\alpha}Q_{L}^{\alpha} &= \left[\delta^{\beta\alpha}\partial_{\mu} + ig_{s}G_{\mu}^{a}(\hat{T}^{a})^{\beta\alpha} + \delta^{\beta\alpha}\frac{ig_{W}}{\sqrt{2}} \left(W_{\mu}^{+}\hat{T}_{L}^{+} + W_{\mu}^{-}\hat{T}_{L}^{-} \right) \right. \\ &+ \delta^{\beta\alpha}ieA_{\mu}\hat{Q} + \delta^{\beta\alpha}ig_{Z}Z_{\mu} \left(\hat{T}_{L}^{3} - \sin^{2}\theta_{W}\hat{Q} \right) \right] \gamma^{\mu}Q_{L}^{\alpha} \\ \mathcal{P}L_{L} &= \left[\partial_{\mu} + \frac{ig_{W}}{\sqrt{2}} \left(W_{\mu}^{+}\hat{T}_{L}^{+} + W_{\mu}^{-}\hat{T}_{L}^{-} \right) + ieA_{\mu}\hat{Q} + ig_{Z}Z_{\mu} \left(\hat{T}_{L}^{3} - \sin^{2}\theta_{W}\hat{Q} \right) \right] \gamma^{\mu}L_{L} \\ \mathcal{P}^{\beta\alpha}u_{R}^{\alpha} &= \left[\delta^{\beta\alpha}\partial_{\mu} + ig_{s}G_{\mu}^{a}(\hat{T}^{a})^{\beta\alpha} + \delta^{\beta\alpha}ieA_{\mu}\hat{Q} - \delta^{\beta\alpha}ig_{Z}\sin^{2}\theta_{W}Z_{\mu}\hat{Q} \right] \gamma^{\mu}u_{R}^{\alpha} \\ \mathcal{P}^{\beta\alpha}d_{R}^{\alpha} &= \left[\delta^{\beta\alpha}\partial_{\mu} + ig_{s}G_{\mu}^{a}(\hat{T}^{a})^{\beta\alpha} + \delta^{\beta\alpha}ieA_{\mu}\hat{Q} - \delta^{\beta\alpha}ig_{Z}\sin^{2}\theta_{W}Z_{\mu}\hat{Q} \right] \gamma^{\mu}d_{R}^{\alpha} \\ \mathcal{P}e_{R} &= \left[\partial_{\mu} + ieA_{\mu}\hat{Q} - ig_{Z}\sin^{2}\theta_{W}Z_{\mu}\hat{Q} \right] \gamma^{\mu}e_{R} \end{split}$$

The index $a = 1, ..., (N_c^2 - 1) = 8$ denotes the SU(3) color generator (in the adjoint representation). In the Yukaway potential, Φ is the scalar doublet introduced in Section 1.7. The field $\tilde{\Phi}$ is its "isospin-hypercharge" conjugate, defined by

$$\tilde{\Phi} \equiv i\sigma^2 \Phi^* = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} \phi^+ \\ \frac{1}{\sqrt{2}}(\phi^0 + i\phi^3) \end{pmatrix} = \begin{pmatrix} \frac{1}{\sqrt{2}}(\phi^0 - i\phi^3) \\ -\phi^- \end{pmatrix}.$$
(1.241)

 y_f^{JI} are the 3 × 3 Yukawa coupling matrices for a fermion species f in generation space, e.g., b- or t-type quarks. Equation (1.238) will be explored in considerable depth throughout the remaining chapters. For now, we focus on how $\mathcal{L}_{\text{Fermion}}$ changes as Φ acquires its vev.

1.8.3 Electroweak Symmetry Breaking II: Fermion Masses

To break EW symmetry in the fermion sector with the Higgs field, we follow the (by now) standard procedure of setting Φ equal to its vev and considering perturbative fluctuations (*h*) around it. We consider the up-type interaction as an example and see

$$y_u^{JI} \overline{Q}_L^{\alpha J} \tilde{\Phi} u_R^{\alpha I} = y_u^{JI} \begin{pmatrix} \overline{u}_L^{\alpha J} & \overline{d}_L^{\alpha J} \end{pmatrix} \begin{pmatrix} \frac{1}{\sqrt{2}} (v+h) \\ 0 \end{pmatrix} u_R^{\alpha I} = \frac{y_u^{JI}}{\sqrt{2}} \overline{u}_L^{\alpha J} (v+h) u_R^{\alpha I}.$$
(1.242)

In the last term of Eq. (1.242) we see two very interesting terms: (i) a three-point point coupling between left- and right-handed fields of the same species type mediated by a Higgs radiation, and (ii) a two-point coupling between left- and right-handed fields proportional to a dimensionful parameter. The second term should be identified as a *fermion mass* term that has been spontaneously by the Higgs field. Making the definition

$$m_f^{JI} \equiv y_f^{JI} \langle \Phi \rangle = \frac{y_f^{JI}}{\sqrt{2}} v \,, \tag{1.243}$$

we now have

$$y_u^{JI} \overline{Q}_L^{\alpha J} \tilde{\Phi} u_R^{\alpha I} + \text{H.c} = m_u^{JI} \overline{u}_L^{\alpha J} u_R^{\alpha I} + \frac{m_u^{JI}}{v} \overline{u}_L^{\alpha J} u_R^{\alpha I} h + \text{H.c.}$$
(1.244)

Applying this systematically, we obtain masses for all SM fermions with RH partners:

$$V_{\text{Yukawa}} = m_u^{JI} \overline{u}_L^{\alpha J} u_R^{\alpha I} + m_d^{JI} \overline{d}_L^{\alpha J} d_R^{\alpha I} + m_e^{JI} \overline{e}_L^{J} e_R^{I} + \frac{m_u^{JI}}{v} \overline{u}_L^{\alpha J} u_R^{\alpha I} h + \frac{m_d^{JI}}{v} \overline{d}_L^{\alpha J} d_R^{\alpha I} h + \frac{m_e^{JI}}{v} \overline{e}_L^{J} e_R^{I} h + \text{H.c}$$
(1.245)

Having broken EW symmetry in the fermion sector, we find ourself at another interesting junction. Our fermion Lagrangian $\mathcal{L}_{\text{Fermion}}$ was written in terms of massless chiral/gauge eigenstates. However, as the broken Lagrangian now only respects

$$\mathrm{SU}(3)_c \times \mathrm{U}(1)_{EM},\tag{1.246}$$

we may conclude that our massive fermion states as they are presently written may no longer be aligned with their mass eigenstates.

1.8.4 Quark and Lepton Mass Mixing

Like in the gauge sector, fermionic gauge states and mass eigenstates before EWSB were aligned. We no long have this luxury and must rotate our states out of the gauge basis in order to obtain mass eigenstates, which are necessary to discuss particle scattering.

Generically, we may decompose our LH and RH chiral fields into mass eigenstates with a unitary transformation:

$$\begin{pmatrix}
u^{1} \\
u^{2} \\
u^{3}
\end{pmatrix}_{\underline{L(R)}} = U_{L(R)} \begin{pmatrix}
u \\
c \\
t
\end{pmatrix}_{\underline{L(R)}}, \qquad \begin{pmatrix}
d^{1} \\
d^{2} \\
d^{3}
\end{pmatrix}_{\underline{L(R)}} = D_{L(R)} \begin{pmatrix}
d \\
s \\
b
\end{pmatrix}_{\underline{L(R)}}, \qquad (1.247)$$
Chiral Basis Chiral Basis Chiral Basis (1.248)
$$\begin{pmatrix}
e^{1} \\
e^{2} \\
e^{3}
\end{pmatrix}_{\underline{L(R)}} = E_{L(R)} \begin{pmatrix}
e \\
\mu \\
\tau
\end{pmatrix}_{\underline{L(R)}}, \qquad (1.248)$$

Our mass and Yukawa matrices m_f^{JI} and y_f^{JI} can now be diagonalized

$$\mathcal{M}_{u} = U_{L}^{-1} m_{u} U_{R} = \begin{pmatrix} m_{u} & 0 & 0\\ 0 & m_{c} & 0\\ 0 & 0 & m_{t} \end{pmatrix} = \frac{v}{\sqrt{2}} \begin{pmatrix} y_{u} & 0 & 0\\ 0 & y_{c} & 0\\ 0 & 0 & y_{t} \end{pmatrix}$$
(1.249)
$$\begin{pmatrix} m_{d} & 0 & 0\\ 0 & 0 & y_{t} \end{pmatrix}$$

$$\mathcal{M}_{d} = D_{L}^{-1} m_{d} D_{R} = \begin{pmatrix} m_{d} & 0 & 0 \\ 0 & m_{s} & 0 \\ 0 & 0 & m_{b} \end{pmatrix} = \frac{v}{\sqrt{2}} \begin{pmatrix} g_{d} & 0 & 0 \\ 0 & y_{s} & 0 \\ 0 & 0 & y_{b} \end{pmatrix}$$
(1.250)

$$\mathcal{M}_{e} = E_{L}^{-1} m_{e} E_{R} = \begin{pmatrix} m_{e} & 0 & 0\\ 0 & m_{\mu} & 0\\ 0 & 0 & m_{\tau} \end{pmatrix} = \frac{v}{\sqrt{2}} \begin{pmatrix} y_{e} & 0 & 0\\ 0 & y_{\mu} & 0\\ 0 & 0 & y_{\tau} \end{pmatrix}.$$
 (1.251)

This allows us to rewrite the Yukawa interactions and mass terms of Eq. (1.245) compactly as

$$V_{\text{Yukawa}} = m_f \overline{f}_L f_R + \frac{m_f}{v} \overline{f}_L f_R h + \text{H.c}, \quad m_f = \frac{y_f v}{\sqrt{2}}, \quad (1.252)$$

Quark Generation					
Ι		II		III	
Species	Mass [MeV]	Species	Mass [GeV]	Species	Mass [GeV]
Up (u)	$2.3^{+0.7}_{-0.5}$	Charm (c)	1.275 ± 0.025	Top (t)	173.21 ± 0.51
Down (d)	$4.8_{-0.3}^{+0.5}$	Strange (s)	0.95 ± 0.05	Bottom (b)	4.18 ± 0.03
Lepton Generation					
Ι		II		III	
Species	Mass	Species	Mass	Species	Mass
Electron-		Muon-	< 0.10 MeV	Tau-	$< 18.2 { m MeV}$
Neutrino (ν_e)	< 2 ev	Neutrino (ν_{μ})	< 0.19 MeV	Neutrino (ν_{τ})	
Electron (e)	$510.998928 \pm$	Muon (μ)	$105.6583715 \pm$	Tau (τ)	$1.77682 \pm$
	$0.000011 { m ~KeV}$		$0.0000035 { m ~MeV}$		$0.00016~{\rm GeV}$

 Table 4: Masses of the Elementary Standard Model Fermions
 [25]

for f = u, d, c, s, t, b, e, μ , τ , and repeated color indices are implicit for quarks. In Table 4, we summarize the SM fermion content after EWSB, in the mass eigenbasis, along with the most precise measurements of their masses presently available.

At this point, it is worth noting that as the Higgs-vector boson couplings originate from the Higgs kinetic term and as Higgs-fermion couplings originate from the Yukawa potential, which are subtracted from kinetic terms in the Lagrangian formalism. Thus, the hVV and hff couplings differ by a relative minus sign. An analysis of $h \rightarrow \gamma\gamma$ decays, which is mediated at LO by the interference between a W boson and t quark loop, and therefore sensitive to this sign difference, concludes that Higgs boson data is consistent with the SM description [32].

For QCD, QED, and Z interactions, as the coupling vertex have the structure

$$\overline{u}_{L}^{\alpha}\gamma^{\mu}u_{L} = \overline{u}_{L}^{'\alpha}U_{L}^{-1}\gamma^{\mu}U_{L}u_{L}^{'} = \overline{u}_{L}^{'\alpha}\gamma^{\mu}u_{L}^{'}, \qquad (1.253)$$

we see that the currents are *flavor-conserving*, and that the interaction basis is still aligned with the mass basis. Thus, they need not be discussed further. We now explore what consequences this rotation in *flavor space* has on our charged current interactions. Recalling the covariant derivatives from above, we have

$$\overline{Q}_{L}^{\beta} i \, \mathcal{D}^{\beta \alpha I} Q_{L}^{\alpha} \quad \ni \quad \overline{Q}_{L}^{\alpha} i \left[\delta^{\beta \alpha} \frac{ig_{W}}{\sqrt{2}} \left(W_{\mu}^{+} \hat{T}_{L}^{+} + W_{\mu}^{-} \hat{T}_{L}^{-} \right) \right] \gamma^{\mu} Q_{L}^{\alpha} \tag{1.254}$$

$$= \delta^{\beta\alpha} \frac{-g_W}{\sqrt{2}} \left(W^+_\mu \overline{u}^\beta_L \gamma^\mu d^\alpha_L + W^-_\mu \overline{d}^\beta_L \gamma^\mu u^\alpha_L \right).$$
(1.255)

Rotating into the mass basis, we get

$$\overline{Q}_{L}^{\beta} i \, \mathcal{D}^{\beta\alpha I} Q_{L}^{\alpha} \quad \ni \quad \delta^{\beta\alpha} \frac{-g_{W}}{\sqrt{2}} \left(W_{\mu}^{+} \overline{u'}_{L}^{\beta} \gamma^{\mu} \left(U_{L}^{-1} D_{L} \right) d_{L}^{\prime \alpha} + W_{\mu}^{-} \overline{d'}_{L}^{\beta} \gamma^{\mu} \left(D_{L}^{-1} U_{L} \right) u_{L}^{\prime \alpha} \right). \quad (1.256)$$

Defining the matrix V such that

$$V \equiv U_L^{-1} D_L, \qquad (1.257)$$

our *flavor-changing charged currents* with (small) intergenerational mixing are governed by

$$\overline{Q}_{L}^{\beta} i \not\!\!\!D^{\beta\alpha I} Q_{L}^{\alpha} \quad \ni \quad \delta^{\beta\alpha} \frac{-g_{W}}{\sqrt{2}} \left(W_{\mu}^{+} \overline{u}_{L}^{\beta} \gamma^{\mu} V_{ud} d_{L}^{\alpha} + W_{\mu}^{-} \overline{d}_{L}^{\beta} \gamma^{\mu} V_{ud}^{\dagger} u_{L}^{\alpha} \right), \tag{1.258}$$

where u_L and d_L now represents mass eigenstates. V is the Cabbibo-Kobayashi-Maskawa (CKM) matrix [33,34]. As a 3×3 unitary matrix, it expressible by three angles and a phase:

$$\begin{split} V^{\text{CKM}} &= \begin{pmatrix} V_{ud} & V_{us} & V_{ub} \\ V_{cd} & V_{cs} & V_{cb} \\ V_{td} & V_{ts} & V_{tb} \end{pmatrix} \\ &= \begin{pmatrix} 1 & 0 & 0 \\ 0 & \cos\theta_{23} & \sin\theta_{23} \\ 0 & -\sin\theta_{23} & \cos\theta_{23} \end{pmatrix} \begin{pmatrix} \cos\theta_{13} & 0 & \sin\theta_{13}e^{-i\delta_{13}} \\ 0 & 1 & 0 \\ -\sin\theta_{13}e^{-i\delta_{13}} & 0 & \cos\theta_{13} \end{pmatrix} \begin{pmatrix} \cos\theta_{12} & \sin\theta_{12} & 0 \\ -\sin\theta_{12} & \cos\theta_{12} & 0 \\ 0 & 0 & 1 \end{pmatrix}. \end{split}$$

The presence of the complex phase $\delta_{13} \neq 0$ in ^{CKM} indicates *CP* violation in weak interactions. The best measurements available at the time of this writing of V^{CKM} are given in Table 5 [25]. Other useful parameterizations can also be found in Ref. [25] and references within.

Gauge invariance bars gluons and photons to undergo flavor changing neutral currents (FCNCs), even at the higher orders of perturbation. However, as the Z is not associated with a good local symmetry, off-diagonal elements of $U_L^{-1}U_L$ may be generated at the loop-level, a process known as the Glashow-Iliopoulos-Maiani (GIM) mechanism [35]. These FCNCs processes, however, face both coupling and phase space suppression.

Table 5: Components of CKM Matrix

$ V_{ud} = 0.97425 \pm 0.00022$	$ V_{us} = 0.2253 \pm 0.0008$	$ V_{ub} = (4.13 \pm 0.49) \times 10^{-3}$		
$ V_{cd} = 0.225 \pm 0.008$	$ V_{cs} = 0.986 \pm 0.016$	$ V_{cb} = (41.1 \pm 1.3) \times 10^{-3}$		
$ V_{td} = (8.4 \pm 0.6) \times 10^{-3}$	$ V_{ts} = (40.0 \pm 2.7) \times 10^{-3}$	$ V_{tb} = 1.021 \pm 0.032$		
Jarlskog invariant $J = 3.06^{+0.21}_{-0.20} \times 10^{-5}$				

1.9 BEYOND THE STANDARD MODEL

In this chapter we have introduced global and local continuous symmetries as well as their spontaneous breakdown via the acquiring of a nonzero vacuum expectation value by a scalar field. Building on these principles, we constructed the Standard Model of particle physics. However, despite the SM's experimental success, it remains an unsatisfactory description of nature. The existence of nonzero neutrino masses, dark matter, a large hierarchy among fermion masses, and a Higgs boson whose mass is unstable under radiative corrections highlight the theory's shortcomings. Extensions of the SM that alleviate these issues vary in size and scope, but commonly predict, among new principles and symmetries, the existence of new gauge bosons (e.g., Left-Right Symmetry), new scalars (e.g., Supersymmetry), and new fermions (e.g., Seesaw Mechanisms). In the following chapters, we will explore several such models, including more phenomenological, semi-model-independent approaches, and derive testable consequences.

2.0 PRINCIPLES OF COLLIDER PHYSICS

2.1 INTRODUCTION

Collider physics and phenomenology explore the manifestation of the SM in high energy and high momentum transfer scattering experiments. It is a deeply rich and enjoyable subject that incorporates perturbative, non-perturbative (all-orders summed and effective field theory), and computational techniques in order to simulate with a reasonably high degree of accuracy the results of lepton-lepton, lepton-hadron, and hadron-hadron collisions. In this chapter, we introduce many fundamental topics of collider physics. Many excellent texts on the topic are available, in particular the classic Barger & Phillips [36] as well as lectures by Han [37] and Willenbrock [38]. The texts Halzen & Martin [24] and Thomson [39] provide an excellent introduction to the field, providing an inordinately large number of useful examples.

2.2 HELICITY AMPLITUDES

We start our study of collider phenomenology with the introduction of helicity amplitudes and helicity eigenstates for representations of the Lorentz group. The theory of scalars, spin one-half fermions, and vector bosons as irreducible representations of the Lorentz group is a very important topic. A rigorous construction from first principles can be found in Weinberg [40]. We now briefly review spin one-half fermions and spin-one bosons. As spin-zero bosons are a trivial representations of the Lorentz group, they transform as scalars; no review of their properties is needed.

2.2.1 Spin One-Half Fermions

To construct the explicit forms of Dirac spinors in the helicity basis, we suppose a fermion propagating in the direction \hat{p}_0 relative to its quantized spin axis. Along this direction, the helicity operator Σ is defined as

$$\Sigma \equiv \sigma \cdot \hat{p}_0 = \begin{pmatrix} \hat{p}_0^3 & \hat{p}_0^1 - i\hat{p}_0^2 \\ \hat{p}_0^1 + i\hat{p}_0^2 & -\hat{p}_0^3 \end{pmatrix} = \begin{pmatrix} \cos\theta & e^{-i\phi}\sin\theta \\ e^{i\phi}\sin\theta & -\cos\theta \end{pmatrix}.$$
 (2.1)

The corresponding helicity eigenstates are the two-component solutions $\chi_{\lambda}(\hat{p}_0)$ to the relationship

$$\Sigma \chi_{\lambda}(\hat{p}_0) = \lambda \chi_{\lambda}(\hat{p}_0), \qquad (2.2)$$

the eigenvalues of which are $\lambda = \pm 1$ and twice the fermion's actual helicity. Conventionally, when the direction of propagation is (anti-)parallel to the spin axis, which results in the eigenvalue $\lambda = (-)1$, we refer to the state as being in its (left-) right-handed helicity eigenstate.

Fixing the spin quantization axis with a definite direction, say \hat{z} , the fermion is aligned with its spin axis when $\hat{p}_0 = \pm \hat{z}$. In such a situation, the four solutions to Eq. (2.2) are

$$\chi_{\lambda=+1}(\hat{z}) = \begin{pmatrix} 1\\ 0 \end{pmatrix}, \quad \chi_{\lambda=-1}(\hat{z}) = \begin{pmatrix} 0\\ 1 \end{pmatrix}, \quad \chi_{\lambda=+1}(-\hat{z}) = \begin{pmatrix} 0\\ 1 \end{pmatrix}, \quad \chi_{\lambda=-1}(-\hat{z}) = \begin{pmatrix} -1\\ 0 \end{pmatrix}. \quad (2.3)$$

Boosting our fermion to an arbitrary reference frame

$$p^{\mu} = (E_0, 0, 0, |\vec{p}_0|) \to p'^{\mu} = p^{\mu} = (E, |\vec{p}| \sin \theta \cos \phi, |\vec{p}| \sin \theta \sin \phi, |\vec{p}| \cos \theta), \quad E^2 = |\vec{p}|^2 + m^2 \quad (2.4)$$

the two-component eigenstates are

$$\chi_{\lambda=+1}(\hat{p}) = \frac{1}{\sqrt{2|\vec{p}|(|\vec{p}|+p_z)}} \begin{pmatrix} |\vec{p}|+p_z\\ p_x+ip_y \end{pmatrix} = \begin{pmatrix} \cos\frac{\theta}{2}\\ e^{i\phi}\sin\frac{\theta}{2} \end{pmatrix}$$
(2.5)

$$\chi_{\lambda=-1}(\hat{p}) = \frac{1}{\sqrt{2|\vec{p}|(|\vec{p}|+p_z)}} \begin{pmatrix} -p_x + ip_y \\ |\vec{p}|+p_z \end{pmatrix} = \begin{pmatrix} -e^{-i\phi}\sin\frac{\theta}{2} \\ \cos\frac{\theta}{2} \end{pmatrix}$$
(2.6)

$$\chi_{\lambda}(-\hat{p}) = -\lambda e^{i\lambda\phi} \chi_{-\lambda}(\hat{p}).$$
(2.7)

The four-component Dirac spinors for a fermion (u_{λ}) and antifermion (v_{λ}) can now be constructed:

$$u_{\lambda}(p) = \begin{pmatrix} \sqrt{E - \lambda |\vec{p}|} \chi_{\lambda}(\hat{p}) \\ \sqrt{E + \lambda |\vec{p}|} \chi_{\lambda}(\hat{p}) \end{pmatrix}, \quad v_{\lambda}(p) = \begin{pmatrix} -\lambda \sqrt{E + \lambda |\vec{p}|} \chi_{-\lambda}(\hat{p}) \\ \lambda \sqrt{E - \lambda |\vec{p}|} \chi_{-\lambda}(\hat{p}) \end{pmatrix}.$$
(2.8)

2.2.1.1 Properties of Dirac Spinors In the high-energy limit, when $E \gg m$, degrees of freedom are decoupled and the Dirac spinors simplify to

$$u_{\lambda=+1}(p) \approx \sqrt{2E} \begin{pmatrix} 0\\ \chi_{\lambda=+1}(\hat{p}) \end{pmatrix} \qquad u_{\lambda=-1}(p) \approx \sqrt{2E} \begin{pmatrix} \chi_{\lambda=-1}(\hat{p})\\ 0 \end{pmatrix}$$
(2.9)

$$v_{\lambda=+1}(p) \approx -\sqrt{2E} \begin{pmatrix} \chi_{\lambda=-1}(\hat{p}) \\ 0 \end{pmatrix}. \qquad v_{\lambda=-1}(p) \approx \sqrt{2E} \begin{pmatrix} 0 \\ \chi_{\lambda=+1}(\hat{p}) \end{pmatrix}.$$
(2.10)

In this limit, u_{λ} and v_{λ} are found to be eigenstates of the chiral projection operators

$$P_L = \frac{1}{2}(1 - \gamma^5), \text{ and } P_R = \frac{1}{2}(1 + \gamma^5)$$
 (2.11)

In particular, the LH fermion and RH antifermion helicity states are LH chiral states

$$P_L u_{\lambda=-1}, \ v_{\lambda=+1} = u_{\lambda=-1}, \ v_{\lambda=+1};$$
 (2.12)

and the RH fermion and LH antifermion helicity states are RH chiral states

$$P_R u_{\lambda=+1}, v_{\lambda=-1} = u_{\lambda=-1}, v_{\lambda=+1}.$$
 (2.13)

It is in this limit that chirality and helicity become equivalent.

2.2.2 Spin-One Vector Bosons

Massive and massless vector bosons are of central importance to broken and unbroken gauge theories, and QFTs in general. Indeed, a non-Abelian gauge theory without additional fermions or scalars represents an entirely nontrivial, self-consistent and self-contained theory with predictive scattering rates. Consider a vector boson with mass M_V and momentum

$$k^{\mu} = (E, k_x, k_y, k_z) = (E, |\vec{k}| \sin \theta \cos \phi, |\vec{k}| \sin \theta \sin \phi, |\vec{k}| \cos \theta)$$
(2.14)

$$= (E, k_T \cos \phi, k_T \sin \phi, |\vec{k}| \cos \theta), \quad E^2 = M_V^2 + |\vec{k}|^2.$$
(2.15)

The transverse momentum is defined by

$$k_T = \sqrt{k_x^2 + k_y^2} = |\vec{k}| \sin \theta.$$
 (2.16)

In the $M_V \to 0$ limit, we have

$$k^{\mu} = (E, k_x, k_y, k_z) = E(1, \sin\theta\cos\phi, \sin\theta\sin\phi, \cos\theta)$$
(2.17)

$$= (E, E_T \cos \phi, E_T \sin \phi, E \cos \theta), \quad E_T = E \sin \theta, \quad (2.18)$$

with transverse energy E_T .

The polarization vectors in the Cartesian representation are given by

$$\varepsilon^{\mu}(k,x) = \frac{1}{|\vec{k}|k_T} (0, k_x k_y, k_y k_z, -k_T^2)$$
(2.19)

$$\varepsilon^{\mu}(k,y) = \frac{1}{k_T}(0, -k_y, k_x, 0)$$
 (2.20)

$$\varepsilon^{\mu}(k,z) = \frac{E}{M_V |\vec{k}|^2} \left(\frac{|\vec{k}|}{E}, k_x, k_y, k_z\right)$$
(2.21)

Checking, we have that the expected orthogonal relationships

$$k_{\mu}\varepsilon^{\mu}(k,x) = \frac{1}{|\vec{k}|k_{T}} (\underbrace{0 - k_{x}^{2}k_{y} - k_{y}^{2}k_{z}}_{-k_{T}^{2}k_{z}} + k_{z}k_{T}^{2}) = 0$$
(2.22)

$$k_{\mu}\varepsilon^{\mu}(k,y) = \frac{1}{k_{T}}\left(0 + k_{x}k_{y} - k_{y}k_{x} + 0\right) = 0$$
(2.23)

$$k_{\mu}\varepsilon^{\mu}(k,z) = \frac{E}{M_{V}|\vec{k}|} \left(\frac{|\vec{k}|^{2}}{E}E - k_{x}^{2} - k_{y}^{2} + k_{z}^{2}\right) = 0$$
(2.24)

In the polar representation, the right- $(\lambda = +1)$, left- $(\lambda = -1)$, and longitudinal $(\lambda = 0)$ polarization vectors are

$$\varepsilon^{\mu}(k,\lambda=\pm) = \frac{1}{\sqrt{2}} \left(\mp \varepsilon^{\mu}(k,x) - i\varepsilon^{\mu}(k,y) \right), \qquad (2.25)$$

$$\varepsilon^{\mu}(k,\lambda=0) = \varepsilon^{\mu}(k,z). \tag{2.26}$$

As these are (at most) linear redefinitions of the Cartesian polarization vectors, inner product relationships hold. For massless vector bosons, there are no longitudinal polarization states.

2.2.3 Decay of Heavy Fermionic Top Quark Partner

Hypothetical, TeV-scale top quark partners represent an excellent example that highlights the differences between chirality and helicity, as well as the interesting roles played by the transverse and longitudinal polarizations of gauge bosons. Such particles are proposed to cancel the large quadratic corrections the SM Higgs' mass receives from the top quark at 1-loop. In these models, the SM gauge state t_L is decomposed into light (~ 173 GeV) and heavy ($\gtrsim 1$ TeV) mass eigenstates, denoted as t and T, respectively. The alignment of the top quark gauge state and the mass eigenstates can be (phenomenologically) parameterized by the angle θ_t :

$$\underbrace{t_L}_{\text{Gauge Basis}} \simeq \underbrace{\cos \theta_t t + \sin \theta_t T}_{\text{Mass Basis}}.$$
(2.27)

The corresponding charged current Feynman rules are then given by

$$tWb : \frac{-ig}{\sqrt{2}}V_{tb}^*\cos\theta_t \tag{2.28}$$

$$TWb : \frac{-ig}{\sqrt{2}} V_{tb}^* \sin \theta_t \tag{2.29}$$

Qualitatively, $\theta_t \sim m_t/m_T$, and thus in the large m_T limit T decouples from the SM.

For the heavy top partner decay into a massive SM W boson and bottom quark,

$$T_{\tau}(p_T) \to W^+_{\lambda}(p_W) + b_{\tau'}(p_b), \qquad (2.30)$$

with helicities $\tau, \tau' = L, R$ and polarizations $\lambda = \pm, 0$, the helicity amplitudes are generically

$$\mathcal{M}_{\lambda\tau'\tau} = \frac{-ig}{\sqrt{2}} V_{tb}^* \cos\theta_t \ [\overline{u}_{\tau'}(p_b) \overline{\varphi}^*_{\lambda}(p_W) P_L u_{\tau}(p_T)]. \tag{2.31}$$

In the rest frame of T, the momenta are

$$p_T = (m_T, \vec{0}) \tag{2.32}$$

$$p_b = (E_b, |\vec{p}_b| \sin \theta \cos \phi, |\vec{p}_b| \sin \theta \sin \phi, |\vec{p}_b| \cos \theta)$$
(2.33)

$$p_W = (E_W, -|\vec{p}_b|\sin\theta\cos\phi, -|\vec{p}_b|\sin\theta\sin\phi, -|\vec{p}_b|\cos\theta)$$
(2.34)

$$E_b = \frac{m_T}{2}(1+r_b-r_W), \quad |\vec{p}_b| = \frac{m_T}{2}(1-r_b-r_W), \quad E_W = \frac{m_T}{2}(1+r_W-r_b), \quad (2.35)$$

where $r_X = m_X^2/m_T^2$. Omitting a universal factor of $(-igV_{tb}^*\cos\theta_t/\sqrt{2})$, the orthogonal helicity amplitudes for transverse W bosons are

$$\mathcal{M}_{-LL} = m_T \sqrt{2(1-r_W)} \sin \frac{\theta}{2}, \qquad \mathcal{M}_{-LR} = m_T \sqrt{2(1-r_W)} e^{i\phi} \cos \frac{\theta}{2}$$
(2.36)

$$\mathcal{M}_{+RL} = -mT\sqrt{2r_b}e^{-i\phi}\cos\frac{\theta}{2} \qquad \qquad \mathcal{M}_{+RR} = m_T\sqrt{2r_b}\sin\frac{\theta}{2} \qquad (2.37)$$

$$\mathcal{M}_{+LL} = \mathcal{M}_{-LR} = \mathcal{M}_{+LR} = \mathcal{M}_{-RR} = 0 \tag{2.38}$$

Several appreciable lessons can be learned from these expressions. In the helicity-conserving cases $(\lambda \tau' \tau) = (-LL)$ and (+RR), zero angular momentum can be carried away by the transversely polarized W when the bottom quark is aligned with its parent fermion. Thus, the amplitudes vanish as θ tends toward zero. Conversely, this is precisely when the helicity-flipping amplitudes $(\lambda \tau' \tau) = (-LR)$ and (+RL) are maximal. As W radiation is a purely LH chiral coupling, the P_L projection operator in Eq. (2.28) collects terms proportional to $m_b = m_T \sqrt{r_b}$ from RH (helicity) bottom quarks. Therefore as the bottom quark is taken massless, its chiral and helicity states align,

and contributions from RH bottom quarks, i.e., $(\lambda \tau' \tau) = (+RL)$ and (+RR), turn off. By angular momentum conservation, the remaining amplitudes for transverse W bosons are identically zero.

Omitting the same coupling factor, the helicity amplitudes for a longitudinally polarized W are

$$\mathcal{M}_{0LL} = \frac{m_T}{2} \sqrt{\frac{1 - r_W}{r_W}} \left(1 - rb + rW + \lambda^{1/2} (1, r_b, r_W) \right) \cos \frac{\theta}{2}$$
(2.39)

$$\mathcal{M}_{0LR} = -\frac{m_T}{2} \sqrt{\frac{1 - r_W}{r_W}} \left(1 - rb + rW + \lambda^{1/2} (1, r_b, r_W) \right) e^{i\phi} \sin\frac{\theta}{2}$$
(2.40)

$$\mathcal{M}_{0RL} = -\frac{m_T}{2} \sqrt{\frac{r_b}{r_W}} \left(1 - rb + rW - \lambda^{1/2} (1, r_b, r_W) \right) e^{-i\phi} \sin\frac{\theta}{2}$$
(2.41)

$$\mathcal{M}_{0RR} = -\frac{m_T}{2} \sqrt{\frac{r_b}{r_W}} \left(1 - rb + rW - \lambda^{1/2} (1, r_b, r_W) \right) \cos \frac{\theta}{2}, \tag{2.42}$$

where $\lambda(x, y, z)$ is the usual kinematic function and simplifies to

$$\lambda(1, r_b, r_W) = 1 - 2(r_b + r_W) + (r_b - r_W)^2.$$
(2.43)

The most striking feature of these amplitudes is the inverse dependence on the W boson mass, which leads to a quadratic growth with respect to m_T in the case of LH (helicity) bottom quarks. Its origin is in the zeroth component of the W polarization vector,

$$\varepsilon_{\mu=0}^{\lambda=0} = E_W / M_W \sim m_T / 2M_W.$$
 (2.44)

If m_T is the result of some Higgs-like mechanism, then it can generically be written as as the product of a Yukawa coupling and scalar vev: $m_T \sim y_T v_T$. We see now that $\varepsilon_{\mu=0} \sim m_T/2M_W \sim y_T v_T/gv$. In other words, for a fixed ratio of vevs, fermionic decays to longitudinally polarized gauge bosons is a measure of the relative coupling strength to their respective Higgs sectors. In the large m_T limit, these amplitudes become the dominant contributions to the T quark decay, a phenomenon known as *longitudinal polarization enhancement*, and has observed in SM top quark decays [41].

2.3 PHASE SPACE

Phase space, abbreviated by PS, is far-reaching concept in physics. It is the set of all allowed configurations in which a system may exist and not forbidden by a symmetry (conservation law). The volume of phase space is a measure of how many unique configurations a system possesses: more available states correspond to a larger phase space volume.

In momentum space, the n-body differential phase space with a total momentum $P_{\text{Tot.}}$ is

$$dPS_n(P_{\text{Tot}}; p_1, p_2 \dots p_n) = (2\pi)^4 \delta^4 \left(P_{\text{Tot}} - \sum_{k=1}^n p_k \right) \prod_{k=1}^n \frac{d^3 p_k}{(2\pi)^3 2E_k}$$
(2.45)

In Eq. (2.45), the Dirac function enforces momentum conservation. For the 1-, 2- and 3-body configurations, the number of d.o.f. are sufficiently constrained by momentum conservation that the differential phase space can be reasonably expressed analytically. For situations with weakly coupled (narrow width) particles propagating intermediately as well as $n \ge 4$ -body systems, it is helpful to apply the phase space recursion relationship, given in Section 2.3.5. When the former is coupled with the narrow width approximation (NWA), on-shell factorization can be applied.

For two-body processes, the Källen kinematic function, also called the " λ " function, is of considerable use and is given by

$$\lambda(x,y,z) = (x-y-z)^2 - 4yz = x^2 + y^2 + z^2 - 2xy - 2xz - 2yz.$$
(2.46)

Quite often, we deal with arguments normalized to the leading variable

$$\lambda\left(1,\frac{y}{x},\frac{z}{x}\right) = \left(1-\frac{y}{x}-\frac{z}{x}\right)^2 - \frac{4yz}{x^2} = \frac{1}{x^2}\left[(x-y-z)^2 - 4yz\right]$$
(2.47)

$$= \frac{1}{x^2}\lambda(x,y,z) \tag{2.48}$$

$$\implies \lambda^{1/2}(x,y,z) = x\lambda^{1/2}\left(1,\frac{y}{x},\frac{z}{x}\right).$$
(2.49)

Physically, for momentum p_i and $r_i = p_i^2/P_{\text{Tot.}}^2$ is the mass ratio (squared) of momentum i = 1, 2and c.m. mass $\sqrt{P_{\text{Tot.}}^2}$, $\lambda^{1/2}(1, r_1, r_2)$ can be interpreted as speed of i = 1 and i = 2 in the parent $P_{\text{Tot.}}$ frame. It is then the case that

$$|\vec{p}_i| = \frac{E_{\text{Tot}}}{2}\beta = \frac{E_{\text{Tot}}}{2}\lambda^{1/2}(1, r_1, r_2).$$
(2.50)

2.3.1 One-Body Phase Space

The one-body final state is a very special scenario because the invariant mass of the final-state momentum must equal the total c.m. energy by momentum conservation. Consider a state with 4-momentum p_1 and mass m_1 . It follows that

$$\int dPS_1 = (2\pi)^4 \int \frac{d^3p_1}{(2\pi)^3 2E_1} \,\delta^4(P_{\text{Tot.}} - p_1) \tag{2.51}$$

$$= (2\pi) \int d^4p \, \delta(p_1^2 - m_1^2) \, \delta^4(P_{\text{Tot.}} - p_1)$$
(2.52)

$$= 2\pi \,\delta(\hat{s} - m_1^2), \tag{2.53}$$

where we have applied the relationship

$$\int \frac{d^3 p_1}{2E_1} = \int d^4 p_1 \,\,\delta(p_1^2 - m_1^2). \tag{2.54}$$

and introduced the Mandelstam collider variable

$$\hat{s} = P_{\text{Tot.}}^2, \tag{2.55}$$

Quantities labeled by the caret (^), colloquially called "hat", denote *partonic* quantities within composite system. Most often this is applied to parton scattering in hadron-hadron and hadron-lepton collisions, it also applied to parton scattering in lepton-lepton collisions when objects are convolved about distribution functions; for example, see Section 2.6.

2.3.2 Solid Angle in *d* Dimension

The derivations of compact expressions for two- and three-body phase space volume elements differ little from their d-dimensional analogs. By introducing this slight but nonetheless additional complexity, we can greatly reap the benefits of having results applicable to higher order calculations. The solid angle volume element for a k-sphere in d-dimensions is given by

$$d^{d}\Omega_{k} = \left[d\theta_{1}\dots d\theta_{k-1}\right] \left[\sin^{k-2}\theta_{k-1}\dots\sin\theta_{2}\right], \quad \theta_{i} \in (0,\pi).$$
(2.56)

We separate the volume element into 2-angle and (k-2)-angle orientations to obtain

$$d^{d}\Omega_{k} = \left(d\theta_{k-1}\sin^{k-2}\theta_{k-1}\right) \left(d\theta_{k-2}\sin^{k-3}\theta_{k-2}\right) \left[d\theta_{1}\dots d\theta_{k-3}\right] \left[\sin^{k-4}\theta_{k-3}\dots\sin\theta_{2}\right] (2.57)$$
$$= \left(d\theta_{k-1}\sin^{k-2}\theta_{k-1}\right) \left(d\theta_{k-2}\sin^{k-3}\theta_{k-2}\right) \times d\Omega_{k-2}.$$
(2.58)

For a k-sphere, the integrated solid angle is

$$\Omega_{k} = \frac{2(\pi)^{k/2}}{\Gamma\left(\frac{k}{2}\right)} \stackrel{\text{E.g.}}{=} \begin{cases} 2, & \text{for } k = 1\\ 2\pi, & \text{for } k = 2\\ 4\pi, & \text{for } k = 3 \end{cases}$$
(2.59)

Integrating over the (k-2) space and relabeling our variables, we get

$$d\Omega_k = \left(d\theta \sin^{k-2}\theta\right) \left(d\phi \sin^{k-3}\phi\right) \frac{2(\pi)^{\frac{k-2}{2}}}{\Gamma\left(\frac{k-2}{2}\right)}, \quad \theta, \phi \in (0,\pi).$$
(2.60)

2.3.3 Two-Body Phase Space in d and 4 Dimensions

For a 2-body phase in d dimensions, we have

$$d^{d}PS_{2}(P_{\text{Tot}};p_{1},p_{2}) = (2\pi)^{d}\delta^{d} \left(P_{\text{Tot}} - p_{1} - p_{2}\right) \frac{d^{d-1}p_{1}}{(2\pi)^{d-1}2E_{1}} \frac{d^{d-1}p_{2}}{(2\pi)^{d-1}2E_{2}},$$
(2.61)

Integrating (without the loss of generality) over the p_2 momentum, we have

$$d^{d}PS_{2}(P_{\text{Tot}}; p_{1}, p_{2}) = \frac{\delta(u)du}{2^{2}(2\pi)^{d-2}}\delta\left(E_{\text{Tot}} - E_{1} - E_{2}\right)\frac{d|\vec{p}_{1}| |\vec{p}_{1}|^{d-2}}{E_{1}E_{2}}d\Omega_{d-1}^{p_{1}}.$$
 (2.62)

Now, defining $u = E_1 + E_2 - E_{\text{Tot}}$ and taking note that $|\vec{p_1}| = |\vec{p_2}|$, we get

$$du = \frac{|\vec{p}_1|d|\vec{p}_1|}{E_1} + \frac{|\vec{p}_1|d|\vec{p}_1|}{E_2} - 0 = \frac{|\vec{p}_1|d|\vec{p}_1|(E_{\text{Tot}} + u)}{E_1E_2}.$$
(2.63)

Plugging this into Eq. (2.61) and using the momentum- λ relationship of Eq. (2.50), we find

$$d^{d}PS_{2}(P_{\text{Tot}}; p_{1}, p_{2}) = \frac{d\Omega_{d-1}^{p_{1}}}{2^{2}(2\pi)^{d-2}} \frac{|\vec{p}_{1}|^{d-3}}{(E_{\text{Tot}} + u)} = \frac{d\Omega_{d-1}^{p_{1}}}{2^{2}(2\pi)^{d-2}} \frac{|\vec{p}_{1}|^{d-3}}{E_{\text{Tot}}}$$
(2.64)

$$= \frac{d\Omega_{d-1}^{p_1}}{2^2(2\pi)^{d-2}} \frac{E_{\text{Tot}}^{d-4}}{2^{d-3}} \lambda^{\frac{d-3}{2}}(1, r_1, r_2), \quad r_i = \frac{p_i^2}{P_{\text{Tot}}^2}.$$
 (2.65)

For the d = 4 case, this simplifies to

$$dPS_2(P_{\text{Tot}}; p_1, p_2) = \frac{d\Omega_3^{p_1}}{2(4\pi)^2} \lambda^{1/2}(1, r_1, r_2), \quad r_i = \frac{p_i^2}{P_{\text{Tot}}^2}, \quad d\Omega_3^{p_1} = d\cos\theta_1 d\phi_1 \,.$$
(2.66)

Useful, equivalent expressions include

$$dPS_2(P_{\text{Tot}}; p_1, p_2) = \frac{d\Omega_3^{p_1}}{2(4\pi)^2} \sqrt{1 - 2(r_1 + r_2) + (r_1 - r_2)^2}$$
(2.67)

$$= \frac{d\Omega_3^{p_1}}{2(4\pi)^2} \sqrt{\left[1 - (\sqrt{r_1} + \sqrt{r_2})^2\right] \left[1 - (\sqrt{r_1} - \sqrt{r_2})^2\right]}.$$
 (2.68)

In the $r_2 \rightarrow 0$ limit,

$$dPS_2(P_{\text{Tot}}; p_1, p_2) = \frac{d\Omega_3^{p_1}}{2(4\pi)^2} (1 - r_1);$$
(2.69)

and when $r_1 = r_2$, we have

$$dPS_2(P_{\text{Tot}}; p_1, p_2) = \frac{d\Omega_3^{p_1}}{2(4\pi)^2} \sqrt{1 - 4r_1}.$$
(2.70)

In four dimensions, the momenta and energies of the final-state particles in the $P_{\rm Tot}$ frame are

$$|\vec{p}_1| = |\vec{p}_2| = \frac{\sqrt{P_{\text{Tot}}^2}}{2} \lambda^{1/2}(1, r_1, r_2), \quad E_1 = \frac{\sqrt{P_{\text{Tot}}^2}}{2} (1 + r_1 - r_2), \quad E_2 = \frac{\sqrt{P_{\text{Tot}}^2}}{2} (1 - r_1 + r_2). \quad (2.71)$$

To generate the two-body phase space via Monte Carlo integration, we define y_1 , y_2 such that

$$\cos \theta = 2y_1 - 1, \quad \phi = 2\pi y_2, \quad y_1, \ y_2 \in [0, 1]. \tag{2.72}$$

Equation (2.66) then simplifies to

$$dPS_2(P_{\text{Tot}}; p_1, p_2) = \frac{dy_1 \, dy_2}{(8\pi)} \lambda^{1/2}(1, r_1, r_2) \approx \Delta y_1 \Delta y_2 \, \frac{1}{(8\pi)} \lambda^{1/2}(1, r_1, r_2).$$
(2.73)

In practice, y_1 , y_2 are randomly generated, which are used to build $\cos \theta$ and ϕ . Δy_1 , Δy_2 represent the finite volume element from which y_1 , y_2 are generated.

2.3.4 Three-Body Phase Space in d and 4 Dimensions

For the three-body case, we follow a procedure similar to the two-body situation. The phase space in d dimensions is given by

$$d^{d}PS_{3}(P_{\text{Tot}}; p_{1}, p_{2}, p_{3}) = (2\pi)^{d}\delta^{d} (P_{\text{Tot}} - p_{1} - p_{2} - p_{3}) \\ \times \frac{d^{d-1}p_{1}}{(2\pi)^{d-1}2E_{1}} \frac{d^{d-1}p_{2}}{(2\pi)^{d-1}2E_{2}} \frac{d^{d-1}p_{3}}{(2\pi)^{d-1}2E_{3}}.$$
(2.74)

As $d^{d-1}p_i/E_i$ is a boost-invariant quantity, we rotate p_2 and p_3 into the $p_{(23)} = p_2 + p_3$ rest frame. Also integrating out (without the loss of generality) p_3 by use of the δ -function, we get

$$d^{d}PS_{3}(P_{\text{Tot}}; p_{1}, p_{2}, p_{3}) = \frac{1}{2^{3}(2\pi)^{2d-3}} \delta\left(E_{\text{Tot}} - E_{1} - (E_{2} - E_{3})\right) \\ \times \frac{d|\vec{p}_{1}||\vec{p}_{1}|^{d-2}d\Omega_{d-1}^{p_{1}}}{E_{1}} \frac{d|\vec{p}_{2}^{(23)}||\vec{p}_{2}^{(23)}|^{d-2}d\Omega_{d-1}^{p_{2}^{(23)}}}{E_{2}^{(23)}E_{3}^{(23)}}$$
(2.75)

In the p_{23} -frame, we have the relationship

$$m_{23} \equiv \sqrt{p_{23}^2} = E_{\text{Tot}}^{(23)} = E_2^{(23)} + E_3^{(23)},$$
 (2.76)

which implies

$$dE_{\text{Tot}}^{(23)} = dE_2^{(23)} + dE_3^{(23)} = \frac{d|\vec{p}_2^{(23)}||\vec{p}_2^{(23)}|}{E_2^{(23)}} + \frac{|\vec{p}_2^{(23)}||\vec{p}_2^{(23)}|}{E_3^{(23)}}$$
(2.77)

$$= d|\vec{p}_{2}^{(23)}||\vec{p}_{2}^{(23)}|\left(\frac{m_{23}}{E_{2}^{(23)}E_{3}^{(23)}}\right).$$
(2.78)

We also note that in the p_{23} frame the momentum- λ relationship gives us

$$|\vec{p}_{2}^{(23)}| = \frac{m_{23}}{2} \lambda^{1/2} \left(1, \frac{p_{2}^{2}}{m_{23}^{2}}, \frac{p_{3}^{2}}{m_{23}^{2}} \right),$$
(2.79)

and more generally

$$\frac{|\vec{p}_2^{(23)}|^{d-3}}{m_{23}} = \frac{m_{23}^{d-4}}{2^{d-3}} \lambda^{\frac{d-3}{2}} \left(1, \frac{p_2^2}{m_{23}^2}, \frac{p_3^2}{m_{23}^2}\right).$$
(2.80)

Now making the appropriate substitutions in Eq. (2.75) gives us

$$d^{d}PS_{3}(P_{\text{Tot}}; p_{1}, p_{2}, p_{3}) = \frac{d\Omega_{d-1}^{p_{1}} d\Omega_{d-1}^{p_{2}^{(23)}}}{2^{3}(2\pi)^{2d-3}} \delta\left(E_{\text{Tot}} - E_{1} - (E_{2} - E_{3})\right) \\ \times \underbrace{\frac{d|\vec{p}_{1}||\vec{p}_{1}|^{d-2}}{E_{1}}}_{2|\vec{p}_{1}|d|\vec{p}_{1}|=2E_{1}dE_{1}} \left[dE_{\text{Tot}}^{(23)} \left(\frac{m_{23}^{d-4}}{2^{d-3}}\right) \lambda^{\frac{d-3}{2}} \left(1, \frac{p_{2}^{2}}{m_{23}^{2}}, \frac{p_{3}^{2}}{m_{23}^{2}}\right) \right] (2.81) \\ = \frac{d\Omega_{d-1}^{p_{1}} d\Omega_{d-1}^{p_{2}^{(23)}}}{2^{3}(2\pi)^{2d-3}} \delta\left(E_{\text{Tot}} - E_{1} - (E_{2} - E_{3})\right) dE_{\text{Tot}}^{(23)} \\ \times dE_{1}|\vec{p}_{1}|^{d-3} \left[\left(\frac{m_{23}^{d-4}}{2^{d-3}}\right) \lambda^{\frac{d-3}{2}} \left(1, \frac{p_{2}^{2}}{m_{23}^{2}}, \frac{p_{3}^{2}}{m_{23}^{2}}\right) \right].$$
(2.82)

In the last line, we made the change of variable from $d|\vec{p_1}|$ to dE_1 . Simplifying, regrouping, and integrating over the final δ -function gives us

$$d^{d}PS_{3}(P_{\text{Tot}}; p_{1}, p_{2}, p_{3}) = \frac{m_{23}^{d-4}}{2^{d+1}(2\pi)^{2d-3}} 2|\vec{p}_{1}|^{d-3} dE_{1} d\Omega_{d-1}^{p_{1}} d\Omega_{d-1}^{p_{2}^{(23)}} \lambda^{\frac{d-3}{2}} \left(1, \frac{p_{2}^{2}}{m_{23}^{2}}, \frac{p_{3}^{2}}{m_{23}^{2}}\right) (2.83)$$

In the d = 4 limit, we have

$$dPS_3(P_{\text{Tot}}; p_1, p_2, p_3) = \frac{1}{(4\pi)^5} 2|\vec{p_1}| dE_1 d\Omega_3^{p_1} d\Omega_3^{p_2^{(23)}} \lambda^{1/2} \left(1, \frac{p_2^2}{m_{23}^2}, \frac{p_3^2}{m_{23}^2}\right).$$
(2.84)

And in the P_{Tot} frame, the maximum momentum and energy of p_1 are

$$|\vec{p}_{1}^{\max}| = \frac{\lambda^{1/2} \left(P_{\text{Tot}}^{2}, m_{1}^{2}, (m_{2} + m_{3})^{2}\right)}{2\sqrt{P_{\text{Tot}}^{2}}}, \quad E_{1}^{\max} = \frac{P_{\text{Tot}}^{2} + m_{1}^{2} - (m_{2} + m_{3})^{2}}{2\sqrt{P_{\text{Tot}}^{2}}}$$
(2.85)

To generate this phase space via Monte Carlo integration, we define y_1, \ldots, y_5 such that

$$E_1 = (E_1^{\max} - m_1)y_1 + m_1, \qquad \cos \theta_1 = 2y_2 - 1, \quad \phi_1 = 2\pi y_3, \tag{2.86}$$

$$\cos \theta_2^{(23)} = 2y_4 - 1, \qquad \phi_2^{(23)} = 2\pi y_5, \quad y_1, \dots, y_5 \in [0, 1].$$
 (2.87)

Equation (2.84) then simplifies to

$$dPS_3(P_{\text{Tot}}; p_1, p_2, p_3) = dy_1 dy_2 dy_3 dy_4 dy_5 \frac{(E_1^{\text{max}} - m_1)}{(4\pi)^3} 2|\vec{p_1}|\lambda^{1/2} \left(1, \frac{p_2^2}{m_{23}^2}, \frac{p_3^2}{m_{23}^2}\right)$$
(2.88)

$$\approx \left[\Delta y_i\right]^5 \frac{(E_1^{\max} - m_1)}{(4\pi)^3} 2|\vec{p}_1| \lambda^{1/2} \left(1, \frac{p_2^2}{m_{23}^2}, \frac{p_3^2}{m_{23}^2}\right),$$
(2.89)

where y_1 , i = 1, ..., 5 are randomly generated and Δy_i represent the finite volume element from which the y_i are generated. After constructing p_1 , the Lorentz invariant m_{23} can be built:

$$m_{23}^2 = (p_2 + p_3)^2 = (P_{\text{Tot}} - p_1)^2 = P_{\text{Tot}}^2 + m_1^2 - 2\sqrt{P_{\text{Tot}}^2}E_1.$$
 (2.90)

Then using Eq. (2.79), p_2 and p_3 can be constructed in the (23)-frame using the procedure outlined in section 2.3.3. Finally, p_2 and p_3 are boosted from the (23)-frame into the P_{Tot} frame.

2.3.5 Phase Space Decomposition

A powerful property of phase space for an arbitrary number of final states is the ability to decompose it into the product of smaller phase spaces. For automated event generator packages, the *phase* space recursion relationship forms the basis of their phase space integration modules. Formally, the relationship states that the n-body phase space volume of P_{Tot} is equivalent to the volume enclosed by (i) an (n-1)-body phase space made by combining two final-state momenta p_i and p_j into p_{ij} , (ii) the corresponding $ij \rightarrow i + j$ 2-body phase space, and (iii) the allowed virtuality of p_{ij}^2 :

$$dPS_n(P_{\text{Tot}}; p_1, \dots, p_i, \dots, p_j, \dots, p_n) = dPS_{n-1}(P_{\text{Tot}}; p_1, \dots, p_{i-1}, p_{i+1}, \dots, p_{j-1}, p_{j+1}, \dots, p_n, p_{ij}) \\ \times dPS_2(p_{ij}; p_i, p_j) \times \frac{d m_{ij}^2}{2\pi}, \quad m_{ij}^2 = p_{ij}^2 = (p_i + p_j)^2.$$
(2.91)

The proof of decomposition is quite general, so we present it *d*-dimensions. We start by factoring the *n*-body phase space into (n - 2)- and 2-particle momentum integrals and factors of 1:

$$dPS_{n}(P_{\text{Tot}}; p_{1}, \dots, p_{n}) = (2\pi)^{d} \delta^{d}(P_{\text{Tot}} - p_{1} \dots - p_{i} \dots - p_{j} \dots - p_{n}) \prod_{f=1}^{n} \frac{d^{d-1}p_{f}}{(2\pi)^{d-1}2E_{f}}$$
(2.92)
$$= (2\pi)^{d} \delta^{d}(P_{\text{Tot}} - p_{1} \dots - p_{i-1} - p_{i+1} \dots - p_{j-1} - p_{j+1} \dots - p_{n} - p_{ij})$$
$$\times \left(\prod_{f=1, \neq i, j}^{n} \frac{d^{d-1}p_{f}}{(2\pi)^{d-1}2E_{f}} \right) \times \frac{d^{d-1}p_{i}}{(2\pi)^{d-1}2E_{i}} \frac{d^{d-1}p_{j}}{(2\pi)^{d-1}2E_{j}}$$
$$\times \underbrace{\delta^{d}(p_{ij} - p_{i} - p_{j}) \ d^{d}p_{ij}}_{=1} \times \underbrace{(2\pi)^{d}2E_{ij}}_{=1} \times \underbrace{\delta(p_{ij}^{2} - m_{ij}^{2}) \ dm_{ij}^{2}}_{=1}$$
(2.92)

Shuffling around terms, we can construct a differential n-1-body phase space volume element:

$$dPS_{n}(P_{\text{Tot}}; p_{1}, \dots, p_{n}) = (2\pi)^{d} \delta^{d}(P_{\text{Tot}} - p_{1} \dots - p_{i-1} - p_{i+1} \dots - p_{j-1} - p_{j+1} \dots - p_{n} - p_{ij}) \\ \times \underbrace{\left(\prod_{f=1, \neq i, j}^{n} \frac{d^{d-1}p_{f}}{(2\pi)^{d-1}2E_{f}}\right) \times \frac{d^{d-1}p_{ij}}{(2\pi)^{d-1}2E_{ij}}}_{(n-1)-\text{momentum integrals}} \times \frac{dE_{ij}}{(2\pi)^{d-1}2E_{j}} 2E_{ij}}{(2\pi)^{d-1}2E_{i}} \\ \times (2\pi)^{d} \delta^{d}(p_{ij} - p_{i} - p_{j}) \frac{d^{d-1}p_{i}}{(2\pi)^{d-1}2E_{i}} \frac{d^{d-1}p_{j}}{(2\pi)^{d-1}2E_{j}}}{(2\pi)^{d-1}2E_{j}} \\ \times \delta(p_{ij}^{2} - m_{ij}^{2}) dm_{ij}^{2} \qquad (2.94) \\ = dPS_{n-1}(P_{\text{Tot}}; p_{1}, \dots, p_{i-1}, p_{i+1}, \dots, p_{j-1}, p_{j+1}, \dots, p_{n}, p_{ij}) \\ \times dPS_{2}(p_{ij}; p_{i}, p_{j}) \\ \times dE_{ij}^{2} \delta(p_{ij}^{2} - m_{ij}^{2}) \frac{dm_{ij}^{2}}{2\pi}, \qquad (2.95)$$

giving us our desired expression

$$dPS_n(P_{\text{Tot}}; p_1, \dots, p_n) = dPS_{n-1}(P_{\text{Tot}}; p_1, \dots, p_{i-1}, p_{i+1}, \dots, p_{j-1}, p_{j+1}, \dots, p_n, p_{ij}) \\ \times dPS_2(p_{ij}; p_i, p_j) \times \frac{d m_{ij}^2}{2\pi}.$$
(2.96)

As alluded, two useful applications phase space decomposition are (i) in automated phase space integration packages and (ii) resonant decays. In the first, the phase space for an arbitrarily high number of final states can be reduced to successive boosts of two final-state particles p_i and p_j into their total momentum frame $(p_i + p_j)$, followed by simple and efficient integration over two-body or three-body solid angles. The explicit formulae for these are given in Sections 2.3.3 and 2.3.4. In the latter case, the invariant mass integral $d m_{ij}^2$ can be interpreted as the virtuality integral for an intermediate resonance. For example: in on-shell top quark decays into a bottom quark, muon, and neutrino, the leading contribution occurs through W boson radiation from the top quark that then splits into leptons. The phase space decomposition

$$dPS_3(t;\mu^+,\nu_{\mu},b) = dPS_2(t;p_{\mu+\nu},b) \times dPS(p_{\mu+\nu};\mu^+,\nu_{\mu}) \times \frac{d \ q_{\mu+\nu}^2}{2\pi}$$
(2.97)

has a physical interpretation as the $t \to W^*b$ two-body phase space, the $W^* \to \mu \nu_{\mu}$ two-body phase space, and the virtuality integral for W^* can spans the entire spectrum of invariant masses that are allowed by conservation of momentum.

2.3.6 One-Particle Phase Space Splitting

It is convenient to write explicitly the one-body phase space for collinear splittings, and similar situations. Consider the process

$$A(p_A) + B(p_B) \to A(p_1) + X \tag{2.98}$$

that is mediated by the subprocess

$$g(p_g) + B(p_B) \to X, \quad p_g = p_A - p_1$$
 (2.99)

where X is some arbitrary n-body final state, g originates from an $A \to Ag$ splitting, and A is an otherwise a spectator in the entire process. Momentum conservation tells us

$$p_A + p_B = \underbrace{p_g + p_1}_{p_A} + p_B = p_1 + p_2 \dots + p_n, \qquad (2.100)$$

and so we may write in d dimensions the δ -function

$$\delta^d(p_A + p_B - p_1 - p_2 - \dots - p_n) = \delta^d(p_g + p_1 + p_B - p_1 - p_2 - \dots - p_n) = \delta^d(p_g + p_B - p_2 - \dots - p_n).$$
(2.101)

Therefore, factoring out the momentum integral for $A(p_1)$, we have

$$dPS_n(A+B\to A+X) = (2\pi)^4 \delta^4 \left(p_A + p_B - p_1 - p_2 - \dots - p_n\right) \prod_{k=1}^n \frac{d^3 p_k}{(2\pi)^3 2E_k}$$
(2.102)

$$= \left[(2\pi)^4 \delta^4 \left(p_g + p_B - p_2 - \dots - p_n \right) \prod_{k=2}^n \frac{d^3 p_k}{(2\pi)^3 2E_k} \right] \frac{d^3 p_1}{(2\pi)^3 2E_1} .103)$$
$$= dPS_n (A + B \xrightarrow{A \to Ag} X) \times \frac{d^3 p_1}{(2\pi)^3 2E_1}, \quad p_1 = p_A - p_g. \quad (2.104)$$

$$= dPS_n(A + B \xrightarrow{A \to Ag} X) \times \frac{a p_1}{(2\pi)^3 2E_1}, \quad p_1 = p_A - p_g. \quad (2.104)$$

2.4 PARTIAL WIDTH

The partial width of an unstable, unpolarized particle A with mass m_A , spin states $(2s_A + 1)$, and SU(3)_c color multiplicity N_c^A , decaying into an n-body final-state f is given by formula

$$\Gamma(A \to f) = \frac{1}{2m_A} \frac{1}{(2s_A + 1)N_c^A} \sum |\mathcal{M}|^2 \cdot dPS_n(p_A; p_1, \dots, p_n).$$
(2.105)

Here, \mathcal{M} is usual $A \to f$ amplitude that can be calculated perturbatively using Feynman Rules. The sum over all partial widths, is the *total width*,

$$\Gamma_{\text{Tot}}^{A} \equiv \sum_{f} \Gamma(A \to f).$$
(2.106)

The fraction of times A will decay into a particular final state X is called the *branching fraction*, and is given as the ratio of the partial and total widths

$$BR(A \to X) \equiv \frac{\Gamma(A \to X)}{\Gamma_{Tot}^{A}} = \frac{\Gamma(A \to X)}{\sum_{f} \Gamma(A \to f)}.$$
(2.107)

The total width is also related to A's mean lifetime, τ , by the expression

$$\tau = \frac{\hbar}{\Gamma_{\rm Tot}^A}.$$
(2.108)

We make explicit the conversion of units from Γ [GeV] to τ [s] for clarity. Hence, partial and total widths are simultaneous estimations of how strongly A couples to its final states (larger coupling, larger width), and its likelihood to decay (larger width, smaller decay time). To appreciate this observable better, we consider the Optical Theorem as derived from unitarity of the S-matrix.

2.4.1 The Optical Theorem and Breit-Wigner Propagators

The S-matrix in QFT can be decomposed into its trivial non-scattering and scattering component, T, by the relationship

$$S = 1 + iT.$$
 (2.109)

To leading order in scattering amplitudes, the unitarity of S tells us

$$1 = S^{\dagger}S = (1 - iT^{\dagger})(1 + iT) = 1 + T^{\dagger}T + i(T - T^{\dagger}), \qquad (2.110)$$

or that the squared norm of the transition operator is equal to its imaginary part:

$$T^{\dagger}T = -i(T - T^{\dagger}).$$
 (2.111)

For initial i, final states f, and arbitrary intermediate n-body state k, this implies

$$\langle f|T^{\dagger}T|i\rangle = \sum_{k} \int dP S_n \langle f|T^{\dagger}|k\rangle \langle k|T|i\rangle$$
 (2.112)

$$= \sum_{k} \int dP S_n \ \mathcal{M}^*(f \to k) \mathcal{M}(i \to k), \qquad (2.113)$$

where by the completeness relationship we sum/integrate over all discrete degrees of freedom and phase space configurations. In words, the result states that the matrix elements of the squared norm transition operator $T^{\dagger}T$ is equal to the sum of transition amplitudes, \mathcal{M} , to all intermediate states. The unitarity condition of Eq. (2.111) also tells use that the matrix element of the imaginary part of the transition operator is the imaginary part of the transition operator matrix elements, i.e.,

$$-i\langle f|(T-T^{\dagger})|i\rangle = -i\left[\mathcal{M}(i\to f) - \mathcal{M}^{*}(f\to i)\right], \qquad (2.114)$$

which ultimately follows from linearity. When combined, we obtain the **Optical Theorem**, which states that *imaginary part of a scattering amplitude is equivalent to the sum of all its intermediate states*:

$$-i\left[\mathcal{M}(i\to f) - \mathcal{M}^*(f\to i)\right] = \sum_k \int dP S_n \ \mathcal{M}^*(f\to k)\mathcal{M}(i\to k).$$
(2.115)

In the special case of $1 \rightarrow 1$ scattering, i.e., particle propagation, of particle A with mass m_A , the initial and final states *i* and *f* are equivalent. The Optical Theorem then stipulates

$$-i\left[\mathcal{M}(A \to A) - \mathcal{M}^*(A \to A)\right] = 2\Im[\mathcal{M}(A \to A)]$$
(2.116)

$$= \frac{2m_A}{2m_A} \sum_{\mathbf{k}} \int dP S_n \ \mathcal{M}^*(A \to k) \mathcal{M}(A \to k). \quad (2.117)$$

$$= 2m_A \sum_{\mathbf{k}} \underbrace{\frac{1}{2m_A} \int dPS_n \ |\mathcal{M}^*(A \to k)|^2}_{\text{Partial Width } \Gamma_k}.$$
 (2.118)

We recognize the last expression as the definition of the total width. In other words, the imaginary part of the $1 \rightarrow 1$ scattering amplitude proportional to the total width of the propagating particle

$$\Im[\mathcal{M}(A \to A)] = m_A \Gamma^A_{\text{Tot}}, \qquad (2.119)$$

with the constant of proportionality being the object's mass.

More significantly is that the $1 \rightarrow 1$ amplitudes are precisely the one-particle irreducible (1PI) correlation function diagrams that constitute the self-energy of A. We denote the self-energy generically (whether A is a scalar, fermion, or vector boson) by $\Pi(q^2)$, where q^2 is the virtuality of A. As A comes on-shell, its inverse propagator (again, generically written)

$$\Delta_A^{-1}(q^2) = q^2 - m_A^2 + \Pi(q^2), \qquad (2.120)$$

takes the form

$$\lim_{q^2 \to m_A^2} \Delta^{-1}(q^2) \approx m_A^2 - m_A^2 + i\Im[\Pi(m_A^2)] = +im_A \Gamma_{\text{Tot}}^A, \qquad (2.121)$$

indicating that when A is on mass-shell, the imaginary part of its self-energy is given by its mass and its total width:

$$\Im[\Pi(m_A^2)] = m_A \Gamma_{\text{Tot}}^A.$$
(2.122)

Furthermore, transition amplitudes are dominantly populated by regions of phase space where particles are close to being on-shell. In other words: in the neighborhood of a pole in the S-matrix. Thus, for intermediate, resonant states with momentum q, mass M, and a well-defined, on-shell self-energy, its all-orders summed propagator is well-modeled in matrix element calculations by making the substitution

$$\frac{i}{(q^2 - m^2) + \Pi(q^2)} \to \frac{i}{(q^2 - M^2) + iM\Gamma}.$$
(2.123)

This is the Breit-Wigner (BW) propagator. As a distribution function, its normalization is set by

$$1 = \int_{-\infty}^{\infty} \frac{d \ q^2 \ N}{(q^2 - M^2)^2 + (M\Gamma)^2} = \frac{N}{M\Gamma} \int_{-\infty}^{\infty} \frac{dx}{1 + x^2}$$
(2.124)

$$= N \frac{\arctan x}{M\Gamma} \bigg|_{-\infty}^{\infty} = N \frac{\pi}{M\Gamma}, \quad x = \frac{q^2 - M^2}{M\Gamma}, \quad (2.125)$$

implying

$$N = \frac{M\Gamma}{\pi}.$$
(2.126)

In application, phase space integration over BW propagators can be make more efficient by making the change of variable

$$q^2 - M^2 \equiv M\Gamma \tan\theta. \tag{2.127}$$

This has the action of *smoothening* the Breit-Wigner resonance distribution

$$\int_{q_{min}^2}^{q_{max}^2} \frac{dp^2}{(q^2 - M^2)^2 + (\Gamma M)^2} = \int_{\theta_{\min}}^{\theta_{\max}} \frac{d\theta}{\Gamma M} = \frac{(\theta_{\max} - \theta_{\min})}{\Gamma M} \int_0^1 dy,$$
(2.128)

where

$$\theta_i = \tan^{-1} \left[\frac{q_i^2 - M^2}{\Gamma m} \right], \quad i \in \{\min, \max\}$$
(2.129)

$$\theta = (\theta_{\max} - \theta_{\min})y + \theta_{\min}.$$
 (2.130)

This has great utility in Monte Carlo or other sampling-based integration techniques, which are adversely affected by sharp peaks in integrands.

2.4.2 Narrow Width Approximation

Generally, widths scale like $\Gamma \sim g^2 M$. Thus, for weakly coupled objects one finds

$$\frac{\Gamma}{M} \sim g^2 \ll 1. \tag{2.131}$$
We label such particles as "narrow" resonances, in reference to their narrow BW distributions. A few examples of SM particles with narrow widths include [25]

$$W^{\pm} : \frac{\Gamma}{M} \approx 0.026 \tag{2.132}$$

$$Z : \frac{\Gamma}{M} \approx 0.027 \tag{2.133}$$

$$\eta : \frac{\Gamma}{M} \approx 2.34 \times 10^{-06}$$
 (2.134)

$$t \quad : \quad \frac{\Gamma}{m} \lesssim 0.012. \tag{2.135}$$

Formally, in the zero width (infinitely stable) limit, the BW distribution function approaches the normal distribution function that in turns approaches a δ -function [42]

$$\lim_{\Gamma \to 0} \frac{1}{\pi} \frac{M\Gamma}{(q^2 - M^2)^2 + (M\Gamma)^2} = \lim_{\Gamma \to 0} \frac{1}{(\alpha M\Gamma)\sqrt{\pi}} e^{-(q^2 - M^2)^2/(\alpha M\Gamma)^2}$$
(2.136)

$$= \int \frac{dp}{2\pi} e^{-i(q^2 - M^2)p} = \delta(q^2 - M^2), \qquad (2.137)$$

where $\alpha^{-1} \approx 1.177M$ is the conversion between total width Γ and the standard deviation σ in normal distributions. In particularly extreme situations where the the total width of an intermediate resonance is much smaller than its mass, e.g., top quarks and W bosons at the few percent accuracy [43], it is often sufficient to approximate BW distributions as δ -functions:

$$\frac{1}{(p^2 - M^2)^2 + (M\Gamma)^2} \to \frac{\pi}{M\Gamma} \delta(p^2 - M^2).$$
(2.138)

This is the narrow width approximation (NWA). Physically, the NWA says that an intermediate particle is sufficiently longed live that its intermediate production can be well-approximated as its on-shell production and subsequent on-shell decay. When used in conjunction with the phase space recursion relationship, the NWA is a very powerful tool that greatly simplifies calculations of cascade decays into on-shell particles. In the top quark decay example of Eq. (2.97), applying the NWA has the affect of putting the W boson on-shell at all times

$$\frac{dPS_3(t;\mu^+,\nu_{\mu},b)}{(p^2-M^2)^2+(M\Gamma)^2} = \frac{\pi}{M\Gamma}\delta(p^2-M^2)dPS_2(t;p_{\mu+\nu},b) \times dPS(p_{\mu+\nu};\mu^+,\nu_{\mu}) \times \frac{d\ q_{\mu+\nu}^2}{2\pi} \\
= \frac{\pi}{M\Gamma}dPS_2(t;W^+,b) \times dPS(W^+;\mu^+,\nu_{\mu}).$$
(2.139)

As an example, we now carry out the full $t \to W^+ b \to \mu^+ \nu_\mu b$ calculation with the NWA.

2.4.3 Example: NWA Applied to Leptonic Decays of Top Quarks

We consider the decay of a top quark into a bottom quark and a pair of massless leptons

$$t \to W^{+*} b \to \ell^+ \nu_\ell b, \qquad (2.140)$$

and model the intermediate W boson propagation using a BW propagator with a momentum transfer q^2 and total with Γ_W . The $t \to \ell^+ \nu_\ell b$ matrix element can then be expressed as

$$\mathcal{M}(t \to \ell^+ \nu_\ell b) = \mathcal{M}^\mu(t \to W^* b) \; \frac{-i \left[g_{\mu\nu} + (\xi - 1) \frac{q_\mu q_\nu}{(q^2 - \xi M_W^2)} \right]}{q^2 - M_W^2 + i M_W \Gamma_W} \, \mathcal{M}^\nu(W^* \to \ell^+ \nu_\ell). \tag{2.141}$$

Physical observables, e.g., cross sections and partial widths, are independent of the gauge parameter ξ , which is sometimes written as η instead of $(\xi - 1)$. A wise choice of ξ can greatly simplify a calculation but at the potential cost of increasing the number of subprocesses (Feynman diagrams) contributing to the process. Common choices of the *gauge fixing* parameter are

$$\xi = \begin{cases} 0, & \text{Landau Gauge} \\ 1, & \text{Feynman Gauge} \\ \infty, & \text{Unitary Gauge for Massive Bosons} \end{cases}$$
(2.142)

However, the gauge-term $q_{\mu}q_{\nu}$ in Eq. (2.141) is unimportant and does not contribute to the final result. This follows from the W propagator contracting with a vector current of massless, external fermions. This can be understood from two semi-independent arguments: (i) Since the W momentum is the sum of the massless lepton momenta

$$q_W = p_\mu + p_\nu, \tag{2.143}$$

by the Dirac equation we have (ignoring factors of $-ig/\sqrt{2})$

$$q_{\nu}\mathcal{M}^{\nu}(W^* \to \ell^+ \nu_{\ell}) = \overline{u}(p_{\nu}) (p'_{\mu} + p'_{\nu}) P_L v(p_{\mu})$$
 (2.144)

$$= \overline{u}(p_{\nu})P_{R}\underbrace{p'_{\mu}v(p_{\mu})}_{=m_{\mu}v(p_{\mu})=0} + \underbrace{\overline{u}(p_{\nu})p'_{\nu}}_{=\overline{u}(p_{\nu})m_{\nu}=0}P_{L}v(p_{\mu}) = 0.$$
(2.145)

(ii) By virtue of being on-shell, massless isospin partners, the leptons respect an unbroken $SU(2)_L$ symmetry and therefore do not couple to the W boson's longitudinal polarizations, i.e., $q_\mu q_\nu$.

Using the completeness relationship in the Unitarity gauge,

$$\sum_{\lambda,\lambda'} \varepsilon^*_{\mu,\lambda}(q) \varepsilon_{\nu,\lambda'}(q) = -g_{\mu\nu} + q_{\mu}q_{\nu}/M_W^2, \qquad (2.146)$$

we can express Eq. (2.141) as the product of matrix elements for two independent processes

$$\mathcal{M}(t \to \ell^+ \nu_\ell b) = \sum_{\lambda, \lambda'} \mathcal{M}^\mu(t \to W^* b) \frac{i\varepsilon^*_{\mu,\lambda}(q)\varepsilon_{\nu,\lambda'}(q)}{q^2 - M_W^2 + iM_W\Gamma_W} \mathcal{M}^\nu(W^* \to \ell^+ \nu_\ell) \quad (2.147)$$

$$= \sum_{\lambda,\lambda'} \mathcal{M}_{\lambda}(t \to W^*b) \quad \frac{i}{q^2 - M_W^2 + iM_W\Gamma_W} \mathcal{M}_{\lambda'}(W^* \to \ell^+ \nu_\ell). \quad (2.148)$$

Squaring the amplitudes and summing over all degrees of freedom, i.e., spins/colors, gives us

$$\sum_{\text{d.o.f.}} |\mathcal{M}(t \to \ell^+ \nu_\ell b)|^2 = \frac{1}{(2s_W + 1)N_c^W} \sum_{\text{d.o.f.}} \frac{|\mathcal{M}_{\lambda\tilde{\lambda}}(t \to W^*b)|^2 |\mathcal{M}_{\lambda'\tilde{\lambda'}}(W^* \to \ell^+ \nu_\ell)|^2}{(q^2 - M_W^2)^2 + (M_W \Gamma_W)^2}.$$
 (2.149)

In order to prevent double-counting of W boson spins, we must average over the number of spin states. Though trivial in this instance, one must also average over the intermediate messenger's color multiplicity in order to avoid double-counting color states. This sees innocuous but systematic practice allows us to write intermediate subprocesses, e.g., W boson splitting, in terms of unpolarized widths and cross sections. We are now in position to make the NWA approximation:

$$\sum_{\mathrm{dof}} |\mathcal{M}(t \to \ell^+ \nu_\ell b)|^2 \approx \frac{1}{(2s_W + 1)N_c^W} \sum_{\mathrm{dof}} |\mathcal{M}_{\lambda\tilde{\lambda}}(t \to W^*b)|^2 |\mathcal{M}_{\lambda'\tilde{\lambda}'}(W^* \to \ell^+ \nu_\ell)|^2 \frac{\pi\delta(q^2 - M_W^2)}{M_W\Gamma_W}.$$

Finally, we evaluate the few remaining steps needed to compute the $t \to W^*b \to \ell^+\nu_\ell b$ partial width. Averaging over the top quark's spin states and colors, as well as integrating over the 3-body phase space, which is immediately decomposed into two two-body spaces, we obtain

$$\begin{split} \Gamma(t \to \ell^{+} \nu_{\ell} b) &= \int dPS_{3}(t; \mu^{+}, \nu_{\mu}, b) \; \frac{1}{2m_{t}(2s_{t}+1)N_{c}^{t}} \sum_{\text{dof}} |\mathcal{M}(t \to \ell^{+} \nu_{\ell} b)|^{2} \quad (2.150) \\ &= \int dPS_{2}(t; W^{*}, b) \; dPS_{2}(W^{*}; \mu^{+}, \nu_{\mu}) \; \frac{dq^{2}}{2\pi} \times \frac{1}{2m_{t}(2s_{t}+1)N_{c}^{t}} \; \frac{1}{(2s_{W}+1)N_{c}^{W}} \\ &\times \; \sum_{\text{dof}} \; |\mathcal{M}_{\lambda\bar{\lambda}}(t \to W^{*}b)|^{2} \; |\mathcal{M}_{\lambda'\bar{\lambda}'}(W^{*} \to \ell^{+} \nu_{\ell})|^{2} \; \frac{\pi}{M_{W}\Gamma_{W}} \delta(q^{2} - M_{W}^{2}) \quad (2.151) \\ &= \; \int dPS_{2}(t; W^{*}, b) \; \frac{1}{2m_{t}(2s_{t}+1)N_{c}^{t}} \sum_{\text{dof}} |\mathcal{M}_{\lambda\bar{\lambda}}(t \to W^{*}b)|^{2} \\ &\times \; \frac{1}{(2s_{W}+1)N_{c}^{W}} \; \frac{\pi}{M_{W}\Gamma_{W}} \; \int \frac{dq^{2}}{2\pi} \delta(q^{2} - M_{W}^{2}) \\ &\times \; \int dPS_{2}(W^{*}; \mu^{+}, \nu_{\mu}) \; \sum_{\text{dof}} |\mathcal{M}_{\lambda'\bar{\lambda}'}(W^{*} \to \ell^{+} \nu_{\ell})|^{2} \quad (2.152) \end{split}$$

Combining the δ -function from the NWA and the virtuality integral from the Recursion Theorem together require that the W be on-shell at all times. With this, we immediately recognize the first line of Eq. (2.152) as the partial width for $t \to Wb$ decay, and the last two lines of Eq. (2.152) provide us the ingredients for the $W \to \ell \nu_{\ell}$ partial width:

$$\Gamma(t \to \ell^+ \nu_\ell b) = \Gamma(t \to Wb) \times \frac{1}{\Gamma_W} \times \frac{1}{2M_W(2s_W + 1)N_c^W} \int dP S_2(W^*; \mu^+, \nu_\mu) \sum_{\text{dof}} |\mathcal{M}_{\lambda'\tilde{\lambda'}}(W^* \to \ell^+ \nu_\ell)|^2 2.153)$$

$$= \Gamma(t \to Wb) \times \frac{1}{\Gamma_W} \Gamma(W \to \ell \nu_\ell)$$
(2.154)

$$= \Gamma(t \to Wb) \times BR(W \to \ell \nu_{\ell})$$
(2.155)

Thus, by combining the NWA and phase space recursion theorem, we can approximate a top quark decay rate to leptons as the rate of a top quark decaying into an on-shell W boson scaled by the probability that an on-shell W will decay to leptons. This probability is given by the branching fraction of W into leptons, and is defined in Eq. (2.107).

Before moving on, it is worth reflecting what we have sacrificed in order to obtain Eq. (2.155). The first is that we have considered only a single slice of phase space, namely when the W boson is on-shell. An entire continuum of phase space for $q^2 \neq M_W^2$ has been neglected and thus Eq. (2.155) is an underestimation, albeit a very good one, of the actual LO $t \to \ell^+ \nu_{\ell} b$ partial width. As the stipulation that $\Gamma/M \ll 1$ breaks down, larger regions of phase space where $q^2 \neq M_W^2$ become increasingly important, and lead to worse estimates of the partial width.

More subtle is the fact that spin correlation between the initial-state top quark and final-state leptons has been lost due to the use of the completeness relationship in Eq. (2.146), which acts to decouple the initial-state and final-state fermion currents. Though the total scattering and decay rates remain unaffected when the NWA holds, angular distributions and "fiducial" rates that are obtained by imposing phase space cuts will be inaccurate. The exception of course being the case of a narrow scalar mediator, for example: a charged Higgs in the 2HDM. In the top quark decay $t \to H^+b$, $H^+ \to \tau^+\nu_{\tau}$, the completeness relationship for scalars can be imposed without any loss of spin correlation because, as a scalar, the scalars carry no such information.

An alternative procedure that preserves spin correlation would be to forgo the use of Eq. (2.146). Starting from the $t \to W^*b \to \ell^+ \nu_\ell b$ matrix element given in Eq. (2.141), we have

$$\mathcal{M}(t \to \ell^+ \nu_\ell b) = \mathcal{M}^\mu(t \to W^* b) \frac{-ig_{\mu\nu}}{q^2 - M_W^2 + iM_W \Gamma_W} \mathcal{M}^\nu(W^* \to \ell^+ \nu_\ell)$$
(2.156)

$$= \mathcal{M}^{\mu}(t \to W^* b) \mathcal{M}_{\mu}(W^* \to \ell^+ \nu_{\ell}) \frac{-i}{q^2 - M_W^2 + iM_W \Gamma_W}$$
(2.157)

Repeating the above procedure will give us

$$\sum_{\text{dof}} |\mathcal{M}(t \to \ell^+ \nu_\ell b)|^2 \approx \sum_{\text{dof}} |\mathcal{M}^{\mu}(t \to W^* b) \mathcal{M}_{\mu}(W^* \to \ell^+ \nu_\ell)|^2 \times \frac{\pi}{M_W \Gamma_W} \delta(q^2 - M_W^2). \quad (2.158)$$

Averaging and integrating over a decomposed phase space gives us

$$\Gamma(t \to \ell^+ \nu_{\ell} b) = \int dP S_2(t; W^*, b) \ dP S_2(W^*; \mu^+, \nu_{\mu}) \ \frac{dq^2}{2\pi} \times \frac{1}{2m_t(2s_t + 1)N_c^t} \\ \times \sum_{\text{dof}} |\mathcal{M}^{\mu}(t \to W^* b) \mathcal{M}_{\mu}(W^* \to \ell^+ \nu_{\ell})|^2 \times \frac{\pi}{M_W \Gamma_W} \delta(q^2 - M_W^2).$$
(2.159)

$$= \frac{1}{2^2 m_t M_W \Gamma_W(2s_t+1) N_c^t} \int dP S_2(t;W,b) \ dP S_2(W;\mu^+,\nu_\mu) \qquad (2.160)$$

$$\times \sum_{\text{dof}} |\mathcal{M}^{\mu}(t \to Wb)\mathcal{M}_{\mu}(W \to \ell^{+}\nu_{\ell})|^{2}.$$
(2.161)

Inserting the closed expressions for each of the two-body phase spaces and grouping together factors of 2π , we obtain the spin-correlated expression

$$\Gamma(t \to \ell^+ \nu_\ell b) = \frac{\sqrt{1 - 2(r_W + r_b) + (r_W - r_b)^2}}{2^{12} \pi^3 m_t M_W \Gamma_W N_c^t} \int d\cos\theta_b \ d\Omega_\mu$$
(2.162)

$$\times \sum_{\text{dof}} |\mathcal{M}^{\mu}(t \to Wb)\mathcal{M}_{\mu}(W \to \ell^{+}\nu_{\ell})|^{2}, \qquad (2.163)$$

where $r_X = m_X/m_t^2$. Despite its apparent bulkiness, the expression above can be evaluated analytically or numerically with little additional effort. The key point is that numerical integration of the original three-body phase space (4 integrals) over a BW propagator is inefficient and can be approximated well by one fewer integrals over zero propagators.

2.5 PARTONIC LEVEL CROSS SECTION

The statistical nature of quantum mechanics lends itself to counting experiments to test predictions made by models. In colliders, antiparallel particle beams are focused onto each other in order to reproduce a type of Rutherford scattering. For a given flux, or luminosity, \mathcal{L} of particles transversing through an effective scattering area, or cross section, σ , the number of scattering events is given schematically by

Number of events =
$$\underbrace{(\text{Number of particles per beam area})}_{\text{Luminosity, }\mathcal{L}} \times \underbrace{(\text{Effective target area})}_{\text{Cross section, }\sigma}$$
(2.164)

For a fixed beam luminosity, we can interpret the cross section as a measure of the likelihood for a particular scattering to occur. Again, schematically, this is given by

$$(\text{Scattering cross section}) = \frac{\text{Number of events}}{\text{Incoming particle flux}}$$
(2.165)
$$= \frac{(\text{Scattering likelihood}) \times (\text{Scattering configurations})}{\text{Incoming particle flux}}$$
(2.166)

From Fermi's Golden Rule, we identify the numerator of this expression as simply the squared matrix element summed over discrete final-state degrees of freedom and integrated over continuous ones, i.e., phase space. For randomly polarized and charged initial states, symmetry factors must be introduced to average over initial-state degrees of freedom. And as is typical for scattering experiments, flux can be factored into the product of number densities of each beam and the relative velocity of the two, implying its invariance under longitudinal boosts (along the beam line). The scattering cross section can now be written (schematically) as

$$(\text{Scattering cross section}) = \frac{1}{(\text{Number density}) \times (\text{Relative velocity}) \times (\text{Symmetry factors})} \\ \times \sum_{\text{Discrete dof}} \int d[\text{Phase space}] (\text{Probability density}) \qquad (2.167)$$

Formally, for incoming particles A and B, with masses m_A , m_B and c.m. energy

$$\sqrt{\hat{s}} = \sqrt{(p_A + p_B)^2},$$
 (2.168)

the $2 \rightarrow n$ scattering rate is given by the formula

$$\sigma(A+B \rightarrow X + \text{anything else}) = \int dP S_n \frac{d\sigma}{dPS_n},$$
 (2.169)

$$\frac{d\sigma}{dPS_n} = \frac{1}{2\hat{s}\lambda^{1/2}(1, r_A, r_B)} \frac{1}{(2s_A + 1)(2s_B + 1)N_c^A N_c^B} \sum_{\text{dof}} |\mathcal{M}|^2, \qquad (2.170)$$

where, for $X = A, B, r_X = m_X^2/\hat{s}, \lambda$ is the kinematic Källen function of Eq. (2.46), $(2s_X + 1)$ represents the number of spin states possessed by particle X, N_c^X is the SU(3)_c color factor of X, dPS_n denotes the Lorentz-invariant *n*-body differential phase space as defined in Eq. (2.45), and \mathcal{M} is the Lorentz-invariant matrix element for scattering process

$$A + B \to X. \tag{2.171}$$

In the $r_B \to 0$ limit, $\lambda^{1/2}(1, r_A, 0) = (1 - r_A)$; and for $r_A, r_B \to 0, \lambda \to 1$.

2.5.1 Example: Zh Production at Electron Colliders

Lepton collider-based Higgs factories are premised on the fact that Z bosons couple directly to both electrons and the Higgs boson, and so Higgs bosons can be produced in the $2 \rightarrow 2$ process

$$e_{\tau}^{-}(p_A) + e_{\tau'}^{+}(p_B) \to Z_{\lambda}(p_Z) + h(p_h),$$
 (2.172)

where $\tau = L, R$ and $\lambda = 0, \pm$ denote the helicity takes of electrons and the Z. For massless electrons, the matrix element is given by

$$\mathcal{M}_{\lambda\tau'\tau}^{Zh} = \kappa_Z \left[\overline{v}_{\tau'}(p_B) \varsigma^*_{\lambda}(p_Z) \left(c_R^e P_R + c_L^e P_L \right) u_{\tau}(p_A) \right] D_Z(s), \qquad (2.173)$$

$$c_R^e = g_V^e + g_A^e = \frac{1}{2} (T_3^e)_L - Q^e \sin^2 \theta_W + \left(\frac{-1}{2}\right) (T_3^e)_L = \sin^2 \theta_W, \qquad (2.174)$$

$$c_L^e = g_V^e - g_A^e = \frac{1}{2} (T_3^e)_L - Q^e \sin^2 \theta_W - \left(\frac{-1}{2}\right) (T_3^e)_L = \sin^2 \theta_W - \frac{1}{2}, \qquad (2.175)$$

$$D_X(p^2) = \frac{1}{(p^2 - M_X^2) + iM_X\Gamma_X}, \quad \kappa_Z = \frac{g^2 M_Z}{\cos^2 \theta_W}$$
(2.176)

where the helicities of e^- , e^+ , and Z are denoted, respectively, by τ , τ' , λ . In the center of mass frame, the 4-momenta can be expressed as

$$p_A = \frac{\sqrt{s}}{2} (1, 0, 0, 1), \quad p_B = \frac{\sqrt{s}}{2} (1, 0, 0, -1)$$
 (2.177)

$$p_Z = (E_Z, |\vec{p}_Z| \sin \theta \cos \phi, |\vec{p}_Z| \sin \theta \sin \phi, |\vec{p}_Z| \cos \theta), \qquad (2.178)$$

$$p_h = (E_h, -|\vec{p}_Z|\sin\theta\cos\phi, -|\vec{p}_Z|\sin\theta\sin\phi, -|\vec{p}_Z|\cos\theta), \qquad (2.179)$$

$$E_Z = \frac{\sqrt{s}}{2}(1+r_Z-r_h), \quad |\vec{p}_Z| = \frac{\sqrt{s}}{2}\lambda^{1/2}(1,r_Z,r_h), \quad E_h = \frac{\sqrt{s}}{2}(1+r_h-r_Z), \quad (2.180)$$

and for X = Z, h we define $r_X = m_X^2/s$. The nonzero amplitudes are given by

$$\mathcal{M}_{0RL} = \frac{-c_L}{2} \kappa_Z D_Z(s) \sqrt{\frac{s}{r_Z}} (1 - r_h + r_Z) e^{-i\phi} \sin\theta \qquad (2.181)$$

$$\mathcal{M}_{0LR} = \frac{-c_R}{2} \kappa_Z D_Z(s) \sqrt{\frac{s}{r_Z}} (1 - r_h + r_Z) e^{i\phi} \sin\theta \qquad (2.182)$$

$$\mathcal{M}_{+RL} = -c_L \kappa_Z D_Z(s) \sqrt{\frac{s}{2}} e^{-i\phi} (1 - \cos\theta)$$
(2.183)

$$\mathcal{M}_{+LR} = c_R \kappa_Z D_Z(s) \sqrt{\frac{s}{2}} e^{i\phi} (1 + \cos\theta)$$
(2.184)

$$\mathcal{M}_{-RL} = -c_L \kappa_Z D_Z(s) \sqrt{\frac{s}{2}} e^{-i\phi} (1 + \cos\theta)$$
(2.185)

$$\mathcal{M}_{-LR} = c_R \kappa_Z D_Z(s) \sqrt{\frac{s}{2}} e^{i\phi} (1 - \cos\theta).$$
(2.186)

As the vector couplings are helicity conserving, the $(\lambda, \tau', \tau) = (\lambda, L, L)$ and (λ, R, R) contributions are zero. We observe longitudinal enhancement in the $(\lambda, \tau', \tau) = (0RL)$ and (0RL) helicity amplitudes. The squared and summed amplitude is then

$$\sum |\mathcal{M}_{Zh}|^2 = \frac{g^4 (c_L^2 + c_R^2) s^2}{8 c_W^4} |D_Z(s)|^2 \times \left[(1 - r_h)^2 - 2r_h r_Z + r_Z (14 + r_Z) + (1 + 2r_h r_Z - (1 - r_h)^2 - (1 - r_Z)^2) \cos(2\theta) \right],$$

$$|D_Z(s)|^2 = \frac{1}{(s - M_Z^2)^2 + (M_Z \Gamma_Z)^2}.$$
(2.187)

The corresponding 2-body phase space is

$$dPS_2(P_{\text{Tot}}; p_Z, p_h) = \frac{d\cos\theta d\phi}{2(4\pi)^2} \lambda^{1/2}(1, r_Z, r_h)$$
(2.188)

$$= \frac{d\cos\theta}{2^4\pi}\sqrt{1-2(r_Z+r_h)+(r_Z-r_h)^2}.$$
 (2.189)

Integrating and averaging over the quantum numbers of the initial states, the Zh production cross section at $\sqrt{s} = 250$ GeV for $m_h = 125$ GeV is

$$\sigma(e^{-}e^{+} \rightarrow Zh) = \underbrace{(\hbar c)^{2}}_{\text{GeV}^{-2} \text{ to fb conversion factor}} \underbrace{\left(\frac{1}{2^{3}s}\right)}_{\text{Flux}} \times \frac{g^{4}s^{2}\left(c_{L}^{2}+c_{R}^{2}\right)}{3c_{W}^{4}} \frac{\left(1+10r_{Z}-2r_{h}+(r_{Z}-r_{h})^{2}\right)}{(s-M_{Z}^{2})^{2}+(M_{Z}\Gamma_{Z})^{2}} \times \underbrace{\left(\frac{\lambda^{1/2}(1,r_{Z},r_{h})}{2^{4}\pi}\right)}_{\text{Phase space}} \approx 240 \text{ fb.}$$

$$(2.190)$$

2.6 INITIAL-STATE PHOTONS FROM ELECTRON-X SCATTERING

On-shell factorization of scattering amplitudes into a product of *universal*, i.e., process-independent, terms and (usually) much simpler (though process dependent) hard scattering matrix element calculations for participants whose masses are much smaller than momentum transfers scales is a central tenet of perturbative QCD and collider experiments. Here we derive γX scattering of a quasi-real photon that originates from a nearly collinear splitting with an electron in high energy eXcollisions. It should be emphasized that this is a property of gauge theories, not unique to photons or electrons, and holds with very minor modifications, though with various physics interpretations, for gg or gq splitting.

As shown in Fig. (4), we consider the eX scattering process

$$e^{-}(p_A) + X(p_B) \to e^{-}(p_1) + \gamma^{*}(q) + X(p_B) \to e^{-}(p_1) + Y$$
 (2.191)

that is mediated by the subprocess

$$\gamma^*(q) + X(p_B) \to Y, \quad q = p_A - p_1.$$
 (2.192)

The total eX and subprocess γ^*X center of mass energies are respectively denoted by

$$s = (p_A + p_B)^2, \quad \hat{s} = (q + p_B)^2.$$
 (2.193)



Figure 4: Diagrammatic representation of quasi-real, initial-state photon from $e\gamma$ splitting.

In essence, the initial-state electron radiates a photon that participates in the $2 \rightarrow 1$ scattering but otherwise acts as a spectator of the process. The matrix element is given by

$$\mathcal{M} = [\overline{u}(p_1)(-ieq_e)\gamma^{\mu}u(p_A)] \frac{(-i)\left(g_{\mu\nu} - (\xi - 1)q_{\mu}q_{\nu}/q^2\right)}{q^2} \mathcal{M}^{\nu}(\gamma^* X \to Y)$$
(2.194)

$$= \frac{(-ieq_e)}{q^2} \left[\overline{u}(p_1)\gamma^{\mu}u(p_A)\right] \left(\sum_{\lambda\lambda'} \varepsilon^*_{\mu}(q,\lambda)\varepsilon_{\nu}(q,\lambda')\right) \mathcal{M}^{\nu}(\gamma^*X \to Y).$$
(2.195)

$$= \frac{(-ieq_e)}{q^2} \sum_{\lambda\lambda'} \left[\overline{u}(p_1) \not = u(p_A) \right] \left[\varepsilon \cdot \mathcal{M}(\gamma X \to Y) \right]$$
(2.196)

$$\equiv \frac{(-ieq_e)}{q^2} \sum_{\lambda\lambda'} \mathcal{M}(e \to e\gamma^*) \mathcal{M}(\gamma^* X \to X).$$
(2.197)

The quantity $\mathcal{M}^{\nu}(\gamma X \to Y)$ is defined such that the scattering process Eq. (2.192) for an on-shell (massless), initial-state photon is given by

$$\mathcal{M}(\gamma_{\lambda}X \to Y) = \varepsilon_{\mu}(q,\lambda) \cdot \mathcal{M}^{\nu}(\gamma_{\lambda}X \to Y).$$
(2.198)

We keep explicit the electron charge as q_e . In the second line, we applied the completeness relationship of Eq. (2.146) and find that we can express the entire matrix element as the product of two subprocess matrix elements: $e \rightarrow \gamma$ splitting and γX scattering. Squaring and summing over final state degrees of freedom gives us

$$\sum_{\text{d.o.f.}} |M|^2 = \frac{(e^2 q_e^2)}{q^4} \sum_{\lambda\lambda'} \sum_{\text{d.o.f.}} |\mathcal{M}(e \to e\gamma^*)|^2 |\mathcal{M}(\gamma X \to X)|^2.$$
(2.199)

However, this is not quite correct as we are inadvertently double counting degrees of freedom of our intermediate γ^* . We must introduce spin-state and (in principle) color-state averaging factors,

which effectively gives us a recipe for an unpolarized intermediate photon with, for now, an arbitrary virtuality. Therefore, we should instead have

$$\sum_{\text{d.o.f.}} |M|^2 = \frac{(e^2 q_e^2)}{q^4} \sum_{\lambda \lambda'} \sum_{\text{d.o.f.}} |\mathcal{M}(e \to e\gamma^*)|^2 \left(\frac{1}{(2s_\gamma + 1)N_c^\gamma}\right) |\mathcal{M}(\gamma X \to X)|^2. \quad (2.200)$$

The obviously trivial color factor N_c^{γ} is present for completeness. We are also free to introduce the Källen function

$$\hat{s}\lambda^{1/2}(1,\tilde{r}_{\gamma},\tilde{r}_{B}), \quad \tilde{r}_{\gamma} = \frac{q^{2}}{\hat{s}}, \tilde{r}_{B} = \frac{p_{B}^{2}}{\hat{s}},$$
(2.201)

which represents the energy available in $\gamma^* X$ scattering. The spin- and color-averaged squared amplitude is then

$$\overline{|M|^2} = \frac{(e^2 q_e^2)}{q^4} \sum_{\lambda\lambda'} \sum_{\text{d.o.f.}} \frac{\hat{s}\lambda^{1/2}(1, \tilde{r}_{\gamma}, \tilde{r}_B)}{(2s_e + 1)N_c^e} |\mathcal{M}(e \to e\gamma^*)|^2 \\ \times \left(\frac{1}{\hat{s}\lambda^{1/2}(1, \tilde{r}_{\gamma}, \tilde{r}_B)(2s_{\gamma} + 1)(2s_B + 1)N_c^{\gamma}B_c^b}\right) |\mathcal{M}(\gamma^* X \to X)|^2.$$
(2.202)

The eX cross section is then given by

$$\sigma(e^{-}X \to e^{-}Y) = \int \frac{d^{3}p_{1}}{(2\pi)^{3}2E_{1}} \sum_{\lambda, \text{d.o.f.}} \frac{\hat{s}\lambda^{1/2}(1, \tilde{r}_{\gamma}, \tilde{r}_{B})}{s\lambda^{1/2}(1, r_{e}, r_{B})} \frac{|\mathcal{M}(e \to e\gamma^{*})|^{2}}{(2s_{e} + 1)N_{c}^{e}} \times \frac{(4\pi\alpha q_{e}^{2})}{q^{4}} \times \hat{\sigma}(\gamma X \to Y), \qquad (2.203)$$

where $r_i = m_i^2/s$, $e^2 = 4\pi\alpha$, we have split the *n*-body phase space into two using Eq. (2.104), and $\hat{\sigma}$ is the subprocess γX cross section. Assuming that the scatting scales *s* and \hat{s} are much larger than any mass relevant initial-state mass, this further refines down to

$$\sigma(e^{-}X \to e^{-}Y) = \int \frac{d^{3}p_{1}}{(2\pi)^{3}2E_{1}} \sum_{\lambda, \text{d.o.f.}} \frac{\hat{s}}{2s} |\mathcal{M}(e \to e\gamma^{*})|^{2} \frac{(4\pi\alpha q_{e}^{2})}{q^{4}} \hat{\sigma}(\gamma X \to Y). \quad (2.204)$$

It now remains to evaluate the $e \rightarrow \gamma$ splitting matrix elements.

2.6.1 $e \rightarrow \gamma$ Splitting

For $e^{-}(p_A) \to e^{-}(p_1)\gamma^*(q)$ splitting, where γ^* with small transverse momentum $q_T \ll E_A$ carries away an energy fraction z from its parent electron, we assign the following momenta

$$p_A = (E_A, 0, 0, E_A) \tag{2.205}$$

$$p_{1} = (1-z)E_{A}(1,\sin\theta_{1},0,\cos\theta_{1})$$

$$= \left((1-z)E_{A},-q_{T},0,(1-z)E_{A},-\frac{q_{T}^{2}}{2}\right)$$
(2.206)

$$q = (zE_A, |\vec{q}| \sin \theta_{\gamma}, 0, |\vec{q}| \cos \theta_{\gamma})$$

$$(2.207)$$

$$= \left(zE_A, q_T, 0, zE_A + \frac{q_T^2}{2(1-z)E_A}\right).$$
(2.208)

The corresponding (complex conjugated) photon polarization vectors in the helicity basis are

$$\varepsilon_{\mu}^{\lambda=+*} = \frac{1}{\sqrt{2}} \left(0, -\cos\theta_{\gamma}, i, -\sin\theta_{\gamma} \right)$$
(2.209)

$$\varepsilon_{\mu}^{\lambda=-*} = \frac{1}{\sqrt{2}} \left(0, \cos \theta_{\gamma}, i, \sin \theta_{\gamma} \right).$$
(2.210)

Then, neglecting terms higher than of $\mathcal{O}\left(\frac{q_T^2}{E_A^2}\right)$, we have

$$p_1^2 = (1-z)^2 E_A^2 - q_T^2 - (1-z)^2 E_A^2 + \frac{2(1-z)E_A q_T^2}{2(1-z)E_A} + \mathcal{O}\left(\frac{q_T^4}{E_A^4}\right) = 0$$
(2.211)

$$q^{2} = z^{2}E_{A}^{2} - q_{T}^{2} - z^{2}E_{A}^{2} - \frac{2zE_{A}q_{T}^{2}}{2(1-z)E_{A}} + \mathcal{O}\left(\frac{q_{T}^{4}}{E_{A}^{4}}\right) = \frac{-q_{T}^{2}}{(1-z)},$$
(2.212)

that is: a massless final-state e^- and internal photon with virtuality proportional to its transverse momentum.

As photon radiation is helicity-conserving, the only nonzero fermion currents are the $e_L^- \rightarrow e_L^- \gamma e_R^- \rightarrow e_R^- \gamma$ channels, given by

$$J_{LL}^{\mu} = 2E_A \sqrt{1-z} \left[\cos\left(\frac{\theta_1}{2}\right), \sin\left(\frac{\theta_1}{2}\right), -i\sin\left(\frac{\theta_1}{2}\right), \cos\left(\frac{\theta_1}{2}\right) \right]$$
(2.213)

$$J_{RR}^{\mu} = 2E_A \sqrt{1-z} \left[\cos\left(\frac{\theta_1}{2}\right), \sin\left(\frac{\theta_1}{2}\right), i\sin\left(\frac{\theta_1}{2}\right), \cos\left(\frac{\theta_1}{2}\right) \right].$$
(2.214)

The permutation of helicity amplitudes (in the small q_T limit) is therefore

$$\mathcal{M}_{LL-} = 2E_A \sqrt{2 - 2z} \cos\left(\frac{\theta_\gamma}{2}\right) \sin\left(\frac{\theta_1 + \theta_\gamma}{2}\right)$$
(2.215)

$$\approx E_A \sqrt{2 - 2z} \left(\theta_1 + \theta_\gamma\right) \tag{2.216}$$

$$\mathcal{M}_{LL+} = -2E_A\sqrt{2-2z}\sin\left(\frac{\theta_\gamma}{2}\right)\cos\left(\frac{\theta_1+\theta_\gamma}{2}\right)$$
(2.217)

$$\approx -E_A \sqrt{2 - 2z} \left(\theta_\gamma\right) \tag{2.218}$$

$$\mathcal{M}_{RR-} = 2E_A \sqrt{2-2z} \sin\left(\frac{\theta_\gamma}{2}\right) \cos\left(\frac{\theta_1+\theta_\gamma}{2}\right)$$
(2.219)

$$\approx E_A \sqrt{2 - 2z} \left(\theta_\gamma \right) \tag{2.220}$$

$$\mathcal{M}_{RR+} = -2E_A\sqrt{2-2z}\cos\left(\frac{\theta_\gamma}{2}\right)\sin\left(\frac{\theta_1+\theta_\gamma}{2}\right)$$
(2.221)

$$\approx -E_A \sqrt{2 - 2z} \left(\theta_1 + \theta_\gamma\right) \tag{2.222}$$

Squaring the amplitudes, making the replacements

$$\theta_{\gamma} \to \frac{q_T}{zE_A}, \quad \theta_1 \to \frac{q_T}{(1-z)E_A},$$
(2.223)

and summing, we obtain

$$\sum |\mathcal{M}(e \to e\gamma^*)|^2 = \frac{4q_T^2}{z(1-z)} P_{\gamma e}(z), \qquad (2.224)$$

where $P_{\gamma e}(z)$ is the universal Altarelli-Parisi splitting function

$$P_{\gamma e}(z) = \frac{1 + (1 - z)^2}{z} \,. \tag{2.225}$$

Its pole represents the soft divergence that appears when q_T , and hence γ^* 's virtuality goes to zero. The function $P_{\gamma e}(z)$ is universal in the sense that it holds for all spin-half-to-internal spin-one bosons splittings in the small transverse momentum limit. Accounting for color factors, the QCD equivalent is in fact given by

$$P_{gq}(z) = C_F \frac{1 + (1 - z)^2}{z}.$$
(2.226)

Had we considered instead an internal electron and on-shell photon, neglecting terms higher than $\mathcal{O}\left(\frac{q_T^2}{E_A^2}\right)$, the $e_i \to e_f$ splitting function, where e_f carries a momentum fraction z from e_i , is

$$P_{ee}(z) = \frac{1+z^2}{1-z},$$
(2.227)

and possesses a collinear divergence. It is interesting to note that the $\gamma \to e$ splitting function

$$P_{e\gamma}(z) = z^2 + (1-z)^2$$
(2.228)

does not have a pole as massless fermion currents turn off (vanish) in the zero fermion energy limit.

2.6.2 Weizsäcker-Williams Approximation

We can now assemble our final result. We start by expressing our one-body phase space in cylindrical coordinates, and ultimately photon virtuality and momentum fraction z

$$\frac{d^3 p_1}{(2\pi)^3 2E_1} = \frac{d\phi_1 \, dp_z \, dq_T^2}{(2\pi)^3 2^2 (1-z)E_A} = \frac{dz \, dq_T^2}{(2^4 \pi^2)(1-z)} = \frac{dz \, dq^2}{(2^4 \pi^2)}, \tag{2.229}$$

where we made use of the fact that

$$p_z \simeq E_1 = (1-z)E_A$$
 and $q^2 = \frac{-q_T^2}{(1-z)}$. (2.230)

Under the working assumptions that masses are negligible compared to the scattering scale and that we are in the collinear (small q_T) $e \to \gamma$ splitting regime, we also have

$$\hat{s} = (q + p_B)^2 = 2q \cdot p_B = zp_A \cdot p_B = zs.$$
 (2.231)

Making the appropriate substitutions, we then have

$$\sigma(e^{-}X \to e^{-}Y) = \int \frac{dz \ dq^2}{(2^4\pi^2)} \frac{z}{2} \frac{4(1-z)q^2}{z(1-z)} P_{\gamma e}(z) \frac{(4\pi\alpha q_e^2)}{q^4} \hat{\sigma}(\gamma X \to Y)$$
(2.232)

$$= \int dz \; \frac{\alpha q_e^2}{2\pi} \; P_{\gamma e}(z) \int \frac{dq^2}{q^2} \hat{\sigma}(\gamma X \to Y) \tag{2.233}$$

A pause is necessary to to address the limits for the virtuality integrations. At zero momentum transfer $(q^2 = 0)$, the cross section diverges but only artificially. We made the assumption that masses are negligible compared to the hard scattering energies, but at zero momentum transfer this is no longer true. Strictly speaking, the photon virtuality is given by

$$q^{2} = (p_{A} - p_{1})^{2} = 2m_{e}^{2} - E_{A}E_{1}(1 - \beta_{A}\beta_{1}\cos\theta_{1}), \qquad (2.234)$$

indicating that the supposed collinear divergence is actually regulated by $\beta_{A,1} < 1$, or in other words, the electron mass. For quark-gluon splitting, this is regulated analogously by the bare quark mass. The electron mass then sets the scale for momentum transfers and we *evolve* our momentum transfer scale starting from $q^2 = m_e^2$. The upper limit of integration must be chosen based on its type of calculation that is being performed. For inclusive cross section calculations, and despite contradicting the small q_T assumption, evolving the integral upwards to \hat{s} is a reasonable estimation [6]. However, as we will discuss shortly later and much detail in later chapters, this can be matched with the deeply inelastic process, rendering the *sum* of the two components relatively scale-independent [3]. For now, we will evolve our integral upwards to $q^2 = Q^2$. The evolution scale Q^2 is also called the *factorization scale*. Doing so gives us

$$\sigma(e^- X \to e^- Y) = \int dz \; \frac{\alpha q_e^2}{2\pi} \; P_{\gamma e}(z) \log\left(\frac{Q^2}{m_e^2}\right) \; \hat{\sigma}(\gamma X \to Y). \tag{2.235}$$

This result, also known as the Weizsäcker-Williams [44, 45] approximation, is expressed more commonly in the form

$$\sigma(e^{-}X \to e^{-}Y) = \int_{z_{\min}}^{1} dz \ f_{\gamma/e}(z, Q^2) \ \hat{\sigma}(\gamma X \to Y), \qquad (2.236)$$

where the *photon distribution function* is given by

$$f_{\gamma/e}(z,Q^2) = \frac{\alpha q_e^2}{2\pi} P_{\gamma e}(z) \log\left(\frac{Q^2}{m_e^2}\right).$$
(2.237)

Written in this form, we interpret $f_{\gamma/e}(z, Q^2)$ as the likelihood of observing a photon in an electron, possessing an energy fraction z of the electron's total energy at a momentum transfer scale Q^2 . The limits of integration are derived from the relation $\hat{s} = zs$, which tells us that

$$\max(z) = \max\left(\frac{\hat{s}}{s}\right) = \frac{\max(\hat{s})}{s} = \frac{s}{s} = 1$$
(2.238)

$$\min(z) = \min\left(\frac{\hat{s}}{s}\right) = \frac{\min(\hat{s})}{s}, \qquad (2.239)$$

and $\min(\hat{s})$ is the minimum invariant mass required for $\gamma B \to X$ to kinematically proceed.

2.6.3 Weak Boson Distribution Functions

Following the identical procedure with W and Z bosons will yield similar results. The polarizationdependent distributions functions for Weak bosons carrying energy fraction z from fermion f evolved to a scale $Q_V \gg M_W$ are given by [36, 46]

$$P_{V/f}(z,Q^2,\lambda=\pm) = \frac{C}{16\pi^2 z} \left[(g_V^f \mp g_A^f)^2 + (g_V^2 \pm g_A^f)^2 (1-z)^2 \right] \log \frac{Q^2}{M_V^2}$$
(2.240)

$$P_{V/f}(z,Q^2,\lambda=0) = \frac{C}{4\pi} \left[(g_V^f)^2 + (g_A^f)^2 \right] \left(\frac{1-z}{z}\right), \qquad (2.241)$$

where for $V = W^{\pm}$ we have

$$C = \frac{g^2}{8}, \quad g_V = -g_A = 1, \tag{2.242}$$

and for V = Z

$$C = \frac{g^2}{\cos^2 \theta_W}, \quad g_V = \frac{1}{2}(T_f^3) - Q_f \sin^2 \theta_W, \quad g_A = \frac{-1}{2}(T_f^3). \tag{2.243}$$

Summing and averaging the transverse W^{\pm} distributions gives

$$f_{W_T/q}(z, Q_V^2) = \frac{C}{8\pi^2} \frac{\left[1 + (1-z)^2\right]}{z} \log\left(\frac{Q_V^2}{M_W^2}\right), \qquad (2.244)$$

2.6.4 Beyond Leading Logarithm

The above *leading logarithm* (LL) result, gives us a result of the form

$$eX$$
-scattering = $e\gamma$ -splitting $\otimes \gamma X$ -scattering. (2.245)

However, the $e\gamma$ -splitting function is universal and will reappear for each successive splitting. This is particularly important for when the momentum transfer is very large, in which case

$$\alpha(Q^2) \log\left(\frac{Q^2}{m_f^2}\right) \sim 1, \tag{2.246}$$

and our perturbative treatment breakdowns. Though $\alpha(M_Z) \sim 1/128$ is quite small, is becomes a considerable problem from QCD where $\alpha_s(M_Z) \sim 0.1$. The solution is actually to consider summing over an arbitrary number of parton splittings

$$qX-\text{scattering} = \sum_{k=1}^{n} \underbrace{qg - \text{splitting} \otimes \cdots qg - \text{splitting}}_{k-\text{splittings}} \otimes gX-\text{scattering.}$$
(2.247)

The result is an expression that can be exponentiated

$$\sum \alpha^k(Q^2) \log^k \left(\frac{Q^2}{m_f^2}\right) \sim \exp\left[\alpha(Q^2) \log\left(\frac{Q^2}{m_f^2}\right)\right].$$
(2.248)

This process, only given schematically here, is called *resummation* and is an all-orders, hence non-perturbative, result. For the the case of QCD, collinear radiation is "resummed" using the Dokshitzer-Gribov-Lipatov-Altarelli-Parisi (DGLAP) equations, resulting in what are known as the *parton distribution functions* (PDFs). These functions give the likelihood of observing a particular parton species, e.g., anti-strange quark or gluon, in a hadron, e.g., proton or Pb nuclei, possessing a fraction x of the hadron's energy at a momentum transfer of Q^2 . The distribution function $f_{\gamma/e}$ in Eq. (2.237) is another example of a PDF. By fixing the PDFs at a particular momentum transfer and energy fraction, in say deeply inelastic scattering (DIS) *ep* experiments, the DGLAP equations are used to evolve the PDFs to a different scale, such as those observed in LHC collisions or potentially at a future 100 TeV collider.

2.6.5 Elastic Photon PDF

It is worth noting that our previous results were reliant on the parton model and dealt with pointparticles. As the proton is charged, at momentum transfers below a couple GeV, it too can give rise initial-state photons in pX collisions. The elastic photon PDF for a proton is given analytically by [47]

$$f_{\gamma/p}^{\rm El}(\xi) = \frac{\alpha_{\rm EM}}{\pi} \frac{(1-\xi)}{\xi} \left[\varphi\left(\frac{\Lambda_{\gamma}^{\rm El^2}}{Q_0^2}\right) - \varphi\left(\frac{Q_{\rm min}^2}{Q_0^2}\right) \right], \quad \alpha_{\rm EM} \approx 1/137, \tag{2.249}$$

$$Q_{\min}^2 = m_p^2 y, \quad y = \frac{\xi^2}{(1-\xi)}, \quad Q_0^2 = 0.71 \text{ GeV}^2, \quad m_p = 0.938 \text{ GeV},$$
 (2.250)

$$\varphi(x) = (1+ay) \left[-\log\left(1+\frac{1}{x}\right) + \sum_{k=1}^{3} \frac{1}{k(1+x)^{k}} \right] + \frac{y(1-b)}{4x(1+x)^{3}} + c\left(1+\frac{y}{4}\right) \left[\log\left(\frac{1+x-b}{1+x}\right) + \sum_{k=1}^{3} \frac{b^{k}}{k(1+x)^{k}} \right], \qquad (2.251)$$

$$a = \frac{1}{4}(1+\mu_p^2) + \frac{4m_p^2}{Q_0^2} \approx 7.16, \quad b = 1 - \frac{4m_p^2}{Q_0^2} \approx -3.96, \quad c = \frac{\mu_p^2 - 1}{b^4} \approx 0.028(2.252)$$

Here, $\Lambda_{\gamma}^{\text{El}}$ is a upper limit on elastic momentum transfers such that $f_{\gamma/p}^{\text{El}} = 0$ for $Q_{\gamma} > \Lambda_{\gamma}^{\text{El}}$. In Eq. (2.249), and later in Eq. (2.270), since $Q_{\gamma} \ll m_Z$, $\alpha(\mu = Q_{\gamma}) \approx \alpha_{\text{EM}} \approx 1/137$ is used. In the hard scattering matrix elements, $\alpha(\mu = M_Z)$ is used. See Ref. [48] for further details.

Equation (2.249) has been found to agree well with data from TeV-scale collisions at $Q_{\gamma} \sim m_{\mu}$ [49]. However, applications to cases with larger momentum transfers and finite angles lead to large errors and increase scale sensitivity. Too large a choice for $\Lambda_{\gamma}^{\text{El}}$ will lead to overestimate of cross sections [47]. However, we observe negligible growth in f_{γ}^{El} at scales well above $\Lambda_{\gamma}^{\text{El}} = 1 - 2 \text{ GeV}$, in agreement with Ref. [50].

Briefly, we draw attention to a typo in the original manuscript that derives Eq. (2.249). This has been only scantly been mentioned in past literature [51,52]. The sign preceding the "y(1-b)" term of φ in Eq. (2.251) is erroneously flipped in Eq. (D7) of Ref. [47]. Both CalcHEP [53–55] and MG5_aMC@NLO [56] have the correct sign in their default PDF libraries.

At these scales, the gauge state γ is a understood to be a linear combination of discrete states: the physical (massless) photon and (massive) vector mesons ($\omega, \phi, ...$), and a continuous mass spectrum, a phenomenon known as generalized vector meson dominance (GVMD) [57]. An analysis of ZEUS measurements of the F_2 structure function at $Q_{\gamma}^2 < m_p^2$ and Bjorken- $x \ll 1$ concludes that GMVD effects are included in the usual dipole parameterizations of the proton's electric and magnetic form factors $G_{\rm E}$ and $G_{\rm M}$ [58]. Thus, the radiation of vector mesons by a proton that are then observed as photon has been folded into Eq. (2.249).

2.7 FACTORIZATION THEOREM, PARTON LUMINOSITIES, AND HADRONIC CROSS SECTION

We are now in position to introduce hadronic level cross sections and the Factorization theorem. We start by considering some partonic-level process

$$a + b \to X.$$
 (2.253)

We suppose that both a and b are massless and possess proton PDFs, denoted by $f_{a/p}(\xi_1, \mu^2)$ and $f_{b/p}(\xi_2, \mu^2)$, where ξ_i is the energy fraction of proton P_i , and $f_{i/p}$ are evolved to factorization scale μ^2 . The partonic center of mass is denoted as $\hat{s} = (p_a + p_b)^2$, and the minimal invariant mass needed for the process to proceed is denoted by \hat{s}_X . The $pa \to X + Y$ scattering rate is then

$$\sigma(pa \to X + Y) = \int_{\xi_2^{\min}}^1 d\xi_2 \ f_{b/p}(\xi_2) \ \hat{\sigma}(ab \to X), \tag{2.254}$$

where

$$\hat{s} = (p_a + p_b)^2 = 2p_a p_b = 2P_2 p_a \xi_2 = s_{pa} \xi_2$$
 (2.255)

is the relation between the partonic and the p-a system's c.m. energies. This last line implies

$$\xi_2^{\min} = \min\left(\frac{\hat{s}}{s_{pa}}\right) = \frac{\hat{s}^{\min}}{s_{pa}} = \frac{m_X^2}{s_{pa}}, \text{ and } s_{pb}^{\min} = \min\left(\frac{\hat{s}}{\xi_a}\right) = m_X^2.$$
 (2.256)

Similarly, we can construct the $pp \to X + Y''$ scattering rate from the semi-partonic pb initial-states. The corresponding limits of integration for the splitting function integrals are

$$\xi_1^{\min} = \min\left(\frac{s_{pa}}{s_{pp}}\right) = \frac{s_{pa}^{\min}}{s_{pp}} = \frac{m_X^2}{s_{pp}} \equiv \tau_{\min}$$
(2.257)

$$\xi_2^{\min} = \frac{m_X^2}{s_{pb}} = \frac{m_X^2}{s_{pp}\xi_1} = \tau_{\min}/\xi_1.$$
(2.258)

This gives us

$$\sigma(pp \to X + Y'') = \int_{\xi_1^{\min}}^1 d\xi_1 \ f_{a/p}(\xi_1) \ \hat{\sigma}(pb \to X + Y), \tag{2.259}$$

$$= \int_{\xi_1^{\min}}^1 d\xi_1 \int_{\xi_2^{\min}}^1 d\xi_2 f_{a/p}(\xi_1) f_{b/p}(\xi_2) \hat{\sigma}(ab \to X), \qquad (2.260)$$

$$= \int_{\tau_{\min}}^{1} d\xi_1 \int_{\tau_{\min}/\xi_1}^{1} d\xi_2 \ f_{a/p}(\xi_1) f_{b/p}(\xi_2) \ \hat{\sigma}(ab \to X).$$
(2.261)

However, our assignment of a to the first proton and b to the second proton was arbitrary and indistinguishable from the reverse assignment. Thus, we obtain the **Factorization Theorem**

$$\sigma(pp \to X + Y'') = \int_{\tau_{\min}}^{1} d\xi_1 \int_{\tau_{\min}/\xi_1}^{1} d\xi_2 \left[f_{a/p}(\xi_1, \mu^2) f_{b/p}(\xi_2, \mu^2) \hat{\sigma}(ab \to X) + (a \leftrightarrow b) \right], \quad (2.262)$$

where τ is the minimal energy fraction required for the process to be kinematically allowed

$$\tau \equiv \frac{\hat{s}}{s} = \xi_1 \xi_2, \quad \tau_{\min} = \frac{\hat{s}_{\min}}{s},$$
(2.263)

and states that a sufficiently inclusive hadronic-level scattering at sufficiently large momentum transfers can be expressed as a convolution of the partonic-level scattering with the probability (PDFs) of observing the participating partons in the hadron. In other words, the likelihood of observing a particular process in hadron collisions can be obtained by "multiplying" (convolving) the probabilities of partons reproducing the desired final-state and the likelihood of finding said probabilities in the scattering hadrons.

Using the relationship $\tau = \xi_1 \xi_2$, we can make the change of variable

$$\int_{\xi_2^{\min}}^1 d\xi_2 = \int_{\tau_{\min}}^1 \frac{d\tau}{\xi_1},$$
(2.264)

allowing us to write

$$\sigma(pp \to X + Y'') = \int_{\tau_{\min}}^{1} d\tau \int_{\tau}^{1} \frac{d\xi_1}{\xi_1} \left[f_{a/p}(\xi_1) f_{b/p}(\xi_2) \hat{\sigma}(ab \to X) + (1 \leftrightarrow 2) \right].$$
(2.265)

For $2 \rightarrow 1$ processes, this readily simplifies to

$$\sigma(pp \to X + Y'') = \int_{\tau_{\min}}^{1} d\tau \int_{\tau}^{1} \frac{d\xi_{1}}{\xi_{1}} \left[f_{a/p}(\xi_{1}) f_{b/p}(\xi_{2}) \frac{d\hat{\sigma}}{dPS_{1}} \int dPS_{1} + (1 \leftrightarrow 2) \right]$$
(2.266)

$$= \int_{\tau_{\min}}^{\tau} d\tau \int_{\tau}^{\tau} \frac{d\xi_1}{\xi_1} \left[f_{a/p}(\xi_1) f_{b/p}(\xi_2) \frac{d\sigma}{dPS_1} \frac{2\pi}{s} \delta(\tau - \tau_{\min}) + (1 \leftrightarrow 2) \right] 2.267)$$

$$= \frac{2\pi}{s} \int_{\tau_{\min}}^{1} \frac{d\xi_1}{\xi_1} \left[f_{a/p}(\xi_1) f_{b/p}(\xi_2) \frac{d\hat{\sigma}}{dPS_1} + (1 \leftrightarrow 2) \right].$$
(2.268)

2.7.1 Inelastic Photon PDF

Following the methodology of Ref. [48], we can extend our discussion on initial-state photons from electrons and protons in Section 2.6 to initial state photons from quarks in protons. The inelastic cross section for producing final-state X is given explicitly by

$$\sigma_{\text{Inel}}(pp \to X + \text{anything}) = \sum_{q,q'} \int_{\tau_0}^1 d\xi_1 \int_{\tau_0/\xi_1}^1 d\xi_2 \int_{\tau_0/\xi_1/\xi_2}^1 dz$$
$$\times \left[f_{q/p} \left(\xi_1, Q_f^2 \right) f_{\gamma/q'} \left(z, Q_\gamma^2 \right) f_{q'/p} \left(\xi_2, Q_f^2 \right) \hat{\sigma} \left(q_1 \gamma_2 \right) + (1 \leftrightarrow 2) \right], \quad (2.269)$$
$$\tau_0 = m_X^2/s, \quad \tau = \hat{s}/s = \xi_1 \xi_2 z.$$

The Weizsäcker-Williams photon structure function [44, 45] is given by

$$f_{\gamma/q}(z,Q_{\gamma}^2) = \frac{\alpha_{\rm EM} \ e_q^2}{2\pi} \left(\frac{1+(1-z)^2}{z}\right) \log\left(\frac{Q_{\gamma}^2}{\Lambda_{\gamma}^{\rm Inel}}\right), \quad \alpha_{\rm EM} \approx 1/137, \tag{2.270}$$

where $e_q^2 = 4/9$ (1/9) for up-(down-)type quarks and $\Lambda_{\gamma}^{\text{Inel}}$ is a low-momentum transfer cutoff. In DGLAP-evolved photon PDFs [59], $\Lambda_{\gamma}^{\text{Inel}}$ is taken as the mass of the participating quark. Ref. [48] argues a low-energy cutoff $\mathcal{O}(1-2)$ GeV so that the associated photon is sufficiently off-shell for the parton model to be valid. Taking $\Lambda_{\gamma}^{\text{Inel}} = \Lambda_{\gamma}^{\text{El}} = \mathcal{O}(1-2)$ GeV allows for the inclusion of non-perturbative phenomena without worry of double counting of phase space [3].

Fixing z and defining $\xi_{\gamma} \equiv \xi_2 z$, we have the relationships

$$\tau_0 = \min\left(\xi_1 \xi_2 z\right) = \min\left(\xi_1 \xi_\gamma\right) \implies \min\left(\xi_\gamma\right) = \frac{\tau_0}{\xi_1} \text{ for fixed } \xi_1. \tag{2.271}$$

Physically, ξ_{γ} is the fraction of proton energy carried by the initial-state photon. Eq. (2.270) can be expressed into the more familiar two-PDF factorization theorem, i.e., Eq. (2.262), by grouping together the convolutions about $f_{q'/p}$ and $f_{\gamma/q'}$:

$$\sum_{q'} \int_{\tau_0/\xi_1}^1 d\xi_2 \int_{\tau_0/\xi_1/\xi_2}^1 dz \ f_{\gamma/q'}(z) \ f_{q'/p}(\xi_2) = \sum_{q'} \int_{\tau_0/\xi_1}^1 \frac{d\xi_\gamma}{z} \int_{z_{\min}}^1 dz \ f_{\gamma/q'}(z) \ f_{q'/p}\left(\frac{\xi_\gamma}{z}\right)$$
(2.272)
$$= \int_{z_{\min}}^1 d\xi_2 \ f_{\min}(\xi_1)$$
(2.273)

$$= \int_{\tau_0/\xi_1}^{\tau} d\xi_{\gamma} f_{\gamma/p}^{\text{Inel}}(\xi_{\gamma})$$
(2.273)

$$f_{\gamma/p}^{\text{Inel}}\left(\xi_{\gamma}, Q_{\gamma}^{2}, Q_{f}^{2}\right) \equiv \sum_{q'} \int_{z_{\min}=\xi_{\gamma}}^{1} \frac{dz}{z} f_{\gamma/q'}\left(z, Q_{\gamma}^{2}\right) f_{q'/p}\left(\frac{\xi_{\gamma}}{z}, Q_{f}^{2}\right) 2.274)$$

The minimal fraction z of energy that can be carried away by the photon from the quark corresponds to when the quark has the maximum fraction ξ_2 of energy from its parent proton. Thus, for a fixed ξ_{γ} , we have

$$1 = \max(\xi_2) = \max\left(\frac{\xi_{\gamma}}{z}\right) = \frac{\xi_{\gamma}}{\min(z)} \implies \min(z) = \xi_{\gamma}.$$
 (2.275)

The resulting expression is

$$\sigma_{\text{Inel}}(pp \to N\ell^{\pm}X) = \sum_{q} \int_{\tau_0}^{1} d\xi_1 \int_{\tau_0/\xi_1}^{1} d\xi_2 \left[f_{q/p}\left(\xi_1, Q_f^2\right) f_{\gamma/p}^{\text{Inel}}\left(\xi_2, Q_\gamma^2, Q_f^2\right) \hat{\sigma}\left(q_1\gamma_2\right) + (1 \leftrightarrow 2) \right]$$

Real, initial-state photons from inelastic quark emissions can be studied in MG5 by linking the appropriate Les Houches accord PDFs (LHAPDF) libraries [60] and using the MRST2004 QED [59] or NNPDF QED [61] PDF sets. With this prescription, sub-leading (but important) photon substructure effects [62], e.g., $P_{g\gamma}$ splitting functions, are included in evolution equations.



Figure 5: Parton luminosity as a function of $\sqrt{\tau}$ at (a) 14 TeV and (b) 100 TeV.

2.7.2 Parton Luminosities

From the Factorization Theorem, we can extract the *parton luminosity* \mathcal{L} , which is a measure of the parton-parton flux in hadron collisions. Parton luminosities are given in terms of the PDFs $f_{i,j/p}$ by the expression

$$\Phi_{ij}(\tau) \equiv \frac{d\mathcal{L}_{ij}}{d\tau} = \frac{1}{1+\delta_{ij}} \int_{\tau}^{1} \frac{d\xi}{\xi} \left[f_{i/p}(\xi, Q_f^2) f_{j/p}\left(\frac{\tau}{\xi}, Q_f^2\right) + (i \leftrightarrow j) \right], \qquad (2.276)$$

where for a process

$$i + j \to X,$$
 (2.277)

we have

$$\sigma(pp \to X + Y) = \sum_{i,j} \int_{\tau_0}^{1} d\xi_a \int_{\frac{\tau_0}{\xi_a}}^{1} d\xi_b \left[f_{i/p}(\xi_a, \mu^2) f_{j/p}(\xi_b, \mu^2) \hat{\sigma}(ij \to X) + (i \leftrightarrow j) \right] (2.278) \\
= \int_{\tau_0}^{1} d\tau \sum_{ij} \frac{d\mathcal{L}_{ij}}{d\tau} \hat{\sigma}(ij \to X). \tag{2.279}$$

In Fig. 5, we plot the parton luminosities for various initial-state pairs in
$$\sqrt{s} = 14$$
 and 100 TeV pp collisions. We include the light quarks (u, d, c, s) and adopt the 2010 update of the CTEQ6L PDFs [63]. We evolve the quark PDFs to half the total partonic energy,

$$Q_f = \frac{\sqrt{\hat{s}}}{2}.\tag{2.280}$$

2.7.3 Parton-Vector Boson Luminosities

We can extend the definition of parton luminosities to quark-V-scattering where V is a spin-1 vector boson that collinearly splits from initial parton i by making the replacement in Eq. (2.278)

$$f_{i/p}(\xi, Q_f^2) \to f_{V/p}(\xi, Q_V^2, Q_f^2),$$
 (2.281)

$$f_{V/p}(\xi, Q_V^2, Q_f^2) = \sum_q \int_{\xi}^{1} \frac{dz}{z} f_{V/q}(z, Q_V^2) f_{q/p}\left(\frac{\xi}{z}, Q_f^2\right)$$
(2.282)

resulting in the following qV luminosity formula

$$\Phi_{qV}(\tau) = \int_{\tau}^{1} \frac{d\xi}{\xi} \int_{\tau/\xi}^{1} \frac{dz}{z} \sum_{q'} \left[f_{q/p}(\xi) f_{V/q'}(z) f\left(\frac{\tau}{\xi z}\right) + f_{q/p}\left(\frac{\tau}{\xi z}\right) f_{V/q'}(z) f_{q'/p}(\xi) \right].$$
(2.283)

We plot the qV parton luminosities at 14 and 100 TeV pp collisions in Fig. 5 and observe that the luminosities are typically $\sim \alpha$ smaller than the qq rates.

2.7.4 Vector Boson Scattering: Double Initial-State Parton Splitting

We further extend luminosities to initial-state VV' scattering by making a substitution of initialstate parton j in Eq. (2.278):

$$f_{j/p}(\xi, Q_f^2) \to f_{V/p}(\xi, Q_V^2, Q_f^2).$$
 (2.284)

The resulting luminosity expression is

$$\Phi_{VV'}(\tau) = \frac{1}{(\delta_{VV'}+1)} \int_{\tau}^{1} \frac{d\xi}{\xi} \int_{\tau/\xi}^{1} \frac{dz_1}{z_1} \int_{\tau/\xi/z_1}^{1} \frac{dz_2}{z_2} \sum_{q,q'}$$

$$\times \left[f_{V/q}(z_2) f_{V'/q'}(z_1) f_{q/p}(\xi) f_{q'/p}\left(\frac{\tau}{\xi z_1 z_2}\right) + f_{V/q}(z_2) f_{V'/q'}(z_1) f_{q/p}\left(\frac{\tau}{\xi z_1 z_2}\right) f_{q'/p}(\xi) \right]$$
(2.285)

We plot the WW parton luminosities at 14 and 100 TeV pp collisions in Fig. 5.

2.8 STATISTICS

2.8.1 Poisson Statistics

To determine the discovery potential at a particular significance, we first translate significance into a corresponding confidence level (CL),¹ e.g.,

$$2\sigma \leftrightarrow 95.45\%$$
 CL, $3\sigma \leftrightarrow 99.73\%$ CL, $5\sigma \leftrightarrow 99.9999\%$ CL. (2.286)

¹We use σ -sensitivity and CL interchangeably in the text.

Given an given integrated luminosity \mathcal{L} , SM background rate σ_{SM} , and CL, say 95.45% CL, we solve for the maximum number of background-only events, denoted by n^b , using the Poisson distribution:

$$0.9545 = \sum_{k=0}^{n^{b}} P\left(k|\mu^{b} = \sigma_{\rm SM}\mathcal{L}\right) = \sum_{k=0}^{n^{b}} \frac{(\sigma_{\rm SM}\mathcal{L})^{k}}{k!} e^{-\sigma_{\rm SM}\mathcal{L}}.$$
(2.287)

The requisite number of signal events at a 95.45% CL (or 2σ significance) is obtained by solving for the mean number of signal events μ^s such that a mean number of total expected events ($\mu^s + \mu^b$) will generate n^b events only 4.55% (= 100% - 95.45%) of the time, i.e., find μ^s such that

$$P\left(k \ge n^b | \mu = \mu^s + \mu^b\right) = \frac{(\mu^s + \mu^b)^{n^b}}{(n^b)!} e^{-(\mu^s + \mu^b)} = 0.455.$$
(2.288)

The 2σ sensitivity to nonzero $S_{\ell\ell}$ is then

$$S_{\ell\ell'}^{2\sigma} = \frac{\mu^s}{\mathcal{L} \times \sigma_{\text{Tot } 0}}.$$
(2.289)

For fixed signal σ_s and background $\sigma_{\rm SM}$ rates, $\mu^s + \mu^b = (\sigma_s + \sigma_{\rm SM}) \times \mathcal{L}$. The required luminosity for a 2σ discovery can then be obtained by solving Eq. (2.288) for \mathcal{L} .

2.8.2 Gaussian Statistics

In the large event limit, we approach Gaussian statistics and the uncertainty greatly simplifies. For given number of expected SM background events N_b and number of actual observed events N_o , the significance estimator is given by

$$\sigma = \frac{N_o - N_b}{\sqrt{N_o}}.\tag{2.290}$$

For a new physics signal, we may replace $N_o - N_b$ by N_s , the number of signal events after a luminosity \mathcal{L} . Both the signal and background processes then have corresponding cross sections, labeled by σ_s and σ_b . The significance can then expressed as

$$\sigma = \frac{N_s}{\sqrt{N_s + N_b}} = \frac{\sigma_s \mathcal{L}}{\sqrt{\sigma_s \mathcal{L} + \sigma_b \mathcal{L}}} = \frac{\sigma_s}{\sqrt{\sigma_s + \sigma_b}} \sqrt{\mathcal{L}},$$
(2.291)

indicating a power-law growth in significance as a function of data.

3.0 HIGGS BOSONS FROM THE TOP DECAYS

3.1 INTRODUCTION

The discovery of a light, Standard Model (SM)-like Higgs boson at the CERN Large Hadron Collider (LHC) [26, 27] is a tremendous step towards understanding the underlying mechanism of electroweak (EW) symmetry breaking (EWSB). The observed signal, consistent with the leading production mechanism $gg \to h$, indicates the existence of the Higgs boson coupling to the topquark [64]. Ultimately, the $t\bar{t}h$ coupling may be determined at the LHC luminosity upgrade and at a high energy e^+e^- linear collider [5]. Regardless of their rarity, a Higgs boson that is less massive than the top quark implies that $t \to h$ transitions exist. With an annual luminosity of $\mathcal{L} = 100 \text{ fb}^{-1}/\text{yr}$, the 14 TeV LHC will produce over 90 million $t\bar{t}$ pairs a year [65]. Thus, searches for $t \to h$ transitions that are sensitive to new physics scenarios are an essential part of the LHC program. For example: the rare decay involving the Flavor Changing Neutral Current (FCNC)

$$t \to ch.$$
 (3.1)

This process is particularly interesting for several reasons. At leading order, it is induced at oneloop in the SM and, due to GIM suppression [35, 66, 67], its branching fraction is very small, about 10^{-14} . NLO QCD contributions increase this by 10% [68]. However, new physics beyond the SM (BSM), such as an extended Higgs sector [66, 69-72] or Supersymmetry (SUSY) [73-75], can significantly enhance this decay, making it a very sensitive channel to new physics.

In this study, we consider another $t \to h$ transition:

$$t \to W^* b \ h, \tag{3.2}$$

where the off-shell W^* decays to a pair of light fermions. We now know that this is kinematically allowed in the SM. Proceeding at tree-level through the diagrams depicted in Fig. 6, Eq. (3.2) has been previously evaluated [76–82]. Both the $t\bar{t}h$ and WWh interactions are simultaneously involved, resulting in a certain subtle, but accidental, cancellation. The predicted branching fraction in the SM is about 10^{-9} . Though still small, the rate is significantly larger than that of Eq. (3.1), thereby representing the leading $t \to h$ transition in the SM. Subsequently, we are motivated to investigate how sensitive Eq. (3.2) is to new physics.

To systematically quantify this sensitivity in a model-independent fashion, we first employ the approach of Effective Field Theory (EFT). In particular, we consider the effects of gauge invariant, dimension-six operators that can alter the $t\bar{t}h$ interaction and take into account constraints on anomalous $t\bar{t}h$ couplings imposed by data.

It is highly probable that the scalar sector responsible for the EWSB extends well beyond a solitary Higgs boson. For example: in the Two Higgs Doublet Model (2HDM), one of the best motivated SM extensions, an additional scalar $SU(2)_L$ doublet is introduced to facilitate EWSB. We extend our study into leading $t \rightarrow h$ transitions by considering CP-conserving variants of the so-called Type I and Type II 2HDM, denoted by 2HDM(I) and 2HDM(II), respectively. The corresponding decay channel is

$$t \to W^* b H \to f_1 \bar{f}_2 \ b H, \tag{3.3}$$

where H is generically either one of the two CP-even (h, H) or the CP-odd (A) Higgs bosons, and f_1, f_2 are the light fermions in the SM. For h/H, Eq. (3.3) proceeds identically though Fig. 6. For A, the middle diagram is absent.

The remainder of this analysis proceeds as follows: In section 3.2, we introduce our theoretical framework and comment on current experimental constraints for each new physics scenario. We then present in section 3.3 the SM, EFT, 2HDM(I), and 2HDM(II) predictions for the top quark branching fraction of Eq. (3.3) over respective parameter spaces. Observation prospects at present and future colliders are briefly addressed in section 3.4. Finally in section 3.5, we summarize our results and conclude.

3.2 THEORETICAL FRAMEWORK

The theoretical frameworks under consideration include the effective field theory (EFT) for $t\bar{t}h$ interactions up to dimension-six operators (Section 3.2.1), the two Higgs doublet model of Type I [2HDM(I)] (Section 3.2.3), and Type II [2HDM(II)] (Section 3.2.4). Current experimental constraints on the model parameters are also presented.



Figure 6: Feynman diagrams representing the leading transition $t \to H$ in Eqs. (3.2) and (3.3). Drawn using the package JaxoDraw [83].

3.2.1 The SM as an Effective Field Theory

To systematically search for new physics beyond the reach of present-day experiments, we employ Effective Field Theory (EFT) to model new physical phenomena and linearly realize the SM gauge symmetries [84–86]. After integrating out heavy degrees of freedom at a scale Λ , the low energy effects can be parameterized by

$$\mathcal{L} = \mathcal{L}_{\rm SM} + \mathcal{L}_{\rm Eff.}, \quad \mathcal{L}_{\rm Eff.} = \sum_{i,j} \frac{f_{i,j}}{\Lambda^i} \mathcal{O}_{i,j}, \qquad (3.4)$$

where \mathcal{L}_{SM} is the SM Lagrangian, the $f_{i,j}$ are real, dimensionless "anomalous couplings" naturally of order $1 \sim 4\pi$, and $\mathcal{O}_{i,j}$ represent $SU(3)_c \times SU(2)_L \times U(1)_Y$ -invariant, dimension-(4+i) operators constructed solely from SM fields. When $f_{i,j} \to 4\pi$, however, one is likely in the strong coupling regime and the EFT approach breaks down. Here, $f_{i,j}$ is assumed to be $\mathcal{O}(1)$. For the remainder of the text, we consider only the next-to-leading interactions at dimension-six and drop the i = 2subscript.

3.2.1.1 EFT framework and parameters Many linearly independent dimension-six operators can affect the $t\bar{t}h$, WWh, $b\bar{b}h$, tWb, htc/u, or 4-point tWbh vertices [84–94]. Results from the ATLAS and CMS experiments indicate that the WWh coupling is close to its SM value [26,27,64], and evidence suggest that the $b\bar{b}h$ coupling cannot be much larger than the SM prediction [95,96]. As dimension-six tWbh vertices originate from terms of the form tWb(v + h) [85,89], the size of anomalous 4-point tWbh couplings are restricted to be small by the stringent limits on anomalous tWb couplings [88,90,97–100]. Anomalous htc/u couplings are constrained to be small [72,101–103].

As we are interesting in the next-to-leading contribution to the $t \to W^*bh$ transition, we consider those operators affecting the weakly constrained $t\bar{t}h$ vertex only. In the basis of Ref. [85], the most general $t\bar{t}h$ interaction Lagrangian one can construct using linearly independent dimensionsix operators requires only two operators [92] (one CP-even and one CP-odd):

$$\mathcal{O}_{t1} = \left(\Phi^{\dagger}\Phi - \frac{v^2}{2}\right) \left(\overline{q_L}t_R\tilde{\Phi} + \tilde{\Phi}^{\dagger}\overline{t_R}q_L\right), \quad \overline{\mathcal{O}}_{t1} = i\left(\Phi^{\dagger}\Phi - \frac{v^2}{2}\right) \left(\overline{q_L}t_R\tilde{\Phi} - \tilde{\Phi}^{\dagger}\overline{t_R}q_L\right), \quad (3.5)$$

where Φ is the SM Higgs $SU(2)_L$ doublet with $U(1)_Y$ hypercharge +1,

$$v = \sqrt{2} \langle \Phi \rangle \approx 246 \text{ GeV}, \quad \overline{q_L} = (\overline{t_L}, \overline{b_L}), \quad \tilde{\Phi} = i\sigma_2 \Phi^*, \quad t_{L/R} = P_{L/R} t,$$
 (3.6)

and $P_{L/R} = \frac{1}{2}(1 \mp \gamma^5)$ is the left/right-handed (LH/RH) chiral projection operator. These respectively lead to anomalous scalar- and pseudoscalar-type interactions and correspond to the operator $Q_{u\varphi}$ in Refs. [86, 93], which assume *complex* Wilson coefficients. To investigate the sensitivity of operators that select out different kinematic features from those listed above, we consider also the two redundant¹ (CP-odd) operators

$$\overline{\mathcal{O}}_{\Phi q}^{(1)} = \left[\Phi^{\dagger}(D_{\mu}\Phi) + (D_{\mu}\Phi)^{\dagger}\Phi\right](\overline{q_{L}}\gamma^{\mu}q_{L}), \quad \overline{\mathcal{O}}_{t2} = \left[\Phi^{\dagger}(D_{\mu}\Phi) + (D_{\mu}\Phi)^{\dagger}\Phi\right](\overline{t_{R}}\gamma^{\mu}t_{R}), \quad (3.7)$$

which respectively lead to anomalous left/right-handed (LH/RH) chiral couplings. We do not consider other operators that can affect the $t \to W^*bh$ decay because their Wilson coefficients are strongly constrained by data.

After EWSB, the $t\bar{t}h$ interaction Lagrangian contains four² new independent terms:

$$\mathcal{L}_{tth} = -\frac{1}{\sqrt{2}}\bar{t}\left(y_t - g^S - ig^P\gamma^5\right)th + \left(\frac{\partial_\mu h}{v}\right)\bar{t}\gamma^\mu\left(g^L P_L + g^R P_R\right)t,\tag{3.8}$$

where y_t is the SM top quark Yukawa coupling,

$$y_t = \frac{gm_t}{\sqrt{2}M_W} \simeq 1, \tag{3.9}$$

and the anomalous couplings g^X beyond the SM (BSM) are

$$g^{S} = f_{t1} \frac{v^{2}}{\Lambda^{2}}, \quad g^{P} = \overline{f}_{t1} \frac{v^{2}}{\Lambda^{2}}, \quad g^{L} = \overline{f}_{\Phi q}^{(1)} \frac{v^{2}}{\Lambda^{2}}, \quad g^{R} = \overline{f}_{t2} \frac{v^{2}}{\Lambda^{2}}.$$
 (3.10)

¹ Using integration by parts and the appropriate equations of motion, e.g., $i \overrightarrow{\mathcal{D}} q_L = y_u u_R \tilde{\Phi} + y_d d_R \Phi$, one finds that the operator $\overline{\mathcal{O}}_{t2}$ is linearly dependent on \mathcal{O}_{t1} and $\overline{\mathcal{O}}_{t1}$ plus the bottom quark analogues. Similarly, $\overline{\mathcal{O}}_{\Phi q}^{(1)}$ is linearly dependent on \mathcal{O}_{t1} and $\overline{\mathcal{O}}_{t1}$ [86].

² The anomalous LH chiral $b\bar{b}h$ coupling from $\overline{\mathcal{O}}_{\Phi q}^{(1)}$ is ignored as its contribution suffers from kinematic and helicity suppression. See the discussion in Sections 3.3 and 3.3.1.

Operator	g^X Bound	$\Lambda/\sqrt{ f_{\mathcal{O}} } \; [\text{GeV}]$
\mathcal{O}_{t1}	$-0.72 < g^S < 0.21$	> 537
	$1.77 < g^S < 2.70$	150 - 185
$\overline{\mathcal{O}}_{t1}$	$-1.4 < g^P < 1.4$	> 208

Table 6: Bounds on EFT couplings

The relative minus signs between y_t and g^X are arbitrary due to the unknown couplings f. To better understand the influence of g^S and g^P on Eq. (3.2), it is useful to rewrite the relevant parts of Eq. (3.42) as

$$y_t - g^S - ig^P \gamma^5 = g^{\text{Eff.}} \left(e^{-i\delta_{\text{CP}}} P_R + e^{i\delta_{\text{CP}}} P_L \right), \qquad (3.11)$$

where the effective coupling, $g^{\text{Eff.}}$, and the CP-violating (CPV) phase, δ_{CP} , are

$$g^{Eff.} \equiv \sqrt{(y_t - g^S)^2 + g^{P/2}}, \quad \delta_{CP} \equiv \sin^{-1} \left[\frac{g^P}{\sqrt{(y_t - g^S)^2 + g^{P/2}}} \right].$$
 (3.12)

3.2.1.2 EFT Constraints Independent of deviations in the $h \rightarrow \gamma \gamma$ channel and with no assumption on the Higgs boson's total width, ATLAS has measured the gluon-gluon fusion (ggF) scale factor to be [64]

$$\kappa_g = 1.08^{+0.32}_{-0.14}, \quad \kappa_g^2 \equiv \sigma(gg \to h) / \sigma^{\text{SM}}(gg \to h).$$
(3.13)

Since ggF is dominated by a top quark loop, we can approximate an anomalous g^S contribution to the observed rate by

$$\sigma(gg \to h) = \kappa_g^2 \times \sigma^{SM}(gg \to h) \approx \frac{(y_t - g^S)^2}{y_t^2} \times \sigma^{SM}(gg \to h), \tag{3.14}$$

implying

$$g^S \in [-0.72, 0.21] \cup [1.77, 2.70]$$
at $2\sigma.$ (3.15)

Similarly, we can relate Eq. (3.13) to g^P by

$$\sigma(gg \to h) = \kappa_g^2 \times \sigma^{SM}(gg \to h) \approx \frac{y_t^2 + (g^P)^2}{y_t^2} \times \sigma^{SM}(gg \to h), \tag{3.16}$$

indicating

$$g^P \in [-1.41, 1.41]$$
 at 2σ . (3.17)

We next translate measurements of κ_g into bounds on the cutoff scale of new physics involving operators \mathcal{O}_{t1} and $\overline{\mathcal{O}}_{t1}$. The bounds on new physics scales $\Lambda/\sqrt{|f_{\mathcal{O}}|}$ are given in Table 6. With the Naive Dimensional Analysis (NDA) [104, 105] and $f_{\mathcal{O}} \sim \mathcal{O}(1)$, the new physics scale is pushed to about $\mathcal{O}(1 \text{ TeV})$. Translating limits on κ_g into bounds on $g^{L/R}$, and hence on $\overline{\mathcal{O}}_{\Phi q}^{(1)}$ and $\overline{\mathcal{O}}_{t2}$, is a nontrivial procedure due to the derivative coupling. Subsequently, such results are not presently available.

3.2.2 Linear Dependence of EFT Operators

Reference [93,94] argue that the operators

$$\mathcal{O}_{t1} = \left(\varphi^{\dagger}\varphi - \frac{v^2}{2}\right) \left(\overline{q_L} t_R \tilde{\varphi} + \tilde{\varphi}^{\dagger} \overline{t_R} q_L\right), \qquad \overline{\mathcal{O}}_{t1} = i \left(\varphi^{\dagger}\varphi - \frac{v^2}{2}\right) \left(\overline{q_L} t_R \tilde{\varphi} - \tilde{\varphi}^{\dagger} \overline{t_R} q_L\right)$$
$$\overline{\mathcal{O}}_{t2} = \left[\varphi^{\dagger} (D_{\mu}\varphi) + (D_{\mu}\varphi)^{\dagger}\varphi\right] (\overline{t_R}\gamma^{\mu} t_R), \qquad \overline{\mathcal{O}}_{\varphi q}^{(1)} = \left[\varphi^{\dagger} (D_{\mu}\varphi) + (D_{\mu}\varphi)^{\dagger}\varphi\right] (\overline{t_L}\gamma^{\mu} t_L), \quad (3.18)$$

are linearly dependent with respect to each other. Following the notation of Ref. [86], φ is the Higgs $SU(2)_L$ doublet, $\tilde{\varphi} = i\sigma^2\varphi^*$, $\overline{q_L} = (\overline{t_L}, \overline{b_L})$, and $t_{L/R} = P_{L/R}t$, where $P_{L/R} = \frac{1}{2}(1 \mp \gamma^5)$ is the LH/RH chiral projection operator. In this basis, all the operators above have real Wilson coefficients. These effective operators introduce (clockwise beginning from \mathcal{O}_{t1}) anomalous scalar, pseudoscalar, LH vector couplings, and RH vector couplings. The bottom two operators introduce derivative couplings of the form

$$(\partial_{\mu}h)\gamma^{\mu}P_{R/L}.$$
(3.19)

We will show that $\overline{\mathcal{O}}_{t2}$ is equivalent to the top two operators, up to an overall coefficient; we will also show that $\overline{\mathcal{O}}_{\varphi q}^{(1)}$ can be expressed in terms of the first two operators and the b_R analogue.

To demonstrate this, the equations of motion (EoM) for quarks will be necessary. They can be obtained from the SM Lagrangian:

$$\mathcal{L}_{Quarks} = i\overline{q_L} \mathcal{D} q_L + i\overline{u_R} \mathcal{D} u_R + i\overline{d_R} \mathcal{D} d_R \tag{3.20}$$

$$-\Gamma_{u}\overline{q_{L}}u_{R}\tilde{\varphi}-\Gamma_{d}\overline{q_{L}}d_{R}\varphi-\Gamma_{u}^{\dagger}\tilde{\varphi}^{\dagger}\overline{u_{R}}q_{L}-\Gamma_{d}^{\dagger}\varphi^{\dagger}\overline{d_{R}}q_{L},\qquad(3.21)$$

where Γ_f represent Yukawa couplings and, for $T^A = \frac{1}{2}\lambda^A$ and $S^I = \frac{1}{2}\sigma^I$,

$$(D_{\mu}q_L)^{\alpha j} = \left(\partial_{\mu} + ig_s T^A_{\alpha\beta} G^A_{\mu} + ig S^I_{jk} W^I_{\mu} + ig' Y_q B_{\mu}\right) q^{\beta k}, \tag{3.22}$$

with weak isospin and color indices j, k = 1, 2 and $\alpha, \beta = 1, 2, 3$.

Taking the appropriate functional derivatives, the EoMs can be obtained. For q_L , the RH up-type quark u_R and the RH down-type quark d_R these are

$$i\overrightarrow{\not{D}}q_L = \Gamma_u u_R \widetilde{\varphi} + \Gamma_d d_R \varphi \quad \iff \quad i\overline{q_L} \overleftarrow{\not{D}} = -i\overline{\left(\overrightarrow{\not{D}}q_L\right)} = -\Gamma_u^{\dagger} \widetilde{\varphi}^{\dagger} \overline{u_R} - \Gamma_d^{\dagger} \varphi^{\dagger} \overline{d_R} \qquad (3.23)$$

$$i\overrightarrow{\not{D}}u_R = \Gamma_u^{\dagger}\left(\widetilde{\varphi}^{\dagger}q_L\right) \quad \Longleftrightarrow \quad i\overline{u_R}\overleftarrow{\not{D}} = -i\overline{\left(\overrightarrow{\not{D}}u_R\right)} = -\Gamma_u\left(\overline{q_L}\widetilde{\varphi}\right) \tag{3.24}$$

$$i\overrightarrow{\mathcal{D}}d_R = \Gamma_d^{\dagger}\left(\varphi^{\dagger}q_L\right) \quad \Longleftrightarrow \quad i\overline{d_R}\overleftarrow{\mathcal{D}} = -i\overline{\left(\overrightarrow{\mathcal{D}}d_R\right)} = -\Gamma_d\left(\overline{q_L}\varphi\right) \tag{3.25}$$

3.2.2.1 \mathcal{O}_{t1} and $\overline{\mathcal{O}}_{t1}$ As noted above, \mathcal{O}_{t1} and $\overline{\mathcal{O}}_{t1}$ possess *real* Wilson coefficients. The operators in Ref. [86] possess *complex* coefficients, and so the operator

$$Q_{u\varphi} = \left(\varphi^{\dagger}\varphi\right) \left(\overline{q_L} d_R \tilde{\varphi}\right) \tag{3.26}$$

in Ref. [86] maps with a one-to-one correspondence to \mathcal{O}_{t1} and $\overline{\mathcal{O}}_{t1}$.

3.2.2.2 $\overline{\mathcal{O}}_{t2}$ For the operator $\overline{\mathcal{O}}_{t2}$, we see

$$\overline{\mathcal{O}}_{t2} = \left[\varphi^{\dagger}(D_{\mu}\varphi) + (D_{\mu}\varphi)^{\dagger}\varphi\right](\overline{t_R}\gamma^{\mu}t_R)$$
(3.27)

$$= \left[\varphi^{\dagger}(\partial_{\mu}\varphi) + (\partial_{\mu}\varphi)^{\dagger}\varphi\right](\overline{t_{R}}\gamma^{\mu}t_{R})$$
(3.28)

$$= \left[\partial_{\mu}(\varphi^{\dagger}\varphi)\right](\overline{t_{R}}\gamma^{\mu}t_{R}) \tag{3.29}$$

$$\stackrel{\text{IBP}}{=} \left[\int \dots \right] + (\varphi^{\dagger} \varphi) D_{\mu} \left(\overline{t_R} \gamma^{\mu} t_R \right)$$
(3.30)

$$= \left[\int \dots \right] + \left(\varphi^{\dagger} \varphi \right) \left[\left(\overline{t_R} \overleftarrow{\not} p t_R \right) + \left(\overline{t_R} \overrightarrow{\not} p t_R \right) \right]$$
(3.31)

$$= \left[\int \dots\right] + \left(\varphi^{\dagger}\varphi\right) \left[i\Gamma_{t}\left(\overline{q_{L}}\tilde{\varphi}\right)t_{R} + i\Gamma_{t}^{\dagger}\overline{t_{R}}\left(\tilde{\varphi}^{\dagger}q_{L}\right)\right]$$
(3.32)

$$= \left[\int \dots \right] + i \left[\Gamma_t Q_{u\varphi} + \text{H.c.} \right], \qquad (3.33)$$

where $[\int \dots]$ denotes a total derivative and has no observable effect on the physical amplitude. Subsequently, we see that this operator is proportional to $Q_{u\varphi}$ and its Hermitian conjugate, and hence is a linear combination of \mathcal{O}_{t1} and $\overline{\mathcal{O}}_{t1}$. **3.2.2.3** $\overline{\mathcal{O}}_{\varphi q}^{(1)}$ For the operator $\overline{\mathcal{O}}_{\varphi q}^{(1)}$, we obtain

$$\overline{\mathcal{O}}_{\varphi q}^{(1)} = \left[\varphi^{\dagger}(D_{\mu}\varphi) + (D_{\mu}\varphi)^{\dagger}\varphi\right](\overline{q_{L}}\gamma^{\mu}q_{L})$$
(3.34)

$$= \left[\partial_{\mu}(\varphi^{\dagger}\varphi)\right](\overline{q_{L}}\gamma^{\mu}q_{L}) \tag{3.35}$$

$$\stackrel{\text{IBP}}{=} \left[\int \dots \right] + (\varphi^{\dagger} \varphi) D_{\mu} \left(\overline{q_L} \gamma^{\mu} q_L \right)$$
(3.36)

$$= \left[\int \dots \right] + \left(\varphi^{\dagger} \varphi \right) \left[\left(\overline{q_L} \overleftarrow{\not{\mathcal{D}}} q_L \right) + \left(\overline{q_L} \overrightarrow{\not{\mathcal{D}}} q_L \right) \right]$$
(3.37)

$$= \left[\int \dots\right] + \left(\varphi^{\dagger}\varphi\right) \left[i\left(\Gamma_{t}^{\dagger} \tilde{\varphi}^{\dagger}\overline{t_{R}} + \Gamma_{b}^{\dagger} \varphi^{\dagger}\overline{b_{R}}\right)q_{L} - i\overline{q_{L}}\left(\Gamma_{t} t_{R}\tilde{\varphi} + \Gamma_{b} b_{R}\varphi\right)\right] \quad (3.38)$$

$$= \left[\int \dots\right] + i(\varphi^{\dagger}\varphi) \left[\Gamma_{t}^{\dagger}(\tilde{\varphi}^{\dagger}\overline{t_{R}}q_{L}) + \Gamma_{b}^{\dagger}(\varphi^{\dagger}\overline{b_{R}}q_{L}) - \Gamma_{t}(\overline{q_{L}}t_{R}\tilde{\varphi}) - \Gamma_{b}(\overline{q_{L}}b_{R}\varphi)\right] (3.39)$$

$$= \left[\int \dots \right] + i \left[\text{H.c.} - \Gamma_t Q_{u\varphi} - \Gamma_b Q_{d\varphi} \right]$$
(3.40)

where $Q_{d\varphi}$ is a Ref. [86] operator and

$$Q_{d\varphi} = \left(\varphi^{\dagger}\varphi\right) \left(\overline{q_L} u_R\varphi\right). \tag{3.41}$$

In this case, $\overline{\mathcal{O}}_{\varphi q}$ is linearly independent only because we do not include an operator analogous to $Q_{d\varphi}$. However, Ref. [92] points out that the most general *tth* Lagrangian constructed from the minimal set of dimension-6 operators has the form

$$\mathcal{L}_{tth} = -\frac{1}{\sqrt{2}}\bar{t}\left(Y_t^V + iY_t^P\gamma^5\right)th \qquad (3.42)$$

for real $Y_t^{V,A}$ because the derivative coupling terms identically vanish due to equations of motion.

The dimension-six operators used here are taken from from Whisnant, et al. [85]. However, the issue of redundant operators reported by Grzadkowski [86], Aguilar-Saavedra [92], and Einhorn & Wudka [93,94] appeared more than a decade after the Whisnant, et al.

3.2.3 Type I Two Higgs Doublet Model

In the generic CP-conserving 2HDM, EWSB is facilitated by two $SU(2)_L$ doublets, Φ_i , for $i \in \{1, 2\}$, each with $U(1)_Y$ hypercharge +1 and a nonzero vacuum expectation value (vev) v_i . A \mathbb{Z}_2 symmetry is applied for $\Phi_1 \leftrightarrow \Phi_2$ to eliminate tree-level FCNC but may be softly broken at loop-level. After EWSB, there are five physical spin-0 states: h, H, A, and H^{\pm} , which are respectively the two CP-even, single CP-odd, and $U(1)_{EM}$ charged Higgs bosons with masses m_h, m_H, m_A , and $m_{H^{\pm}}$. By convention, we fix the ordering of h and H by taking

$$m_h < m_H.$$

Vertex	SM	2HDM I	$\sin(\beta - \alpha) = 1 - \Delta_V$
$hu\overline{u}/d\overline{d}$	0 or 1	$\frac{\cos \alpha}{\sin \beta}$	$1 - \Delta_V + \sqrt{2\Delta_V - \Delta_V^2} \cot \beta$
hW^+W^-	0 or 1	$\sin(\beta - \alpha)$	$1 - \Delta_V$
$Hu\overline{u}/d\overline{d}$	0 or 1	$\frac{\sin \alpha}{\sin \beta}$	$(\Delta_V - 1)\cot\beta + \sqrt{2\Delta_V - \Delta_V^2}$
HW^+W^-	0 or 1	$\cos(\beta - \alpha)$	$\sqrt{2\Delta_V - \Delta_V^2}$
$Au\overline{u}$	-	\coteta	\coteta
$Ad\overline{d}$	-	$-\cot\beta$	$-\coteta$

Table 7: Neutral Scalar Boson Couplings in the 2HDM(I) Relative to the SM Higgs Couplings

Two angles, α and β , remain as free parameters. α measures the mixing between the two CP-even Higgs fields to form the mass eigenstates (h, H) and spans $\alpha \in [-\pi/2, \pi/2]$. β represents the relative size of $\langle \Phi_i \rangle$ and is defined by

$$\tan \beta \equiv \langle \Phi_2 \rangle / \langle \Phi_1 \rangle = v_2 / v_1, \quad \beta \in [0, \pi/2].$$
(3.43)

Reviews of various 2HDMs and their phenomenologies can be found in Refs. [106–108].

3.2.3.1 Type I 2HDM framework and parameters In the 2HDM(I), much like in the SM, only one Higgs doublet is responsible for generating fermion masses and couples accordingly; the second CP-even Higgs boson interacts with fermions through mixing. The interaction Lagrangian relevant to this study is

$$\mathcal{L} \ni - \frac{gm_u}{2M_W} \overline{u} \left(h \frac{\cos \alpha}{\sin \beta} + H \frac{\sin \alpha}{\sin \beta} - i\gamma^5 A \cot \beta \right) u - \frac{gm_d}{2M_W} \overline{d} \left(h \frac{\cos \alpha}{\sin \beta} + H \frac{\sin \alpha}{\sin \beta} + i\gamma^5 A \cot \beta \right) d + gM_W W_{\mu} W^{\mu} \left[h \sin(\beta - \alpha) + H \cos(\beta - \alpha) \right].$$
(3.44)

In Eq. (3.44), $u_{L(R)}$ is the LH (RH) up-type quark spinor, $d_{L(R)}$ is the down-type quark analogue, and g is the weak coupling constant in the SM.

Discovering a Higgs boson with SM-like couplings greatly impacts the 2HDM. In particular, the

measured couplings to weak bosons [26, 27, 64] imply either

$$\sin(\beta - \alpha) \approx 1$$
 for *h* to be SM-like, (3.45)

or
$$\cos(\beta - \alpha) \approx 1$$
 for *H* to be SM-like. (3.46)

Generally, we may parameterize how far $\sin(\beta - \alpha)$ is away from one and define Δ_V such that

$$\sin(\beta - \alpha) \equiv 1 - \Delta_V, \quad 0 \le \Delta_V \le 1. \tag{3.47}$$

We restrict the couplings to have the same sign as those of the SM [32] and limit Δ_V up to one. Eq. (3.47) maps to the parameterization used by the SFitter Collaboration [109] by taking $\Delta_V \rightarrow -\Delta_V$ and allowing $\Delta_V < 0$. After substituting α by Δ_V in Eq. (3.44), we have

$$\mathcal{L} \ni - \frac{gm_u}{2M_W} \overline{u} \left[h \left(1 - \Delta_V + \sqrt{2\Delta_V - \Delta_V^2} \cot \beta \right) + H \left((\Delta_V - 1) \cot \beta + \sqrt{2\Delta_V - \Delta_V^2} \right) \right] u - \frac{gm_d}{2M_W} \overline{d} \left[h \left(1 - \Delta_V + \sqrt{2\Delta_V - \Delta_V^2} \cot \beta \right) + H \left((\Delta_V - 1) \cot \beta + \sqrt{2\Delta_V - \Delta_V^2} \right) \right] d + \frac{gm_u}{2M_W} \overline{u} \left[i \gamma^5 A \cot \beta \right] u - \frac{gm_d}{2M_W} \overline{d} \left[i \gamma^5 A \cot \beta \right] d + gM_W W_\mu W^\mu \left[h (1 - \Delta_V) + H \sqrt{2\Delta_V - \Delta_V^2} \right].$$
(3.48)

Table 7 summarizes the bosonic and fermionic couplings to the neutral scalar in the 2HDM(I) relative to those in the SM, i.e., the 2HDM(I) coupling coefficient divided by the SM coupling coefficient. In the small (large) Δ_V limit, h(H) becomes SM-like and H(h) becomes non-SM-like. At $\Delta_V = 0$ ($\Delta_V = 1$), H(h) decouples from the gauge bosons. The relevant tree-level couplings to A are independent of Δ_V as they are initially independent of α . In the large tan β limit, A decouples from the theory. For all parameter scenarios considered, we identify the SM-like Higgs as the one with stronger couplings to WW, ZZ, and having a mass of 125.5 GeV.

3.2.3.2 Type I 2HDM Constraints Since the Higgs boson's discovery, many reports have appeared investigating the 2HDMs' compatibility with data [32, 109–122]. We list here constraints relevant to the 2HDM(I) and note when a result is applicable to other types. The following bounds assume one SM-like Higgs boson at approximately 126 GeV.

(i) $\cos(\beta - \alpha) - \tan\beta$ Parameter Space: A global fit of available LHC data, in particular from $h \to \gamma\gamma$, VV, $b\bar{b}$, $\tau^+\tau^-$, has set stringent bounds [121]. Representative values at 95%CL are

$$\cos(\beta - \alpha) < 0.3 \ (0.40) \ [0.42] \quad \text{for} \quad \tan \beta = 2.4 \ (10) \ [100].$$
 (3.49)

Similar conclusions have been reached by Refs. [115, 117, 119, 120, 122].

(ii) $m_{H^{\pm}} - \tan \beta$ Parameter Space: For all 2HDMs, flavor observables exclude at 95% CL [123,124]

$$\tan \beta < 1 \quad \text{for} \quad m_{H^{\pm}} < 500 \text{ GeV.}$$
(3.50)

Values of $\tan \beta < 1$ are allowed given a sufficiently heavy H^{\pm} [114, 116, 123, 124]. Due to the particular $\tan \beta$ dependence, no absolute lower bound on $m_{H^{\pm}}$ from flavor constraints exists in the 2HDM(I) [123]. An observation of excess $B \to D^* \tau \nu$ decays [125] has yet to be confirmed and is not considered.

(iii) Additional Higgs Masses: For both 2HDM(I) and (II), additional CP-even scalars below LEP bounds [126–128] are allowed given sufficiently decoupled H^{\pm} and A [112]. A second CP-even Higgs is incompatible with LHC data for mass

$$180 \text{ GeV} < m_H < 350 \text{ GeV}, \tag{3.51}$$

but allowed outside this range [113]. Direct searches for H^{\pm} and A exclude [127–130]

$$m_{H^{\pm}}, \ m_A \lesssim 80 \text{ GeV}.$$
 (3.52)

Additional considerations include the compatibility of a SM-like Higgs boson with EW precision data in general 2HDMs [110], the perturbative unitarity limits on the heavy Higgs masses in a general, CP-conserving 2HDM [118,131,132], and perturbative unitarity limits on tan β in an exact \mathcal{Z}_2 -symmetric, CP-conserving 2HDM [111,119]. Since FCNC do exist in nature and the SM, it is unnecessary to impose the severe constraints on tan β associated with an exact \mathcal{Z}_2 symmetry.

3.2.4 Type II Two Higgs Doublet Model

3.2.4.1 Type II 2HDM framework and parameters In the 2HDM(II), one Higgs doublet is assigned a hypercharge +1, giving masses to fermions with weak isospin $T_L^3 = +\frac{1}{2}$, and the second is assigned a hypercharge -1, giving masses to $T_L^3 = -\frac{1}{2}$ fermions. The doublets are denoted respectively by Φ_u and Φ_d , and β is written as

$$\tan \beta \equiv \langle \Phi_u \rangle / \langle \Phi_d \rangle = v_u / v_d. \tag{3.53}$$

After EWSB, the CP-conserving interaction Lagrangian relevant to Eq. (3.3) is similar to Eq. (3.44), with the only difference being the down-type quark Yukawa couplings:

$$\mathcal{L} \ni - \frac{gm_d}{2M_W} \overline{d} \left(-h \frac{\sin \alpha}{\cos \beta} + H \frac{\cos \alpha}{\cos \beta} - i\gamma^5 A \tan \beta \right) d.$$
(3.54)

Vertex	SM	2HDM II	$\sin(\beta - \alpha) = 1 - \Delta_V$
$hu\overline{u}$	0 or 1	$\frac{\cos \alpha}{\sin \beta}$	$1 - \Delta_V + \sqrt{2\Delta_V - \Delta_V^2} \cot\beta$
$hd\overline{d}$	0 or 1	$-\frac{\sin \alpha}{\cos \beta}$	$1 - \Delta_V - \sqrt{2\Delta_V - \Delta_V^2} \tan\beta$
hW^+W^-	0 or 1	$\sin(\beta - \alpha)$	$1 - \Delta_V$
$Hu\overline{u}$	0 or 1	$\frac{\sin \alpha}{\sin \beta}$	$(\Delta_V - 1) \cot \beta + \sqrt{2\Delta_V - \Delta_V^2}$
$Hd\overline{d}$	0 or 1	$\frac{\cos \alpha}{\cos \beta}$	$(1-\Delta_V)\tan\beta + \sqrt{2\Delta_V - \Delta_V^2}$
HW^+W^-	0 or 1	$\cos(\beta - \alpha)$	$\sqrt{2\Delta_V - \Delta_V^2}$
$Au\overline{u}$	_	\coteta	\coteta
$Ad\overline{d}$	_	an eta	aneta

Table 8: Neutral Scalar Boson Couplings in the 2HDM(II) Relative to the SM Higgs Couplings

The notation used in Eq. (3.54) is the same as the 2HDM(I) Lagrangian Eq. (3.44). Using Eq. (3.47), and similar to Eq. (3.48), the preceding line becomes

$$\mathcal{L} \ni - \frac{gm_d}{2M_W} \overline{d} \left[h \left(1 - \Delta_V - \sqrt{2\Delta_V - \Delta_V^2} \tan \beta \right) + H \left((1 - \Delta_V) \tan \beta + \sqrt{2\Delta_V - \Delta_V^2} \right) \right] d + i \frac{gm_d}{2M_W} \overline{d} \gamma^5 d A \tan \beta.$$
(3.55)

Table 8 summarizes the bosonic and fermionic couplings to the neutral scalars in the 2HDM(II) relative to those in the SM. Like the 2HDM(I), in the small (large) Δ_V limit, h (H) becomes SM-like and H (h) becomes non-SM-like. At $\Delta_V = 0$ ($\Delta_V = 1$), H (h) decouples from the gauge bosons. In this same limit, the h (H) Yukawa couplings become independent of tan β . Unlike the 2HDM(I), A only decouples from the theory if taken to be infinitely heavy.

An important feature for the Higgs couplings to fermions is that the down-type quark couplings are enhanced at higher values of $\tan \beta$, while the up-type quark couplings are suppressed. For the charged Higgs however, there is an interplay between the two and the particular value $\tan \beta = \sqrt{m_t^{\overline{\text{MS}}}(m_t)/m_b^{\overline{\text{MS}}}(m_t)} \approx 7.6$ minimizes the decay $t \to H^+b$. Though no such minima occur in the 2HDM(I), sensitivity to $\tan \beta = 7.6$ will be investigated in both 2HDM scenarios.

3.2.4.2 Type II 2HDM Constraints Constraints relevant to the 2HDM(II) are listed here. See Section 3.2.3 for generic 2HDM bounds.

(i) $\cos(\beta - \alpha) - \tan \beta$ Parameter Space: A global fit of available LHC data, in particular from $h \rightarrow \gamma \gamma$, VV, $b\bar{b}$, $\tau^+\tau^-$, has set stringent bounds [121]. Representative values at 95% CL are

$$\cos(\beta - \alpha) < 0.06 \ (0.01) \quad \text{for} \quad \tan \beta = 2.4 \ (10).$$
 (3.56)

Similar conclusions have been reached by Refs. [115, 117, 119, 120, 122].

(ii) $m_{H^{\pm}} - \tan \beta$ Parameter Space: Flavor observables, and in particular BR $(B \to X_s \gamma)$, exclude at 95% CL [124, 133, 134]

$$m_{H^{\pm}} < 327 \text{ GeV} \quad \text{for all } \tan \beta$$
 (3.57)

From BR($B \rightarrow \tau \nu$) measurements, the **UT***fit* Collaboration [135] has determined the absolute bound

$$\tan\beta < 7.4 \frac{m_{H^{\pm}}}{100 \text{ GeV}}.$$
(3.58)

3.3 BRANCHING RATIOS

The discovery of a SM-like Higgs boson at 126 GeV [26,27,64] implies that

$$t \to W^{+*}bh, \quad W^{+*} \to f_1\bar{f}_2$$

$$(3.59)$$

is kinematically allowed and proceeds through the diagrams given in Fig. 6. Following Ref. [81], we define the $t \to W^*bh$ partial width as

$$\Gamma(t \to Wbh) = \frac{\Gamma(t \to \mu^+ \nu_\mu bh)}{\mathrm{BR}(W \to \mu \nu_\mu)},\tag{3.60}$$

and the $t \to W^*bh$ branching ratio by

$$BR(t \to Wbh) = \frac{\Gamma(t \to Wbh)}{\Gamma_{Tot.}}, \quad \Gamma_{Tot.} \equiv \Gamma(t \to Wb).$$
(3.61)

With CalcHEP 3.4.2 [53–55], we find excellent agreement with Ref. [81]. With updated parameters [26, 27, 64, 136]:

$$m_t^{\rm MS}(m_t) = 173.5 \text{ GeV}, \quad m_b^{\rm MS}(m_t) = 3.01 \text{ GeV}, \quad m_h = 125.5 \text{ GeV}, \quad m_\mu = 0 \text{ GeV},$$

$$M_W = 80.385 \text{ GeV}, \quad M_Z = 91.1876 \text{ GeV}, \quad G_F = 1.1663787 \times 10^{-5} \text{ GeV}^{-1},$$

$$\Gamma_W = 2.085 \text{ GeV}, \quad \text{BR}(W \to \mu\nu) = 0.1057, \quad (3.62)$$

we calculate $\Gamma_{Tot.}$ at leading order to be

$$\Gamma_{\rm Tot.} = 1.509 \,\,{\rm GeV},$$
(3.63)

g^X		$BR(t \to Wbh)$
g^S	0.5	1.075×10^{-9}
	-0.5	3.078×10^{-9}
g^P	0.5	1.929×10^{-9}
	-0.5	1.928×10^{-9}
g^L	0.5	1.812×10^{-9}
	-0.5	1.812×10^{-9}
g^R	0.5	1.927×10^{-9}
	-0.5	1.928×10^{-9}

Table 9: BR $(t \to Wbh)$ for Benchmark Values of Anomalous $t\bar{t}h$ Couplings

and find that the SM predicts

$$BR^{SM}(t \to Wbh) = 1.80 \times 10^{-9}.$$
(3.64)

The smallness of this branching fraction falls from several features, including phase space suppression associated with the three-body final state, kinematic suppression due to the off-shell W boson, and an accidental cancellation between the leading $t\bar{t}h$ and subleasing WWh diagrams. Nevertheless, this decay rate is $\mathcal{O}(10^5)$ larger than the well-studied [66,67] two-body $t \to ch$ transition. This is due to the GIM suppression for the FCNC.

In the remainder of this section, we investigate how the branching fraction can change in the context of EFT, 2HDM(I), and 2HDM(II).

3.3.1 EFT $BR(t \rightarrow Wbh)$

We present first the behavior of BR $(t \to Wbh)$ as a function of anomalous $t\bar{t}h$ couplings. For one non-zero anomalous coupling from Eq. (3.42) at a time, we calculate the branching fraction over the domain $g^X \in [-2, +2]$ and set all other anomalous couplings to zero. Bounds on g^S and g^P , Eqs. (3.15) and (3.17) respectively, are applied. The results are shown in Fig. 7. To investigate the sensitivity of operators that select out different kinematic features, we include the redundant operators listed in Eq. (3.7), which give rise to anomalous g^L and g^R . In all plots, the SM prediction


Figure 7: BR $(t \to Wbh)$ as a function of (a) g^S , (b) g^P , (c) g^L , (d) g^R . The solid line denotes the SM prediction, Eq. (3.64). The shaded region is excluded at 95% C.L.

as in Eq. (3.64) is shown as a (black) solid line labeled by "SM". Table 9 lists values of the branching fraction for various benchmark values of g^X .

In Fig. 7(a), BR($t \to Wbh$) as a function of the anomalous scalar coupling g^S is shown. From the Lagrangian in Eq. (3.11), it is clear that $(y_t - g^S)$ acts as an effective Yukawa coupling. For $g^S < 0$, the anomalous coupling enhances the already dominant top-Higgsstrahlung diagram. For $g^S > 0$, an accidental cancellation among the anomalous scalar, Yukawa, and gauge terms results in a minimum at $g^S \approx 0.92$. When $g^S \gtrsim 0.92$, the quadratic term takes over and causes the branching fraction to grow. An observed transition rate smaller than the SM prediction thus implies that $g^S > 0$. Indirect measurements of the $t\bar{t}h$ coupling, as seen in Fig. 7(a), indicate that

$$BR(t \to Wbh) = (0.8 \sim 2.1) \times BR^{SM}(t \to Wbh).$$
(3.65)

Figure 7(b) shows the influence of an anomalous pseudoscalar coupling, g^P , on BR($t \to Wbh$). From the Lagrangian in Eq. (3.11), similar to the discussions in the previous session, the $t \to h$ transition is symmetric with respect to g^P due to the dominance of the quadratic term. Both couplings contribute greatest when the intermediate, off-shell top quark propagates in its RH helicity state, which gives an m_t enhancement over other diagrams. The CPV associated with $\delta_{\rm CP}$ is unobservable here because the asymmetry is proportional to interference terms, which are small.

The linear dependence on g^S in interference terms from the previous case and the strict quadratic dependence on g^P here implies that that branching fraction is less sensitive to small values of g^P than it is to small values of g^S . The rate therefore grows more slowly as a function of g^P than g^S . As seen in Figure 7(b), the bounds on g^P allow

$$BR(t \to Wbh) = (1 \sim 1.5) \times BR^{SM}(t \to Wbh).$$
(3.66)

In Fig. 7(c), we see the branching fraction as a function of an anomalous LH vector current with coupling g^L . Over the domain investigated, the contribution is rather small. We turn to kinematics to elucidate this behavior. First, the anomalous contribution is proportional to k_{μ}/v , where k_{μ} is the momentum of the Higgs. Since the energy budget for this process is fixed at m_t , and since we require a final state Higgs $(E_h \gtrsim m_h)$, $k_{\mu}/v \sim E_h/v$ ranges between 0.5 ~ 0.6, leading to kinematic suppression of anomalous contributions. Second, note that a fermion participating in two sequential LH chiral interactions necessarily propagates in its LH helicity state. Hence, the anomalous contribution is proportional to the internal, off-shell top quark momentum and leads to helicity suppression of anomalous contributions. We consequently expect and observe very small growth in the branching fraction over the range of g^L .

Figure 7(d) displays the results for BR($t \to Wbh$) as a function of anomalous RH vector current with coupling g^R . Unlike the LH case, the anomalous contribution has a large effect over the domain considered, comparable to g^S and g^P . As in the previous case, there is kinematic suppression; however, there is no longer helicity suppression. A massive fermion participating in a RH chiral interaction followed by a LH chiral interaction propagates in its RH helicity state. Hence, as in the g^P case, the anomalous contribution is proportional to m_t . Comparatively, there is a faster rise in the transition rate as a function of g^R than g^L .

H (125.5 GeV)	Δ_V	$\tan\beta$	$BR^{2HDM(I)}(t \to WbH)$
h	0.05	3	1.840×10^{-9}
h		7.6	1.714×10^{-9}
Н	0.7	3	1.460×10^{-9}
Н		7.6	1.567×10^{-9}
h	0.7	3	4.643×10^{-10}
h		7.6	2.573×10^{-10}
A (100 GeV)		3	1.814×10^{-9}
A (100 GeV)		7.6	2.829×10^{-10}

Table 10: $BR(t \rightarrow WbH)$ for Benchmark Values of Higgses in the 2HDM(I)



Figure 8: The 2HDM(I) BR($t \to WbH$) as a function of (a) $\tan \beta$ for SM-like *h* (long dash), *H* (dash-dot), and non-SM-like *h* (short dash), *H* (dot); (b) Δ_V for *h* at $\tan \beta = 3$ 7.6, 15 (short dash, long dash, dash-dot), and for *H* (dot, long dash, dash-dot). The solid line denotes the SM prediction, Eq. (3.64).

3.3.2 Type I 2HDM $BR(t \rightarrow WbH)$

The behavior of BR $(t \to WbH)$, where H represents h, H, or A in the 2HDM(I), is presented in this section. To explore sensitivity to the anomalous WWH coupling, Δ_V , we consider

$$\tan \beta = 3, 7.6, 15 \text{ for } \Delta_V \in [0, 1].$$
 (3.67)

For these values of $\tan \beta$, the largest deviation in the WWH coupling allowed by present data corresponds to a light SM-like Higgs with $\cos(\beta - \alpha) = 0.3$, i.e.,

$$\Delta_V = 0.05 \ (0.7) \quad \text{for} \quad h \ (H) \approx h^{SM}.$$
 (3.68)

To determine the mass sensitivity, we focus on the mass windows

$$m_h \in [95 \text{ GeV}, 126 \text{ GeV}], \ m_H \in [126 \text{ GeV}, 155 \text{ GeV}], \ m_A \in [95 \text{ GeV}, 155 \text{ GeV}].$$
 (3.69)

Below 95 GeV, the SM Z boson background becomes relevant, making observation of the transition very difficult; above 155 GeV the kinematic suppression of $t \to H/A$ becomes too great for practical purposes. However, it is straightforward to extrapolate these results in the event of a neutral scalar's discovery in these peripheral ranges.

Table 10 lists values of BR $(t \rightarrow WbH)$ for several Higgses and benchmark parameter values.

3.3.2.1 BR $(t \to Wbh, H)$ vs tan β , Δ_V The decay rates for $t \to W^*bh$ and $t \to W^*bH$ as a function of (a) tan β and (b) Δ_V are shown in Fig. 8. Except for low value of tan $\beta < 3$, the rates are always smaller than the SM rate. Beyond tan $\beta \approx 3$, the SM-like CP-even Higgs rates become independent of tan β and converge to asymptotic values; for the non-SM-like Higgses, this occurs at tan $\beta \approx 15$. To see how this happens, note that the Yukawa couplings in the 2HDM(I) (Table 7) take the simple form

$$c_1 \cot \beta + c_2, \tag{3.70}$$

where $c_{1,2}$ are elementary functions of Δ_V , as seen in Table 7. In the large $\tan \beta$ limit, the c_1 part vanishes, leaving the asymptotic value c_2 . In the SM-like limit, the c_2 terms are larger than the c_1 contributions, whereas the reverse holds in the non-SM-like limit. We extract asymptotic values by observing that for a given CP-even Higgs the c_2 terms and WWH couplings are the identical.



Figure 9: The 2HDM(I) BR($t \rightarrow WbH$) as a function of mass for a non-SM-like (a) h and (b) H assuming $\tan \beta = 3$, 7.6, 15 (short dash, long dash, dash-dot). The solid line denotes the SM prediction, Eq. (3.64).



Figure 10: The 2HDM(I) BR($t \to WbA$) as a function of (a) tan β and (b) m_A for tan $\beta = 3, 7.6, 15$ (short dash, long dash, dash-dot). The solid line denotes the SM prediction, Eq. (3.64).

Consequently,

$$\lim_{\tan\beta\to\infty} \mathrm{BR}^{2HDM(I)}(t\to Wbh) = (1-\Delta_V)^2 \mathrm{BR}^{SM}(t\to Wbh)$$
$$= \sin^2(\beta-\alpha) \mathrm{BR}^{SM}(t\to Wbh)$$
(3.71)

$$\lim_{\tan\beta\to\infty} \mathrm{BR}^{2HDM(I)}(t\to WbH) = (2\Delta_V - \Delta_V^2)\mathrm{BR}^{SM}(t\to Wbh)$$
$$= \cos^2(\beta - \alpha)\mathrm{BR}^{SM}(t\to Wbh). \tag{3.72}$$

For our choices of Δ_V , the asymptotic rates in Fig. 8(a) are

$$\lim_{\tan\beta\to\infty} \mathrm{BR}^{2HDM(I)}_{\Delta_V=0.7}(t\to WbH) = 0.910 \times \mathrm{BR}^{SM}(t\to Wbh), \tag{3.73}$$

$$\lim_{\tan\beta\to\infty} \mathrm{BR}^{2HDM(I)}_{\Delta_V=0.05}(t\to Wbh) = 0.903 \times \mathrm{BR}^{SM}(t\to Wbh), \qquad (3.74)$$

$$\lim_{\tan\beta\to\infty} \operatorname{BR}_{\Delta_V=0.05}^{2HDM(I)}(t\to WbH) = 0.098 \times \operatorname{BR}^{SM}(t\to Wbh),$$
(3.75)

$$\lim_{\tan\beta\to\infty} \mathrm{BR}^{2HDM(I)}_{\Delta_V=0.7}(t\to Wbh) = 0.090 \times \mathrm{BR}^{SM}(t\to Wbh),$$
(3.76)

and agree well with numerical calculations.

The Δ_V dependence in Fig 8(b) and the relationship between h and H is indicative of much broader behavior found in all 2HDM variants. To saturate the sum rule for the electroweak symmetry breaking [106], the hWW coupling (g_{hWW}) and the HWW coupling (g_{HWW}) obey

$$g_{hWW}^2 + g_{HWW}^2 = g_{h^{SM}WW}^2, aga{3.77}$$

where $g_{h^{SM}WW}$ is the SM hWW coupling. For h and H with degenerate masses,

$$BR(t \to Wbh) + BR(t \to WbH) = BR^{SM}(t \to Wbh) + \mathcal{O}(\cot^2\beta).$$
(3.78)

Indeed, Eqs. (3.71) and (3.72) satisfy this relationship. Furthermore, this can be extended to an arbitrary number of scalar $SU(2)_L$ doublets and singlets [106]. Though mass splittings, etc., will break this equality, it provides a useful estimate for processes involving transitions in models with additional scalar $SU(2)_L$ doublets and singlets.

3.3.2.2 BR $(t \to Wbh/H)$ vs $m_{h/H}$ As a function of mass, we plot in Fig. 9 the decay rates for $t \to W^*bH$ where H is a non-SM-like CP-even Higgs; the mass of the SM-like Higgs is taken to be 125.5 GeV. For a mass below (above) 110 GeV, we observe that transition rate to a non-SMlike Higgs remains above (below) the SM rate. As the scalar mass decreases and the W^* comes closer to being on-shell, the availability of phase space greatly ameliorates the coupling suppression

H (125.5 GeV)	Δ_V	aneta	$BR^{2HDM(II)}(t \to WbH)$
h	5×10^{-5}	3	1.813×10^{-9}
h		7.6	1.809×10^{-9}
Н	0.99	3	1.798×10^{-9}
Н		7.6	1.802×10^{-9}
h	0.99	3	1.440×10^{-10}
h		7.6	4.990×10^{-11}
A (100 GeV)		3	1.760×10^{-9}
A (100 GeV)		7.6	1.007×10^{-9}

Table 11: $BR(t \rightarrow WbH)$ for Benchmark Values of Higgses in the 2HDM(II)

associated with Δ_V . However, despite this relief, the transition rate to a non-SM-like H stays below the SM rate for much of the parameter space. The insensitivity to large and moderate tan β seen in Fig. 9 is consistent with previous discussions.

3.3.2.3 BR $(t \to WbA)$ **vs** $\tan \beta$, m_A Here, we consider the decay rate to the CP-odd Higgs, $t \to W^*bA$. Fig. 10 shows BR $(t \to WbA)$ as a function of (a) $\tan \beta$ and (b) m_A . Except for very low $\tan \beta$ and mass, the branching fraction remains well below the SM prediction for much of the parameter space, approximately equaling it at $\tan \beta \simeq 3$ for $m_A = 100$ GeV. Due to CP-invariance in the gauge sector there is no tree-level AWW contribution. And since the $f\bar{f}A$ couplings are independent of Δ_V , the decay rate is fixed entirely by m_A and $\tan \beta$. Destructive interference still exists, however, since the $t\bar{t}A$ and $b\bar{b}A$ vertices differ by a minus sign. A quadratic dependence on $\cot \beta$ is the consequence the $f\bar{f}A$ coupling ($\propto \cot \beta$). See Table 7. Despite this monotonic dependence on $\tan \beta$, which implies that BR $(t \to WbA)$ is a direct measure of $\tan \beta$ were it to be measured, the recuperation of available phase space at low m_A is unable to compensate for the $\cot^2 \beta$ suppression.



Figure 11: The 2HDM(II) BR $(t \to WbH)$ as a function of (a) $\tan\beta$ for SM-like *h* (long dash), *H* (dash-dot), and non-SM-like *h* (short dash), *H* (dot); (b) Δ_V for *h* at $\tan\beta = 3$ 7.6, 15 (short dash, long dash, dash-dot), and for *H* (dot, long dash, dash-dot). The solid line denotes the SM prediction, Eq. (3.64).



Figure 12: The 2HDM(I) BR $(t \to WbH)$ as a function of mass for a non-SM-like (a) h and (b) H assuming tan $\beta = 3$, 7.6, 15 (short dash, long dash, dash-dot). The solid line denotes the SM prediction, Eq. (3.64).

3.3.3 Type II 2HDM $BR(t \rightarrow WbH)$

We report here the behavior of BR $(t \to WbH)$, where H represents h, H, or A in the 2HDM(II). The same values of tan β are used here as the Type I case. To avoid constraints, we choose a Δ_V that corresponds to a light Higgs with $\cos(\beta - \alpha) = 0.01$, i.e.,

$$\Delta_V = 5. \times 10^{-5} \ (0.99) \quad \text{for} \quad h \ (H) \approx h^{SM}. \tag{3.79}$$

Table 11 lists values of the branching fraction for several Higgses and benchmark parameter values. In the following figures, the predicted SM decay rate is shown as a black, solid line labeled by "SM".

3.3.3.1 BR $(t \to Wbh, H)$ vs tan β , Δ_V Figure 11 depicts the branching ratio BR $(t \to WbH)$ for both of the CP-even Higgses as a function of (a) tan β for SM-like and non-SM-like h and H, and (b) Δ_V for small and large values of tan β .

In Fig 11(a), for SM-like Higgses, the branching fraction is indistinguishable from the SM prediction as a function of tan β ; for non-SM-like Higgses, however, the rates minimize for tan $\beta = 7 \sim 8$. This dependence on tan β is indicative of a playoff between the $t\bar{t}H$ and the $b\bar{b}H$ couplings. In the SM limit, this specific behavior is suppressed for SM-like Higgs bosons because the couplings to these bosons grow independent of tan β . When h is non-SM-like ($\Delta_V = 0.99$), sensitivity to tan β maximizes because the tan β -independent parts of the fermionic Higgs couplings nearly cancel. As tan β grows, the contribution from $t\bar{t}h (\propto \cot \beta)$ runs BR($t \rightarrow Wbh$) down until the $b\bar{b}h$ contribution ($\propto \tan \beta$) takes over at tan $\beta \approx 7.6$. When H is non-SM-like ($\Delta_V = 5. \times 10^{-5}$) we expect and observe similar behavior as the non-SM-like h case.

Much of the relationship between h and H observed in in Fig. 11(b) is type-independent and the discussion can be found in the Type I scenario. For a light Higgs, we indeed see that at $\tan \beta = 7.6$ transition rates are minimized for all values of Δ_V . For a heavy Higgs, however, this value of $\tan \beta$ only minimizes the rate in the $h \to h^{SM}$ limit, in which case the $t \to H$ transition rate vanishes. The $t \to H$ rate minimum occurs at larger Δ_V with decreasing $\tan \beta$ because the $t\bar{t}H$ ($b\bar{b}H$) contribution becomes numerically larger (smaller).

3.3.3.2 BR $(t \to Wbh, H)$ vs m_H Figure 12 presents the $t \to W^*bH$ branching ratio for a non-SM-like Higgs boson as a function of mass. For the mass window given in Eq. (3.69), we find considerable enhancement in the decay rate relative to the SM rate due to the increase in available



Figure 13: The 2HDM(II) BR($t \to WbA$) as a function of (a) $\tan \beta$ and (b) m_A for $\tan \beta = 3$, 7.6, 15 (short dash, long dash, dash-dot). The solid line denotes the SM prediction, Eq. (3.64).

phase space, overcoming the coupling suppression associated with scalars that have non-SM-like coupling.

3.3.3.3 BR $(t \to WbA)$ vs tan β , m_A Turning to the CP-odd Higgs decay channel, $t \to W^*bA$, we note that many of the arguments made in the 2HDM(I) case carry over to this situation. Unlike the Type I scenario, however, there is only *constructive* interference between the fermion contributions. Figure 13 shows BR $(t \to WbA)$ as a function of (a) tan β and (b) m_A .

In Fig. 13(a), due to an accidental cancellation, the branching fraction minimizes at $\tan \beta \approx 5.8$, which is unsurprisingly close to the $t \to H^+ b$ minimum at $\tan \beta = \sqrt{m_t/m_b} \approx 7.6$. At $\tan \beta \approx 5.8$, the $t\bar{t}A$ coupling ($\propto \cot \beta$) and the $b\bar{b}A$ coupling ($\propto \tan \beta$) contribute equally. At smaller values of $\tan \beta$, $t\bar{t}A$ is the dominant term but is driven down by an increasing $\tan \beta$; and at larger values, $b\bar{b}A$ is the dominant term, which ramps up the rate. In the large $\tan \beta$ limit, the $t\bar{t}A$ graph becomes negligible and the rate becomes quadratically with $\tan \beta$.

In Fig. 13(b), we observe a similarity between A and the non-SM-like Higgs boson, H_X . We attribute this to a similarity of contributing diagrams. For example: the WWA vertex does not exist because of CP-invariance, and by virtue of being *non*-SM-like, the WWH_X vertex is considerably suppressed. In this domain, fermionic couplings to A and H_X also have the same dependence on

 $\tan\beta$.

3.4 OBSERVATION PROSPECTS AT COLLIDERS

In this section, we estimate observation prospects at current and future colliders. The 14 TeV LHC $t\bar{t}$ production cross section at NNLO in QCD has been calculated [65] to be

$$\sigma_{LHC14}^{NNLO}(t\bar{t}) = 933 \text{ pb.}$$
 (3.80)

The SM $pp \rightarrow t\bar{t} \rightarrow WW^*b\bar{b}h$ cross section at the LHC is thus estimated to be

$$\sigma_{LHC14}(pp \to t\bar{t} \to WW^*b\bar{b}h) \approx 2 \times \sigma_{LHC14}^{NNLO}(t\bar{t}) \times BR(t \to Wbh) = 3.4 \text{ ab.}$$
(3.81)

The factor of two in the preceding line accounts for either top or antitop quark decaying into the Higgs. To assure a clear trigger and to discriminate against the large SM backgrounds, we require at least one W boson decaying leptonically ($\ell = e, \mu$), i.e.

$$BR(WW^* \to \ell^+ \ell^{'-} \nu_\ell \overline{\nu}_{\ell'} + jj\ell^{\pm} \nu_\ell^{(-)}) \approx 0.33.$$
 (3.82)

The total cross section for an arbitrarily decaying h is therefore estimated to be

$$\sigma_{LHC14}(pp \to t\bar{t} \to WW^*b\bar{b}h \to h(\ell^+\ell^{\prime-}\nu_\ell\bar{\nu}_{\ell^\prime} + jj\ell^{\pm}\overset{(-)}{\nu_\ell})) \approx 1.1 \text{ ab.}$$
(3.83)

Higgs branching fractions and detector efficiencies will further suppress this rate. Such a small cross section means that observing this SM process will be challenging. Following the same procedure, we estimate Eq. (3.83) for several proposed colliders and collider upgrades; the results are given in Table 12.

3.5 SUMMARY AND CONCLUSION

Given the discovery of a SM-like Higgs boson, we have recalculated the rare top quark decay mode $t \rightarrow W^*bh$, where *h* represents the SM Higgs boson. We have extended this calculation to include the effects of anomalous $t\bar{t}h$ couplings originating from effective operators as well as both CP-even and the single CP-odd scalars in the CP-conserving 2HDM Types I and II. The most updated model constraints have been reported. We summarize our results:

Table 12: Cross sections for $t\bar{t}$ and $t\bar{t} \to WW^*b\bar{b}h \to h(\ell^+\ell'^-\nu_\ell\bar{\nu}_{\ell'} + jj\ell^\pm\nu_\ell)$ at 14 [65], 33 [137], and 100 [138] TeV pp, and 350 GeV e^+e^- [139] Colliders.

Process	14 TeV pp	33 TeV pp	100 TeV pp	$350 \text{ GeV} e^+e^-$
$\sigma(tar{t})[{ m pb}]$	933	5410	$2.7 imes 10^4$	0.45
$\sigma(t\bar{t} \to h(\ell^+ \ell'^- \nu_\ell \overline{\nu}_{\ell'} + jj\ell^\pm \overset{(-)}{\nu_\ell})) \text{ [ab]}$	1.1	6.5	32	5×10^{-4}

(i) The SM predicts a $t \to W^*bh$ branching ratio of

$$BR(t \to Wbh) = 1.80 \times 10^{-9} \text{ for } m_h = 125.5 \text{ GeV.}$$
 (3.84)

This is the leading $t \to h$ transition, five orders of magnitude larger than the next channel $t \to ch$. See Eq. (3.64).

- (ii) Present LHC Higgs constraints on anomalous $t\bar{t}h$ couplings permit up to a factor of two enhancement of the $t \to W^*bh$ transition. See Eq. (3.65).
- (iii) The operator $\overline{\mathcal{O}}_{t2}$, which selects different kinematic features than either \mathcal{O}_{t1} or $\overline{\mathcal{O}}_{t1}$, results in comparable enhancement of the $t \to W^*bh$ transition. See Fig. 7.
- (iv) In the 2HDM(I), decays to CP-even Higgses do not decouple in the large $\tan \beta$ limit and their rates approach asymptotic values that are functions of the anomalous WWh coupling. They are given in Eqs. (3.71) and (3.72).
- (v) In the Type I (II) 2HDM, due to the increase in available phase space, the branching ratio to a light, non-SM-like Higgs boson can as much as 2 (7) times larger than Eq. (3.84).
- (vi) In the Type I (II) 2HDM, the branching ratio to a light, CP-odd Higgs can be as much as 1.6 (3) times larger than Eq. (3.84).
- (vii) The $pp \to t\bar{t} \to WW^*b\bar{b}h \to h(\ell^+\ell^{\prime-}\nu_\ell\bar{\nu}_{\ell'} + jj\ell^{\pm}\overset{(-)}{\nu_\ell})$ production cross section at the 14 TeV LHC and future colliders have been estimated [Eq. (3.83)]; a few $t \to W^*bh$ events over the full LHC lifetime. Due to enhancements in gluon distribution functions, any increase in collision energies can greatly increase this rate.

4.0 INCLUSIVE HEAVY MAJORANA NEUTRINO PRODUCTION AT HADRON COLLIDERS

4.1 INTRODUCTION

The discovery of the Higgs boson completes the Standard Model (SM). Yet, the existence of nonzero neutrino masses remains one of the clearest indications of physics beyond the Standard Model (BSM) [140–147] The simplest SM extension that can simultaneously explain both the existence of neutrino masses and their smallness, the so-called Type I seesaw mechanism [148–157], introduces a right handed (RH) neutrino N_R . Via a Yukawa coupling y_{ν} , the resulting Dirac mass is $m_D = y_{\nu} \langle \Phi \rangle$, where Φ is the SM Higgs SU(2)_L doublet. As N_R is a SM-gauge singlet, one could assign N_R a Majorana mass m_M without violating any fundamental symmetry of the model. Requiring that $m_M \gg m_D$, the neutrino mass eigenvalues are

$$m_1 \sim m_D \frac{m_D}{m_M}$$
 and $m_2 \sim m_M$. (4.1)

Thus, the apparent smallness of neutrino masses compared to other fermion masses is due to the suppression by a new scale above the EW scale. Taking the Yukawa coupling to be $y_{\nu} \sim \mathcal{O}(1)$, the Majorana mass scale must be of the order 10^{13} GeV to recover sub-eV light neutrinos masses. However, if the Yukawa couplings are as small as the electron Yukawa coupling, i.e., $y_{\nu} \leq \mathcal{O}(10^{-5})$, then the mass scale could be at $\mathcal{O}(1)$ TeV or lower [158–161].

Given the lack of guidance from theory of lepton flavor physics, searches for Majorana neutrinos must be carried out as general and model-independent as possible. Low-energy phenomenology of Majorana neutrinos has been studied in detail [1,160-175]. Studied first in Ref. [162] and later in Refs. [163–168], the production channel most sensitive to heavy Majorana neutrinos (N) at hadron colliders is the resonant Drell-Yan (DY) process,

$$pp \to W^{\pm *} \to N \ \ell^{\pm}, \quad \text{with} \ N \to W^{\mp} \ \ell^{'\pm}, \quad W^{\mp} \to j \ j,$$

$$(4.2)$$



Figure 14: Diagram representing resonant heavy Majorana neutrino production through the DY process and its decay into same-sign leptons and dijet. All diagrams drawn using JaxoDraw [83].

in which the same-sign dilepton channel violates lepton number L by two units ($\Delta L = 2$); see figure 14. Searches for Eq. (4.2) are underway at LHC experiments [176–178]. Non-observation in the dimuon channel has set a lower bound on the heavy neutrino mass of 100 (300) GeV for mixing $|V_{\mu N}|^2 = 10^{-2} (-1)$ [177]. Bounds on mixing from $0\nu\beta\beta$ [179,180] and EW precision data [181–184] indicate that the 14 TeV LHC is sensitive to Majorana neutrinos with mass between 10 and 375 GeV after 100 fb⁻¹ of data [166]. Recently renewed interest in a very large hadron collider (VLHC) with a center of mass (c.m.) energy about 100 TeV, which will undoubtedly extend the coverage, suggests a reexamination of the search strategy at the new energy frontier.

Production channels for heavy Majorana neutrinos at higher orders of α were systematically cataloged in Ref. [165]. Recently, the vector boson fusion (VBF) channel $W\gamma \rightarrow N\ell^{\pm}$ was studied at the LHC, and its *t*-channel enhancement to the total cross section was emphasized [174]. Along with that, they also considered corrections to the DY process by including the tree-level QCD contributions to $N\ell^{\pm}$ +jets. Significant enhancement was claimed over both the leading order (LO) DY signal [166,168] and the expected next-to-next-to-leading order (NNLO) in QCD-corrected DY rate [185], prompting us to revisit the issue.

We carry out a systematic treatment of the photon-initiated processes. The elastic emission (or photon emission off a nucleon) at colliders, as shown in figure 15(a), is of considerable interest for both SM [47, 51, 186–190] and BSM processes [48, 50, 52, 62, 191–195], and has been observed at electron [196], hadron [49, 197], and lepton-hadron [198, 199] colliders. The inelastic (collinear photon off a quark) and deeply inelastic (large momentum transfer off a quark) channels, as depicted in figure 15(b), may take over at higher momentum transfers [59, 188, 200]. Comparing with the



Figure 15: Diagrammatic description of (a) elastic and (b) inelastic/deeply inelastic γp scattering.

DY production $qq' \to W^* \to N\ell^{\pm}$, we find that the $W\gamma$ fusion process becomes relatively more important at higher scales, taking over the QCD-corrected DY mechanism at $\gtrsim 1$ TeV (770 GeV) at the 14-TeV LHC (100 TeV VLHC). At $m_N \sim 375$ GeV, a benchmark value presented in [168], we find the $W\gamma$ contribution to be about 20% (30%) of the LO DY cross section.

NNLO in QCD corrections to the DY processes are well-known [185] and the K-factor for the inclusive cross sections are about 1.2-1.4 (1.2-1.5) at LHC (VLHC) energies. Taking into account all the contributions, we present the state-of-the-art results for the inclusive production of heavy neutrinos in 14 and 100 TeV pp collisions. We further perform a signal-versus-background analysis for a 100 TeV collider of the fully reconstructible and L-violating final state in Eq. (4.2). With the currently allowed mixing $|V_{\mu N}|^2 < 6 \times 10^{-3}$, we find that the 5σ discovery potential of Ref. [168] can be extended to $m_N = 530$ (1070) GeV at the 14 TeV LHC (100 TeV VLHC) after 1 ab⁻¹. Reversely, for $m_N = 500$ GeV and the same integrated luminosity, a mixing $|V_{\mu N}|^2$ of the order 1.1×10^{-3} (2.5×10^{-4}) may be probed. Our results are less optimistic than reported in [174]. We attribute the discrepancy to their significant overestimate of the signal in the tree-level QCD calculations, as quantified in section 4.3.3.4.

The rest of paper is organized as follows: In section 4.3, we describe our treatment of the several production channels considered in this study, address the relevant scale dependence, and present the inclusive $N\ell^{\pm}$ rate at the 14 TeV LHC and 100 TeV VLHC. In section 4.4, we perform the signal-versus-background analysis at a future 100 TeV *pp* collider and report the discovery potential. Finally summarize and conclude in section 4.5.

4.2 NEUTRINO MIXING FORMALISM

Our formalism and notation follow Ref. [1,168]. We assume that there are three left-handed (L.H.) neutrinos (denoted by ν_{aL} , a = 1, 2, 3) with three corresponding light mass eigenstates (denoted by m), and n right-handed (R.H.) neutrinos (denoted by $N_{a'R}$, a' = 1, ..., n) with n corresponding heavy mass eigenstates (denoted by m'). The mixing between chiral states and mass eigenstates may then be parameterized [168] by

$$\begin{pmatrix} \nu_L \\ N_L^c \end{pmatrix} = \begin{pmatrix} U_{3\times3} & V_{3\times n} \\ X_{n\times3} & Y_{n\times n} \end{pmatrix} \begin{pmatrix} \nu_m \\ N_{m'}^c \end{pmatrix},$$
(4.3)

where $\psi^c = C \overline{\psi}^T$ denotes the charge conjugate of the spinor field ψ , with C labeling the charge conjugation operator, and the chiral states satisfy $\psi_L^c \equiv (\psi^c)_L = (\psi_R)^c$. Expanding the L.H. and R.H. chiral states, we obtain:

$$\overline{\nu_{aL}} = \sum_{m=1}^{3} \overline{\nu_m} U_{ma}^* + \sum_{m'=4}^{n+3} \overline{N_{m'}^c} V_{m'a}^*, \qquad \overline{N_{a'L}^c} = \sum_{m=1}^{3} \overline{\nu_m} X_{ma'}^* + \sum_{m'=4}^{n+3} \overline{N_{m'}^c} Y_{m'a'}^* \qquad (4.4)$$

$$\overline{\nu_{aR}^c} = \sum_{m=1}^3 \overline{\nu_m^c} U_{ma} + \sum_{m'=4}^{n+3} \overline{N_{m'}} V_{m'a}, \qquad \overline{N_{a'R}} = \sum_{m=1}^3 \overline{\nu_m^c} X_{ma'} + \sum_{m'=4}^{n+3} \overline{N_{m'}} Y_{m'a'}.$$
(4.5)

Under this formalism, one expects diagonal mixing of order 1,

$$UU^{\dagger} \text{ and } YY^{\dagger} \sim \mathcal{O}(1);$$

$$(4.6)$$

and suppressed off-diagonal mixing,

$$VV^{\dagger} \text{ and } XX^{\dagger} \sim \mathcal{O}(m_m/m_{m'}).$$
 (4.7)

For simplicity, we consider only the lightest, heavy mass eigenstate neutrino N. The SM W coupling to heavy neutrino N and charged lepton ℓ can now be written as

$$\mathcal{L} = - \frac{g}{\sqrt{2}} \sum_{\ell=e}^{\tau} W_{\mu}^{+} \left[\sum_{m=1}^{3} \overline{\nu_{m}} U_{\ell m}^{*} + \overline{N^{c}} V_{\ell N}^{*} \right] \gamma^{\mu} P_{L} \ell^{-} + \text{H.c..}$$
(4.8)

4.3 HEAVY N PRODUCTION AT HADRON COLLIDERS

For the production of a heavy Majorana neutrino at hadron colliders, the leading channel is the DY process at order α^2 (LO) [162]

$$q \ \overline{q}' \to W^{\pm *} \to N \ \ell^{\pm}. \tag{4.9}$$

The QCD corrections to DY-type processes up to α_s^2 (NNLO) are known [185], and will be included in our analysis. Among other potential contributions, the next promising channel perhaps is the VBF channel [165]

$$W \gamma \to N \ \ell^{\pm},$$
 (4.10)

due to the collinear logarithmic enhancement from t-channel vector boson radiation. Formally of order α^2 , there is an additional α suppression from the photon coupling to the radiation source. Collinear radiation off charged fermions (protons or quarks) leads to significant enhancement but requires proper treatment. In our full analysis, Ws are not considered initial-state partons [165] and all gauge invariant diagrams, including non-VBF contributions, are included.

We write the production cross section of a heavy state X in hadronic collisions as

$$\sigma(pp \to X + \text{anything}) = \sum_{i,j} \int_{\tau_0}^1 d\xi_a \int_{\frac{\tau_0}{\xi_a}}^1 d\xi_b \left[f_{i/p}(\xi_a, Q_f^2) f_{j/p}(\xi_b, Q_f^2) \hat{\sigma}(ij \to X) + (i \leftrightarrow j) \right] .$$

$$= \int_{\tau_0}^1 d\tau \sum_{ij} \frac{d\mathcal{L}_{ij}}{d\tau} \hat{\sigma}(ij \to X).$$

$$(4.12)$$

where $\xi_{a,b}$ are the fractions of momenta carried by initial partons (i, j), Q_f is the parton factorization scale, and $\tau = \hat{s}/s$ with \sqrt{s} ($\sqrt{\hat{s}}$) the proton beam (parton) c.m. energy. For heavy neutrino production, the threshold is $\tau_0 = m_N^2/s$. Parton luminosities are given in terms of the parton distribution functions (PDFs) $f_{i,j/p}$ by the expression

$$\Phi_{ij}(\tau) \equiv \frac{d\mathcal{L}_{ij}}{d\tau} = \frac{1}{1+\delta_{ij}} \int_{\tau}^{1} \frac{d\xi}{\xi} \left[f_{i/p}(\xi, Q_f^2) f_{j/p}\left(\frac{\tau}{\xi}, Q_f^2\right) + (i \leftrightarrow j) \right].$$
(4.13)

We include the light quarks (u, d, c, s) and adopt the 2010 update of the CTEQ6L PDFs [63]. Unless stated otherwise, all quark (and gluon) factorization scales are set to half the c.m. energy:

$$Q_f = \sqrt{\hat{s}}/2. \tag{4.14}$$

For the processes with initial state photons (γ) , their treatment and associated scale choices are given in section 4.3.3.

For the heavy neutrino production via the SM charged current coupling, the cross section is proportional to the mixing parameter (squared) between the mass eigenstate N and the charged lepton ℓ (e, μ, τ). Thus it is convenient to factorize out the model-dependent parameter $|V_{\ell N}|^2$

$$\sigma(pp \to N\ell^{\pm}) \equiv \sigma_0(pp \to N\ell^{\pm}) \times |V_{\ell N}|^2, \qquad (4.15)$$

where σ_0 will be called the "bare cross section". Using the phase space slicing method [201–204], the heavy Majorana neutrino production can be evaluated at next-to-leading order (NLO) in QCD



Figure 16: (a) 14 TeV LHC (b) 100 TeV VLHC $N\ell^{\pm}$ cross section, divided by $|V_{\ell N}|^2$, and NLO *K*-factor as a function of m_N at LO DY (solid) and NLO in QCD(bash).

accuracy. Using the 2012 update of the CT10 PDFs [205] and factorization, renormalization scales $\mu_f = \mu_r = m_N$, we plot in Fig. 16 the LO and NLO bare cross section and NLO K-factor¹ as a function of Majorana neutrino mass m_N at the (a) 14 TeV LHC and (b) 100 TeV VLHC. At 14 (100) TeV, for $m_N = 100 - 600$ GeV, the bare NLO rate ranges from 0.03 - 30 pb (0.6 - 250 pb). The corresponding K-factor spans 1.13 - 1.17 (1.15 - 1.2).

The branching fraction of a heavy neutrino to a particular lepton flavor ℓ is proportional to $|V_{N\ell}|^2 / \sum_{\ell'} |V_{N\ell'}|^2$. Thus for neutrino production and decay into same-sign leptons with dijet, it is similarly convenient to factorize out this ratio [166]:

$$\sigma(pp \to \ell^{\pm} \ell^{'\pm} + 2j) \equiv \sigma_0(pp \to \ell^{\pm} \ell^{'\pm} + 2j) \times S_{\ell\ell'}, \tag{4.16}$$

$$S_{\ell\ell'} = \frac{|V_{\ell N}|^2 |V_{\ell' N}|^2}{\sum_{\ell''} |V_{\ell'' N}|^2}.$$
(4.17)

The utility of this approach is that all the flavor-model dependence is encapsulated into a single, measurable number. Factorization into a bare rate and mixing coefficient holds generally for QCD and EW corrections as well.

¹The $N^n LO$ K-factor is defined as $K = \sigma^{N^n LO}(N\ell) / \sigma^{LO}(N\ell)$, where $\sigma^{N^n LO}(N\ell)$ is the NⁿLO-corrected cross section and $\sigma^{LO}(N\ell)$ is the lowest order (n = 0), or Born, cross section.

4.3.1 Constraints on Heavy Neutrino mixing

As seen above in Eq. (4.15), one of the most important model-dependent parameters to control the signal production rate is the neutrino mixing $V_{\ell N}$. Addressing the origin of lepton flavor is beyond the scope of this study, so masses and mixing factors are taken as independent, phenomenological parameters. We consider only the lightest, heavy neutrino mass eigenstate and require it to be kinematically accessible. Updates on heavy neutrino constraints can be found elsewhere [1,168,206]. Here we list only the most stringent bounds relevant to our analysis.

• Bounds from $0\nu\beta\beta$: For heavy Majorana neutrinos with $M_i \gg 1$ GeV, the absence of $0\nu\beta\beta$ decay restricts the mixing between heavy mass and electron-flavor eigenstates [179, 180]:

$$\sum_{m'} \frac{|V_{em'}|^2}{M_{m'}} < 5 \times 10^{-5} \text{ TeV}^{-1}.$$
(4.18)

• Bounds from EW Precision Data: Mixing between a SM singlet above a few hundred GeV in mass and lepton flavor eigenstates is constrained by EW data [183]:

$$|V_{\mu N}|^2 < 3.2 \times 10^{-3}, \quad |V_{\tau N}|^2 < 6.2 \times 10^{-3} \text{ at } 90\% \text{ C.L.}$$
 (4.19)

We consider the existence of only the lightest heavy Majorana neutrino, which is equivalent to the decoupling limit where heavier eigenstates are taken to have infinite mass. Thus, for representative neutrino masses

$$m_N = 300 \ (500) \ [1000] \ \text{GeV},$$
 (4.20)

we use the following mixing coefficients

$$|V_{eN}|^2 = 1.5 \ (2.5) \ [5] \times 10^{-5}, \qquad |V_{\mu N}|^2 = 3.2 \times 10^{-3}, \qquad |V_{\tau N}|^2 = 6.2 \times 10^{-3}, \tag{4.21}$$

corresponding to a total neutrino width of

$$\Gamma_N = 0.303 \ (1.50) \ [12.3] \ \text{GeV}.$$
 (4.22)

As $\Gamma_t/m_N \approx 0.1\% - 1\%$, the heavy neutrino resonance is very narrow and application of the narrow width approximation (NWA) is justified. For $S_{\ell\ell}$, these mixing parameters imply

$$S_{ee} = 2.4 \ (6.6) \ [26] \times 10^{-8}$$
 for $m_N = 300 \ (500) \ [1000] \ \text{GeV}$ (4.23)

$$S_{e\mu} = S_{\mu e} = 5.1 \ (8.5) \ [17] \times 10^{-6}$$
 for $m_N = 300 \ (500) \ [1000] \ \text{GeV}$ (4.24)

$$S_{\mu\mu} = 1.1 \times 10^{-3}$$
 for $m_N \in [100, 1000] \text{ GeV}$ (4.25)

Though the bound on $|V_{eN}|$ varies with m_N , $S_{\mu\mu}$ changes at the per mil level over the masses we investigate and is taken as constant. The allowed sizes of $S_{e\mu}$, $S_{\mu\mu}$, and $S_{\tau\ell}$ demonstrate the complementarity to searches for *L*-violation at $0\nu\beta\beta$ experiments afforded by hadron colliders. To make an exact comparison with Ref. [168], we also consider the bound [181, 182]

$$S_{\mu\mu} \approx \frac{|V_{\mu N}|^4}{|V_{\mu N}|^2} = |V_{\mu N}|^2 = 6 \times 10^{-3}$$
(4.26)

However, bare results, which are mixing-independent, are presented wherever possible.

4.3.2 N Production via the Drell-Yan Process at NNLO

Before presenting the production cross sections, it is informative to understand the available parton luminosities (Φ_{ij}) as defined in Eq. (4.13). We show $\Phi_{q\bar{q}'}$ versus $\sqrt{\tau}$ for $q\bar{q}'$ annihilation summing over light quarks (u, d, c, s) by the solid (black) curves in figures 17(a) and 17(b) for the 14 TeV LHC and 100 TeV VLHC, respectively. The upper horizontal axis labels the partonic c.m. energy $\sqrt{\hat{s}}$. As expected, at a fixed $\sqrt{\hat{s}}$ the DY luminosity at 100 TeV significantly increases over that at 14 TeV. At $\sqrt{\hat{s}} \approx 500$ GeV (2 TeV), the gain is a factor of 600 (1.8 × 10³), and the discovery potential of heavy Majorana neutrinos is greatly expanded. Luminosity ratios with respect to $\Phi_{q\bar{q}'}$ are given in figure 17(c) and 17(d), and will be discussed when appropriate.

Cross sections for resonant N production via the charged current DY process in Eq. (4.2) and shown in figure 14 are calculated with the usual helicity amplitudes at the LO α^2 . Monte Carlo integration is performed using CUBA [207]. Results are checked by implementing the heavy Majorana neutrino model into FeynRules 2.0.6 [208, 209] and MG5_aMC@NLO 2.1.0 [56] (MG5). For simplicity, percent-level contributions from off-diagonal Cabbibo-Kobayashi-Maskawa (CKM) matrix elements are ignored and the diagonal elements are taken to be unity. SM inputs $\alpha^{\overline{\text{MS}}}(M_Z)$, M_Z , and $\sin^2_{\overline{\text{MS}}}(\theta_W)$ are taken from the 2012 Particle Data Group (PDG) [136].

We estimate the 14 and 100 TeV pp NNLO K-factor by using FEWZ 2.1 [210, 211] to compute the equivalent quantity for the SM process

$$pp \to W^* \to \mu^{\pm} \nu,$$
 (4.27)

and impose only an minimum invariant mass cut, $\sqrt{\hat{s}^{\min}}$. Because LO $N\ell$ production and Eq. (4.27) are identical DY processes (up mass effects) with the same color structure, K-factors calculated with a fixed \hat{s} are equal.





Figure 17: Parton luminosities at (a) 14 TeV and (b) 100 TeV for the DY (solid), elastic (dot), inelastic (dash), and DIS (dash-diamond) $N\ell X$ processes; Ratio of parton luminosities to the DY luminosity in (c) and (d).

Table 13 lists² the LO and NNLO cross sections as well as the NNLO K-factors for several representative values of $\sqrt{\hat{s}^{\min}}$. At $\sqrt{\hat{s}^{\min}} = 1$ TeV, the QCD-corrected charged current rate can

 $^{^2\}mathrm{As}$ no NNLO CTEQ6L PDF set exists, we have adopted the MSTW2008 series to obtain a self-consistent estimate of the NNLO $K\text{-}\mathrm{factor}.$

$\sqrt{\hat{s}^{\min}}$	14 TeV LO [pb]	NNLO [pb]	K	100 TeV LO [pb]	NNLO [pb]	K
100 GeV	152	209	1.38	1150	1420	1.23
$300 {\rm GeV}$	1.54	1.90	1.23	17.0	25.6	1.50
$500 { m ~GeV}$	0.248	0.304	1.22	3.56	4.97	1.40
$1 { m TeV}$	17.0×10^{-3}	20.5×10^{-3}	1.20	0.380	0.485	1.28

Table 13: LO and NNLO cross sections for $pp \to W^* \to \mu^{\pm}\nu$ at 14 and 100 TeV with successive invariant mass cuts using MSTW2008LO and NNLO PDF Sets.

reach tens (several hundreds) of fb at 14 (100) TeV. Over the range from $\sqrt{\hat{s}^{\min}} = 100 \text{ GeV} - 1 \text{ TeV}$,

$$K = 1.20 - 1.38$$
 at 14 TeV, (4.28)

$$= 1.23 - 1.50$$
 at 100 TeV. (4.29)

This agrees with calculations for similar DY processes [212, 213]. We see that the higher order QCD corrections to the DY channel are quite stable, which will be important for our discussions in section 4.3.3. Throughout the study, independent of neutrino mass, we apply to the DY-process a K-factor of

$$K = 1.2 (1.3)$$
 for 14 (100) TeV. (4.30)

Including the QCD K-factor, we show the NNLO total cross sections [called the "bare cross section σ_0 " by factorizing out $|V_{\ell N}|^2$ as defined in Eq. (4.15)] as a function of heavy neutrino mass in figures 18(a) and 18(b) for the 14-TeV LHC and 100-TeV VLHC, respectively. The curves are denoted by the (black) solid lines. Here and henceforth, we impose the following basic acceptance cuts on the transverse momentum and pseudorapidity of the charged leptons for 14 (100) TeV,

$$p_T^{\ell} > 10 \; (30) \; \text{GeV}, \quad |\eta^{\ell}| < 2.4 \; (2.5).$$
 (4.31)

The motive to include these cuts is two-fold. First, they are consistent with the detector acceptance for our future simulations and the definition of "fiducial" cross section. Second, they serve as kinematical regulators for potential collinear singularities, to be discussed next. The p_T and η criteria at 100 TeV follow the 2013 Snowmass benchmarks [214].



Figure 18: (a) 14 TeV LHC (b) 100 TeV VLHC $N\ell X$ cross section, divided by $|V_{\ell N}|^2$, as a function of the N mass for the NNLO DY (solid), elastic (dot), inelastic (dash), DIS (dash-diamond), and summed γ -initiated (dash-dot) processes. (c,d) Ratio of cross sections relative to NNLO DY rate.

4.3.3 Photon-Initiated Processes

After the dominant DY channel, VBF via $W\gamma$ fusion, as introduced in Eq. (4.10), presents a promising additional contribution to the heavy N production. We do not make any approximation for the initial state W and treat its radiation off the light quarks with exact matrix element



Figure 19: Feynman diagrams for photon-initiated process $q\gamma \to N\ell^{\pm}q'$.

calculations. In fact, we consistently treat the full set of diagrams, shown in figure 19, for the photon-initiated process at order α^3

$$q \ \gamma \to N \ \ell^{\pm} \ q'. \tag{4.32}$$

Obviously, diagrams figure 19(c) and (d) do not add to $W\gamma$ fusion and are just small QED corrections.³ Diagram figure 19(b) involves a massless *t*-channel charged lepton. The collinear pole is regularized by the basic acceptance cuts in Eq. (4.31). What is non-trivial, however, is how to properly treat initial-state photons across the different sources depicted in figure 15. We now discuss the individual channels in detail.

4.3.3.1 Elastic Scattering: Intact Final-State Nucleons Here and henceforth, the virtuality for the incoming photon in $W\gamma$ fusion is denoted as $Q_{\gamma} > 0$. In the collinear limit that results in momentum transfers on the order of the proton mass or less, $Q_{\gamma}^2 \leq m_p^2$, initial-state photons are appropriately described as massless radiation by an elastic proton, i.e., does not break apart and remains as an on-shell nucleon, as indicated in figure 15(a). To model this, we use the "Improved" Weizsäcker-Williams approximation [47] and factorize the photon's collinear behavior into a structure function of the proton to obtain the elastic photon PDF $f_{\gamma/p}^{\text{El}}$. In Eq. (4.11), this entails replacing one $f_{i/p}$ with $f_{\gamma/p}^{\text{El}}$:

$$f_{i/p}(\xi, Q_f^2) \to f_{\gamma/p}^{\text{El}}(\xi). \tag{4.33}$$

The expression for $f_{\gamma/p}^{\text{El}}$, given in Section 2.6.5, is dependent on a cutoff scale $\Lambda_{\gamma}^{\text{El}}$, above which the description of elastic $p \to \gamma$ emission starts to break down. Typically, the scale is taken to

³ Diagram 19(d) involves a collinear singularity from massless quark splitting. It is unimportant for our current consideration since its contribution is simply a QED correction to the quark PDF. For consistency and with little change to our results, $\Lambda_{\gamma}^{\text{DIS}} = 15 \text{ GeV}$ [defined in Eq. (4.40)] is applied as a regulator.

be $\mathcal{O}(m_p - 2 \text{ GeV})$ [47, 50–52, 190, 193–195] but should be insensitive to small variations if an appropriate scale is chosen. Based on analysis of ep scattering at low Q_{γ} [58], we take

$$\Lambda_{\gamma}^{\rm El} = \sqrt{1.5 \ {\rm GeV}^2} \approx 1.22 \ {\rm GeV}. \tag{4.34}$$

The scale dependence associated with $\Lambda_{\gamma}^{\text{El}}$ is discussed in section 4.3.5.

In figure 17, the elastic luminosity spectrum ($\Phi_{\rm El}$) is denoted by the (green) dot line. For the range studied, $\Phi_{\rm El}$ is roughly 2 – 4% of the $q\bar{q}'$ DY luminosity at 14 and 100 TeV.

We calculate the matrix element for the diagrams in figure 19 in the same manner as the DY channel. The results are checked with MG5 using the elastic, asymmetric $p\gamma$ beam mode. In figures 18(a) and 18(b), we plot the bare cross section for the elastic process, denoted by a (green) dot line, as a function of neutrino mass. The rate varies between 1 - 30 (40 - 100) fb at 14 (100) TeV for $m_N = 100$ GeV-1 TeV. As seen in figures 18(c) and 18(d), where the cross sections are normalized to the DY rate, it reaches about 30 (40)% of the DY rate for large m_N .

4.3.3.2 Inelastic Scattering: Collinear Photons From Quarks For momentum transfers above the proton mass, the parton model is valid. When this configuration coincides with the collinear radiation limit, initial-state photons are appropriately described as being radiated by quark partons. To model a quark splitting to a photon, we follow the methodology of Ref. [48] and use the (original) Weizsäcker-Williams approximation [44, 45] to obtain the inelastic photon PDF $f_{\gamma/p}^{\text{Inel}}$. Unlike the elastic case, factorization requires us to convolve about a splitting function. The inelastic $N\ell^{\pm}X$ cross section is obtained by making the replacement in Eq. (4.11)

$$f_{i/p}(\xi, Q_f^2) \rightarrow f_{\gamma/p}^{\text{Inel}}(\xi, Q_\gamma^2, Q_f^2), \qquad (4.35)$$

$$f_{\gamma/p}^{\text{Inel}}(\xi, Q_{\gamma}^2, Q_f^2) = \sum_j \int_{\xi}^1 \frac{dz}{z} f_{\gamma/j}(z, Q_{\gamma}^2) f_{j/p}\left(\frac{\xi}{z}, Q_f^2\right), \qquad (4.36)$$

where $f_{\gamma/j}$ is the Weizsäcker-Williams $j \to \gamma$ distribution function, with Q_{γ} and Q_f being the factorization scales for the photon and quark distributions, respectively. The summation is over all charged quarks. Details regarding Eq. (4.36) can be found in Section 2.7.1.

Clearly, the scale for the photon momentum transfer should be above the elastic bound $Q_{\gamma} \geq \Lambda_{\gamma}^{\text{El}}$. What is not clear, however, is how high we should evolve Q_{γ} . If we crudely consider the total inclusive cross section, we could simply choose the kinematical upper limit $Q_{\gamma}^2 \approx Q_f^2 \approx \hat{s}/4$ or $\hat{s}/4 - m_N^2$, which is a quite common practice in the literature [48]. However, we do not consider this a satisfactory treatment. Well below the kinematical upper limit, the photon virtuality Q_{γ}

becomes sufficiently large so that the collinear photon approximation as in figure 19 breaks down. Consequently, "deeply inelastic scattering" (DIS), as in figure 20, becomes the dominant feature. For a brief review of DIS, see Ref. [215]. Thus, a more reasonable treatment is to introduce an upper limit for the inelastic process $\Lambda_{\gamma}^{\text{DIS}}$, above which a full DIS calculation of figure 20 should be applied. We adopt the following scheme

$$Q_{\gamma} = \Lambda_{\gamma}^{\text{DIS}} = \begin{cases} 15 \text{ GeV} & \text{for 14 TeV} \\ 25 \text{ GeV} & \text{for 100 TeV} \end{cases}$$
(4.37)

Sensitivity to variations $\Lambda_{\gamma}^{\text{DIS}}$ are discussed in section 4.3.5.

Consistent with $\Phi_{ij}(\tau)$ in Eq. (4.13), we define the inelastic γq parton luminosity Φ_{Inel} to be

$$\Phi_{\text{Inel}}(\tau) = \int_{\tau}^{1} \frac{d\xi}{\xi} \int_{\tau/\xi}^{1} \frac{dz}{z} \sum_{q,q'} \left[f_{q/p}(\xi) f_{\gamma/q'}(z) f_{q'/p}\left(\frac{\tau}{\xi z}\right) + f_{q/p}\left(\frac{\tau}{\xi z}\right) f_{\gamma/q'}(z) f_{q'/p}(\xi) \right].$$
(4.38)

In figure 17, we give the Φ_{Inel} spectrum as a function of $\sqrt{\tau}$, denoted by the (red) dash curve, for 14 and 100 TeV. For the range investigated, Φ_{Inel} ranges between 2 – 4% of the DY luminosity. Compared to its elastic counterpart, the smallness of the inelastic luminosity is attributed the limited Q_{γ}^2 evolution.

The inelastic matrix element is identical to the elastic case. In figures 18(a) and 18(b), we show the bare cross section for the inelastic process, denoted by the (red) dash line, as a function of the neutrino mass. The rate varies between 0.7 – 30 (40 – 260) fb at 14 (100) TeV for $m_N = 100 \text{ GeV} - 1 \text{ TeV}$. As seen in figures 18(c) and 18(d), where the cross sections are normalized to the DY rate, it reaches about 10 (50)% of the DY rate at large m_N .

4.3.3.3 Deeply Inelastic Scattering: High p_T Quark Jet As discussed in the previous section, at a sufficiently large momentum transfer the collinear photon description breaks down and the associated final-state quark emerges as an observable jet. The electroweak process at α^4

$$q_1 q_2 \to N \ \ell^{\pm} \ q'_1 \ q'_2.$$
 (4.39)

becomes DIS, as shown by the Feynman diagrams in figure 20. The top row of figure 20 can be identified as the DIS analog of those diagrams in figure 19. Again, the first two diagrams represent the $W\gamma$ fusion with collinear log-enhancement from *t*-channel W exchange. At these momentum transfers, the WZ fusion channel [165] turns on but is numerically smaller; see figure 20, bottom row, first diagram. The center row and two bottom-rightmost diagrams in figure 20 represent onshell W/Z production at α^3 with subsequent $W/Z \rightarrow q\bar{q}'$ decay. Those processes, however, scale



Figure 20: Feynman diagrams for the DIS process $q_1q_2 \rightarrow N\ell^{\pm}q'_1q'_2$.

as $1/\hat{s}$ and are not log-enhanced. A subset of these last diagrams also represent higher-order QED corrections to the DY process.

To model DIS, we use MG5 and simulate Eq. (4.39) at order α^4 . We impose⁴ at the generator level a minimum on momentum transfers between initial-state and final-state quarks

$$\min_{i,j=1,2} \sqrt{|(q_i - q'_j)^2|} > \Lambda_{\gamma}^{\text{DIS}}.$$
(4.40)

This requirement serves to separate the elastic and inelastic channels from DIS. Sensitivity to this cutoff is addressed in section 4.3.5.

In figure 17, we show the quark-quark parton luminosity spectrum $\Phi_{qq'}$, the source of the DIS processes, and represented by the (orange) dash-diamond curves. Though possessing the largest parton luminosity, the channel must overcome its larger coupling and phase space suppression. At 14 and 100 TeV, $\Phi_{qq'}$ ranges 3 – 5 times larger than $\Phi_{q\bar{q}'}$. The difference in size between $\Phi_{qq'}$ and $\Phi_{\rm El~(Inel)}$ is due to the additional coupling $\alpha_{\rm EM}$ in $f_{\gamma/p}^{\rm El~(Inel)}$.

In figures 18(a) and 18(b), we plot bare cross section as in Eq. (4.15), denoted by the (orange) dash-diamond curve. In figures 18(c) and 18(d), the same curves are normalized to the DY rate. At 14 (100) TeV, the cross section ranges from 1 - 60 (80 - 500) fb, reaching about 35% (80%) of

⁴For consistency, we also require the lepton cuts given in Eq. (4.31) and a jet separation $\Delta R_{jj} > 0.4$ to regularize irrelevant $\gamma^* \to q\bar{q}$ diagrams, where $\Delta R \equiv \sqrt{\Delta \phi^2 + \Delta \eta^2}$ with $y = \eta \equiv -\log[\tan(\theta/2)]$ in the massless limit.

$\sigma_{14 \text{ TeV LHC}}/ V_{\ell N} ^2 \text{ [fb]}$	$m_N = 300 \text{ GeV}$	$m_N = 500 \text{ GeV}$	$m_N = 1 \text{ TeV}$
$pp \rightarrow N\ell^{\pm} \text{ LO DY } [K = 1.2]$	293 (352)	47.3 (56.8)	2.87(3.44)
$pp \to N\ell^{\pm}X$ Elastic	10.8971	5.16756	1.23693
$pp \to N\ell^{\pm}X$ Inelastic	8.32241	3.44245	0.65728
$pp \to N\ell^{\pm}X$ DIS	11.7	5.19	1.21
$\sigma_{\gamma-{\rm Initiated}}/\sigma_{\rm DY}^{K=1.2}$	0.09	0.24	0.90
$\sigma_{100 \text{ TeV VLHC}}/ V_{\ell N} ^2 \text{ [fb]}$	$m_N = 300 \text{ GeV}$	$m_N = 500 \text{ GeV}$	$m_N = 1 \text{ TeV}$
$\boxed{\begin{array}{c} \sigma_{100 \text{ TeV VLHC}}/ V_{\ell N} ^2 \text{ [fb]} \\ pp \to N\ell^{\pm} \text{ LO DY } [K=1.3] \end{array}}$	$m_N = 300 \text{ GeV}$ 2540 (3300)	$m_N = 500 \text{ GeV}$ 583 (758)	$m_N = 1 \text{ TeV}$ 70.5 (91.6)
$ \boxed{ \begin{array}{c} \sigma_{100 \ \text{TeV VLHC}} / V_{\ell N} ^2 \ \text{[fb]} \\ \hline pp \rightarrow N \ell^{\pm} \ \text{LO DY } [K = 1.3] \\ \hline pp \rightarrow N \ell^{\pm} X \ \text{Elastic} \end{array} } $	$m_N = 300 \text{ GeV}$ 2540 (3300) 85.8	$m_N = 500 \text{ GeV}$ 583 (758) 65.5	$m_N = 1 \text{ TeV}$ 70.5 (91.6) 36.4
$ \begin{aligned} \sigma_{100 \text{ TeV VLHC}} / V_{\ell N} ^2 \text{ [fb]} \\ \hline pp \to N\ell^{\pm} \text{ LO DY } [K = 1.3] \\ \hline pp \to N\ell^{\pm} X \text{ Elastic} \\ \hline pp \to N\ell^{\pm} X \text{ Inelastic} \end{aligned} $	$m_N = 300 \text{ GeV}$ 2540 (3300) 85.8 144	$m_N = 500 \text{ GeV}$ 583 (758) 65.5 96.0	$m_N = 1 \text{ TeV}$ 70.5 (91.6) 36.4 42.7
$ \begin{aligned} \sigma_{100 \text{ TeV VLHC}} / V_{\ell N} ^2 \text{ [fb]} \\ \hline pp &\to N\ell^{\pm} \text{ LO DY } [K = 1.3] \\ pp &\to N\ell^{\pm} X \text{ Elastic} \\ \hline pp &\to N\ell^{\pm} X \text{ Inelastic} \\ \hline pp &\to N\ell^{\pm} X \text{ DIS} \end{aligned} $	$m_N = 300 \text{ GeV}$ 2540 (3300) 85.8 144 210	$m_N = 500 \text{ GeV}$ 583 (758) 65.5 96.0 145	$m_N = 1 \text{ TeV}$ 70.5 (91.6) 36.4 42.7 76.7

Table 14: Total cross sections of various $pp \to N\ell^{\pm}X$ channels for representative values of m_N after applying minimal acceptance cuts of Eqs. (4.31).

the DY rate.

To compare channels, we observe that the DIS (elastic) process increases greatest (least) with increasing collider energies. This is due to the increase likelihood for larger momentum transfers in more energetic collisions. A similar conclusion was found for elastic and inelastic $\gamma\gamma$ scattering at the Tevatron and LHC [192].

4.3.3.4 Total Neutrino Production from γ **-Initiated Processes** The total heavy neutrino production cross section from γ -initiated processes may be obtained by summing the elastic, inelastic, and DIS channels [48, 192]:

$$\sigma_{\gamma-\text{Initiated}}(N\ell^{\pm}X) = \sigma_{\text{El}}(N\ell^{\pm}X) + \sigma_{\text{Inel}}(N\ell^{\pm}X) + \sigma_{\text{DIS}}(N\ell^{\pm}X), \qquad (4.41)$$

We plot Eq. (4.41) as a function of m_N in figures 18(a) and 18(b) at 14 and 100 TeV, denoted by the (blue) dash-dot curve. In figures 18(c) and 18(d), the same curves are normalized to the DY rate. For $m_N = 100 \text{ GeV} - 1 \text{ TeV}$, the total rate spans 3 - 100 (150 - 1000) fb at 14 (100) TeV,



Figure 21: (a) The tree-level differential cross section for $N\ell^{\pm}j$ at $\alpha^{2}\alpha_{\rm s}$ with respect to p_{T}^{j} ; (b) Integrated cross section $\sigma(N\ell^{\pm}j)$ versus the minimum p_{T}^{j} cutoff. The solid line denotes the LO DY rate.

reaching about 90 (110)% of the DY rate at large m_N . We find that the $W\gamma$ fusion represents the largest heavy neutrino production mechanism for $m_N > 1$ TeV (770) GeV at 14 (100) TeV. We expect for increasing collider energy this crossover will occur earlier at lighter neutrino masses. Cross sections for representative values of m_N for all channels at 14 and 100 TeV are given in Table 14.

Before closing the discussion for the heavy N production at hadron colliders, an important remark is in order. We have taken into account the *inclusive* QCD correction at NNLO as a K-factor. In contrast, Ref. [174] included only the tree-level process at order $\alpha^2 \alpha_s^2$ and α^4

$$pp \to N\ell^{\pm}jj.$$
 (4.42)

When calculating the exclusive $N\ell^{\pm}jj$ cross section, kinematical cuts of $p_{Tj} > 10$ GeV and $\Delta R_{jj} > 0.4$ were applied to regularize the cross section. For $m_N = 300$ GeV, the exclusive cross section was found to exceed the LO DY channel at 14 TeV, whereas we find that the NNLO correction to the inclusive cross section is only 20% with DIS contributing 3%. More recently [216], the tree-level rate for $N\ell j$ with $p_T^j > 30$ GeV was calculated to be 80% of the LO DY rate at $m_N = 500$ GeV; at NNLO, we find the inclusive correction to be only 20%. We attribute these discrepancies to their

too low a p_T^j cut that overestimate the contribution of initial-state radiation based on a tree-level calculation.

To make the point concrete, we consider the tree-level QCD correction to the DY process at order $\alpha^2 \alpha_s$

$$p \ p \to N \ \ell^{\pm} \ j,$$
 (4.43)

where the final-state jet originates from an initial-state quark or gluon. MG5 is used to simulated Eq. (4.43). In figure 21(a), the differential cross section of p_T^j is shown for a minimal p_T at 5 GeV. The singularity at the origin is apparent. In figure 21(b), the 14 TeV LHC cross section as a function of minimum p_T cut on the jet is presented. A representative neutrino mass of $m_N = 300$ GeV is used; no additional cut has been imposed. At $p_T^{j \min} = 10$ GeV, as adopted in Ref. [174], the $N\ell j$ rate is nearly equal to the DY rate, well above the NNLO prediction for the inclusive cross section [185].

4.3.4 Kinematic Features of N Production with Jets at 14 TeV

To explore the kinematic distributions of the inclusive neutrino production, we fix $\sqrt{s} = 14$ TeV and $m_N = 500$ GeV. At 100 TeV, we observe little change in the kinematical features and our conclusions remain the same. The most notable difference, however, is a broadening of rapidity distributions. This is due an increase in longitudinal momentum carried by the final states, which follows from the increase in average momentum carried by initial-state partons. For $m_N \ge 100$ GeV, we observe little difference from the 500 GeV case we present. Throughout this study, jets are ranked by p_T , namely, the jet with the largest (smallest) p_T is referred to as hardest (softest).

In figure 22, we plot the (a) p_T and (b) η distributions of the hardest jet in p_T produced in association with N for the various $W\gamma$ fusion channels. Also shown are (c) p_T and (d) η distributions of the sub-leading jet for the DIS channel. For the hardest jet, we observe a plateau at $p_T \sim M_W/2$ and a rapidity concentrated at $|\eta| \sim 3.5$, suggesting dominance of t-channel W boson emission. For the soft jet, we observe a rise in cross section at low p_T and a rapidity also concentrated at $|\eta| \sim 3.5$, indicating t-channel emission of a massless vector boson. We conclude that VBF is the driving contribution γ -initiated heavy neutrino production.

In figure 23, we plot the (a) p_T and (b) η distributions of the charged lepton produced in association with N for all channels contributing to $N\ell$ production. Also shown are the (c) p_T and (d) y distribution of N. For both leptons, we observed a tendency for softer p_T and broader



Figure 22: Stacked (a) p_T and (b) η differential distributions, divided by $|V_{\ell N}|^2$, at 14 TeV LHC of the leading jet in the elastic (solid fill), inelastic (dot fill), and DIS (crosshatch fill) processes. (c) p_T and (d) η of the sub-leading jet in DIS.

rapidity distributions in γ -initiated channels than in the DY channel. As DY neutrino production proceeds through the *s*-channel, *N* and ℓ possess harder p_T than the γ -initiated states, which proceed through *t*-channel production and are thus more forward.



Figure 23: Stacked (a) p_T and (b) η differential distributions at 14 TeV LHC of the charged lepton produced in association with N for the DY (line fill), elastic, inelastic and DIS processes. (c) p_T and (d) y of N for the same processes. Fill style and normalization remain unchanged from figure 22.

4.3.5 Scale Dependence

For the processes under consideration, namely DY and $W\gamma$ fusion, there are two factorization scales involved: Q_f and Q_γ . They characterize typical momentum transfers of the physical processes. For the γ -initiated channels, we separate the contributions into three regimes using $\Lambda_{\gamma}^{\text{El}}$ and $\Lambda_{\gamma}^{\text{DIS}}$.

	Default at	т	TT	Variation	
Scale Parameter	$14 (100) { m TeV}$	Lower	Upper	at 14 (100) TeV	
$\Lambda_{\gamma}^{\rm El} \; [{\rm Eq.} \; (4.33)]$	1.22 CoV	m_p	$2.3 \mathrm{GeV}$	O(10%) (12%)	
	1.22 Gev	m_p	$5~{ m GeV}$	${\cal O}(22\%)~(28\%)$	
$\Lambda_{\gamma}^{\mathrm{DIS}}$ [Eq. (4.37)]	$15~{\rm GeV}~(25~{\rm GeV})$	$5~{ m GeV}$	$50~{\rm GeV}$	O(10%) (15%)	
		$5~{ m GeV}$	$150~{\rm GeV}$	${\cal O}(18\%)~(27\%)$	
$Q_f^{\rm DY}$ [Eq. (4.11)]	$\sqrt{\hat{s}}/2$	$m_N/2$	$\sqrt{\hat{s}}$	O(10%) (5%)	
$Q_f^{\rm DIS}$ [Eq. (4.11)]	$\sqrt{\hat{s}}/2$	$m_N/2$	$\sqrt{\hat{s}}$	O(15%) (8%)	

Table 15: Summary of scale dependence in $N\ell^{\pm}X$ production at 14 TeV and 100 TeV.

Though the quark parton scale Q_f is present in all channels, we assume it to be near m_N and set it as in Eq. (4.14).

To quantify the numerical impact of varying these scales, each relevant cross section as a function of m_N is computed with one scale varied while all other scales are held at their default values. The test ranges are taken as

$$m_p \le \Lambda_{\gamma}^{\text{El}} \le 5 \text{ GeV}, \quad 5 \text{ GeV} \le Q_{\gamma} = \Lambda_{\gamma}^{\text{DIS}} \le 150 \text{ GeV}, \quad \frac{m_N}{2} \le Q_f \le \sqrt{\hat{s}},$$
 (4.44)

In figure 24, we plot the variation band in each production channel cross section due to the shifting scale. For a given channel, rates are normalized to the cross section using the default scale choices, as discussed in the previous sections and summarized in the first column of Table 15. High-(low-) scale choices are denoted by a solid line with right-side (upside-down) up triangles.

For the 14 TeV LO DY process, we observe in figure 24(a) maximally a 9% upward (7% downward) variation for the range of m_N investigated. Below $m_N \approx 300$ GeV, the default scale scheme curve is below (above) the high (low) scale scheme curve. The trend is reversed for above $m_N \approx 300$ GeV. At 100 TeV, the crossover point shifts to much higher values of m_N . Numerically, we observe a smaller scale dependence at the 5% level.

In figure 24(b), we plot scale variation associated with the factorization scale Q_f for DIS. Maximally, we observe a 16% upward (8% downward) shift. We observe that the crossover between the high and low scale schemes now occurs at $m_N \leq 100$ GeV. This is to be expected as \hat{s} for the



Figure 24: Cross section ratios relative to the default scale scheme, as a function of m_N , for the high-scale (triangle) and low-scale (upside-down triangle) Q_f scheme in (a) DY and (b) DIS. The same quantity as a function of (c) $\Lambda_{\gamma}^{\text{El}}$ in elastic (dot), inelastic (dash), elastic+inelastic (dash-dot) scattering; (d) $\Lambda_{\gamma}^{\text{DIS}}$ in inelastic (dash), DIS (dash-diamond), and inelastic+DIS (dash-dot).

4-body DIS at a fixed neutrino mass is much larger than that for the 2-body DY channel. Similarly, as $\sqrt{\hat{s}}$ and m_N are no longer comparable, as in the DY case, an asymmetry between the high- and low-scale scheme curves emerges. At 100 TeV, we observe a smaller dependence at the 10% level.

In figure 24(c), we show the dependence on $\Lambda_{\gamma}^{\text{El}}$ in the elastic (dot) and inelastic (dash) channels, as well as the sum of the two channels (dash-dot). For the elastic channel we find very small depen-

dence on $\Lambda_{\gamma}^{\text{El}}$ between m_p and 5 GeV, with the analytical expression for $f_{\gamma/p}^{\text{El}}$ given in Section 2.6.5. For the inelastic channel, on the other hand, we find rather large dependence on $\Lambda_{\gamma}^{\text{El}}$ between m_p and 5 GeV. Since $\Lambda_{\gamma}^{\text{El}}$ acts as the regulator of the inelastic channel's collinear logarithm, this large sensitivity is expected; see Section 2.7.1 for details regarding $f_{\gamma/p}^{\text{Inel}}$. We find that the summed rate is slightly more stable. In the region $m_p < \Lambda_{\gamma}^{\text{El}} < 2.3 \text{ GeV}$, the variation is below the 10% level. Over the entire range studied, this grows to 20%. At 100 TeV, similar behavior is observed and the dependence grows to the 30% level over the whole range.

In figure 24(d), for $m_N = 500$ GeV, we plot the scale dependence on $\Lambda_{\gamma}^{\text{DIS}}$ in the inelastic (dash) and DIS (dash-diamond) channels, as well as the sum of the two channels (dash-dot). Very large sensitivity on the scale is found for individual channels, ranging 40% - 60% over the entire domain. However, as the choice of $\Lambda_{\gamma}^{\text{DIS}}$ is arbitrary, we expect and observe that their sum is considerably less sensitive to $\Lambda_{\gamma}^{\text{DIS}}$. For $\Lambda_{\gamma}^{\text{DIS}} = 5 - 50$ (5 – 150) GeV, we find maximally a 10% (18%) variation. The stability suggests the channels are well-matched for scales in the range of 5 – 50 GeV. Results are summarized in Table 15.

4.4 HEAVY NEUTRINO OBSERVABILITY AT HADRON COLLIDERS

4.4.1 Kinematic Features of Heavy N Decays to Same-Sign Leptons with Jets at 100 TeV

We consider at a 100 TeV pp collider charged current production of a heavy Majorana neutrino N in association with n = 0, 1 or 2 jets, and its decay to same-sign leptons and a dijet via the subprocess $N \to \ell W \to \ell j j$:

$$p p \to N \ell^{\pm} + nj \to \ell^{\pm} \ell'^{\pm} + (n+2)j, \quad n = 0, 1, 2.$$
 (4.45)

Event simulation for the DY and DIS channels was handled with MG5. A NNLO K-factor of K = 1.3 is applied to the LO DY channel; kinematic distributions are not scaled by K. Elastic and inelastic channels were handled by extending neutrino production calculations to include heavy neutrino decay. The NWA with full spin correlation was applied. The elastic channel matrix element was again checked with MG5.

Detector response was modeled by applying a Gaussian smearing to jets and leptons. For jet

energy, the energy resolution is parameterized by [217]

$$\frac{\sigma_E}{E} = \frac{a}{\sqrt{E/\text{ GeV}}} \oplus b, \qquad (4.46)$$

with a = 0.6 (0.9) and b = 0.05 (0.07) for $|\eta| \le 3.2$ (> 3.2), and where the terms are added in quadrature, i.e., $x \oplus y = \sqrt{x^2 + y^2}$. For muons, the inverse- p_T resolution is parameterized by [217]

$$\frac{\sigma_{1/p_T}}{(1/p_T)} = \frac{0.011 \text{ GeV}}{p_T} \oplus 0.00017.$$
(4.47)

We will eventually discuss the sensitivity to the $e^{\pm}\mu^{\pm}$ final state and thus consider electron p_T smearing. For electrons,⁵ the p_T resolution is parameterized by [217]

$$\frac{\sigma_{p_T}}{p_T} = 0.66 \times \left(\frac{0.10}{\sqrt{p_T / \text{ GeV}}} \oplus 0.007\right).$$
(4.48)

Both the muon $1/p_T$ and electron p_T smearing are translated into an energy smearing, keeping the polar angle unchanged. We only impose the cuts on the charged leptons as listed in Eq. (4.31).

In figure 25, we show the transverse momentum and pseudorapidity distributions of the finalstate jets and same-sign dileptons for the processes in Eq. (4.45), for $m_N = 500$ GeV. Jets originating from N decay are denoted by j_{W_i} , for i = 1, 2, and are ranked by $p_T (p_T^{j_{W_1}} > p_T^{j_{W_2}})$. As the three-body $N \rightarrow \ell j j$ decay is preceded by the two-body $N \rightarrow \ell W$ process, $p_T^{j_W}$ scales like $m_N/4$, as seen in figure 25(a). The jets produced in association with N are denoted by j_3 or j_4 , and also ranked by p_T . As VBF drives these channels, we expect j_3 (associated with W^*) and j_4 (associated with γ^*) to scale like $M_W/2$ and $\Lambda_{\gamma}^{\text{DIS}}$, respectively. In figure 25(b), the η distributions of all final-state jets are shown. We see that j_3 and j_4 are significantly more forward than j_{W_1} and j_{W_2} , consistent with jets participating in VBF. The high degree of centrality of j_{W_1} and j_{W_2} follows from the central W decay.

In figures 25(c) and 25(d), we plot the p_T and η distributions of the final-state leptons. The charged lepton produced in association with N is denoted by ℓ_1 and the neutrino's child lepton by ℓ_N . As a decay product, $p_T^{\ell_N}$ scales like $(m_N - M_W)/2$, whereas $p_T^{\ell_1}$ scales as $(\sqrt{\hat{s}} - m_N)/2$. ℓ_1 tends to be soft and more forward in the γ -initiated channels.

4.4.2 Signal Definition and Event Selection: Same-Sign Leptons with Jets

For simplicity, we restrict our study to electrons and muons. We design our cut menu based on the same-sign muon channel. Up to detector smearing effects, the analysis remains unchanged for

⁵ For this group of exotic searches, the dominant lepton uncertainty stems from p_T mis-measurement. The energy uncertainty is only 1% versus a 20% uncertainty in the electron p_T resolution [217].


Figure 25: (a) p_T and (b) η differential distributions of the final-state jets for the processes in Eq. (4.45), for $m_N = 500$ GeV; (c,d) the same for final-state same-sign dileptons.

electrons. A summary of imposed cuts are listed in Table 16. Jets and leptons are identified by imposing an isolation requirement; we require

$$\Delta R_{jj} > 0.4, \quad \Delta R_{\ell\ell} > 0.2. \tag{4.49}$$

We define our signal as two muons possessing the same electric charge and at least two jets satisfying the following fiducial and kinematic requirements:

$$|\eta^{\ell}| < 2.5, \quad p_T^{\ell} > 30 \text{ GeV}, \quad |\eta^j| < 2.5, \quad p_T^j > 30 \text{ GeV}.$$
 (4.50)

Lepton Cuts	Jet Cuts	Other Cuts
$\Delta R_{\ell\ell} > 0.2$	$\Delta R_{jj} > 0.4$	$\Delta R_{\ell j}^{\rm Min} > 0.6$
$p_T^{\ell} \ (p_T^{\ell \operatorname{Max}}) > 30 \ (60) \ \mathrm{GeV}$	$p_T^j \ (p_T^{j \text{ Max}}) > 30 \ (40) \text{ GeV}$	$\not\!\!\!E_T < 50~{ m GeV}$
$ \eta^{\ell} < 2.5$	$ \eta^j < 2.5$	$ m_N^{\text{Candidate}} - m_N < 20 \text{ GeV}$
	$ M_W^{\text{Candidate}} - M_W < 20 \text{ GeV}$	
	$ m_{jjj} - m_t < 20 \text{ GeV} (\text{Veto})$	

Table 16: Parton-level cuts on 100 TeV $\mu^{\pm}\mu^{\pm}jjX$ Analysis.

Table 17: Acceptance rates and percentage efficiencies for the signal $\mu^{\pm}\mu^{\pm}jjX$ at 100 TeV VLHC.

$\sigma_0 \text{ [Eq. (4.16)] [fb]} \setminus m_N \text{ [GeV]}$	300	500	1000
Fiducial + Kin. + Smearing [Eq. (4.50)]	281 (41%)	83.9 (45%)	11.6 (28%)
Leading p_T Minimum [Eq. (4.51)]	278~(99%)	83.8 (>99%)	11.6 (>99%)
$\Delta R_{\ell j}$ Separation [Eq. (4.53)]	264~(95%)	79.3~(95%)	10.7~(92%)
$\not\!$	263 (>99%)	78.1 (99%)	10.1~(95%)
M_W Reco. [Eq. (4.55)]	252~(96%)	74.1 (95%)	9.51~(94%)
m_t Veto [Eq. (4.56)]	251~(99%)	73.5~(99%)	9.42~(99%)
m_N Reco. [Eq. (4.57)]	244 (98%)	$64.7 \ (88\%)$	7.79~(83%)
Acceptance $[\mathcal{A}] = \sigma_0^{\text{All Cuts}} / \sigma_0^{\text{Fid.+Kin.+Sm.}}$	87%	77%	67%

The bare cross sections [defined by factorizing out $S_{\ell\ell}$ as defined in Eq. (4.16)] after cuts listed in Eqs. (4.50) and (4.49) and smearing are given in the first row of Table 17, for representative masses $m_N = 300, 500, \text{ and } 1000 \text{ GeV}$. Events with additional leptons are rejected. Events with additional jets are kept; we have not tried to utilize the VBF channel's high-rapidity jets. About 30-45% of all $\ell^{\pm}\ell'^{\pm}jjX$ events survive these cuts. As learned from figure 25, the η requirement given in Ref. [214] considerably reduces selection efficiency. Extending the fiducial coverage to $\eta^{\text{Max}} = 3$ or larger, though technically difficult, can be very beneficial experimentally.

We plot the maximum p_T of jets in figure 26(a) and of charged leptons in figure 26(b), for $m_N =$



Figure 26: (a) Maximum jet p_T , (b) maximum charged lepton p_T , (c) minimum $\Delta R_{\ell j}$, (d) $\Delta R_{j_{W1}j_{W2}}$ distributions for $m_N = 300$, 500, and 1000 GeV.

300, 500, and 1000 GeV. One finds that the $p_T^{j \text{ Max}}$ scale is $m_N/4$ and is set by the $N \to W \to jj$ chain. For the lepton case, $p_T^{\ell \text{ Max}}$ is set by the neutrino decay and scales as $m_N/2$. In light of these, we apply the following additional selection cuts to reduce background processes:

$$p_T^{j \text{ Max}} > 40 \text{ GeV}, \quad p_T^{\ell \text{ Max}} > 60 \text{ GeV}.$$
 (4.51)

The corresponding rate is given in the second row of Table 17 and we find that virtually all events



Figure 27: (a) p_T for individual contributions to $pp \to \ell^{\pm} \ell^{\pm} j j X$ at $m_N = 500$ GeV. (b) Total p_T for same m_N as figure 26.

pass Eq. (4.51). As both p_T^{Max} are sensitive to m_N , searches can be slightly optimized by instead imposing the variable cut

$$p_T^{j \operatorname{Max}} \gtrsim \mathcal{O}\left(\frac{m_N}{4}\right), \quad p_T^{\ell \operatorname{Max}} \gtrsim \mathcal{O}\left(\frac{m_N}{2}\right).$$
 (4.52)

In each of the several production channels, the final-state charged leptons and jets are widely separated in ΔR ; see figure 26(c). With only a marginal effect on the signal rate, we impose the following cut that greatly reduce heavy quarks backgrounds such as $t\bar{t}$ production [166]:

$$\Delta R_{\ell j}^{\min} > 0.6. \tag{4.53}$$

The corresponding rate is given in the third row of Table 17. If needed, Eq. (4.53) can be set as high as 1.0 and still maintain a high signal efficiency.

In figure 26(d), the separation between the jets in the N decay is shown. For increasing m_N , the separation decreases. This is the result of the W boson becoming more boosted at larger m_N , resulting in more collimated jets. For TeV-scale N, substructure techniques become necessary for optimize event identification and reconstruction. We reserve studying the inclusive same-sign leptons with at least one (fat) jet for a future analysis.



Figure 28: Reconstructed invariant mass of the (a) W boson and (b) heavy N candidates for same m_N as figure 26.

For the signature studied here, no light neutrinos are present in the final state. For the heavy neutrino widths listed in Eq. (4.22), the decay length $\beta c\tau$ is from $10^{-2} - 1$ fm, indicating that Nis very short-lived. Thus, there is no source of missing transverse momentum (MET) in the samesign leptons with (n + 2)j aside from detector-level mis-measurements, which are parameterized by Eqs. (4.46)-(4.48). With this smearing parameterization, forward (large η) jets are observed with less precision than central (small η) jets. Due to the naturally larger energies associated with forward jets participating in VBF at 100 TeV, the energy-dependent term in Eq. (4.46) provides a potentially large source of momentum mis-measurements in our analysis. This channel-dependent behavior can be seen in figure 27(a) for $m_N =500$ GeV. The increase in MET is found to be modest. In figure 27(b), we plot the combined MET differential distribution for representative m_N . To maximize the contributions to our signal rate, we impose the loose criterion

$$p_T < 50 \text{ GeV.}$$
 (4.54)

The corresponding rate is given in the fourth row of Table 17 and show that most events pass. Though technically difficult, tightening this cut can greatly enhance the signal-to-noise ratio.

To identify the heavy neutrino resonance in the complicated $\ell^{\pm}\ell^{\pm} + (n+2)j$ topology, we exploit that the $N \to \ell^{\pm}jj$ decay results in two very energetic jets that remain very central and possess a resonant invariant mass. In the 4j final-state channel, (rare) contributions from $N\ell^{\pm}W^{\mp}$ can lead to the existence of a second W boson in our signal. To avoid identifying a second W (or a continuum distribution) as the W boson from our heavy neutrino decay, we employ the following algorithm: (i) First consider all jets satisfying Eq. (4.50) and require that at least one pair possesses an invariant mass close to M_W , i.e.,

$$|m_{j_m j_n} - M_W| < 20 \text{ GeV.}$$
(4.55)

(ii) If no such pair has an invariant mass within 20 GeV of M_W , then the event is rejected. (iii) If more than one pair satisfies Eq. (4.55), including the situation where one jet can satisfy Eq. (4.55) with multiple partners, we identify the jj-system with the highest p_T as the child W boson from the heavy neutrino decay. This last step is motived by the fact that the p_T of neutrino's decay products scale like $p_T \sim m_N/2$, and thus at larger values of m_N the W boson will become more boosted. This is contrary to $N\ell^{\pm}W^{\mp}$ and continuum events, in which all states are mostly produced close to threshold. In figure 28(a), we plot the reconstructed invariant mass of the dijet system satisfying this procedure and observe a very clear resonance at M_W . The corresponding rate is given in the fifth row of Table 17 and show most events pass.

To remove background events from $t\bar{t}W$ production, events containing four or more jets with any three jets satisfying

$$|m_{jjj} - m_t| < 20 \text{ GeV}$$
 (4.56)

are rejected. As this is a non-resonant distribution in the $N\ell + nj$ channels, its impact on the signal rate is minimal. The corresponding rate is given in the sixth row of Table 17 and show that nearly all events pass. A top quark-veto can be further optimized by introducing high-purity anti-*b*-tagging, e.g., Ref. [218].

We identify N by imposing the m_N -dependent requirement on the two ($\ell_i, W_{\text{Cand.}}$) pairs and choose whichever system possesses an invariant mass closer to m_N . In figure 28(b), we plot the reconstructed invariant mass of this system observing very clear peaks at m_N . It is important to take into account that the width of the heavy neutrino grows like m_N^3 , and reaches the 10 GeV-level at $m_N = 1$ TeV. Therefore, we apply the following width-sensitive cut:

$$|m_{N \text{ Cand.}} - m_{N}| < \text{Max}(20 \text{ GeV}, 3\Gamma_{N}).$$

$$(4.57)$$

The corresponding rate is given in the seventh row of Table 17 and show most events pass.

$m_N \; [\text{GeV}]$	100	200	300	400	500	600
$\sigma_0^{\text{All Cuts}} \text{ [fb]}$	205	588	244	118	64.7	48.1
$\sigma_{\rm Tot}^{\rm SM}$ [ab]	16.3	115	53.2	22.2	11.4	6.01
$n_{2\sigma}^{b+\delta_{\rm Sys}}(100 \ {\rm fb}^{-1})$	4	18	9	5	3	2
$n_{2\sigma}^{s}(100 \text{ fb}^{-1})$	8	16	11	9	7	6
$m_N \; [\text{GeV}]$	700	800	900	1000	1100	1200
$\sigma_0^{\text{All Cuts}} \text{ [fb]}$	23.4	14.4	10.5	7.79	4.61	4.01
$\sigma_{\rm Tot}^{\rm SM}$ [ab]	3.47	1.94	1.57	1.25	0.795	0.649
$n_{2\sigma}^{b+\delta_{\rm Sys}}(100 \ {\rm fb}^{-1})$	2	1	1	1	1	1
$n_{2\sigma}^{s}(100 \text{ fb}^{-1})$	7	5	5	5	5	5

Table 18: Expected $\mu^{\pm}\mu^{\pm}jjX$ (bare) signal and SM background rates at 100 TeV VLHC after cuts. Number of background events and required signal events for 2σ sensitivity after 100 fb⁻¹.

Table 19: Same as Table 18 for $e^{\pm}\mu^{\pm}jjX$.

$m_N \; [\text{GeV}]$	100	200	300	400	500	600
$\sigma_0^{\text{All Cuts}} \text{ [fb]}$	408	1160	480	230	125	93.2
$\sigma_{\rm Tot}^{\rm SM}$ [ab]	196	4000	578	82.2	17.7	8.20
$n_{2\sigma}^{b+\delta_{\rm Sys}}(100 \ {\rm fb}^{-1})$	27	434	71	13	4	3
$n_{2\sigma}^{s}(100 \text{ fb}^{-1})$	18	71	30	13	8	8
$m_N \; [\text{GeV}]$	700	800	900	1000	1100	1200
$\sigma_0^{\text{All Cuts}} \text{ [fb]}$	44.9	27.7	20.3	15.1	8.98	7.86
$\sigma_{\rm Tot}^{\rm SM}$ [ab]	4.79	2.68	2.07	1.87	1.29	0.932
$n_{2\sigma}^{b+\delta_{\rm Sys}}(100 \ {\rm fb}^{-1})$	2	1	1	1	1	1
$n_{2\sigma}^{s}(100 \text{ fb}^{-1})$	6	5	5	5	5	5

The acceptance \mathcal{A} of our signal rate, defined as

$$\mathcal{A} = \sigma_{\text{All Cuts}} / \sigma_{\text{Fidcuial Cuts+Kinematic Cuts+Smearing}}, \tag{4.58}$$

is given in the last row of Table 17. The total bare rate for the $\mu\mu$ and μe channels at representative values of m_N are given, respectively, in the Tables 18 and 19.

4.4.3 Background

Although there are no lepton-number violating processes in the SM, there exist rare processes with final-state, same-sign leptons as well as "faked" backgrounds from detector mis-measurement. Here we describe our estimate of the leading backgrounds to the final-state

$$pp \to \ell^{\pm} \ell^{'\pm} + n \ge 2j + X \tag{4.59}$$

for the $\mu\mu$ and $e\mu$ channels. The principle SM processes are $t\bar{t}X$, $W^{\pm}W^{\pm}X$, and electron charge misidentification. We model the parton-level matrix elements of these processes using MG5_aMC@ NLO [56] and the CTEQ6L PDFs [63] with factorization and renormalization scales $Q = \sqrt{\hat{s}}/2$. We perform the background analysis in the same manner as for the signal-analysis.

4.4.3.1 $t\bar{t}$ At 100 TeV, radiative EW processes at $\alpha_s^2 \alpha$ such as

$$p p \rightarrow t \bar{t} W^{\pm} \rightarrow b \bar{b} W^{+} W^{-} W^{\pm} \rightarrow \ell^{\pm} \ell^{'\pm} b \bar{b} j j \nu_{\ell} \nu_{\ell'}, \qquad (4.60)$$

possess non-negligible cross sections. At LO, $\sigma(t\bar{t}W \to \mu^{\pm}\mu^{\pm}b\bar{b}jj\nu_{\mu}\nu_{\mu}) \approx 40$ fb, and threatens discovery potential. At 14 TeV, $t\bar{t}W$ possesses a NLO K-factor of K = 1.2 [219]. As an estimate, this is applied at 100 TeV. As shown in Table 20, the tight acceptance cuts reduce the rate by roughly 75%. Unlike the signal process, $t\bar{t}W$ produces two light neutrinos, an inherent source of MET. After the MET cut, the background rate is reduced to the 2 fb level. Lastly, the decay chain

$$t \to b W \to b j j \tag{4.61}$$

can be reconstructed into a top quark. Rejecting any event with a three-jet invariant mass near the top quark mass, i.e., Eq. (4.56), dramatically reduces this background to the tens of ab level. At this point, approximately 0.2% of events passing initial selection criteria survive.

$\sigma(t\bar{t}W)$ [fb]	$e\mu$	$\mu\mu$	
Fiducial + Kinematics + Smearing $[K = 1.2]$ [Eq. (4.50)]	20.5	10.3~(26%)	
Leading p_T Minimum [Eq. (4.51)]	16.5	8.23~(80%)	
$\Delta R_{\ell j}$ Separation [Eq. (4.53)]	11.8	5.91 (72%)	
$\not\!$	3.58	1.78 (30%)	
M_W Reconstruction [Eq. (4.55)]	2.54	1.27~(72%)	
m_t Veto [Eq. (4.56)]	0.0452	0.0213 (2%)	
$\sigma(t\bar{t})$ (Electron Charge Mis-ID) [fb]	$e\mu$		
Fiducial + Kinematics + Smearing [Eq. (4.50)] [$K = 0.96$]	94.5	$\times 10^3 (21\%)$	
Leading p_T Minimum [Eq. (4.51)]	$67.0 \times 10^3 (71\%)$		
$\Delta R_{\ell j}$ Separation [Eq. (4.53)]	55.2	$\times 10^3 (82\%)$	
$\not\!$	21.4 :	$\times 10^3 (39\%)$	
M_W Reconstruction [Eq. (4.55)]	3.12	$\times 10^3 (15\%)$	
m_t Veto [Eq. (4.56)]	3.12 ×	$(10^3 (100\%))$	
Charge Mis-ID $[\epsilon_{e \text{ Mis-ID}}]$ [Eq. (4.64)]	10.	9~(0.4%)	

Table 20: Acceptance rates for SM $t\bar{t}$ at 100 TeV pp collider.

At 100 TeV, the NLO $t\bar{t}$ cross section is estimated to be $\sigma(t\bar{t}) \approx 1.8 \times 10^7$ fb [214]. Hence, rare top quark decays have the potential to spoil our sensitivity, e.g.,

$$pp \rightarrow t \,\overline{t} \rightarrow b \,\overline{b} \,W^+ \,W^- \rightarrow b \,\overline{c} \,\ell^+ \,\ell^{+'} \,\nu_\ell \,\nu_{\ell'} \,W^- + \text{c.c.}, \qquad (4.62)$$

where a *b*-quark hadronizes into a *B*-meson that then decays semi-leptonically through the $b \to c\ell\nu_{\ell}$ subprocess, which is proportional to the small mixing $|V_{cb}|^2$. The MET and $\Delta R_{\ell j}$ cuts render the rate negligible [168]. Usage of high-purity anti-*b* tagging techniques [218] can further suppresses this process. The $b \to u$ transition offers a similar background but is $|V_{ub}/V_{cb}|^2 \sim (0.1)^2$ smaller [136].

4.4.3.2 Electron Charge Misidentification An important source of background for the $e^{\pm}\mu^{\pm}$ channel is from electron charge misidentification in fully leptonic decays of top quark pairs:

$$p \ p \to t \ \overline{t} \to b \ \overline{b} \ W^+ \ W^- \to b \ \overline{b} \ e^{\pm} \ \ell^{\mp} \ \nu_e \nu_\ell, \quad \ell = e, \ \mu.$$

$$(4.63)$$

Such misidentification occurs when an electron undergoes bremsstrahlung in the tracker volume and the associated photon converts into an e^+e^- pair. If the electron of opposite charge carries a large fraction of the original electron's energy, then the oppositely charged electron may be misidentified as the primary electron. For muons, this effect is negligible due the near absence of photons converting to muons [220, 221]. At the CMS detector, the electron charge misidentification rate, $\epsilon_{\rm e\ Mis-ID}$, has been determined as a function of generator-level η [221]. We assume a conservative, uniform rate of

$$\epsilon_{e \text{ Mis-ID}} = 3.5 \times 10^{-3}.$$
 (4.64)

To estimate the effect of electron charge mis-ID at 100 TeV, we consider Eq. (4.63), normalized to NLO. Other charge mis-ID channels, including Z + nj, are coupling/phase space suppressed compared to $t\bar{t}$. The $t\bar{t}$ rate after selection cuts is recorded in Table 20, and exists at the 100 pb level. We find that the electron charge mis-ID rate for Eq. (4.63) can be as large as 11 fb before the m_N cand cut is applied. As either electron in the $e^{\pm}e^{\pm}$ channel can be tagged, the mis-ID background is the same size as the $e^{\pm}\mu^{\pm}$ channel. Applying the m_N cand cut we observe that the background quickly falls off for $m_N \gtrsim 200$ GeV. As with other backgrounds possessing final-state bottoms, high purity anti-*b*-tagging offers improvements. We conclude that the effects of charge misidentification are the dominant background in electron-based final states.

4.4.3.3 $W^{\pm}W^{\pm}$ The QCD and EW processes at orders $\alpha_s^2 \alpha^2$ and α^3 , respectively,

$$p p \rightarrow W^{\pm} W^{\pm} j j$$

$$(4.65)$$

$$p p \rightarrow W^{\pm} W^{\pm} W^{\mp}$$
 (4.66)

present a challenging background due to their sizable rates and kinematic similarity to the signal process. The triboson production rate at NLO in QCD for 14 TeV LHC has been calculated [222]. As an estimate, we apply the 14 TeV K-factor of K = 1.8 to the 100 TeV LO $W^{\pm}W^{\pm}W^{\mp}$ channel. After requiring the signal definition criteria, we find the $W^{\pm}W^{\pm}$ backgrounds are present at the several fb-level. Like $t\bar{t}$, the $W^{\pm}W^{\pm}X$ final states possess light neutrinos and non-negligible MET. Imposing a maximum on the allowed MET further reduces the background by about 35%. As no $W \rightarrow jj$ decay exists in the QCD process, the reconstructed M_W requirement drops the rate considerably. After the m_t veto, the SM $W^{\pm}W^{\pm}X$ rate is 0.4 (0.9) fb for the $\mu\mu$ ($e\mu$) channel.

For all background channels, we apply the m_N -dependent cut given in Eq. (4.57) on the invariant mass of the reconstructed W candidate with either charged lepton. The total expected SM

$\sigma(W^{\pm}W^{\pm} + 2j) \text{ [fb]}$	$e\mu$	$\mu\mu$
Fiducial + Kinematics + Smearing [Eq. (4.50)]	11.6	5.78 (11%)
Leading p_T Minimum [Eq. (4.51)]	9.45	4.72 (82%)
$\Delta R_{\ell j}$ Separation [Eq. (4.53)]	7.46	3.63~(77%)
$\not\!$	2.56	1.28~(35%)
M_W Reconstruction [Eq. (4.55)]	0.132	0.0664~(5%)
m_t Veto [Eq. (4.56)]	0.132	0.0664 (100%)
$\sigma(W^{\pm}W^{\pm}W^{\mp})$ [fb]	$e\mu$	$\mu\mu$
Fiducial + Kinematics + Smearing $[K = 1.8]$ [Eq. (4.50)]	3.35	1.68 (13%)
Leading p_T Minimum [Eq. (4.51)]	2.53	1.26~(75%)
$\Delta R_{\ell j}$ Separation [Eq. (4.53)]	2.31	1.11 (87%)
$\not\!$	0.754	0.375~(34%)
M_W Reconstruction [Eq. (4.55)]	0.735	0.368~(98%)
m_t Veto [Eq. (4.56)]	0.735	0.368~(100%)

Table 21: Acceptance rates for SM $W^{\pm}W^{\pm}$ at 100 TeV pp collider.

background after all selection cuts as a function of m_N are given for the $\mu\mu$ channel in figure 29(a), and the $e\mu$ channel in figure 29(b). The total expected SM background for representative values of m_N are given in Tables 18 and 19, respectively. For these channels, we find a SM background of 1-115 ab and 1-4000 ab for the neutrino masses considered. For both channels, the background is greatest for $m_N \leq 400$ GeV and become comparable for $m_N \gtrsim 600$ GeV.

4.4.4 Discovery Potential at 100 TeV

We now estimate the discovery potential at the 100 TeV VLHC of *L*-violation via same-sign leptons and jets. We quantify this using Poisson statistics. Details of our treatment can be found in Section 2.8. The total neutrino cross section is related to the total bare cross section by the expression

$$\sigma(\ell^{\pm}\ell^{'\pm}jj + X) = S_{\ell\ell'} \times \sigma_0(\ell^{\pm}\ell^{'\pm}jj + X).$$
(4.67)



Figure 29: Total SM background versus m_N for (a) $\mu^{\pm}\mu^{\pm}$ and (b) $e^{\pm}\mu^{\pm}$ channels at 100 TeV.

We consider two scenarios for $S_{\mu\mu}$, one used by Ref. [168], dubbed the "optimistic" scenario,

$$S_{\mu\mu} = 6 \times 10^{-3}, \tag{4.68}$$

and the more stringent value obtained in Eq. (4.25), dubbed the "pessimistic" scenario,

$$S_{\mu\mu} = 1.1 \times 10^{-3}. \tag{4.69}$$

For $S_{e\mu}$, we use the m_N -dependent quantity obtained in Eq. (4.24), i.e., $10^{-5} - 10^{-6}$. We introduce a 20% systematic uncertainty by making the following scaling to the SM background cross section

$$\sigma_{\rm SM} \to \delta_{\rm Sys} \times \sigma_{\rm SM}, \quad \delta_{\rm Sys} = 1.2.$$
 (4.70)

For the $\mu\mu$ and $e\mu$ channels, respectively, the maximum number of background events and requisite number of signal events at a 2σ significance after 100 fb⁻¹ are given in Tables 18 and 19. For the $\mu\mu$ channel, these span 1 – 18 background and 5 – 16 signal events; for $e\mu$, 1 – 434 and 5 – 71 events.

We translate this into sensitivity to the mixing parameter $S_{\ell\ell'}$ and plot the 2σ contours in $S_{\ell\ell'} - m_N$ space assuming 100 fb⁻¹ (dash-diamond) and 1 ab⁻¹ (dash-star) for the $\mu\mu$ [figure 30(a)] and $e\mu$ [figure 30(b)] channels. In the $\mu\mu$ scenario and $m_N = 500$ GeV, a mixing at the level of $S_{\mu\mu} = 1.2 \times 10^{-3} (2.5 \times 10^{-4})$ with $100^{-1} (1 \text{ ab}^{-1})$ can be probed. The optimistic (pessimistic)



Figure 30: At 100 TeV and as a function of m_N , the 2σ sensitivity to $S_{\ell\ell'}$ after 100 fb⁻¹ (dash-diamond) and 1 ab⁻¹ (dash-star) for the (a) $\mu^{\pm}\mu^{\pm}$ and (b) $e^{\pm}\mu^{\pm}$ channels. The optimistic (pessimistic) bound is given by the solid (short-dash) horizontal line. (c) The required luminosity for a 3 (dash-circle) and 5σ (dash-star) discovery in the $\mu^{\pm}\mu^{\pm}$ channel

bound is given by the solid (short-dash) horizontal line. In the $e\mu$ scenario and the same mass, we find sensitivity to $S_{e\mu} = 7.2 \ (1.5) \times 10^{-4}$. For the $e\mu$ channel, the EW+ $0\nu\beta\beta$ bound is at the $10^{-6} - 10^{-5}$ level. Sensitivity to $S_{\ell\ell'}$ at 100 TeV is summarized in Table 22.

Comparatively, we observe a slight "dip" (broad "bump") in the $\mu\mu$ (e μ) curve around 200

	L	$S_{e\mu}(100 \text{ TeV})$	$S_{\mu\mu}(100 \text{ TeV})$	$S_{\mu\mu}(14 \text{ TeV})$
2-	$100 {\rm ~fb^{-1}}$	$4.9 imes 10^{-4}$	$2.7 imes 10^{-4}$	1.4×10^{-4}
2σ	1 ab^{-1}	1.4×10^{-4}	$7.5 imes 10^{-5}$	3.1×10^{-5}
275 C	$100 {\rm ~fb^{-1}}$	6×10^{-4}	$7.5 imes 10^{-4}$	3×10^{-3}
375 GeV	1 ab^{-1}	$1.7 imes 10^{-4}$	$1.8 imes 10^{-4}$	$5.5 imes 10^{-4}$
500 C-V	$100 {\rm ~fb^{-1}}$	$7.2 imes 10^{-4}$	1.2×10^{-3}	8×10^{-3}
ouu Gev	$1 {\rm ~ab^{-1}}$	$1.5 imes 10^{-4}$	$2.5 imes 10^{-4}$	1.1×10^{-3}

Table 22: Sensitivity to the mixing parameter $S_{\ell\ell'}$ at the 14 TeV LHC and 100 TeV VLHC

GeV. For the $\mu\mu$ channel, this is due to the low signal acceptance rates for Majorana neutrinos very close to the W threshold; the search methodology for m_N near or below the M_W has been studied elsewhere [166, 168]. For $m_N \geq 200$ GeV, the signal acceptance rate grows rapidly, greatly increasing sensitivity. In the $e\mu$ channel, the electron charge mis-ID background is greatest in the region around 200 GeV and quickly dwindles for larger m_N . In the low-mass regime, we find greater sensitivity in the $\mu\mu$ channel. However, due to flavor multiplicity and comparable background rates, the $e\mu$ channel has greater sensitivity in the large- m_N regime.

In figure 30(c), we plot as a function of m_N the required luminosity for a 3σ (circle) and 5σ (star) discovery in the $\mu\mu$ channel for the optimistic (red, dash) and pessimistic (purple, dash-dot) mixing scenarios. With 100 fb⁻¹(1 ab⁻¹) and in the optimistic scenario, a Majorana neutrino with $m_N = 580$ (1070) GeV can be discovered at 5σ significance; with the same integrated luminosity but in the pessimistic scenario, the reach is $m_N = 215$ (615) GeV. In the optimistic (pessimistic) scenario, for a 375 GeV Majorana neutrino, a benchmark used by Ref. [168], a 5σ discovery can be achieved with 40 (350) fb⁻¹; for 500 GeV, this is 80 (750) fb⁻¹. Sensitivity to m_N at 100 TeV is summarized in Table 23.

4.4.5 Updated Discovery Potential at 14 TeV LHC

We update the 14 TeV LHC discovery potential to heavy Majorana neutrinos above the W boson threshold decaying to same-sign muons. Our procedure largely follows the 100 TeV scenario but numerical values are based on Ref. [168]. Signal-wise, we require exactly two same-sign muons

-					
100 TeV	$2\sigma(100 \text{ fb}^{-1})$	$5\sigma(100 \text{ fb}^{-1})$	$5\sigma(1 \text{ ab}^{-1})$	$\mathcal{L}_{5\sigma}(375 \text{ GeV})$	$\mathcal{L}_{5\sigma}(500 \text{ GeV})$
Optimistic	$980~{ m GeV}$	$580~{\rm GeV}$	$1070 { m ~GeV}$	$40 {\rm ~fb^{-1}}$	80 fb^{-1}
Pessimistic	$470 {\rm GeV}$	$215~{\rm GeV}$	$615~{ m GeV}$	$380 { m ~fb^{-1}}$	$750 { m ~fb^{-1}}$
14 TeV	$2\sigma(100 \text{ fb}^{-1})$	$5\sigma(100 \text{ fb}^{-1})$	$5\sigma(1 \text{ ab}^{-1})$	$\mathcal{L}_{5\sigma}(375 \text{ GeV})$	$\mathcal{L}_{5\sigma}(500 \text{ GeV})$
Optimistic	$465~{\rm GeV}$	$270~{\rm GeV}$	$530 { m ~GeV}$	$300 {\rm ~fb^{-1}}$	$810 { m ~fb^{-1}}$
Pessimistic	$255~{ m GeV}$	$135~{\rm GeV}$	$280 { m ~GeV}$	2.6 ab^{-1}	6.9 ab^{-1}

Table 23: Sensitivity to heavy neutrino production in the $\mu\mu$ channel at 14 and 100 TeV.

Table 24: Parton-level cuts on 14 TeV $\mu^{\pm}\mu^{\pm}jjX$ Analysis

Lepton Cuts	Jet Cuts	Other Cuts
$\Delta R_{\ell\ell} > 0.2$	$\Delta R_{jj} > 0.4$	$\Delta R_{\ell j}^{\rm Min} > 0.5$
$p_T^{\ell} \ (p_T^{\ell \text{ Max}}) > 10 \ (30) \text{ GeV}$	$p_T^j \ (p_T^{j \text{ Max}}) > 15 \ (40) \text{ GeV}$	$\not\!\!\!E_T < 35~{ m GeV}$
$ \eta^\ell < 2.4$	$ \eta^j < 2.4$	$ m_N^{\rm Candidate} - m_N < 20 {\rm GeV}$
	$ M_W^{\rm Candidate} - M_W < 20 {\rm ~GeV}$	
	$ m_{jjj} - m_t < 20 \text{ GeV (Veto)}$	

Table 25: Same as Table 18 for 14 TeV LHC.

$\sigma \setminus m_N \; [{ m GeV}]$	100	200	300	400	500	600	700
$\sigma_0^{\text{All Cuts}} \text{ [fb]}$	576	132	36.0	14.0	6.28	3.05	1.55
$\sigma_{ m Tot}^{ m SM}$ [ab]	14.1	18.6	5.62	2.05	0.837	0.393	0.195
$n_{2\sigma}^{b+\delta_{\rm Sys}}(100 \ {\rm fb}^{-1})$	4	4	2	1	1	0	0
$n_{2\sigma}^{s}(100 \text{ fb}^{-1})$	8	8	6	5	5	4	4

(vetoing additional leptons) and at least two jets (allowing additional jets) satisfying the cuts listed in Table 24. Differences from the analysis introduced by Ref. [168] include: updated smearing



Figure 31: At 14 TeV, (a) same as figure 30(a); (b) same as figure 30(c).

parameterization given in Eqs. (4.46) and (4.47); an $\not\!\!E_T$ requirement based on the ATLAS detector capabilities given in Ref. [177]; cuts on the leading charged lepton and jet; and more stringent requirements on the W and N candidate masses. These differences sacrifice sensitivity to $m_N \lesssim$ 100 GeV for high-mass reach. For our NNLO in QCD K-factor, we use K = 1.2, as given in Eq. (4.28). We report the bare heavy neutrino rate after all cuts for representative m_N in the first row of Table 25. The total bare rate ranges from 2 - 580 fb for $m_N = 100 - 700$ GeV.

As previously discussed or shown, the $t\bar{t}$ background for the dimuon channel is negligible, so we focus on $W^{\pm}W^{\pm}$ pairs. For triboson production, an NLO in QCD K factor of K = 1.8 is applied [222]. After all cuts, the expected SM background for representative m_N is given in the second row Table 25. After the m_N -dependent cut, the expected SM background rate reaches at most 19 ab. Like the 100 TeV case, a 20% systematic is introduced into the background. For the $\mu\mu$ and $e\mu$ channels, respectively, The maximum number of background events and requisite number of signal events at a 2σ significance after 100 fb⁻¹ are given in the third and fourth rows, respectively, of Table 25.

In figure 31(a), we plot the 2σ sensitivity to the mixing coefficient $S_{\mu\mu}$ after 100 fb⁻¹ (dashdiamond) and 1 ab⁻¹ (dash-star). For the benchmark $m_N = 375$ GeV, a mixing at the level of $S_{\mu\mu} = 3 \times 10^{-3} (5.5 \times 10^{-4})$ with $100^{-1} (1 \text{ ab}^{-1})$ can be probed; for $m_N = 500$ GeV, we find sensitivity to be $S_{\mu\mu} = 8 \times 10^{-3} (1.1 \times 10^{-4})$. The optimistic (pessimistic) bound is given by the solid (short-dash) horizontal line. Sensitivity to $S_{\mu\mu}$ at 14 TeV is summarized in Table 22.

In figure 31(b), we plot as a function of m_N the required luminosity for a 3σ (circle) and 5σ (star) discovery in the $\mu\mu$ channel for the optimistic (red, dash) and pessimistic (purple, dash-dot) mixing scenarios. With 100 fb⁻¹ (1 ab⁻¹) and in the optimistic scenario, a Majorana neutrino with $m_N = 270$ (530) GeV can be discovered at 5σ significance; in the pessimistic scenario, the reach is $m_N = 135$ (280) GeV. In the optimistic (pessimistic) scenario, for the 375 GeV benchmark, a 5σ discovery can be achieved with 300 (2600) fb⁻¹; for 500 GeV, this is 810 (6900) fb⁻¹. Sensitivity to m_N at 14 TeV is summarized in Table 23.

4.5 SUMMARY

The search for a heavy Majorana neutrino at the LHC is of fundamental importance. It is complimentary to the neutrino oscillation programs and, in particular, neutrinoless double-beta decay experiments. We have studied the production of a heavy Majorana neutrino at hadron colliders and its lepton-number violating decay as in Eq. (4.45), including the NNLO DY contribution, the elastic and inelastic $p\gamma \rightarrow N\ell j$ processes, and the DIS $pp \rightarrow N\ell j j$ process via $W\gamma^*$ fusion. We have determined the discovery potential of the same-sign dilepton signal at a future 100 TeV ppcollider, and updated the results at the 14 TeV LHC. We summarize our findings as follows:

- Vector boson fusion processes, e.g., Wγ → Nℓ, become increasingly more important at higher collider energies and larger mass scales due to collinear logarithmic enhancements of the cross section. At the 14 TeV LHC, the three contributing channels of elastic, inelastic and DIS are comparable in magnitude, while at the 100 TeV VLHC, the tendency, in descending importance, is DIS, inelastic, and elastic; see figures 18(a) and 18(b).
- We approximately computed the QCD corrections up to NNLO of the DY production of $N\ell$ to obtain the K-factor. We found it to span 1.2 1.5 for m_N values between 100 GeV and 1 TeV at 14 and 100 TeV pp collisions, and is summarized in Table 13.
- The $W\gamma$ fusion processes surpasses the DY mechanism at $m_N \sim 1$ TeV (770 GeV) at the 14 TeV LHC (100 TeV VLHC); see figure 18(c) [18(d)]. However, we disagree with the results of Refs. [174], where higher order contributions dominating over the LO DY production at

 $m_N \ge 200$ GeV were claimed. The discrepancy is attributed to their too low a p_T^j cut that overestimates the contribution of initial-state radiation based on a tree-level calculation.

- We have introduced a systematic treatment for combining initial-state photons from various channels and predict cross sections that are rather stable against the scale choices, typically less than 20%. The exception is the inelastic process, which is rather sensitive to the scale $\Lambda_{\gamma}^{\text{El}}$ where the elastic and inelastic processes are separated. Variation of this scale could lead to about a 30% uncertainty. Scale dependence is shown in figure 24 and the results summarized in Table 15.
- We quantified the signal observability by examining the SM backgrounds. We conclude that, with the currently allowed mixing $|V_{\mu N}|^2 < 6 \times 10^{-3}$, a 5 σ discovery can be made via the samesign dimuon channel for $m_N = 530$ (1070) GeV at the 14 TeV LHC (100 TeV VLHC) after 1 ab⁻¹; see Table 23. Reversely, for $m_N = 500$ GeV and the same integrated luminosity, a mixing $|V_{\mu N}|^2$ of the order 1.1×10^{-3} (2.5×10^{-4}) may be probed; see Table 22. This study represents the first investigation into heavy Majorana neutrino production in 100 TeV *pp* collisions.

5.0 LEPTON NUMBER VIOLATION AND W' CHIRAL COUPLINGS AT THE LARGE HADRON COLLIDER

5.1 INTRODUCTION

Neutrino experiments, over the past decade, have shown undeniably that neutrinos are massive and have large mixing angles [140–147] In the Standard Model (SM) of particle physics, neutrino masses can be accommodated by a non-renormalizable dimension-5 operator containing left-handed (L.H.) neutrinos, ν_L [223]. Such an operator can be generated at low energy by including heavy righthanded (R.H.) neutrinos, ν_R . However, the R.H. neutrinos are gauge singlets and so Majorana mass terms should also be present without violating any gauge symmetry. The consequences of massive Majorana neutrinos are well-known [224–226], and have been incorporated into many models, such as left-right symmetric theories [227]; supersymmetric (SUSY) SO(10) grand unified theories (GUTs) [228] and other GUTs [229]; R-parity violating SUSY [230]; and extra dimensions [231]. A recent review of TeV scale neutrino mass models can be found in Ref. [232].

Many of the aforementioned models contain an extended gauge group or Keluza-Klein (KK) excitations of SM gauge bosons. We refer to additional vector bosons charged under the $U(1)_{EM}$ gauge group collectively as "W'". If the masses of the W' and the lightest heavy neutrino mass eigenstate, N, are both on the order of a few TeV, then they can be produced in tandem at the Large Hadron Collider (LHC). As first observed by Ref. [162], a W' with mass greater than a Majorana neutrino's mass allows the possibility of observing the spectacular lepton number (L) violating process

$$pp \to W' \to \ell^{\pm} N \to \ell^{\pm} \ell^{\pm} jj.$$
 (5.1)

If a W' is discovered at the LHC [233], it is obviously imperative to measure its chiral coupling to fermions. In a previous work [234], three of the present authors proposed measuring the W' chiral couplings to quarks by studying the process

$$pp \to W' \to t\bar{b} \to \ell^+ \nu_\ell b\bar{b}.$$
 (5.2)

It was found that the couplings could be establish as being purely left- or purely right-handed by analyzing the polar angle of the charged lepton in the top's rest-frame with respect to the top's direction of motion in the partonic center of momentum (c.m.) frame.

We now extend this prior analysis into the leptonic sector via the L-violating cascade decay of Eq. (5.1). More specifically, by reconstructing the polar angle of the lepton originating from the neutrino decay in the neutrino rest-frame and with respect to the direction of motion of the neutrino in the partonic c.m. frame, it can be uniquely determined if the W' coupling to leptons is purely left-handed, purely right-handed, or a mixture of the two. We show that the distribution of the angle made between N's production plane and its sequential decay plane is sensitive to the W'chiral coupling with the initial-state quarks but *independent* of the W' coupling to leptons. These results are demonstrated through a combination of analytical calculations and event simulations, assuming nominal LHC parameters.

Majorana neutrinos can decay into either leptons or antileptons, and so W' and N may also contribute to the *L*-conserving collider signature

$$pp \to W' \to \ell^{\pm} N \to \ell^{+} \ell^{-} jj.$$
 (5.3)

For completeness, we have analyzed the polar angular distributions of the unlike-sign process and comment on the important differences between the *L*-conserving and *L*-violating cases.

This paper is structured as follows: First, in section 5.2, we present our notation for the W' couplings to SM particles and neutrino mass eigenstates, and list current constraints on both W''s and N's. In section 5.4, we discuss the production and decay of W''s and N's at the LHC. The like-sign lepton signature, $pp \rightarrow \ell^{\pm} \ell^{\pm} jj$, its reconstruction, and suppressed background are fully analyzed in section 5.5. In 5.6, we propose methods to measure independently the chiral couplings of the W' to leptons and to the initial-state quarks. Finally, in section 5.7, we provide a few comments on the contribution of W' and N to the L-conserving process $pp \rightarrow W' \rightarrow \ell^+ \ell^- jj$ regarding the difference between the Majorana and Dirac neutrinos. We conclude and summarize our results in section 5.8. Two appendices are additionally included. The first addresses neutrino mass mixing in the context of W' couplings, and the second presents a derivation of the matrix element and angular distributions for our like- and unlike-sign dilepton signals.

5.2 THEORETICAL FRAMEWORK AND CURRENT CONSTRAINTS

There are many Beyond the Standard Model (BSM) theories containing additional vector bosons that couple to SM fermions, for example: left-right symmetric theories [235] with a new $SU(2)_R$ symmetry and an associated W'_R ; Little Higgs models with enlarged gauge symmetries [236]; extra dimensional theories with KK excitations [237–239]. Heavy Majorana neutrinos in BSM theories [227–231], and in particular those with TeV-scale masses [161,240–242], are just as common.

In this analysis, we assume the existence of a new heavy electrically charged vector boson, W'^{\pm} with mass $M_{W'}$, and a right-handed neutrino, N_R . We denote the corresponding heavy neutrino mass eigenstate as N with mass m_N . We stipulate that $M_{W'}$ is of the order of a few TeV and $M_{W'} > m_N$ so as the $W' \to N\ell$ decay is kinematically accessible by the LHC, but do not otherwise tailor to a specific theory. Regarding the parameterization of mixing between neutrino mass eigenstates with SM flavor eigenstates, we adopt the notation of Ref. [168], and extend it to include coupling to a model-independent W' in Section 5.2.1. This parameterization is accomplished with a minimum amount of parameters.

5.2.1 Neutrino Mixing with W' Couplings

5.2.1.1 Model-Independent W' Charged Current Couplings The goal of this paper is to explore the feasibility of quantifying the properties of a new charged gauge boson, W', at the LHC. For this purpose, we relax the W' interactions to include both left-handed and right-handed leptons,

$$L_{aL} = \begin{pmatrix} \nu_a \\ l_a \end{pmatrix}_L, \quad R_{bR} = \begin{pmatrix} N_b \\ l_b \end{pmatrix}_R, \quad (5.4)$$

with a, b = 1, 2, 3. The L.H. neutrinos and charged leptons that are members of $SU(2)_L$ doublets in the Standard Model (SM) are denoted by ν_{aL} and l_a . The R.H. neutrinos, which are SM singlets, and R.H. charged leptons are denoted by N_{bR} and l_b . To grasp the feature of Left-Right symmetric models for a W', we pair N_{bR} and l_b into the an $SU(2)_R$ doublet. Though there may be more "sterile" neutrinos, i.e., b > 3, we consider only b = 3 and one new mass eigenstate in our phenomenological presentation. The mass mixing matrix in Eq. (4.3), in the present case, becomes a 6×6 matrix with several repeating entries. With this assignment, the resulting charged current interactions are

$$\mathcal{L} = \left(-\frac{g_L^{\ell}}{\sqrt{2}} W_{\mu L}^{'+} \sum_{a=1}^{3} \overline{\nu_{aL}} \gamma^{\mu} P_L l_a^- - \frac{g_R^{\ell}}{\sqrt{2}} W_{\mu R}^{'+} \sum_{b=1}^{3} \overline{N_{bR}} \gamma^{\mu} P_R l_b^- \right) + h.c.$$
(5.5)

We have explicitly included the couplings of left- and right-charged currents with new gauge interactions via $W'_{L,R}$.

The gauge state leptons, l_a and l_b , may be rotated into the mass eigenstates, which are defined to be the flavors eigenstates $\ell = e, \mu, \tau$. This amounts to the rotation

$$l_a^{-} = \sum_{\ell=e}^{\tau} O_{a\ell} \ell^{-}.$$
 (5.6)

With the SM-like simplest Higgs mechanism, this transformation is trivial and we will make it implicit without loss of generality. By simultaneously expanding into the neutrinos' mass basis and into the charged leptons' flavor basis, we obtain

$$\mathcal{L} = - \sum_{\ell=e}^{\tau} \frac{g_{L}^{\ell}}{\sqrt{2}} W_{\mu}^{\prime +} \left[\sum_{m=1}^{3} \overline{\nu_{m}} U_{m\ell}^{*} + \sum_{m'=4}^{n+3} \overline{N_{m'}^{c}} V_{m'\ell}^{*} \right] \gamma^{\mu} P_{L} \ell^{-} + h.c.$$

$$- \sum_{\ell=e}^{\tau} \frac{g_{R}^{\ell}}{\sqrt{2}} W_{\mu}^{\prime +} \left[\sum_{m=1}^{3} \overline{\nu_{m}^{c}} X_{m\ell} + \sum_{m'=4}^{n+3} \overline{N_{m'}} Y_{m'\ell} \right] \gamma^{\mu} P_{R} \ell^{-} + h.c., \qquad (5.7)$$

where

$$U_{m\ell}^* \equiv \sum_{a=1}^3 U_{ma}^* O_{a\ell}, \quad V_{m'\ell}^* \equiv \sum_{a=1}^3 V_{m'a}^* O_{a\ell}, \quad X_{m\ell} \equiv \sum_{b=1}^3 X_{mb}^* O_{b\ell}, \quad Y_{m'\ell}^* \equiv \sum_{b=1}^3 Y_{m'b} O_{b\ell}.$$
(5.8)

These are the general couplings for the W' charged currents that we follow in this study. Leptonic couplings to the SM W^{\pm} boson can be recovered from Eq. (5.7) by identifying $W'^{\pm} \to W^{\pm}$ and by setting

$$g_L^\ell = g \text{ and } g_R^\ell = 0, \tag{5.9}$$

where g is the $SU(2)_L$ coupling constant in the SM. Similarly, we arrive at the $SU(2)_R$ charged current coupling by identifying $W' \to W_R^{\pm}$ and by setting

$$g_L^\ell = 0 \text{ and } g_R^\ell \neq 0.$$
 (5.10)

In the quark sector, we do not plan to go through a fully-fledged construction for the charged current couplings. Instead, we take the simplest approach and just parameterize the modelindependent W' Lagrangian by

$$\mathcal{L} = \frac{-1}{\sqrt{2}} \sum_{i,j=1}^{3} W_{\mu}^{'+} \overline{u_i} V_{ij}^{CKM'} \gamma^{\mu} \left[g_L^q P_L + g_R^q P_R \right] d_j + h.c.,$$
(5.11)

where $V^{CKM'}$ is an unknown flavor mixing matrix.

5.2.2 W' Chiral Coupling to Fermions

The model-independent Lagrangian that governs the interaction between SM quarks and a new, massive, electrically charged vector boson, W', is given by

$$\mathcal{L} = -\frac{1}{\sqrt{2}} \sum_{i,j=1}^{3} W_{\mu}^{'+} \overline{u_i} V_{ij}^{CKM'} \gamma^{\mu} \left[g_R^q P_R + g_L^q P_L \right] d_j + h.c.,$$
(5.12)

where u_i (d_j) denotes the Dirac spinor of an up-(down-)type quark with flavor i (j); $V^{CKM'}$ parameterizes the mixing between flavors i and j for the new charged current interactions just as the Cabibbo-Kobayashi-Maskawa (CKM) matrix does in the SM; $g_{R,L}^q$ is the W''s universal coupling strength to right-(left-) handed quarks; and $P_{R,L} = \frac{1}{2} (1 \pm \gamma_5)$ denotes the R, L-handed chiral projection operator.

We parameterize the new boson's coupling to charged leptons with flavor ℓ and neutral leptons with mass m_m (for the three light states) or m_N (for the heavy state) in the following way:

$$\mathcal{L} = -\sum_{\ell=e}^{\tau} \frac{g_{R}^{\ell}}{\sqrt{2}} W_{\mu}^{'+} \left[\sum_{m=1}^{3} \overline{\nu_{m}^{c}} X_{\ell m} + \overline{N} Y_{\ell N} \right] \gamma^{\mu} P_{R} \ell^{-} -\sum_{\ell=e}^{\tau} \frac{g_{L}^{\ell}}{\sqrt{2}} W_{\mu}^{'+} \left[\sum_{m=1}^{3} \overline{\nu_{m}} U_{\ell m}^{*} + \overline{N^{c}} V_{\ell N}^{*} \right] \gamma^{\mu} P_{L} \ell^{-} + h.c.$$
(5.13)

 $g_R^\ell (g_L^\ell)$ is the W's coupling strength to R.H. (L.H.) leptons; $X_{\ell m} (U_{\ell m})$ parameterizes the mixing between light neutrino mass eigenstates and R.H. (L.H.) interactions; and $Y_{\ell N} (V_{\ell N})$ parameterizes the mixing between the heavy neutrino mass eigenstate and R.H. (L.H.) interactions. Lastly, $\psi^c = C\overline{\psi}^T$ denotes the charge conjugate of the field ψ , with C being the charge conjugate operator, and the chiral states satisfy $P_L(\psi^c) = (P_R\psi)^c$. In Section 5.2.1, our choice of parameterization is discussed in detail. From a viewpoint of the model construction as discussed in Refs. [140–147,162, 168], one may expect that UU^{\dagger} , $YY^{\dagger} \sim \mathcal{O}(1)$ and VV^{\dagger} , $XX^{\dagger} \sim \mathcal{O}(m_m/m_N)$. Since we prefer a model-independent approach, we will not follow rigorously the above argument and will take the parameters as

$$UU^{\dagger}, YY^{\dagger} \sim \mathcal{O}(1), \text{ and } VV^{\dagger}, XX^{\dagger} \sim \mathcal{O}(10^{-3}),$$
 (5.14)

which is guided by the current constraints as presented later in this section.

In Eq. (5.13), the W' is allowed to have both independent right-handed $(g_R^{q,\ell})$ and left-handed $(g_L^{q,\ell})$ couplings. Subsequently, the pure gauge states W'_R and W'_L are special cases of W' when

$$g_R^{q,\ell} \neq 0 \quad \text{and} \quad g_L^{q,\ell} = 0,$$
 (5.15)

and

$$g_R^{q,\ell} = 0 \quad \text{and} \quad g_L^{q,\ell} \neq 0,$$
 (5.16)

respectively. Additionally, the SM W coupling to leptons can be recovered from Eq. (5.13) by setting

$$g_R^{\ell} = 0, \quad \text{and} \quad g_L^{\ell} = g.$$
 (5.17)

Here, g is the usual SM $SU(2)_L$ coupling constant.

5.2.3 Current Constraints on W'

We list only the most stringent, most relevant constraints to our analysis here and refer the reader to Ref. [136, 243] for a more complete review.

• Bounds from CMS: The CMS Experiment has searched for W_R and heavy N, where $M_{W_R} > m_N$, with the $\ell^{\pm}\ell^{\pm}jj$ collider signature [244], assuming $g_R = g$. With 5.0 fb⁻¹ of 7 TeV and 3.7 fb⁻¹ of 8 TeV pp collisions, the present mass bounds for W'_R and N are

$$M_{W_R} > 2.9 \text{ TeV} (m_N \approx 0.8 \text{ TeV}) \text{ and } m_N > 1.9 \text{ TeV} (M_{W_R} \approx 2.4 \text{ TeV}.)$$
 (5.18)

$$M_{W_{SSM}} > 2.85 \text{ TeV.}$$
 (5.19)

• Bounds from ATLAS: The ATLAS Experiment has also searched for W_R and heavy N, under the same stipulations as the CMS Experiment [246]. With 2.1 fb⁻¹ of 7 TeV pp collisions, the present mass bounds for W'_R and N are

$$M_{W_R} > 2.5 \text{ TeV} \ (m_N \approx 0.8 \text{ TeV}) \text{ and } m_N > 1.6 \text{ TeV} \ (M_{W_R} \approx 1.8 \text{ TeV}.)$$
 (5.20)

 Global Fit Analysis: The effects of a generic Z' boson on EW precision observables place bounds [247] of

$$M_{Z'}/g_{Z'} \gtrsim 2.7 - 6.7 \text{ TeV}.$$
 (5.21)

For Z' and W' bosons originating from the same broken symmetry, we expect similar constraints on $M_{W'}/g_{W'}$ since

$$M_{W'} \sim M_{Z'} \times \mathcal{O}(1). \tag{5.22}$$

Bounds on W_L – W_R Mixing: Non-leptonic Kaon decays [248] and universality in Weak decays [249] constrain W_L – W_R mixing. The present bound for the L-R mixing angle ζ [227] is

$$|\zeta| \le 1 \sim 4 \times 10^{-3}.\tag{5.23}$$

5.2.4 Current Constraints on N

More complete lists of constraints on low and high mass neutrinos, respectively, are available [136, 168].

Bounds from 0νββ: For m_N ≫ 1 GeV, a lack of evidence for neutrinoless double beta decay bounds the mixing between heavy neutrino states and the electron-flavor state at [179,180,250, 251]

$$\sum_{m'} \frac{|V_{em'}|^2}{m_{m'}} < 5 \times 10^{-5} \,\mathrm{TeV}^{-1},\tag{5.24}$$

where the sum is over all heavy Majorana neutrinos.

• Bounds from EW Precision Data: A TeV scale singlet neutrino mixing with the SM flavor states is constrained [183] by

$$|V_{eN}|^2, |V_{\mu N}|^2 < 0.003 \text{ and } |V_{\tau N}|^2 < 0.006$$
 (5.25)

5.3 DERIVATION OF PARTONIC LEVEL ANGULAR DISTRIBUTIONS

We strive clarify a few subtleties that arise when calculating observables involving Majorana fermions. To do so, we present a detailed derivation of the matrix element for the lepton-number (L) violating process:

$$u_i(p_A) + \overline{d}_j(p_B) \to W'^+ \to \ell_1^+(p_1) + \ell_2^+(p_2) + q_m(p_3) + \overline{q}_n(p_4), \tag{5.26}$$

with an intermediate Majorana neutrino of mass m_N , and governed by the Lagrangian given in Section 5.2. As discussed in Section 5.5, and shown in Fig. 35, there are two interfering Feynman diagrams associated with our $2\ell^+2j$ final state. The interference term may be neglect safely when calculating the amplitude squared, $|\mathcal{M}|^2$, since the heavy neutrino's width is very narrow and thus the interference is expected to be small. When constructing and evaluating $|\mathcal{M}|^2$, we focus



Figure 32: The partonic-level process for heavy W'^+ production and decay into like-sign leptons and quarks in hadronic collisions. The longer, black arrow not touching the Feynman diagram denotes fermion flow (FF).

on only a single diagram (Fig. 32) but stress that the two diagrams can be treated identically. Additionally, the narrowness of the SM W boson's width allows us to further apply the Narrow Width Approximation (NWA). The NWA stipulates that, due to its small width compared to its mass, the W boson will dominantly be produced on-shell, and further implies

$$\hat{\sigma}(u_i \overline{d}_j \to \ell_1^+ \ell_2^+ q \overline{q}') \approx \hat{\sigma}(u_i \overline{d}_j \to \ell_1^+ \ell_2^+ W^-) \times BR(W \to q \overline{q}'), \tag{5.27}$$

where $BR(X \to Y)$ is the branching fraction of X going into Y. Since $BR(W \to q\overline{q}')$ is well-known, our work is reduced to determining the analytical expression for

$$\hat{\sigma}(u_i \overline{d}_j \to \ell_1^+ \ell_2^+ W^-). \tag{5.28}$$

5.3.1 Determination of the Spin-Summed, Polarization-Dependent, Squared Matrix Element

The usefulness of Feynman rules stems from the ability to assign specific multiplicative factors to each component of a Feynman diagram. However, Dirac field Feynman rules are dependent on Wick's Theorem, which is a statement on field contractions. For Dirac fields, only combinations of the form $\overline{\psi}\psi$ can contract, where as for Majorana fields, $\psi\psi$ and $\overline{\psi}\overline{\psi}$ are allowed to contract. In short, Feynman rules for Dirac fermions do not account for all possible Majorana interactions. We therefore adopt the Feynman rules developed in Ref. [252] for a two-fold reason. The first is that the rules for diagram segments not involving Majorana fermions do not change. The second is that for parts that do involve Majorana fermions, the new Feynman rules reduce to (a) treating the Majorana fermion like a Dirac fermion and modifying the vertex factor for an ordinary Dirac fermion with an appropriately placed factor of -1, and/or (b) making a single $u \leftrightarrow v$ spinor substitution. The placement of the additional minus sign and possible spinor substitution is based on the direction of fermion flow (FF) relative to the traditionally chosen fermion number flow (FNF). When the fermion flow and fermion number flow are equal, the newer rules simplify to the usual rules. Computationally, these rules provide a desirable technique that can be automated in a straight forward manner.

In the present case, we identify the relevant FF as being identical to the lepton number-changing current. The FF current starts at ℓ_1 , the charged lepton produced in the W' boson decay, and points anti-parallel to ℓ_1 's momentum; the current then continues parallel to the Majorana neutrino's momentum; and finally terminates at ℓ_2 , the charged lepton produced in the N decay, and points parallel to ℓ_2 's momentum. See the curved black arrow in Fig. 32. With this orientation, the FF is parallel to the FNF at the $W'\ell_1N$ vertex, and anti-parallel to it at the $N\ell_2W$ vertex. This change in relative current orientation causes two modifications, the first of which is to the spinor of the outgoing lepton originating from the $N\ell_2W$ vertex:

$$\overline{v}_{\ell_2}(p_2) \to \overline{u}_{\ell_2}(p_2),\tag{5.29}$$

and accounts explicitly for the change in lepton number. The second modification is to the $N\ell_2W$ vertex itself and occurs in the following way:

$$C_{N\ell_2}^{\rho} = \frac{-ig}{\sqrt{2}} V_{\ell_2 N} \gamma^{\rho} P_L \to C_{N\ell_2}^{\prime \rho} = (-1)^2 \frac{ig}{\sqrt{2}} V_{\ell_2 N} \gamma^{\rho} P_R, \qquad (5.30)$$

where g is the SM SU(2)_L coupling constant, $P_{R,L} \equiv \frac{1}{2}(1 \pm \gamma_5)$, and, as defined in Ref. [252], the primed-vertex convention indicates

$$\Gamma' \equiv \mathcal{C}\Gamma^T \mathcal{C}^{-1} = \eta \Gamma, \tag{5.31}$$

where C is the charge conjugation operator and for which

$$\eta = \begin{cases} 1, & \Gamma \in \{1, i\gamma^5, \gamma^{\mu}\gamma^5\} \\ -1, & \Gamma \in \{\gamma^{\mu}, \sigma^{\mu\nu}\} \end{cases}.$$
(5.32)

As a result, we find that the matrix element describing the $u_i \overline{d}_j \to \ell_1^+ \ell_2^+ W_\lambda^-$ scattering process, for an outgoing SM W^- boson with polarization λ , and in the Feynman Gauge, is

$$i\mathcal{M}_{\lambda} = \varepsilon_{\lambda\rho}^{*}(p_{W}) \cdot \frac{\left[\overline{v}_{Bj}A_{ji}^{\mu}u_{Ai}\right] \cdot \left[\overline{u}_{2}C_{N\ell_{2}}^{\prime\rho}(\not p_{N}+m_{N})B_{\mu\,\ell_{1}N}v_{1}\right]}{\left(\hat{s}-M_{W^{\prime}}^{2}+i\Gamma_{W^{\prime}}M_{W^{\prime}}\right)\left(p_{N}^{2}-m_{N}^{2}+i\Gamma_{N}m_{N}\right)},\tag{5.33}$$

where the vertex terms are given by

$$A_{ji}^{\mu} = \frac{1}{\sqrt{2}} V_{ji}^{CKM'} \gamma^{\mu} \left[g_R^q P_R + g_L^q P_L \right]$$
(5.34)

$$B_{\ell_1 N}^{\nu} = \frac{1}{\sqrt{2}} \gamma^{\nu} \left[g_R^{\ell} P_R Y_{\ell_1 N} + g_L^{\ell} P_L V_{\ell_1 N}^* \right]$$
(5.35)

To be explicit: $\varepsilon_{\lambda\rho}^*(p_W)$ denotes the outgoing polarization vector of the on-shell W boson with momentum p_W , mass M_W , and polarization λ ; \overline{v}_{Bj} represents the spinor \overline{v} of an initial-state antiquark of flavor j and momentum p_B ; similarly, u_{Ai} represents the spinor u of an initial-state quark of flavor i and momentum p_A ; \overline{u}_2 denotes the spinor of our final-state *antilepton* with flavor ℓ_2 and momentum p_2 ; and likewise, v_1 denotes the spinor of our final-state antilepton with flavor ℓ_1 and momentum p_1 . The W' mass, width, and momentum-squared are respectively given by $M_{W'}$, $\Gamma_{W'}$, and the Mandelstam variable

$$\hat{s} = (p_A + p_B)^2 = (p_1 + p_2 + p_W)^2.$$
 (5.36)

The heavy neutrino's mass, width, and momentum are similarly given by m_N , Γ_N , and

$$p_N = p_A + p_B - p_1 = p_W + p_2. (5.37)$$

After squaring and summing over external spins, diagrams, and colors (N_C) , but not external boson polarizations (λ) , the polarization-dependent squared amplitude is

$$\sum |\mathcal{M}_{\lambda}|^{2} = \frac{4N_{C}^{2} g^{2} |V_{ji}^{CKM'}|^{2} |V_{\ell_{2}N}|^{2} \operatorname{Tr}\left[\not{p}_{A}\gamma^{\sigma} \not{p}_{B}\gamma^{\mu}\left(g_{R}^{q}{}^{2}P_{R}+g_{L}^{q}{}^{2}P_{L}\right)\right]}{2^{3}(1+\delta_{\ell_{1}\ell_{2}})\left[\left(\hat{s}-M_{W'}^{2}\right)^{2}+\left(\Gamma_{W'}M_{W'}\right)^{2}\right]\left[\left(p_{N}^{2}-m_{N}^{2}\right)^{2}+\left(\Gamma_{N}m_{N}\right)^{2}\right]} \times \operatorname{Tr}\left[\not{p}_{1}\gamma_{\sigma}\left(\not{p}_{N}+m_{N}\right)\not{q}_{\lambda}\not{p}_{2}\not{q}_{\lambda}^{*}P_{R}\left(\not{p}_{N}+m_{N}\right)\gamma_{\mu}\left(g_{R}^{\ell}{}^{2}P_{R}|Y_{\ell_{1}N}|^{2}+g_{L}^{\ell}{}^{2}P_{L}|V_{\ell_{1}N}|^{2}\right)\right] (5.38)$$

$$=\frac{2^{3}N_{C}^{2} g^{2} |V_{ji}^{CKM'}|^{2} |V_{\ell_{2}N}|^{2}}{\left(1+\delta_{\ell_{1}\ell_{2}}\right)\left[\left(\hat{s}-M_{W'}^{2}\right)^{2}+\left(\Gamma_{W'}M_{W'}\right)^{2}\right]\left[\left(p_{N}^{2}-m_{N}^{2}\right)^{2}+\left(\Gamma_{N}m_{N}\right)^{2}\right]} \times \left[|Y_{\ell_{1}N}|^{2} \left(g_{R}^{q}g_{R}^{\ell}\right)^{2}\mathcal{A}_{\lambda}+|Y_{\ell_{1}N}|^{2} \left(g_{L}^{q}g_{R}^{\ell}\right)^{2}\mathcal{B}_{\lambda}+|V_{\ell_{1}N}|^{2} \left(g_{L}^{q}g_{L}^{\ell}\right)^{2}\mathcal{C}_{\lambda}+|V_{\ell_{1}N}|^{2} \left(g_{R}^{q}g_{L}^{\ell}\right)^{2}\mathcal{D}_{\lambda}\right], (5.39)$$

where

$$\mathcal{A}_{\lambda} = 2(p_A \cdot p_1)(p_B \cdot p_N) \left[(p_N \cdot p_2) + 2(p_N \cdot \varepsilon_{\lambda})(\varepsilon_{\lambda} \cdot p_2) \right] - m_N^2(p_A \cdot p_1) \left[(p_B \cdot p_2) + 2(p_B \cdot \varepsilon_{\lambda})(\varepsilon_{\lambda} \cdot p_2) \right],$$
(5.40)

$$\mathcal{B}_{\lambda} = 2(p_B \cdot p_1)(p_A \cdot p_N) \left[(p_N \cdot p_2) + 2(p_N \cdot \varepsilon_{\lambda})(\varepsilon_{\lambda} \cdot p_2) \right]$$

$$- m_N^2(p_B \cdot p_1) \left[(p_A \cdot p_2) + 2(p_A \cdot \varepsilon_\lambda)(\varepsilon_\lambda \cdot p_2) \right], \qquad (5.41)$$

$$C_{\lambda} = m_N^2(p_A \cdot p_1) \left[(p_B \cdot p_2) + 2(p_B \cdot \varepsilon_{\lambda})(\varepsilon_{\lambda} \cdot p_2) \right], \qquad (5.42)$$

$$\mathcal{D}_{\lambda} = m_N^2 (p_B \cdot p_1) \left[(p_A \cdot p_2) + 2(p_A \cdot \varepsilon_{\lambda})(\varepsilon_{\lambda} \cdot p_2) \right], \qquad (5.43)$$

and ε_{λ} is taken to be real.

The Majorana neutrino's width, Γ_N , is expected to be very small. Therefore, to simplify analytic integration, we again apply the Narrow Width Approximation such that

$$\frac{1}{(p_N^2 - m_N^2)^2 + (\Gamma_N m_N)^2} \approx \frac{\pi}{\Gamma_N m_N} \,\delta\left(p_N^2 - m_N^2\right). \tag{5.44}$$

We are motivated to make this additional approximation to highlight and emphasize the analyzing power of the angular distributions. Our reported numerical results do not reflect this extra stipulation; see Eq. (5.72). Consequentially, the squared and summed amplitude becomes

$$\sum |\mathcal{M}_{\lambda}|^{2} \approx \frac{2^{3} \pi N_{C} g^{2} |V_{ji}^{CKM'}|^{2} |V_{\ell_{2}N}|^{2} \delta \left(p_{N}^{2} - m_{N}^{2}\right)}{\left(1 + \delta_{\ell_{1}\ell_{2}}\right) \left(\Gamma_{N}m_{N}\right) \left[\left(\hat{s} - M_{W'}^{2}\right)^{2} + \left(\Gamma_{W'}M_{W'}\right)^{2}\right]}$$
(5.45)

$$\times \left[|Y_{\ell_1 N}|^2 \left(g_R^q g_R^\ell \right)^2 \mathcal{A}_{\lambda} + |Y_{\ell_1 N}|^2 \left(g_L^q g_R^\ell \right)^2 \mathcal{B}_{\lambda} + |V_{\ell_1 N}|^2 \left(g_L^q g_L^\ell \right)^2 \mathcal{C}_{\lambda} + |V_{\ell_1 N}|^2 \left(g_R^q g_L^\ell \right)^2 \mathcal{D}_{\lambda} \right]$$

5.3.2 Phase Space Volume Element

We calculate the partonic-level cross section using the usual formula,

$$d\hat{\sigma} = \frac{1}{2\hat{s}} \frac{1}{4N_C^2} \sum |\mathcal{M}|^2 \cdot dPS_n.$$
(5.46)

Here, the factor of $4N_C^2$ comes from averaging over initial-state colors and spins. The factor dPS_n represents the *n*-body phase space volume element,

$$dPS_n(P; p_1 \dots p_n) = \prod_{k=1}^n \frac{d^3 p_k}{(2\pi)^3 2E_k} (2\pi)^4 \delta^4 \left(P - p_1 - \dots - p_n\right),$$
(5.47)

which can be decomposed using the recursion formula

$$dPS_n(P; p_1, \dots, p_n) = dPS_{n-1}(P; p_1, \dots, p_{n-1,n}) \times dPS_2(p_{n-1,n}; p_{n-1}, p_n) \times \frac{d\, p_{n-1,n}^2}{2\pi}, \quad (5.48)$$

where $P = \sum_{m=1}^{n} p_m$ and $p_{i,j} = p_i + p_j$. In the present case, dPS_3 is expressible as

$$dPS_3(p_A + p_B; p_1, p_2, p_W) = dPS_2(p_A + p_B; p_1, p_N) \times dPS_2(p_N; p_2, p_W) \times \frac{d p_N^2}{2\pi}.$$
 (5.49)

Since each dPS_k is individually Lorentz invariant, the two phase space elements in Eq. (5.49) can be evaluated in different reference frames. When $dPS_2(p_1, p_N)$ is evaluated in the partonic c.m. frame and $dPS_2(p_2, p_W)$ in the neutrino rest-frame, the full volume element is found to be

$$dPS_3(p_A + p_B; p_1, p_2, p_W) = d\Omega_N \frac{(1 - \tilde{\mu}_N^2)}{2(4\pi)^2} \times d\Omega_{\ell_2} \frac{(1 - \rho_W^2)}{2(4\pi)^2} \times \frac{d\,p_N^2}{2\pi},\tag{5.50}$$

with

$$\mu_N^2 = \frac{m_N^2}{\hat{s}}, \qquad \tilde{\mu}_N^2 = \frac{p_N^2}{\hat{s}}, \qquad \rho_W^2 = \frac{M_W}{p_N^2}, \tag{5.51}$$

and, in the on-shell limit,

$$\mu_N, \tilde{\mu}_N \to x_N = \frac{m_N}{M_{W'}}, \qquad \rho_W \to y_W = \frac{M_W}{m_N}.$$
(5.52)

The solid angle element $d\Omega_N$ is defined as the angle made by N with respect to the direction of propagation of the initial-state quark in the c.m. frame; $d\Omega_{\ell_2}$ is defined as the angle made by ℓ_2^+ with respect to the heavy neutrino spin axis in the neutrino's rest-frame.

5.3.3 Partonic-Level Angular Distributions

The angular distribution of the charged lepton from the neutrino decay is most efficiently determined by evaluating $\sum |\mathcal{M}|^2$ in the neutrino rest-frame. Like individual dPS_k volume elements, $|\mathcal{M}|^2$ is separately Lorentz invariant and thus can be evaluated in its own reference frame.

In order to evaluate Eq. (5.45) in the neutrino rest-frame, we must first rotate and boost the four-momenta of the initial-state quarks from the c.m. frame. Without the loss of generality, we assume that the initial-state (anti)quark is originally traveling in the positive (negative) \hat{z} -axis and that the $\ell_1^+ N$ pair propagate in $\hat{y} - \hat{z}$ plane. This allows us to rotate the entire $2 \rightarrow 2$ system such that the neutrino's momentum is aligned with the \hat{z} -axis, and then boost into the neutrino rest-frame. Since we are applying the NWA and immediately integrating over dp_N^2 , we will take N to be on-shell. After boosting, our four-momenta are:

$$p_{A} = \frac{\hat{s}}{4m_{N}} \left((1 - \cos\theta_{N}) + \mu_{N}^{2} (1 + \cos\theta_{N}), 0, -2\mu_{N}\sin\theta_{N}, \mu_{N}^{2} (1 + \cos\theta_{N}) - (1 - \cos\theta_{N}) \right),$$

$$p_{B} = \frac{\hat{s}}{4m_{N}} \left((1 + \cos\theta_{N}) + \mu_{N}^{2} (1 - \cos\theta_{N}), 0, 2\mu_{N}\sin\theta_{N}, \mu_{N}^{2} (1 - \cos\theta_{N}) - (1 + \cos\theta_{N}) \right),$$

$$p_N = (m_N, 0, 0, 0), \text{ and } p_1 = \frac{\hat{s}}{2m_N} (1 - \mu_N^2)(1, 0, 0, -1),$$
 (5.53)

where θ_N represents the polar angle between \vec{p}_N and \vec{p}_A in the c.m. frame. In the neutrino restframe, the $N \to \ell_2^+ W^-$ decay products' momenta are

$$p_{2} = |\vec{p}_{2}| \left(1, \sin \theta_{\ell_{2}} \cos \phi_{\ell_{2}}, \sin \theta_{\ell_{2}} \sin \phi_{\ell_{2}}, \cos \theta_{\ell_{2}}\right), \quad |\vec{p}_{2}| = |\vec{p}_{W}| = \frac{m_{N}}{2} (1 - y_{W}^{2}),$$

$$p_{W} = |\vec{p}_{2}| \left(\frac{E_{W}}{|\vec{p}_{2}|}, -\sin \theta_{\ell_{2}} \cos \phi_{\ell_{2}}, -\sin \theta_{\ell_{2}} \sin \phi_{\ell_{2}}, -\cos \theta_{\ell_{2}}\right), \quad E_{W} = \frac{m_{N}}{2} (1 + y_{W}^{2}), \quad (5.54)$$

where θ_{ℓ_2} and ϕ_{ℓ_2} are defined with respect to the neutrino spin axis in the c.m. frame. Explicitly, $\hat{z} = \hat{p}_N$, where $\hat{p}_N = \vec{p}_N / |\vec{p}_N|$ is measured in the c.m. frame, and ϕ_{ℓ_2} w.r.t. to the $+\hat{y}$ axis. This is consistent with Eq. (5.50). The polarization vectors for the SM W boson are subsequently:

$$\varepsilon_{0}^{\mu}(p_{W}) = \frac{E_{W}}{m_{W}} \left(\frac{|\vec{p}_{2}|}{E_{W}}, -\sin\theta_{\ell_{2}}\cos\phi_{\ell_{2}}, -\sin\theta_{\ell_{2}}\sin\phi_{\ell_{2}}, -\cos\theta_{\ell_{2}} \right), \\
\varepsilon_{T1}^{\mu}(p_{W}) = (0, -\cos\theta_{\ell_{2}}\cos\phi_{\ell_{2}}, -\cos\theta_{\ell_{2}}\sin\phi_{\ell_{2}}, \sin\theta_{\ell_{2}}), \\
\varepsilon_{T2}^{\mu}(p_{W}) = (0, \sin\phi_{\ell_{2}}, -\cos\phi_{\ell_{2}}, 0).$$
(5.55)

Here the labels 0, T1, and T2 denote the longitudinal and transverse polarizations of the outgoing vector boson. After combining Eqs. (5.45), (5.46), (5.50), and integrating over dp_N^2 , as well as $d\Omega_N$, for the *L*-violating process $u_i \overline{d}_j \rightarrow \ell_1^+ N \rightarrow \ell_1^+ \ell_2^+ W^-$ with a longitudinally polarized $W^$ boson the angular distribution is

$$\frac{d\hat{\sigma}_{0}}{d\Omega_{\ell_{2}}} = \frac{\hat{\sigma}(W_{0})}{2^{4}\pi} \times \left\{ 4 \left[1 + \left(\frac{2 - \mu_{N}^{2}}{2 + \mu_{N}^{2}} \right) \left(\frac{g_{R}^{\ell} |Y_{\ell_{1}N}|^{2} - g_{L}^{\ell} |V_{\ell_{1}N}|^{2}}{g_{R}^{\ell} |Y_{\ell_{1}N}|^{2} + g_{L}^{\ell} |V_{\ell_{1}N}|^{2}} \right) \cos \theta_{\ell_{2}} \right] - \frac{3\pi \ \mu_{N}}{\left(2 + \mu_{N}^{2}\right)} \left(\frac{g_{R}^{q} |2 - g_{L}^{q}|^{2}}{g_{R}^{q} |2 + g_{L}^{q}|^{2}} \right) \sin \theta_{\ell_{2}} \cos \phi_{\ell_{2}} \right\}.$$
(5.56)

Accordingly, for transversely polarized W bosons the angular distributions are

$$\frac{d\hat{\sigma}_{T1}}{d\Omega_{\ell_2}} = \frac{d\hat{\sigma}_{T2}}{d\Omega_{\ell_2}} = \frac{\hat{\sigma}(W_T)}{2^5\pi} \times \left\{ 4 \left[1 - \left(\frac{2 - \mu_N^2}{2 + \mu_N^2} \right) \left(\frac{g_R^{\ell_2} |Y_{\ell_1N}|^2 - g_L^{\ell_2} |V_{\ell_1N}|^2}{g_R^{\ell_2} |Y_{\ell_1N}|^2 + g_L^{\ell_2} |V_{\ell_1N}|^2} \right) \cos \theta_{\ell_2} \right] \\
+ \frac{3\pi \,\mu_N}{\left(2 + \mu_N^2\right)} \left(\frac{g_R^{q_2} - g_L^{q_2}}{g_R^{q_2} + g_L^{q_2}} \right) \sin \theta_{\ell_2} \cos \phi_{\ell_2} \right\}.$$
(5.57)

In the preceding lines, we have used the following quantities

$$\hat{\sigma}(W_{0}) \equiv \hat{\sigma}(u\bar{d} \to \ell_{1}^{+}N \to \ell_{1}^{+}\ell_{2}^{+}W_{0}^{-}) \\
= \frac{g^{2} |V_{ji}^{CKM'}|^{2} |V_{\ell_{2}N}|^{2}}{3N_{C} 2^{10} \pi^{2} (1 + \delta_{\ell_{1}\ell_{2}})} \left(g_{R}^{q}{}^{2} + g_{L}^{q}{}^{2}\right) \left(g_{R}^{\ell}{}^{2}|Y_{\ell_{1}N}|^{2} + g_{L}^{\ell}{}^{2}|V_{\ell_{1}N}|^{2}\right) \\
\times \frac{m_{N}}{\Gamma_{N}} \frac{\hat{s}}{\left[(\hat{s} - M_{W'}^{2})^{2} + (\Gamma_{W'}M_{W'})^{2}\right]} (1 - \mu_{N}^{2})^{2} (1 - g_{W}^{2})^{2} (2 + \mu_{N}^{2}) \left(\frac{1}{2g_{W}^{2}}\right) \quad (5.58) \\
\hat{\sigma}(W_{T}) \equiv \hat{\sigma}(u\bar{d} \to \ell_{1}^{+}N \to \ell_{1}^{+}\ell_{2}^{+}W_{T}^{-}) \\
= \hat{\sigma}(W_{0}) \times 2g_{W}^{2} \quad (5.59)$$

Integrating over the azimuthal angle, the polar distributions are calculated to be

$$\frac{d\hat{\sigma}_0}{d\cos\theta_{\ell_2}} = \frac{\hat{\sigma}(W_0)}{2} \left[1 + \left(\frac{2-\mu_N^2}{2+\mu_N^2}\right) \left(\frac{g_R^{\ell\,2}|Y_{\ell_1N}|^2 - g_L^{\ell\,2}|V_{\ell_1N}|^2}{g_R^{\ell\,2}|Y_{\ell_1N}|^2 + g_L^{\ell\,2}|V_{\ell_1N}|^2}\right) \cos\theta_{\ell_2} \right] \tag{5.60}$$

and

$$\frac{d\hat{\sigma}_T}{d\cos\theta_{\ell_2}} \equiv \frac{d(\hat{\sigma}_{T1} + \hat{\sigma}_{T1})}{d\cos\theta_{\ell_2}} = \frac{\hat{\sigma}(W_T)}{2} \left[1 - \left(\frac{2 - \mu_N^2}{2 + \mu_N^2}\right) \left(\frac{g_R^{\ell\,2} |Y_{\ell_1N}|^2 - g_L^{\ell\,2} |V_{\ell_1N}|^2}{g_R^{\ell\,2} |Y_{\ell_1N}|^2 + g_L^{\ell\,2} |V_{\ell_1N}|^2}\right) \cos\theta_{\ell_2} \right] \quad (5.61)$$

After combining the two, we find that the polarization-summed polar distribution for the full $u_i \overline{d}_j \rightarrow \ell_1^+ \ell_2^+ q \overline{q}'$ process is

$$\frac{d\hat{\sigma}_{Tot.}}{d\cos\theta_{\ell_2}} \equiv \frac{d(\hat{\sigma}_0 + \hat{\sigma}_T)}{d\cos\theta_{\ell_2}} \\
= \frac{\hat{\sigma}_{Tot.}}{2} \left[1 + \frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \left(\frac{2 - \mu_N^2}{2 + \mu_N^2} \right) \left(\frac{g_R^{\ell\,2} |Y_{\ell_1N}|^2 - g_L^{\ell\,2} |V_{\ell_1N}|^2}{g_R^{\ell\,2} |Y_{\ell_1N}|^2 + g_L^{\ell\,2} |V_{\ell_1N}|^2} \right) \cos\theta_{\ell_2} \right] (5.62)$$

where

$$\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} = \frac{\hat{\sigma}(W_0) - 2y_W^2 \hat{\sigma}(W_0)}{\hat{\sigma}(W_0) + 2y_W^2 \hat{\sigma}(W_0)} = \frac{1 - 2y_W^2}{1 + 2y_W^2},\tag{5.63}$$

and the total partonic-level cross section is

$$\hat{\sigma}_{Tot.} \equiv \hat{\sigma}(u_i \overline{d}_j \to \ell_1^+ \ell_2^+ q \overline{q}')$$
(5.64)

$$= (\hat{\sigma}(W_0) + \hat{\sigma}(W_T)) \times BR(W \to q\bar{q}')$$
(5.65)

$$= \hat{\sigma}(W_0)(1+2y_W^2) \times BR(W \to q\bar{q}')$$

$$g^2 |V_{ji}^{CKM'}|^2 |V_{\ell_2N}|^2 (g^2 + g^2) (g^2 + g^2)$$

$$= \frac{1}{3N_C} \frac{1}{2^{10} \pi^2} \frac{1}{(1+\delta_{\ell_1\ell_2})} \left(g_R^{q^2} + g_L^{q^2}\right) \left(g_R^{\ell_2} |Y_{\ell_1N}|^2 + g_L^{\ell_2} |V_{\ell_1N}|^2\right) \left(\frac{M_N}{\Gamma_N}\right) \\ \times \frac{\hat{s} (1-y_W^2)^2 (1-y_W^2)^2 (2+\mu_N^2)}{\left[(\hat{s}-M_{W'}^2)^2 + (\Gamma_{W'}M_{W'})^2\right]} \left(\frac{1+2y_W^2}{2y_W^2}\right) \times BR(W \to q\bar{q}').$$
(5.67)

Having instead chosen to integrate first over the polar angle before the azimuthal angle, the polarization-dependent azimuthal distributions for the $u_i \overline{d}_j \rightarrow \ell_1^+ N \rightarrow \ell_1^+ \ell_2^+ W^-$ process are

$$\frac{d\hat{\sigma}_0}{d\phi_{\ell_2}} = \frac{\hat{\sigma}(W_0)}{2\pi} \left[1 - \frac{3\pi^2}{16} \frac{\mu_N}{\left(2 + \mu_N^2\right)} \left(\frac{g_R^{q\,2} - g_L^{q\,2}}{g_R^{q\,2} + g_L^{q\,2}} \right) \cos\phi_{\ell_2} \right],\tag{5.68}$$

and

$$\frac{d\hat{\sigma}_T}{d\phi_{\ell_2}} \equiv \frac{(d\hat{\sigma}_{T_1} + \hat{\sigma}_{T_2})}{d\phi_{\ell_2}} = \frac{\hat{\sigma}(W_T)}{2\pi} \left[1 + \frac{3\pi^2}{16} \frac{\mu_N}{(2+\mu_N^2)} \left(\frac{g_R^{q\,2} - g_L^{q\,2}}{g_R^{q\,2} + g_L^{q\,2}} \right) \cos\phi_{\ell_2} \right]. \tag{5.69}$$

Similarly, after combining the azimuthal distributions, the total polarization-summed azimuthal distribution for the full $u_i \overline{d}_j \rightarrow \ell_1^+ \ell_2^+ q \overline{q}'$ process is

$$\frac{d\hat{\sigma}_{Tot.}}{d\phi_{\ell_2}} = \frac{\hat{\sigma}_{Tot.}}{2\pi} \left[1 - \frac{3\pi^2}{16} \frac{\mu_N}{\left(2 + \mu_N^2\right)} \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{g_R^{q\,2} - g_L^{q\,2}}{g_R^{q\,2} + g_L^{q\,2}} \right) \cos\phi_{\ell_2} \right]. \tag{5.70}$$

Under the definition of the azimuthal angle, Φ , in Eq. (5.119), we have $\Phi = -\phi_{\ell_2}$, and consequentially recover Eq. (5.120):

$$\frac{d\hat{\sigma}_{Tot.}}{d\Phi} = \frac{\hat{\sigma}_{Tot.}}{2\pi} \left[1 - \frac{3\pi^2}{16} \frac{\mu_N}{\left(2 + \mu_N^2\right)} \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{g_R^{q\,2} - g_L^{q\,2}}{g_R^{q\,2} + g_L^{q\,2}} \right) \cos \Phi \right]. \tag{5.71}$$

Lastly, were the NWA never applied to N, the differential cross section for the $u_i \overline{d}_j \rightarrow \ell_1^+ \ell_2^+ W^$ process is

$$\frac{d\hat{\sigma}}{dp_N^2} = \frac{g^2 |V_{ji}^{CKM'}|^2 |V_{\ell_2N}|^2}{3N_C 2^{11} \pi^3 M_W^2 (1+\delta_{\ell_1\ell_2})} \left(g_R^{q^2} + g_L^{q^2}\right) \left(p_N^2 g_R^{\ell^2} |Y_{\ell_1N}|^2 + m_N^2 |V_{\ell_1N}|^2 g_L^{\ell^2}\right) \\
\times \frac{\hat{s} (1-\tilde{\mu}_N^2)^2 (2+\tilde{\mu}_N^2)}{\left[(\hat{s}-M_{W'}^2)^2 + (\Gamma_{W'}M_{W'})^2\right]} \frac{p_N^2 (1-\rho_W^2)^2 (1+2\rho_W^2)}{\left[(p_N^2-m_N^2)^2 + (\Gamma_N m_N)^2\right]},$$
(5.72)

where $\tilde{\mu}_N^2 \equiv p_N^2/\hat{s}$ and $\rho_W^2 \equiv M_W^2/p_N^2$.

5.3.4 Partonic-Level Angular Distributions: L-Conserving Case

For comparison, we consider the case where the heavy neutrino decays through the following Lconserving process:

$$u_i \overline{d}_j \to W' \to \ell_1^+ N \to \ell_1^+ \ell_2^- W^+.$$
(5.73)

Following the identical arguments specified in the preceding section, the subsequent polarizationdependent angular distributions are

$$\frac{d\hat{\sigma}_{0}}{d\Omega_{\ell_{2}}} = \frac{\hat{\sigma}(W_{0})}{2^{4}\pi} \times \left\{ 4 \left[1 - \left(\frac{2 - \mu_{N}^{2}}{2 + \mu_{N}^{2}} \right) \left(\frac{g_{R}^{\ell_{2}} |Y_{\ell_{1}N}|^{2} - g_{L}^{\ell_{2}} |V_{\ell_{1}N}|^{2}}{g_{R}^{\ell_{2}} |Y_{\ell_{1}N}|^{2} + g_{L}^{\ell_{2}} |V_{\ell_{1}N}|^{2}} \right) \cos \theta_{\ell_{2}} \right] \\
+ \frac{3\pi\mu_{N}}{\left(2 + \mu_{N}^{2}\right)} \left(\frac{g_{R}^{q^{2}} - g_{L}^{q^{2}}}{g_{R}^{q^{2}} + g_{L}^{q^{2}}} \right) \sin \theta_{\ell_{2}} \cos \phi_{\ell_{2}} \right\},$$
(5.74)

and

$$\frac{d\hat{\sigma}_{T1}}{d\Omega_{\ell_2}} = \frac{d\hat{\sigma}_{T2}}{d\Omega_{\ell_2}} = \frac{\hat{\sigma}(W_T)}{2^4\pi} \times \left\{ 4 \left[1 + \left(\frac{2 - \mu_N^2}{2 + \mu_N^2} \right) \left(\frac{g_R^{\ell_2} |Y_{\ell_1N}|^2 - g_L^{\ell_2} |V_{\ell_1N}|^2}{g_R^{\ell_2} |Y_{\ell_1N}|^2 + g_L^{\ell_2} |V_{\ell_1N}|^2} \right) \cos \theta_{\ell_2} \right] - \frac{3\pi\mu_N}{\left(2 + \mu_N^2\right)} \left(\frac{g_R^{q_2} - g_L^{q_2}}{g_R^{q_2} + g_L^{q_2}} \right) \sin \theta_{\ell_2} \cos \phi_{\ell_2} \right\}.$$
(5.75)

The polarization-summed distributions for the polar and azimuthal cases are therefore

$$\frac{d\hat{\sigma}_{Tot.}}{d\cos\theta_{\ell_2}} = \frac{\hat{\sigma}_{Tot.}}{2} \left[1 - \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{2 - \mu_N^2}{2 + \mu_N^2} \right) \left(\frac{g_R^{\ell_2} |Y_{\ell_1N}|^2 - g_L^{\ell_2} |V_{\ell_1N}|^2}{g_R^{\ell_2} |Y_{\ell_1N}|^2 + g_L^{\ell_2} |V_{\ell_1N}|^2} \right) \cos\theta_{\ell_2} \right], \quad (5.76)$$

and

$$\frac{d\hat{\sigma}_{Tot.}}{d\Phi} = \frac{\hat{\sigma}_{Tot.}}{2\pi} \left[1 + \frac{3\pi^2}{16} \frac{\mu_N}{\left(2 + \mu_N^2\right)} \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{g_R^{q\,2} - g_L^{q\,2}}{g_R^{q\,2} + g_L^{q\,2}} \right) \cos \Phi \right],\tag{5.77}$$

respectively, where $\sigma_{Tot.}$ is still given by Eq.(5.67). Comparison to Eqs. (5.62) and (5.71) demonstrates that the slopes of the angular distributions differ in sign for the *L*-violating and *L*-conserving cases. Consequentially, adding the *L*-conserving and *L*-violating distributions together results in the quantitative feature

$$\hat{\sigma}_{Tot.} = \frac{d\hat{\sigma}_{Tot.}^L}{d\cos\theta_{\ell_2}} + \frac{d\hat{\sigma}_{Tot.}^{L}}{d\cos\theta_{\ell_2}} = \pi \left[\frac{d\hat{\sigma}_{Tot.}^L}{d\Phi} + \frac{d\hat{\sigma}_{Tot.}^{L}}{d\Phi} \right],$$
(5.78)

where $L(\not\!\!L)$ denotes the lepton number-conserving (violating) angular distributions.

5.4 W' AND N PRODUCTION AND DECAY AT THE LHC

For the remainder of this analysis, we consider for our various benchmark calculations only the pure gauge states W'_R and W'_L , respectively given by Eq. (5.15) and Eq. (5.16), and with SM coupling strength

$$g_{R,L}^{q,\ell} = g.$$
 (5.79)

More general results can be obtained by simple scaling. Unless explicitly stated otherwise, we take

$$M_{W'} = 3 \text{ TeV}, \quad m_N = 500 \text{ GeV}, \quad |V_{ud}^{CKM'}|^2 = 1,$$
 (5.80)

and use the CTEQ6L1 parton distribution functions (pdfs) [63] for all hadronic-level cross section calculations. Explicitly, we consider only the $ud \to W'$ production mode.

Regarding our choice of neutrino mixing parameters, for mixing between L.H. gauge states and light mass eigenstates, we use the Pontecorvo-Maki-Nakagawa-Sakata (PMNS) matrix with mixing angles taken from Ref. [136], which includes recent measurements of θ_{13} , and take δ_{CP} , α_1 , $\alpha_2 = 0$. The bounds from $0\nu\beta\beta$ decay are quite severe and discourage collider searches for L-violation in the electronic channel. However, neutrino mixing between the mu- or tau-flavor state and lightest heavy mass eigenstate can still be considerably larger in L.H. interactions. Therefore, we use

$$|V_{eN}|^2 = 2.5 \times 10^{-5}, \quad |V_{\mu N}|^2 = 1 \times 10^{-3}, \quad \text{and} \quad |V_{\tau N}|^2 = 1 \times 10^{-3}.$$
 (5.81)

These numerical values are in line with Eqs. (5.24), (5.25), and (5.80); and furthermore, mimic the observed $\mu - \tau$ symmetry seen in mixing between flavor states and light mass eigenstates. Where necessary, for mixing between R.H gauge states and light mass eigenstates, we apply the unitarity condition

$$\sum_{m=1}^{3} |X_{\ell m}|^2 = 1 - \sum_{m=1}^{3} |U_{\ell m}|^2, \quad \text{for} \quad \ell = e, \mu, \tau.$$
(5.82)

For mixing between R.H. gauge states and the lightest, heavy mass eigenstate, we apply Eq. (5.14) and take

$$|Y_{\ell N}|^2 = 1, \quad \text{for} \quad \ell = e, \mu, \tau.$$
 (5.83)

5.4.1 W' Production and Decay

Under our parameterization, the partial widths for W' decaying into a pairs of quarks are

$$\Gamma(W' \to \bar{q}q') = 3|V_{qq'}^{\text{CKM}'}|^2 (g_L^{q^2} + g_R^{q^2}) \frac{M_{W'}}{48\pi},$$

$$\Gamma(W' \to tb) = 3|V_{tb}^{\text{CKM}'}|^2 (g_L^{q^2} + g_R^{q^2}) \frac{M_{W'}}{48\pi} \left(1 - x_t^2\right)^2 \left(1 + \frac{1}{2}x_t^2\right),$$
(5.84)

where $x_i = m_i/M_{W'}$, and the factors of three represent color multiplicity. Likewise, the partial widths of the W' decaying to leptons are

$$\Gamma(W' \to \ell \nu_m) = \left(g_R^{\ell} |X_{\ell m}|^2 + g_L^{\ell} |U_{\ell m}|^2 \right) \frac{M_{W'}}{48\pi},$$
(5.85)

$$\Gamma(W' \to \ell N) = \left(g_R^{\ell} |Y_{\ell N}|^2 + g_L^{\ell} |V_{\ell N}|^2\right) \frac{M_{W'}}{48\pi} \left(1 - x_N^2\right)^2 \left(1 + \frac{1}{2}x_N^2\right).$$
(5.86)

Summing over the partial widths, the full widths are found to be

$$\Gamma_{W'_{R}} = \frac{M_{W'}}{32\pi} \left[4 + (1 - x_{t}^{2})^{2}(2 + x_{t}^{2}) + (1 - x_{N}^{2})^{2}(2 + x_{N}^{2}) \frac{1}{3} \sum_{\ell=e}^{\tau} |Y_{\ell N}|^{2} + \frac{2}{3} \sum_{m=1,\ell=e}^{3,\tau} |X_{\ell m}|^{2} \frac{1}{3} \frac{1}{2} \sum_{m=1,\ell=e}^{T} |Y_{\ell N}|^{2} + \frac{2}{3} \sum_{m=1,\ell=e}^{3,\tau} |X_{\ell m}|^{2} \frac{1}{3} \frac{1}{2} \sum_{m=1,\ell=e}^{T} |Y_{\ell N}|^{2} + \frac{2}{3} \sum_{m=1,\ell=e}^{3,\tau} |U_{\ell m}|^{2} \frac{1}{3} \frac{1}{2} \sum_{m=1,\ell=e}^{T} |W_{\ell N}|^{2} + \frac{2}{3} \sum_{m=1,\ell=e}^{3,\tau} |U_{\ell m}|^{2} \frac{1}{3} \frac{1}{2} \sum_{m=1,\ell=e}^{T} |W_{\ell N}|^{2} + \frac{2}{3} \sum_{m=1,\ell=e}^{3,\tau} |U_{\ell m}|^{2} \frac{1}{3} \frac{1}{2} \sum_{m=1,\ell=e}^{T} |W_{\ell N}|^{2} + \frac{2}{3} \sum_{m=1,\ell=e}^{3,\tau} |W_{\ell m}|^{2} \frac{1}{3} \frac{1}{2} \sum_{m=1,\ell=e}^{T} |W_{\ell m}|^{2} \frac{1}{3} \frac{1}{2} \sum_{m=1,\ell=e}^{T} |W_{\ell m}|^{2} \frac{1}{3} \sum_{m=1,\ell=e}^{T} |W_{\ell m}|^{2}$$



Figure 33: (a) The total decay width for W'_R (solid) and W'_L (dash); (b) the branching ratio of $W'_{R,L} \to N\ell^+$, with subsequent $W'_R \to N\mu^+$ (dot) and $W'_L \to N\mu^+$ (dash-dot) ratios; and the production cross sections at the (c) 8 and (d) 14 TeV LHC of W'_R (solid), W'_L (dash), $W'_R \to N\ell^+$ (dot), and $W'_L \to N\ell^+$ (dash-dot).

As a function of $M_{W'}$, Fig. 33 shows (a) the total W' decay width; (b) the branding ratio (BR) of $W' \to N\ell$, for $\ell = e, \mu, \tau$, defined as the ratio of the partial width to the total W' width, Γ'_W :

$$BR(W' \to \ell N) = \frac{\Gamma(W' \to \ell N)}{\Gamma_{W'}};$$
(5.89)

and the production cross sections for the pure gauge eigenstates $W'_{R,L}$, along with $pp \to W'_{R,L} \to N\ell^+$ in (c) 8 TeV and (d) 14 TeV pp collisions.

The production cross section of the W' and its subsequent decay to N is calculated in the


Figure 34: As a function of heavy neutrino mass, (a) the total N width and the $N \to \ell^+ W_{\lambda}^$ partial widths, and (b) the combined $N \to \ell^+ W^-$ and individual $N \to \ell^+ W_{\lambda}^-$ branching ratios for longitudinal ($\lambda = 0$) and transverse ($\lambda = T$) W polarizations.

usual fashion [252]. The treatment of our full $2 \to 4$ process, on the other hand, is addressed in Section 5.3. Since the *u*-quark is more prevalent in the proton than the *d*-quark, and since the dominate subprocess of W'^+ (W'^-) production at the LHC is $u\bar{d} \to W'^+$ ($d\bar{u} \to W'^-$), the production cross section of W'^+ is greater than the W'^- cross section. In a similar vein, the mixing between L.H. interaction states and heavy neutrino mass eigenstates is suppressed by $|V_{\ell N}|^2 \sim \mathcal{O}(10^{-3})$, whereas the mixing between R.H. interaction states and heavy neutrino mass eigenstates is proportional to $|Y_{\ell N}|^2 \sim \mathcal{O}(1)$. Consequently, the $W'_L \to N\ell$ branching ratio, and hence the $pp \to W'_L \to N\ell$ cross section, is roughly three orders of magnitude smaller than the W'_R rates.

5.4.2 Heavy Neutrino Decay

A heavy neutrino with mass of a few hundred GeV or more can decay through on-shell SM gauge and Higgs bosons. The partial widths of the lightest heavy neutrino are

$$\Gamma(N \to \ell^{\pm} W_0^{\mp}) \equiv \Gamma_0 = \frac{g^2}{64\pi M_W^2} |V_{\ell N}|^2 m_N^3 (1 - y_W^2)^2
\Gamma(N \to \ell^{\pm} W_T^{\mp}) \equiv \Gamma_T = \frac{g^2}{32\pi} |V_{\ell N}|^2 m_N \left(1 - y_W^2\right)^2
\Gamma(N \to \nu_\ell Z) \equiv \Gamma_Z = \frac{g^2}{64\pi M_W^2} |V_{\ell N}|^2 m_N^3 (1 - y_Z^2)^2 \left(1 + 2y_Z^2\right)
\Gamma(N \to \nu_\ell H) \equiv \Gamma_H = \frac{g^2}{64\pi M_W^2} |V_{\ell N}|^2 m_N^3 (1 - y_H^2)^2$$
(5.90)

where $W_{0,T}$ are longitudinally and transversely polarized W's, respectively, and $y_i = M_i/m_N$. The decays of the heavy neutrino through a W' are not kinematically accessible. The total width is

$$\Gamma_{\rm N} = \sum_{\ell=e}^{\tau} \left(2(\Gamma_0 + \Gamma_T) + \Gamma_Z + \Gamma_H \right) \tag{5.91}$$

where the factor of two in front of $\Gamma_{0,T}$ is from the sum over positively and negatively charged leptons.

Figure 34(a) shows the total decay width (solid) and the partial decay widths to positively charged lepton (dashed) normalized to the sum over the mixing matrices. For this plot the mass of the SM Higgs boson is set to 125 GeV. The normalized width grows dramatically with mass due to decays into longitudinally polarized W's and Z's and the Higgs boson. Although the width appears to be large at high neutrino mass, for mixing angles on the order of a percent or less the width is still narrow.

Also of interest is the branching ratio (BR) of heavy neutrinos into charged leptons:

$$BR\left(N \to \ell^{\pm} W^{\mp}\right) = \frac{\sum_{\ell=e}^{\tau} \left(\Gamma_0 + \Gamma_T\right)}{\Gamma_{Tot}}$$
(5.92)

Figure 34(b) shows the total BR of the heavy neutrino into positively charged leptons (solid) and individually the BR into longitudinally (dashed) and transversely (dotted) polarize W's as a function of neutrino mass. The BR's into negatively charged leptons are the same. As the mass of the neutrino increases the Z and Higgs decay channels open, hence the branching ratio into charged leptons decreases. Since Γ_0 grows more quickly with neutrino mass than Γ_T , for $m_N \gg M_W$ the total BR converges to the BR into longitudinally polarized W's. Also, at high neutrino masses

$$\Gamma_0 \approx \Gamma_H \approx \Gamma_Z \tag{5.93}$$



Figure 35: The partonic level process for a heavy W'^+ production and decay to like sign leptons in hadronic collisions.

Hence the total width approaches $4\Gamma_0$ and, from Eq. (5.92), the branching ratio into a positively charged leptons is approximately 0.25. This is a manifestation of the Goldstone Equivalence Theorem when taking m_N and $V_{\ell N}$ as independent parameters.

5.5 LIKE-SIGN DILEPTON SIGNATURE

A distinctive feature of Majorana neutrinos is that they facilitate L-violating processes, and to study this behavior at the LHC we consider the L-violating cascade

$$u(p_A) \ \bar{d}(p_B) \to W_{R,L}^{'+}(q) \to \ell_1^+(p_1) \ N(p_N) \to \ell_1^+(p_1) \ \ell_2^+(p_2) \ q(p_3) \ \bar{q}'(p_4).$$
(5.94)

The two diagrams that contribute to this process are shown in Fig. 35. Figure 36 shows the total production cross section for the like-sign dimuon process as a function of m_N . In it, the solid line denotes the pure W'_R gauge state while the dashed line represents the pure W'_L state. Since the $W'_R \to N\mu$ branching ratio is larger than $W'_L \to N\mu$ ratio, the cross section for W'_R is systematically larger than for W'_L . Additionally, as the neutrino mass approaches the W' mass the cross section drops precipitously due to phase space suppression.

In principle, the conjugate process, $\bar{u}d \to W'^-$, should also be possible at the LHC. However, it will possess a much smaller production rate because the $\bar{u}d$ initial-state has a smaller parton



Figure 36: Total cross section of $pp \to W'^+ \to \mu^+ \mu^+ W^-$ times $W^- \to q\bar{q}'$ branching ratio versus heavy neutrino mass at (a) 8 and (b) 14 TeV. Solid (dashed) line corresponds to $W'_R(W'_L)$ gauge state.

luminosity than $u\bar{d}$. Despite this, all reconstruction methods and observables discussed below are applicable to both processes.

5.5.1 Event Selection

For simplicity, we restrict our study to like-sign muons. There is no change in the analysis if extended to electrons; however, $\not\!\!\!E_T$ requirements must be reassessed for inclusion of unstable τ 's [170, 172]. Consequently, our signal consists strictly of two positively charged leptons and two jets, a fact that allows for considerable background suppression. In simulating this like-sign leptons plus dijet signal, to make our analysis more realistic, we smear the lepton and jet energies to emulate real detector resolution effects. These effects are assumed to be Gaussian and parameterized by

$$\frac{\sigma(E)}{E} = \frac{a}{\sqrt{E}} \oplus b, \tag{5.95}$$

where $\sigma(E)/E$ is the energy resolution, a is a sampling term, b is a constant term, \oplus represents addition in quadrature, and all energies are measured in GeV. For leptons we take a = 5% and b = 0.55%, and for jets we take a = 100% and b = 5% [253].

After smearing, we define our candidate event as two positively charged leptons and two jets

$\sigma({ m fb})$	8 TeV		14 TeV	
	W'_L	W'_R	W'_L	W'_R
Reco. without Cuts or Smearing	4.6×10^{-5}	0.046	$9.3 imes 10^{-4}$	0.95
+ Smearing + Fiducial + Kinematics (Eq. (5.96))	4.0×10^{-5}	0.035	8.2×10^{-4}	0.71
+ Isolation (Eq. (5.98))	2.1×10^{-5}	0.027	$3.2 imes 10^{-4}$	0.50
$+\not\!\!\!E_T + m_{jj}$ Requirements (Eq. (5.104))	$1.7 imes 10^{-5}$	0.023	$2.6 imes 10^{-4}$	0.42
+ Mass Req. (Eq. (5.105))	7.2×10^{-6}	0.012	2.0×10^{-4}	0.35
$\sigma(\text{All Cuts})/\sigma(\text{Smearing} + \text{Fid.} + \text{Kin.})$	18%	35%	25%	49%

Table 26: Cross section for $pp \to W'_{L,R}^+ \to \mu^+ \mu^+ q \overline{q}'$ after consecutive cuts for 8 and 14 TeV LHC.

passing the following basic kinematic and fiducial cuts on the transverse momentum, p_T , and pseudorapidity, η :

$$p_T^j \ge 30 \text{ GeV}, \ p_T^\ell \ge \ 20 \text{ GeV}, \ \eta_j \le 3.0, \ \eta_\ell \le 2.5.$$
 (5.96)

Table 26 lists the cross sections for Eq. (5.94) assuming the pure $W'_{R,L}$ gauge states at the 8 and 14 TeV LHC without smearing or acceptance cuts (row 1), and with smearing plus acceptance cuts from Eq. (5.96) (row 2). Here and henceforth, we assume a 100% efficiency for lepton and jet identification.

The goal of this analysis is to unambiguously determine the properties of W' and N. To do so, our candidate leptons and jets must be well-defined and well-separated, that latter of which is measured by

$$\Delta R_{ij} = \sqrt{(\Delta \phi_{ij})^2 + (\Delta \eta_{ij})^2},\tag{5.97}$$

where $\Delta \phi_{ij}$ and $\Delta \eta_{ij}$ are the difference in the azimuthal angles and rapidities, respectively, of particles *i* and *j*. Subsequently, we apply isolation cuts on our candidate objects:

$$\Delta R_{\ell j}^{\min} \ge 0.4, \quad \Delta R_{jj} \ge 0.3 \tag{5.98}$$

for all lepton and jet combinations, where $\Delta R_{\ell j}^{\min}$ is defined as

$$\Delta R_{\ell j}^{\min} = \min_{i=W',N} \Delta R_{\ell_i j}^{\min}.$$
(5.99)



Figure 37: Invariant mass distribution of $m_{\ell_1 j j}$ and $m_{\ell_2 j j}$, where ℓ_i for i = 1, 2 originates from either the W' or N. Cuts from Eqs. (5.96) and (5.98) as well as the energy smearing are applied.

In Eq. (5.99), the subscript i = W', N on ℓ_i denotes the identified parent particle of ℓ_i . The effects of the isolation cuts applied at both the 8 and 14 TeV LHC are shown in the third row of Table 26. To understand the origin of these precise numbers and parent-particle identification, we digress to succinctly connect properties of our chiral Lagrangian to the final-state kinematical distributions.

5.5.2 Characteristics of Kinematical Distributions

Our signal suffers from a very evident ambiguity: either lepton can originate from the neutrino decay. The origin of each lepton must thus be determined in order to fully reconstruct an event. As noted in section 5.4, the width of N is narrow. Consequently, there is a very small probability for the phase space of each diagram in Fig. 35 to overlap, meaning that the interference of the two diagrams is negligible. In fact, in the W'_R case, the interference is exactly zero because the charged lepton from the N decay is left-handed while the charged lepton from the W'_R is righthanded. Furthermore, since the two diagrams add incoherently, it is reasonable to expect that only one diagram contributes at a time. Intuitively, this means that only one of the two following momentum combinations will closely reconstruct the heavy neutrino mass:

$$m_{1jj}^2 = (p_1 + p_3 + p_4)^2$$
 or $m_{2jj}^2 = (p_2 + p_3 + p_4)^2$, (5.100)

where p_3 and p_4 are the momenta of our final-state jets.



Figure 38: Transverse momentum distributions for (a,b) the lepton identified as originating from the W' (dashed) and neutrino (solid), and (c,d) the hardest (dashed) and softest (solid) jets in $pp \to W' \to \ell^+ \ell^+ j j$ production. The W'_L case is represented in (a,c) and the W'_R case in (b,d). The energy smearing has been applied.

After calculating both permutations of m_N (Fig. 37), the appearance of the N mass peak is stark. Using the central value of the mass peak, $m_N^{Reco.}$, we identify the charged lepton from the N decay as the charged lepton from our candidate event that most closely recovers $m_N^{Reco.}$, i.e.,

$$\Delta m_{min} = \min_{i=1,2} |m_{ijj} - m_N^{Reco.}|, \qquad (5.101)$$

where m_{ijj} for i = 1, 2 is defined by Eq. (5.100).

Independent of reconstructing N, the charged lepton associated with the W' decay can be

identified by analyzing the transverse momentum, p_T , distributions of our final-state objects. In Fig. 38, the p_T distributions of the charged leptons (a,b) and jets (c,d) for the W'_L (a,c) and W'_R (b,d) gauge states. As expected, the lepton identified as originating from the W' has a Jacobian peak around $M_{W'}/2$ for both the W'_L and W'_R cases. To understand the other distributions, we consider spin correlations.

Figure 39 shows the spin correlations of the process in Eq. (5.94) with the single arrowed lines representing momentum direction and double arrowed lines spin. The direction \hat{z} is defined as the direction of motion of the neutrino in the W' rest-frame. Each column indicates the spin and momentum of the particles in their parents' rest-frame with the first column in the neutrino rest-frame. Note that for the W'_R (W'_L) the heavy neutrino is in a mostly right-(left-) handed helicity state. Hence, for the W'_R (W'_L) the neutrino spin points with (against) the \hat{z} direction. The decays of the neutrino through longitudinal W are shown in Fig. 39(a) and 39(b) for W'_L and W'_R , respectively, and the decays through a transversely polarized W are shown in Fig. 39(c) for W'_L and Fig. 39(d) for W'_R .

As shown in Fig. 34, 500 GeV neutrino preferentially decays into longitudinally polarized W's. We therefore focus on that case. For the W'_R , the lepton from the heavy neutrino decay moves preferentially along the \hat{z} direction. Hence, the boost into the partonic c.m. frame will be along the charged lepton's momentum. In the W'_L case, the charged lepton moves in negative \hat{z} direction and the boost into the partonic c.m. frame is against the lepton's momentum. Therefore, the lepton from the heavy neutrino decay is harder in the W'_R case than in the W'_L case. The contribution from decay into transversely polarized W's is in the opposite direction. However, as noted previously, this contribution is smaller than the decays into longitudinally polarized W's. Similar arguments can be made to explain that the two jets are softer in the W'_R case than in the W'_L case.

As previously stated, identifying well-separated objects in our event is paramount to measuring our observables. For 14 TeV LHC collisions, Fig. 40 shows (a) the separation between the two jets, ΔR_{jj} , and (b) the minimum separation between the leptons identified as originating from the heavy neutrino and W' and the two jets defined by

$$\Delta R_{\ell_i j}^{\min} = \min_{k=1,2} \Delta R_{\ell_i j_k}, \tag{5.102}$$

where i = W' for the lepton coming from the W' and i = N for the lepton coming from the neutrino decay. The solid lines are for W'_R and the dashed lines for W'_L . The ΔR_{jj} distributions peak at low values for both the left- and right-handed cases. This is due to the W from the heavy



Figure 39: Helicity and spin correlations in the chains $N_{L,R} \to \ell^+ W^- \to \ell^+ q \bar{q}'$ from W'_L decay in (a), (c); and from W'_R decay in (b), (d). Figures (a) and (b) are for longitudinally polarized SM W's, and Figs. (c) and (d) are for transversely polarized SM W's. The decay goes from left to right as labeled by the particle names. The momenta (single arrow lines) and spins (double arrow lines) are in the parent rest-frame in the direction of the heavy neutrino's motion (\hat{z}) in the W'rest-frame.

neutrino decay being highly boosted and its decay products therefore collimated. Also, as can be seen from Fig. 38, in the W'_R case the lepton from the neutrino decay is harder and hence the SM W softer than in the W'_L case. Since the SM W is less boosted in the right-handed case, the jets are less collimated and the ΔR_{jj} distribution has a longer tail for W'_R than for W'_L . Also, since the neutrino is highly boosted, its decay products are expected to land opposite in the transverse plane from the lepton from W' decay. Hence, $\Delta R_{\ell_W j}^{\min}$ peaks near π for both the the left-handed and right-handed case. Finally, $\Delta R_{\ell_N j}^{\min}$ is peaked near $2m_N/E_N \approx 0.7$ for both the W'_L and W'_R



Figure 40: (a) ΔR_{jj} distribution and (b) $\Delta R_{\ell j}^{\min}$ distributions for both the lepton identified as originating from N and W'. The solid lines are for the W'_R case and dashed lines W'_L . Energy smearing has been applied.

cases. The ΔR distributions at the 8 TeV LHC are peaked at similar values, but are more narrow than the 14 TeV distributions. Based on these arguments, we define the isolation cuts given by Eq. (5.98).

The isolation cuts more severely affect the W'_L cross section since the ΔR_{jj} distribution is strongly peaked at low values for W'_L . As the mass of the W' increases, the SM W from the heavy neutrino decay will become more boosted. Hence, the two jets will become more collimated and the effects of the isolation cuts will be even more significant. Since we will only be interested in the angular distributions of the lepton, it is possible to relax the ΔR_{jj} cut and look for one or two jets with two like sign leptons. Also, the separation between the lepton and jets from the heavy neutrino decay depend on the ratio of $m_N/M_{W'}$. As $m_N/M_{W'}$ increases (decreases) the lepton and jets become more (less) well separated.

5.5.3 Background Reduction and Statistical Significance

The SM background for our $\ell^+\ell^+jj$ signature has been thoroughly studied for the 14 TeV LHC by Ref. [168]. The largest background to our process was found to be from $t\bar{t}$ events with the cascade



Figure 41: Missing energy distribution for $pp \to W'_{L,R}^+ \to \mu^+ \mu^+ q \overline{q}'$ at the LHC. Energy smearing has been applied.

decays,

$$t \to W^+ b \to \ell^+ \nu_m b, \quad \bar{t} \to W^- \bar{b} \to W^- \bar{c} \nu_m \ell^+,$$
(5.103)

and was also found to be greatly suppressed by the lepton isolation cuts in Eq. (5.98). The background can be further suppressed by noting that leptonic $t\bar{t}$ events contain a final state light neutrino and therefore a considerable amount of missing transverse energy, $\not\!\!\!E_T$. This is in direct comparison with our signal where all the $\not\!\!\!E_T$ is due to detector resolution effects. The $\not\!\!\!E_T$ for our like-sign leptons + dijet events is shown in Fig. 41 for both the right- (solid) and left-handed (dashed) W' cases. Furthermore, the two jets in our process originate from a SM W whereas the jets in the top background do not. Hence $\not\!\!\!\!E_T$ and dijet invariant mass, m_{jj} , cuts are also applied:

$$E_T < 30 \text{ GeV}, \quad 60 \text{ GeV} < m_{jj} < 100 \text{ GeV}.$$
 (5.104)

The effect of these cuts on the signal rate are seen in the fourth line of Table 26.

Having obtained a measurement of m_N from Eq. (5.100) and $M_{W'}$ from the W's Jacobian peak, if desired, invariant mass cuts on $m_{\ell_N jj}$ and \hat{s} can be imposed to further isolate the signal:

$$|m_{\ell_N jj} - m_N| \le 0.1 \ m_N \text{ and } |\hat{s} - M_{W'}| \le 0.1 \ M_{W'}.$$
 (5.105)

The effects of these cuts are shown in the fifth line of Table 26.



Figure 42: Integrated luminosity needed at 14 TeV LHC for (a) achievable statistical significance for W'_R with $M_{W'} = 3$ TeV and $m_N = 500$ GeV, and (b) reachable W'_R mass at 3σ and 5σ sensitivity.

As \sqrt{s} increases from 8 TeV to 14 TeV, the percentage of events passing the selection cuts also increases. See the final line of Table 26. In particular, we note that relatively fewer events are failing the cuts imposed on the reconstructed masses [Eq. (5.105)]. To understand this effect, consider that increasing the c.m. energy also enlarges the phase space. Consequently, our internal propagators are more likely to be on-shell.

The contribution from the irreducible background for our $\ell^{\pm}\ell^{\pm}jj$ signal,

$$pp \to W^{\pm}W^{\pm}W^{\mp}, \quad pp \to W^{\pm}W^{\pm}jj, \quad pp \to t\bar{t}$$
 (5.106)

events and

$$pp \to jjZZ, \quad pp \to jjZW,$$
 (5.107)

wherein leptons from the Z boson escape from a detector, are estimated [168] to be at most $\sigma = 0.08$ fb using a comparable list of selection cuts. However, this previous analysis does not impose any restriction on the invariant mass of the system as done in Eq. (5.105), and therefore, realistically, the background will be much less than 0.08 fb. In either case, our W'_R signal is clearly

$$W'_{L}: \xrightarrow{q} \overbrace{\ell^{+}}^{\overline{q}'} W'_{R}: \xrightarrow{q} \overbrace{\ell^{+}}^{\overline{q}'} \overbrace{N}^{\ell^{+}} \overbrace{N}^{\ell^{+}} \overbrace{N}^{\ell^{+}} \overbrace{N}^{N} \overbrace{I}^{\ell^{+}} \overbrace{N}^{\ell^{+}} \overbrace{I}^{\ell^{+}} \overbrace{N}^{N} \overbrace{I}^{\ell^{+}} \overbrace{I$$

Figure 43: Spin correlations for $q\bar{q}' \to W' \to N\ell^+$ for (a) left-handed and (b) right-handed couplings. Single arrow lines represent momentum directions and double arrow lines represent spin directions.

above background. Using $\sigma = 0.08$ fb as an estimation for our background, we calculate the significance and reachability of our W'_R signal at the 14 TeV LHC as shown in Fig. 42. With 100 fb⁻¹ integrated luminosity, a W'_R signal via the lepton-number violating process can be observed at a 5σ level up to a mass of 3 TeV. As evident, the required integrated luminosity for a discovery at the LHC grows rapidly with increasing $M_{W'_R}$. This is expected if we again consider that the W boson becomes increasingly boosted as $M_{W'_R}$ grows. A more boosted W leads to more collimated jets, which have more difficulty passing the isolation cuts [Eq. (5.98)] than their less collimated counterparts.

5.6 W' CHIRAL COUPLINGS FROM ANGULAR CORRELATIONS AT THE LHC

Once a new gauge boson W' is observed at the LHC, it is of fundamental importance to determined the nature of its coupling to the SM fermions. Here, we identify various kinematical quantities that depend on the chiral couplings of the fermions to a W'. Each quantity will have a different dependence on the W' chiral couplings and so will provide independent measurements of the chiral couplings.

5.6.1 W' Chiral Couplings To Leptons

Figure 43 shows the spin correlations for the process $q\bar{q}' \to W' \to N\ell^+$ in the partonic c.m. frame for both the (a) left-handed and (b) right-handed cases. Double arrowed lines represent spin and single arrowed lines momentum. As it is well-known, although the preferred charged lepton momentum direction leads to a clear distribution of parity violation, it cannot reveal more detailed nature of the chiral coupling. On the other hand, the nature of the W' leptonic chiral couplings is encoded in polarization of the heavy neutrino, i.e., in the $W'_R(W'_L)$ case the heavy neutrino is preferentially right-handed (left-handed). Hence, if the polarization of the neutrino can be determined, the lefthanded and right-handed cases can be distinguished. Spin observables such as $\langle \hat{s}_N \cdot \hat{a} \rangle$, where s_N is the spin of the heavy neutrino and \hat{a} is an arbitrary spin quantization axis, are sensitive to the polarization of the heavy neutrino. Defining the angle θ^* between the \hat{a} and the direction of motion of the charged lepton originating from the heavy neutrino decay, \hat{p}_{ℓ_2} , the angular distribution of the partial width of the neutrino decaying into a charged lepton and two jets is [254]

$$\frac{1}{\Gamma} \frac{d\Gamma}{d\cos\theta^*} (N \to \ell^{\pm} jj) = \frac{1}{2} \left(1 + 2 A^{\ell^{\pm}} \cos\theta^* \right), \qquad (5.108)$$

where $A^{\ell^+} = -A^{\ell^-} \equiv A$ due to the CP invariance. The coefficient A is related to $\langle \hat{s}_N \cdot \hat{a} \rangle$ and is the forward-backward asymmetry of the charged lepton with respect to the direction \hat{a} . We will refer to A as the analyzing power. The angular distribution of either of the two jets from the neutrino decay will also have a similar linear form and may be used to perform this analysis, although uncertainties in jet measurements may cause more complications.

A highly boosted neutrino from a heavy W' decay will be produced mostly in a helicity state; hence, it is natural to choose $\hat{a} = \hat{p}_N$, the direction of motion of the neutrino in the partonic c.m. frame, and measure \hat{p}_{ℓ_2} in the neutrino rest-frame. At the partonic level, the angular distribution of the lepton from neutrino decay in the reconstructible neutrino rest-frame is (See App. 5.3)

$$\frac{d\hat{\sigma}(u\bar{d} \to \ell_{1}^{+}\ell_{2}^{+}W^{-})}{d\cos\theta_{\ell_{2}}} = \frac{\hat{\sigma}_{Tot.}}{2} \left[1 + \left(\frac{\hat{\sigma}(W_{0}) - \hat{\sigma}(W_{T})}{\hat{\sigma}(W_{0}) + \hat{\sigma}(W_{T})} \right) \left(\frac{2 - \mu_{N}^{2}}{2 + \mu_{N}^{2}} \right) \left(\frac{g_{R}^{\ell\,2}|Y_{\ell_{1}N}|^{2} - g_{L}^{\ell\,2}|V_{\ell_{1}N}|^{2}}{g_{R}^{\ell\,2}|Y_{\ell_{1}N}|^{2} + g_{L}^{\ell\,2}|V_{\ell_{1}N}|^{2}} \right) \cos\theta_{\ell_{2}} \right].$$
(5.109)

Here $\hat{\sigma}(W_0)$ and $\hat{\sigma}(W_T)$ are the partonic level $u\overline{d} \to W'^+ \to \ell_1^+ \ell_2^+ W_\lambda^-$ cross sections with N decaying into longitudinally ($\lambda = 0$) and transversely ($\lambda = T$) polarized W's, respectively. They are given by

$$\times \frac{1}{\left[(\hat{s} - M_{W'}^2)^2 + (\Gamma_{W'}M_{W'})^2\right]} (1 - y_W^2)^2 (1 - \mu_N^2)^2 (2 + \mu_N^2) \left(\frac{1}{2y_W^2}\right)$$
(5.111)

$$\hat{\sigma}(W_T) \equiv \hat{\sigma}(ud \to \ell_1^+ N \to \ell_1^+ \ell_2^+ W_T^-)$$
(5.112)

$$= \hat{\sigma}(W_0) \times 2y_W^2. \tag{5.113}$$

where $\mu_N = m_N / \sqrt{\hat{s}}$, $y_W = M_W / m_N$, and $\hat{\sigma}_{Tot.} = (\hat{\sigma}(W_0) + \hat{\sigma}(W_T)) \times BR(W \to q\bar{q}')$ is the total partonic cross section. As W' comes on-shell, $\mu_N \to x_N$. In this reference frame, θ^* from Eq. (5.108) satisfies

$$\cos\theta^* = \cos\theta_{\ell_2} \equiv \hat{p}_{\ell_2} \cdot \hat{p}_N, \tag{5.114}$$

where, again, \hat{p}_{ℓ_2} is measured in the neutrino rest-frame and \hat{p}_N is measured in the partonic c.m. frame.

For an on-shell W', the analyzing power at the partonic and hadronic level are the same. In such a case, after comparing Eqs. (5.108) and (5.109), we find that the analyzing power is

$$A = \frac{1}{2} \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{2 - x_N^2}{2 + x_N^2} \right) \left(\frac{g_R^{\ell 2} |Y_{\ell_1 N}|^2 - g_L^{\ell 2} |V_{\ell_1 N}|^2}{g_R^{\ell 2} |Y_{\ell_1 N}|^2 + g_L^{\ell 2} |V_{\ell_1 N}|^2} \right) = \frac{1}{2} \left(\frac{1 - 2y_W^2}{1 + 2y_W^2} \right) \left(\frac{2 - x_N^2}{2 + x_N^2} \right) \left(\frac{g_R^{\ell 2} |Y_{\ell_1 N}|^2 - g_L^{\ell 2} |V_{\ell_1 N}|^2}{g_R^{\ell 2} |Y_{\ell_1 N}|^2 + g_L^{\ell 2} |V_{\ell_1 N}|^2} \right).$$
(5.115)

The different signs for the analyzing power between the neutrino decays to the two different W polarizations and between the $W'_{L,R}$ cases can be understood via the spin correlation in Fig. 39. For the W'_R case, a heavy neutrino decaying to a longitudinal (transverse) W will have the charged lepton preferentially moving with (against) \hat{p}_N . For the W'_L case the helicity of the neutrino, and therefore the direction of the charged lepton, is reversed. Hence the analyzing power is proportional to $(\hat{\sigma}(W_0) - \hat{\sigma}(W_T))(g_R^{\ell 2}|Y_{\ell_1N}|^2 - g_L^{\ell 2}|V_{\ell_1N}|^2)$.

In the analysis of Fig. 39, the left- and right-chiral neutrinos at the $W' \to N\ell^+$ vertex are approximated as the left-handed and right-handed helicity states in the partonic c.m. frame. As the neutrino becomes more massive relative to the W', the approximation of the chiral basis by the helicity basis begins to break down, i.e., the left- (right-) helicity state makes a larger contribution to the right- (left-) chiral state. In Eq. (5.109), this is reflected by the $\cos \theta_{\ell_2}$ ($\cos \theta_{\ell}$ for simplicity) coefficient

$$\frac{2 - x_N^2}{2 + x_N^2} = \frac{2M_{W'}^2 - m_N^2}{2M_{W'}^2 + m_N^2}.$$
(5.116)



Figure 44: The angular distribution of the charged lepton originating from neutrino decay in the heavy neutrino rest-frame with respect to the neutrino moving direction in the partonic c.m. frame at the LHC with $M_{W'}$, m_N set by Eq. (5.80). Distribution (a) without smearing or cuts, (b) with energy smearing and cuts in Eqs. (5.96), (5.98), (5.104), and (5.105), and (c) with all cuts applied to (b) except the ΔR_{jj} cuts in Eq. (5.98). The solid lines are for the Monte Carlo simulation results and in (a) and (c) the dashed lines are for the analytical result in Eq. (5.109).

As x_N increases, the distribution flattens due to the right-handed (left-handed) neutrino helicity state, thereby making a larger contribution to the W'_L (W'_R) distributions.

Figure 44 shows the hadronic level angular distribution of the lepton in the neutrino's restframe for both W'_L and W'_R at the LHC. The case without smearing or cuts is shown in Fig. 44(a), and contains both the analytical results (dashed line) and Monte Carlo simulation (solid line) histograms. As can be clearly seen, the analytical and numerical results are in good agreement. Figure 44(b) shows the leptonic angular distribution after energy smearing and cuts in Eqs. (5.96), (5.98), (5.104), and (5.105). Notice that there is a small depletion of events for $\cos \theta_{\ell} \approx 1$ and a large depletion when $\cos \theta_{\ell} < 0$. First, when $\cos \theta_{\ell} \approx 1$ the charged lepton is moving with and the jets against the direction of motion of the neutrino in the partonic c.m. frame. Hence, with boost back to the partonic c.m. frame, the jets are softest at this point and the jet p_T cuts in Eq. (5.96) lead to a depletion of event in this region. When $\cos \theta_{\ell} < 0$, the lepton is moving against and the SM W is moving with the neutrino's direction of motion. Hence, with the boost back to the partonic c.m. frame, the W is boosted and its decay products highly collimated. Consequently, the ΔR_{jj} cuts in Eq. (5.98) lead to a large depletion of events. Figure 44(c) shows lepton angular distribution with the same cuts as Fig. 44(b) except the ΔR_{jj} cuts. For comparison, both the Monte Carlo simulation with cuts (solid) and analytical results without cuts (dashed) are shown. It is clear that the discriminating power of the lepton angular distribution would increases and the Monte Carlo distribution approaches the analytical results if the jet isolation cuts are relaxed.

The analyzing power in Eq. (5.115) can additionally be related to the forward backward asymmetry

$$\mathcal{A} = \frac{\sigma(\cos\theta_{\ell} \ge 0) - \sigma(\cos\theta_{\ell} < 0)}{\sigma(\cos\theta_{\ell} \ge 0) + \sigma(\cos\theta_{\ell} < 0)}.$$
(5.117)

Without cuts or smearing, $\mathcal{A} = A$; and for the values of m_N , $M_{W'}$ stipulated in Eq. (5.80),

,

$$A = \begin{cases} +0.43, & W' = W'_R \\ -0.43, & W' = W'_L \end{cases}.$$
(5.118)

The simulated values for the forward backward asymmetry with consecutive cuts are shown in Table 27. Again, simulations are in good agreement with the theoretical prediction for the forward backward asymmetry for no smearing or cuts. As the cuts become more severe, the simulated and theoretical values deviate more, however the W'_L and W'_R cases can still be distinguished clearly. Furthermore, as shown in the final row, if the ΔR_{jj} cuts in Eq. (5.98) are relaxed, the discriminating power of the asymmetry is greatly increased, and the theory and simulation are in much better agreement.

Table 27: Forward-backward asymmetry for $pp \to W'_{L,R}^+ \to \mu^+ \mu^+ q\bar{q}'$ with consecutive cuts at 8 and 14 TeV LHC. The last row has the same cuts applied as the previous row with the removal of the ΔR_{jj} cuts in Eq. (5.98).

4	8 TeV		14 TeV	
		W'_R	W'_L	W'_R
Reco. without Cuts or Smearing	-0.42	0.42	-0.43	0.43
+ Smearing + Fiducial + Kinematics (Eq. (5.96))	-0.46	0.33	-0.47	0.34
+ Isolation (Eq. (5.98))	-0.11	0.59	0.083	0.72
$+\not\!$	-0.078	0.62	0.11	0.75
+ Mass Reco. (Eq. (5.105))	0.16	0.77	0.18	0.77
$-\Delta R_{jj}$	-0.34	0.49	-0.34	0.49

5.6.2 W' Chiral Couplings to Initial-State Quarks

Thus far, we have only presented the results to test the chiral coupling of W' to the final state leptons. It is equally important to examine its couplings to the initial state quarks. Define an azimuthal angle

$$\cos \Phi = \frac{\hat{p}_N \times \vec{p}_{\ell_2}}{|\hat{p}_N \times \vec{p}_{\ell_2}|} \cdot \frac{\hat{p}_N \times \vec{p}_q}{|\hat{p}_N \times \vec{p}_q|},$$
(5.119)

as the angle between the $qq' \to N\ell_1^+$ production plane and $N \to W^-\ell_2^+$ decay plane in the neutrino rest-frame, where \vec{p}_{ℓ_2} is the three momentum of ℓ_2 , the charged lepton identified as originating from the neutrino; \hat{p}_N is the direction of motion of the neutrino in the partonic c.m. frame; and \vec{p}_q is the initial-state quark momentum. The definition of Φ is invariant under boosts along \hat{p}_N , hence the quark and charged lepton momenta can be evaluated either in the partonic c.m. or the neutrino rest-frame. The angular distribution between the two planes is thus calculated to be

$$\frac{d\hat{\sigma}}{d\Phi} = \frac{\sigma_{Tot.}}{2\pi} \left[1 + \frac{3\pi^2}{16} \frac{\mu_N}{2 + \mu_N^2} \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{g_R^{q\,2} - g_L^{q\,2}}{g_R^{q\,2} + g_L^{q\,2}} \right) \cos \Phi \right]. \tag{5.120}$$

The distribution for W'_L is 180° out of phase with the W'_R distribution and the slope only depends on the W' chiral coupling to the initial-state quarks. Hence, the phase of this distribution determines the chirality of the initial-state quarks couplings to the W' independently of the leptonic chiral



Figure 45: Spin correlations for neutrino production in the neutrino rest-frame. Single arrowed lines represent momentum and double arrowed lines represent spin in the helicity basis. The \hat{z} -axis is defined to be the neutrino's direction of motion in the partonic c.m. frame and the \hat{y} -axis is defined such that y-component of the initial-state quark momentum is always positive.

couplings to the W'.

To understand the distribution in Eq. (5.120), we consider the spin correlations between the initial and final states. As noted previously, the angle Φ is invariant under the boosts along \hat{p}_N . So for simplicity, we consider the spin correlations in the heavy neutrino rest-frame. Figure 45 shows the spin correlations of the neutrino production in the neutrino's rest-frame for both the (a) W'_L and (b) W'_R cases. Like before, single arrowed lines represent momentum directions and double arrowed lines spin in the helicity basis. Also, we define the production plane to be oriented in the $\hat{y} - \hat{z}$ plane such that the \hat{y} -component of the quark momentum always points along the positive \hat{y} -axis and that $\hat{z} = \hat{p}_N$. With this axis convention, $\Phi = -\phi_{\ell_2}$, where ϕ_{ℓ_2} is the azimuthal angle of ℓ_2 as measured from the positive \hat{y} -axis.

Figure 46 shows the spin correlations for the heavy neutrino production and decay with the spin quantization axis chosen to be the \hat{y} direction as defined above. The W'_L case is shown in Figs. 46(a,c) and the W'_R case in (b,d). The solid dots next to the N and ℓ_1 indicate that they have no momentum in the \hat{y} -direction. In the W'_R case, the initial-state quark must be right-handed and the initial-state antiquark left-handed. Hence, the total spin of the initial-state points in the positive \hat{y} -direction, causing the spin of the neutrino to also point in the positive \hat{y} -direction.

Figure 46: Spin correlations in the neutrino rest-frame as described in Fig. 45. Double arrowed lines represent spin with \hat{y} being the quantization axis and single arrowed lines are the \hat{y} component of the particles.

When the neutrino decays to a longitudinal or transverse W, the lepton from the neutrino decay has spin along or against the \hat{y} -axis, respectively. For the W'_R case, figures 46(b) and (d) show the decay into longitudinal and transverse W's, respectively. Therefore, for the decay into W_0 (W_T) case, the lepton prefers to move in the same (opposite) direction as the initial-state quark and Φ peaks at 0 ($\pm \pi$). In the W'_L case, the direction of motion of ℓ_2 relative to the direction of motion of the initial-state quark is reversed and the peaks in the Φ distribution are shifted by π . This explains the 180° phase difference in the angular distribution, Eq. (5.120), between the W'_L and W'_R cases, and between the neutrino decay to W_0 and W_T . Also, notice that this argument only relies on the W' - q - q' coupling and not the $W' - N - \ell$ chiral couplings. Hence, measuring the



Figure 47: Φ distributions at the 14 TeV LHC with $M'_W = 3$ TeV and $m_N = 1.5$ TeV for fully reconstructed events (solid), the analytical result in Eq. (5.120) (dashed), and Monte Carlo truth (dash-dot). Figure (a) is without energy smearing or cuts, (b) with energy smearing and cuts in Eqs. (5.96), (5.98), (5.104), and (5.105), and (c) with the same cuts as (b) without the ΔR_{jj} cut in Eq. (5.98).

distribution of the angle between the $qq' \to N\ell_1$ production and the $N \to \ell_2^+ W^-$ decay planes can determine the chiral couplings of a W' to light quarks independently from the chiral couplings of the W' to leptons.

Most of the angular definition and analysis depend on the initial state quark momentum direction. Since the LHC is a symmetric pp machine, this is not known *a priori*. However, at the LHC u and d quarks are valence and antiquarks are sea. Hence, the initial-state quark generally has a larger momentum fraction than the initial-state antiquark; and the initial-state quark direction can be identified as the direction of motion of the fully reconstructed partonic c.m. frame. Similar techniques have been used for studying forward-backward asymmetries associated with new heavy gauge bosons [234, 255].

Figure 47 shows the Φ distributions at the 14 TeV LHC with $M'_W = 3$ TeV for both W'_L and W'_R . From Eq. (5.120), the amplitude of the Φ distribution depends on the ratio m_N/M'_W , and therefore increase m_N to 1.5 TeV. The solid line is the Φ distribution with the initial state quark moving direction identified as the partonic c.m. frame boost direction; the dashed lines is the theoretical distribution given in Eq. (5.120); and in (a) the dash-dot lines are the Monte Carlo truth, i.e. using the known direction of the initial-state quark.

Figure 47(a) does not include cuts or smearing; as can be seen, the Monte Carlo truth and theoretical calculation agree very well. The reconstructed distribution has a smaller amplitude than the theoretical distribution due to the direction of the initial-state quark being misidentified. Figure 47(b) shows the theoretical prediction and reconstructed distribution with smearing and the cuts in Eqs. (5.96,5.98,5.104,5.105) applied. For $\Phi = 0$, the SM W is maximally boosted and its decay products are maximally collimated. Consequently, the ΔR_{jj} cut in Eq. (5.98) causes a large depletion of events in the central region. Figure 47(c) shows the reconstructed distribution with the same cuts as (b) minus the ΔR_{jj} cut. With the relaxation of this cut, the W'_L and W'_R cases become reasonably discernible with the W'_L distribution nearly the same as the theoretical prediction. The continued depletion of events at $\Phi = 0$ and $\Phi = \pm \pi$ are due to the rapidity cuts on leptons and jets, respectively.

5.7 UNLIKE-SIGN DILEPTON ANGULAR DISTRIBUTIONS

Intrinsically, Majorana neutrinos can decay to positively or negatively charged leptons, and therefore also contribute to the L-conserving process

$$pp \to W' \to \ell_1^+ \ell_2^- jj. \tag{5.121}$$

These events can be reconstructed similarly to the method described in Section 5.5. However, the SM backgrounds for this process, particularly $pp \rightarrow Zjj$, will be larger. Our purpose here is not to do a full signal versus backgrounds study, but to comment on the differences between the like-sign and unlike-sign lepton cases. Again, $u\bar{d}$ has a larger parton luminosity than $d\bar{u}$, so we focus only

on W'^+ production:

$$pp \to W'^+ \to N\ell_1^+ \to \ell_1^+ \ell_2^- jj \tag{5.122}$$

5.7.1 W' Chiral Coupling from Angular Distributions

For the unlike-sign case, we mimic our entire like-sign analysis and reconstruct the polar angular distribution of the lepton originating from neutrino decay in the heavy neutrino rest-frame (App. 5.3.4). Respectively, the polar and azimuthal distributions are similar to those in Eqs. (5.109)and (5.120) up to a opposite sign in front of the angular dependence.

$$\frac{d\hat{\sigma}}{d\cos\theta_{\ell_2}} = \frac{\hat{\sigma}_{Tot.}}{2} \left[1 - \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{2 - \mu_N^2}{2 + \mu_N^2} \right) \left(\frac{g_R^{\ell\,2} |Y_{\ell_1N}|^2 - g_L^{\ell\,2} |V_{\ell_1N}|^2}{g_R^{\ell\,2} |Y_{\ell_1N}|^2 + g_L^{\ell\,2} |V_{\ell_1N}|^2} \right) \cos\theta_{\ell_2} \right],\tag{5.123}$$

$$\frac{d\hat{\sigma}}{d\Phi} = \frac{\hat{\sigma}_{Tot.}}{2\pi} \left[1 - \frac{3\pi^2}{16} \frac{\mu_N}{2 + \mu_N^2} \left(\frac{\hat{\sigma}(W_0) - \hat{\sigma}(W_T)}{\hat{\sigma}(W_0) + \hat{\sigma}(W_T)} \right) \left(\frac{g_R^{q\,2} - g_L^{q\,2}}{g_R^{q\,2} + g_L^{q\,2}} \right) \cos \Phi \right]. \tag{5.124}$$

Figure 48 shows the Φ distributions for the unlike-sign process and follows the identical procedure as for the like-sign case. The solid line is the Φ distribution with the initial-state quark propagation direction identified as the partonic c.m. frame boost direction; the dashed lines are the theoretical distributions given by Eq. (5.124); and in (a) the dashed-dotted lines are the Monte Carlo truth, i.e., using the known direction of the initial-state quark. Figure 48(a) does not include cuts or smearing. Figure 48(b) shows the theoretical prediction and reconstructed distribution with smearing and cuts in Eqs. (5.96), (5.98), (5.104), and (5.105) applied. Figure 48(c) shows the reconstructed distribution with the same cuts as 48(b) minus the ΔR_{jj} isolation cut.

To understand why the sign of the slope for the *L*-conserving distributions differ from the *L*-violating distributions, we turn to spin correlations. For W'^+ , the spin correlations for $u\bar{d} \rightarrow W'^+ \rightarrow N\ell^+$ are shown in Fig. 43 without yet specifying N's decay. However, we only need to analyze the angular correlation in the neutrino decay. The spin correlations are simply obtained by replacing the right-handed antilepton in Fig. 39 with a left-handed lepton. Since the direction of the spin of the lepton is completely determined by the neutrino spin, which is unchanged between the two cases, the effect of the helicity flip is to reverse the direction of the final state lepton momentum relative to the \hat{z} direction. Therefore, the slopes of the lepton angular distribution are opposite for the like-sign and unlike-sign lepton cases. These same arguments can be made to show that the phases of the Φ distribution in Eqs. (5.120) and (5.124) differ by 180°.



Figure 48: For the opposite sign lepton case, the angular distribution of the charged lepton originating from neutrino decay in the heavy neutrino rest-frame with respect to the neutrino moving direction in the partonic c.m. frame at the LHC with $M_{W'}$, m_N set by Eq. (5.80). Distribution (a) without smearing or cuts, (b) with energy smearing and cuts in Eqs. (5.96), (5.98), (5.104), and (5.105), and (c) with all cuts applied to (b) except the ΔR_{jj} cuts in Eq. (5.98). The solid lines are for the Monte Carlo simulation results and in (a) and (c) the dashed lines are for the analytical result in Eq. (5.109).

The analysis of the two cases also reveals that, unlike the angular distributions, the total cross section is independent of having like-sign or unlike-sign leptons in the final state. This may be understood by recognizing that the difference between the two final states is tantamount to a charge conjugation. Having integrated out the angular dependence, the total cross section is invariant under parity inversion. Consequently, by CP-invariance, the total rate is invariant under charge conjugation. This behavior is evident in Eq. (5.90) and Fig. 34, which show that N decays to ℓ^+W^- and ℓ^-W^+ equally.

5.8 SUMMARY

The nature of the neutrino mass remains one of most profound puzzles in particle physics. The possibility of its being Majorana-like is an extremely interesting aspect since it may have farreaching consequences in particle physics, nuclear physics and cosmology.

Given the outstanding performance of the LHC, we are motivated to study the observability for a heavy Majorana neutrino N along with a new charged gauge boson W' at the LHC. We first parameterized their couplings in a model-independent approach in Section 5.2 and presented the current constraints on the mass and coupling parameters.

We studied the production and decay of W' and N at the LHC, and optimized the observability of the like-sign dilepton signal over the SM backgrounds. We emphasized the complementarity of these two particles by exploiting the characteristic kinematical distributions resulting from spincorrelations to unambiguously determine their properties. Our phenomenological results can be summarized as follows.

- 1. The heavy neutrino is likely to have a large R.H. component and thus the W'_R would likely yield a larger signal rate than that for W'_L , governed by the mixing parameters as discussed in Section 5.2. Under these assumptions, we found that at the 14 TeV LHC a 5σ signal, via the clean channels $\ell^{\pm}\ell^{\pm}jj$, may be reached for $M_{W'_R} = 3$ TeV (4 TeV) with 90 fb⁻¹ (1 ab⁻¹) integrated luminosity, as seen in Fig. 42.
- 2. The chiral coupling of W' to the leptons can be inferred by the polar angle distribution of the leptons in the reconstructed neutrino frame, as seen in Fig. 44, owing to the spin correlation from the intermediate state N.

- 3. The chiral coupling of W' to the initial state quarks can be inferred by the azimuthal angular distribution of the neutrino production and decay planes, as seen in Fig. 47.
- 4. The kinematical distributions for the like-sign and unlike-sign cases have been found to be quite sensitive to spin correlations and are complementary. In particular, the angular distributions differ by a minus sign and provide qualitative differences for a Majorana and a Dirac N. Thus in addition to observing final states that violate lepton-number, comparison of the two scenarios provides a means to differentiate the Majorana nature of N.

Overall, if the LHC serves as a discovery machine for a new gauge boson W', then its properties and much rich physics will await to be explored. Perhaps a Majorana nature of a heavy neutrino may be first established associated with W' physics.

BIBLIOGRAPHY

- T. Han, I. Lewis, R. Ruiz and Z. g. Si, Phys. Rev. D 87, no. 3, 035011 (2013) [arXiv:1211.6447 [hep-ph]].
- [2] T. Han and R. Ruiz, Phys. Rev. D 89, no. 7, 074045 (2014) [arXiv:1312.3324 [hep-ph]].
- [3] D. Alva, T. Han and R. Ruiz, JHEP **1502**, 072 (2015) [arXiv:1411.7305 [hep-ph]].
- [4] K. Agashe et al. [Top Quark Working Group Collaboration], arXiv:1311.2028 [hep-ph].
- [5] S. Dawson, A. Gritsan, H. Logan, J. Qian, C. Tully, R. Van Kooten, A. Ajaib and A. Anastassov et al., arXiv:1310.8361 [hep-ex].
- [6] M. E. Peskin and D. V. Schroeder, Reading, USA: Addison-Wesley (1995)
- [7] J. Goldstone, Nuovo Cim. 19, 154 (1961); J. Goldstone, A. Salam and S. Weinberg, Phys. Rev. 127, 965 (1962).
- [8] F. Englert and R. Brout, Phys. Rev. Lett. 13, 321 (1964).
- [9] P. W. Higgs, Phys. Lett. **12**, 132 (1964).
- [10] P. W. Higgs, Phys. Rev. Lett. **13**, 508 (1964).
- [11] G. S. Guralnik, C. R. Hagen and T. W. B. Kibble, Phys. Rev. Lett. 13, 585 (1964).
- [12] P. W. Higgs, Phys. Rev. **145**, 1156 (1966).
- [13] T. W. B. Kibble, Phys. Rev. **155**, 1554 (1967).
- [14] G. 't Hooft and M. J. G. Veltman, Nucl. Phys. B 44, 189 (1972).
- [15] M. Gell-Mann, Phys. Lett. 8, 214 (1964).
- [16] G. Zweig, CERN-TH-401.
- [17] G. Zweig, Developments in the Quark Theory of Hadrons, Volume 1. Edited by D. Lichtenberg and S. Rosen. pp. 22-101
- [18] D. J. Gross and F. Wilczek, Phys. Rev. Lett. 30, 1343 (1973).
- [19] H. D. Politzer, Phys. Rev. Lett. **30**, 1346 (1973).

- [20] S. L. Glashow, Nucl. Phys. **22**, 579 (1961).
- [21] S. Weinberg, Phys. Rev. Lett. **19**, 1264 (1967).
- [22] A. Salam, Conf. Proc. C 680519, 367 (1968).
- [23] S. Willenbrock, hep-ph/0410370.
- [24] F. Halzen and A. D. Martin, New York, Usa: Wiley (1984) 396p
- [25] K. A. Olive *et al.* [Particle Data Group Collaboration], Chin. Phys. C 38, 090001 (2014).
- [26] G. Aad et al. [ATLAS Collaboration], Phys. Lett. B 716, 1 (2012) [arXiv:1207.7214 [hep-ex]].
- [27] S. Chatrchyan *et al.* [CMS Collaboration], Phys. Lett. B **716**, 30 (2012) [arXiv:1207.7235 [hep-ex]].
- [28] G. Aad *et al.* [CMS s Collaboration], arXiv:1503.07589 [hep-ex].
- [29] G. Aad *et al.* [ATLAS Collaboration], Phys. Lett. B **726**, 88 (2013) [Erratum-ibid. B **734**, 406 (2014)] [arXiv:1307.1427 [hep-ex]].
- [30] V. Khachatryan *et al.* [CMS Collaboration], arXiv:1412.8662 [hep-ex].
- [31] S. Chatrchyan *et al.* [CMS Collaboration], Phys. Lett. B **725**, 36 (2013) [arXiv:1302.1764 [hep-ex]].
- [32] J. Ellis and T. You, JHEP **1306**, 103 (2013) [arXiv:1303.3879 [hep-ph]].
- [33] N. Cabibbo, Phys. Rev. Lett. **10**, 531 (1963).
- [34] M. Kobayashi and T. Maskawa, Prog. Theor. Phys. 49, 652 (1973).
- [35] S. L. Glashow, J. Iliopoulos and L. Maiani, Phys. Rev. D 2, 1285 (1970).
- [36] V. D. Barger and R. J. N. Phillips, REDWOOD CITY, USA: ADDISON-WESLEY (1987) 592 P. (FRONTIERS IN PHYSICS, 71)
- [37] T. Han, hep-ph/0508097.
- [38] S. Willenbrock, In *Boulder 1989, Proceedings, From actions to answers* 323-362 and Brookhaven Nat. Lab. Upton - BNL-43793 (90, rec. Mar.) 40 p
- [39] M. Thomson,
- [40] S. Weinberg, Cambridge, UK: Univ. Pr. (1995) 609 p
- [41] T. Aaltonen et al. [CDF Collaboration], Phys. Lett. B 722, 48 (2013) [arXiv:1205.0354 [hepex]].
- [42] T. Plehn, Lect. Notes Phys. 844, 1 (2012) [arXiv:0910.4182 [hep-ph]].

- [43] J. M. Campbell and R. K. Ellis, J. Phys. G 42, no. 1, 015005 (2015) [arXiv:1204.1513 [hepph]].
- [44] E. J. Williams, Nature of the high-energy particles of penetrating radiation and status of ionization and radiation formulae, Phys. Rev. 45, 729 (1934).
- [45] C. F. von Weizsacker, Radiation emitted in collisions of very fast electrons, Z. Phys. 88, 612 (1934).
- [46] S. Dawson, Nucl. Phys. B **249**, 42 (1985).
- [47] V. M. Budnev, I. F. Ginzburg, G. V. Meledin and V. G. Serbo, The Two photon particle production mechanism. Physical problems. Applications. Equivalent photon approximation, Phys. Rept. 15, 181 (1975).
- [48] M. Drees, R. M. Godbole, M. Nowakowski and S. D. Rindani, Phys. Rev. D 50, 2335 (1994) [hep-ph/9403368].
- [49] S. Chatrchyan et al. [CMS Collaboration], Exclusive photon-photon production of muon pairs in proton-proton collisions at $\sqrt{s} = 7$ TeV, JHEP **1201**, 052 (2012) [arXiv:1111.5536].
- [50] I. Sahin and M. Koksal, Search for electromagnetic properties of the neutrinos at the LHC, JHEP 1103, 100 (2011) [arXiv:1010.3434 [hep-ph]].
- [51] J. de Favereau de Jeneret, V. Lemaitre, Y. Liu, S. Ovyn, T. Pierzchala, K. Piotrzkowski, X. Rouby and N. Schul et al., High energy photon interactions at the LHC, arXiv:0908.2020.
- [52] E. Chapon, C. Royon and O. Kepka, Anomalous quartic W W gamma gamma, Z Z gamma gamma, and trilinear WW gamma couplings in two-photon processes at high luminosity at the LHC, Phys. Rev. D 81, 074003 (2010) [arXiv:0912.5161 [hep-ph]].
- [53] A. Pukhov, E. Boos, M. Dubinin, V. Edneral, V. Ilyin, D. Kovalenko, A. Kryukov and V. Savrin *et al.*, hep-ph/9908288;
- [54] A. Pukhov, hep-ph/0412191;
- [55] A. Belyaev, N. D. Christensen and A. Pukhov, CalcHEP 3.4 for collider physics within and beyond the Standard Model, Comput. Phys. Commun. 184, 1729 (2013) [arXiv:1207.6082].
- [56] J. Alwall, R. Frederix, S. Frixione, V. Hirschi, F. Maltoni, O. Mattelaer, H. -S. Shao and T. Stelzer et al., The automated computation of tree-level and next-to-leading order differential cross sections, and their matching to parton shower simulations, arXiv:1405.0301.
- [57] J. J. Sakurai and D. Schildknecht, Generalized vector dominance and inelastic electron proton scattering, Phys. Lett. B 40, 121 (1972).
- [58] J. Alwall and G. Ingelman, Interpretation of electron proton scattering at low Q^{**}2, Phys. Lett. B 596, 77 (2004) [hep-ph/0402248].
- [59] A. D. Martin, R. G. Roberts, W. J. Stirling and R. S. Thorne, Parton distributions incorporating QED contributions, Eur. Phys. J. C 39, 155 (2005) [hep-ph/0411040].

- [60] M. R. Whalley, D. Bourilkov and R. C. Group, The Les Houches accord PDFs (LHAPDF) and LHAGLUE, hep-ph/0508110.
- [61] R. D. Ball et al. [NNPDF Collaboration], Parton distributions with QED corrections, Nucl. Phys. B 877, no. 2, 290 (2013) [arXiv:1308.0598 [hep-ph]].
- [62] M. Drees and K. Grassie, Parametrizations of the Photon Structure and Applications to Supersymmetric Particle Production at HERA, Z. Phys. C 28, 451 (1985).
- [63] J. Pumplin, D. R. Stump, J. Huston, H. L. Lai, P. M. Nadolsky and W. K. Tung, New generation of parton distributions with uncertainties from global QCD analysis, JHEP 0207, 012 (2002) [hep-ph/0201195].
- [64] S. Chatrchyan et al. [CMS Collaboration], JHEP 1306, 081 (2013) [arXiv:1303.4571 [hep-ex]]; G. Aad et al. [ATLAS Collaboration], Phys. Lett. B 726, 88 (2013) [arXiv:1307.1427 [hep-ex]].
- [65] M. Czakon, P. Fiedler and A. Mitov, Phys. Rev. Lett. 110, 252004 (2013) [arXiv:1303.6254 [hep-ph]].
- [66] G. Eilam, J. L. Hewett and A. Soni, Phys. Rev. D 44, 1473 (1991) [Erratum-ibid. D 59, 039901 (1998)].
- [67] B. Mele, S. Petrarca and A. Soddu, Phys. Lett. B **435**, 401 (1998) [hep-ph/9805498].
- [68] C. Zhang and F. Maltoni, Phys. Rev. D 88, 054005 (2013) [arXiv:1305.7386 [hep-ph]].
- [69] W.-S. Hou, Phys. Lett. B **296**, 179 (1992).
- [70] D. Atwood, L. Reina and A. Soni, Phys. Rev. D 55, 3156 (1997) [hep-ph/9609279].
- [71] J. A. Aguilar-Saavedra, Acta Phys. Polon. B 35, 2695 (2004) [hep-ph/0409342].
- [72] K. -F. Chen, W. -S. Hou, C. Kao and M. Kohda, Phys. Lett. B 725, 378 (2013) [arXiv:1304.8037 [hep-ph]].
- [73] J. M. Yang, B.-L. Young and X. Zhang, Phys. Rev. D 58, 055001 (1998) [hep-ph/9705341].
- [74] J. J. Cao, G. Eilam, M. Frank, K. Hikasa, G. L. Liu, I. Turan and J. M. Yang, Phys. Rev. D 75, 075021 (2007) [hep-ph/0702264].
- [75] G. Eilam, A. Gemintern, T. Han, J. M. Yang and X. Zhang, Phys. Lett. B 510, 227 (2001) [hep-ph/0102037].
- [76] T. G. Rizzo, Phys. Rev. D 35, 1067 (1987).
- [77] V. D. Barger and W.-Y. Keung, Phys. Lett. B **202**, 393 (1988).
- [78] V. D. Barger, W.-Y. Keung and T. G. Rizzo, Phys. Rev. D 40, 2274 (1989).
- [79] R. Decker, M. Nowakowski and A. Pilaftsis, Mod. Phys. Lett. A 06, 3491 (1991) [Erratumibid. A 07, 819 (1992)].

- [80] R. Decker, M. Nowakowski and A. Pilaftsis, Z. Phys. C 57, 339 (1993) [hep-ph/9301283].
- [81] G. Mahlon and S. J. Parke, Phys. Lett. B **347**, 394 (1995) [hep-ph/9412250].
- [82] E. O. Iltan, Fizika B **13**, 675 (2004) [hep-ph/0208230].
- [83] D. Binosi and L. Theussl, Comput. Phys. Commun. 161, 76 (2004) [hep-ph/0309015].
- [84] C. J. C. Burges and H. J. Schnitzer, Nucl. Phys. B 228, 464 (1983); C. N. Leung, S. T. Love and S. Rao, Z. Phys. C 31 (1986) 433; W. Buchmuller and D. Wyler, Nucl. Phys. B 268 (1986) 621; K. Hagiwara, S. Ishihara, R. Szalapski and D. Zeppenfeld, Phys. Rev. D 48, 2182 (1993).
- [85] K. Whisnant, J.-M. Yang, B.-L. Young and X. Zhang, Phys. Rev. D 56, 467 (1997) [hep-ph/9702305];
- [86] B. Grzadkowski, M. Iskrzynski, M. Misiak and J. Rosiek, JHEP 1010, 085 (2010) [arXiv:1008.4884 [hep-ph]].
- [87] A. De Rujula, M. B. Gavela, P. Hernandez and E. Masso, Nucl. Phys. B 384, 3 (1992).
- [88] G. J. Gounaris, D. T. Papadamou and F. M. Renard, Z. Phys. C 76 (1997) 333 [hep-ph/9609437];
- [89] J. M. Yang and B.-L. Young, Phys. Rev. D 56, 5907 (1997) [hep-ph/9703463].
- [90] T. Han, T. Huang, Z. H. Lin, J. X. Wang and X. Zhang, Phys. Rev. D 61, 015006 (1999) [hep-ph/9908236].
- [91] J. A. Aguilar-Saavedra, Nucl. Phys. B 812, 181 (2009) [arXiv:0811.3842 [hep-ph]].
- [92] J. A. Aguilar-Saavedra, Nucl. Phys. B 821, 215 (2009) [arXiv:0904.2387 [hep-ph]].
- [93] M. B. Einhorn and J. Wudka, Nucl. Phys. B 876 (2013) 556 [arXiv:1307.0478 [hep-ph]].
- [94] M. B. Einhorn and J. Wudka, Nucl. Phys. B 877, 792 (2013) [arXiv:1308.2255 [hep-ph]].
- [95] S. Chatrchyan *et al.* [CMS Collaboration], Phys. Rev. D 89 (2014) 012003 [arXiv:1310.3687 [hep-ex]];
- [96] The ATLAS collaboration, ATLAS-CONF-2013-108.
- [97] V. M. Abazov et al. [D0 Collaboration], Phys. Rev. D 83, 032009 (2011) [arXiv:1011.6549 [hep-ex]];
- [98] V. M. Abazov et al. [D0 Collaboration], Phys. Lett. B 713, 165 (2012) [arXiv:1204.2332 [hep-ex]];
- [99] T. Aaltonen *et al.* [CDF Collaboration], Phys. Rev. D 87, 031104 (2013) [arXiv:1211.4523 [hep-ex]].
- [100] G. Aad et al. [ATLAS Collaboration], JHEP 1206, 088 (2012) [arXiv:1205.2484 [hep-ex]];
 [CMS Collaboration], CMS-PAS-TOP-12-004 (2012), CMS-PAS-TOP-12-015 (2013)

- [101] N. Craig, J. A. Evans, R. Gray, M. Park, S. Somalwar, S. Thomas and M. Walker, Phys. Rev. D 86, 075002 (2012) [arXiv:1207.6794 [hep-ph]].
- [102] D. Atwood, S. K. Gupta and A. Soni, arXiv:1305.2427 [hep-ph].
- [103] The ATLAS collaboration, ATLAS-CONF-2013-081.
- [104] A. Manohar and H. Georgi, Nucl. Phys. B 234, 189 (1984); H. Georgi, Weak Interactions and Modern Particle Theory, (Benjamin/Cummings, Menlo Park, USA, 1984); H. Georgi and L. Randall, Nucl. Phys. B 276, 241 (1986); H. Georgi, Phys. Lett. B 298, 187 (1993). [hep-ph/9207278];
- [105] M. A. Luty, Phys. Rev. D 57, 1531 (1998) [hep-ph/9706235]; A. G. Cohen, D. B. Kaplan and A. E. Nelson, Phys. Lett. B 412, 301 (1997) [hep-ph/9706275].
- [106] J. F. Gunion, H. E. Haber, G. L. Kane and S. Dawson, Front. Phys. 80, 1 (2000).
- [107] J. F. Gunion and H. E. Haber, Phys. Rev. D 67, 075019 (2003) [hep-ph/0207010].
- [108] G. C. Branco, P. M. Ferreira, L. Lavoura, M. N. Rebelo, M. Sher and J. P. Silva, Phys. Rept. 516, 1 (2012) [arXiv:1106.0034 [hep-ph]].
- [109] M. Klute, R. Lafaye, T. Plehn, M. Rauch and D. Zerwas, Phys. Rev. Lett. 109, 101801 (2012) [arXiv:1205.2699 [hep-ph]].
- [110] M. Baak, M. Goebel, J. Haller, A. Hoecker, D. Ludwig, K. Moenig, M. Schott and J. Stelzer, Eur. Phys. J. C 72, 2003 (2012) [arXiv:1107.0975 [hep-ph]].
- [111] B. Gorczyca and M. Krawczyk, arXiv:1112.5086 [hep-ph].
- [112] P. M. Ferreira, R. Santos, M. Sher and J. P. Silva, Phys. Rev. D 85, 035020 (2012) [arXiv:1201.0019 [hep-ph]].
- [113] W. Altmannshofer, S. Gori and G. D. Kribs, Phys. Rev. D 86, 115009 (2012) [arXiv:1210.2465 [hep-ph]].
- [114] F. Mahmoudi and T. Hurth, PoS ICHEP 2012, 324 (2013) [arXiv:1211.2796 [hep-ph]].
- [115] A. Azatov and J. Galloway, Int. J. Mod. Phys. A 28, 1330004 (2013) [arXiv:1212.1380].
- [116] C.-Y. Chen and S. Dawson, Phys. Rev. D 87, no. 5, 055016 (2013) [arXiv:1301.0309 [hep-ph]].
- [117] C. -W. Chiang and K. Yagyu, JHEP 1307, 160 (2013) [arXiv:1303.0168 [hep-ph]].
- [118] J. Chang, K. Cheung, C. -T. Lu and T. -C. Yuan, Phys. Rev. D 87, 093005 (2013) [arXiv:1303.6335 [hep-ph]].
- [119] A. Barroso, P. M. Ferreira, R. Santos, M. Sher and J. P. Silva, arXiv:1304.5225 [hep-ph].
- [120] C. -Y. Chen, S. Dawson and M. Sher, Phys. Rev. D 88, 015018 (2013) [arXiv:1305.1624 [hep-ph]].
- [121] N. Craig, J. Galloway and S. Thomas, arXiv:1305.2424 [hep-ph].

- [122] V. Barger, L. L. Everett, H. E. Logan and G. Shaughnessy, arXiv:1307.3676 [hep-ph].
- [123] F. Mahmoudi and O. Stal, Phys. Rev. D 81, 035016 (2010) [arXiv:0907.1791 [hep-ph]].
- [124] E. Barberio et al. [Heavy Flavor Averaging Group Collaboration], arXiv:0808.1297 [hep-ex].
- [125] J. P. Lees et al. [BaBar Collaboration], Phys. Rev. D 88, 072012 (2013) [arXiv:1303.0571 [hep-ex]].
- [126] R. Barate et al. [LEP Working Group for Higgs boson searches and ALEPH and DELPHI and L3 and OPAL Collaborations], Phys. Lett. B 565, 61 (2003) [hep-ex/0306033].
- [127] G. Abbiendi et al. [OPAL Collaboration], Eur. Phys. J. C 40 (2005) 317 [hep-ex/0408097].
- [128] J. Abdallah et al. [DELPHI Collaboration], Eur. Phys. J. C 38, 1 (2004) [hep-ex/0410017].
- [129] G. Abbiendi et al. [ALEPH and DELPHI and L3 and OPAL and LEP Collaborations], Eur. Phys. J. C 73, 2463 (2013) [arXiv:1301.6065 [hep-ex]].
- [130] B. Aubert et al. [BaBar Collaboration], Phys. Rev. Lett. 103, 081803 (2009) [arXiv:0905.4539 [hep-ex]]
- [131] S. Kanemura, T. Kubota and E. Takasugi, Phys. Lett. B **313**, 155 (1993) [hep-ph/9303263].
- [132] A. G. Akeroyd, A. Arhrib and E.-M. Naimi, Phys. Lett. B **490**, 119 (2000) [hep-ph/0006035].
- [133] M. Misiak, H. M. Asatrian, K. Bieri, M. Czakon, A. Czarnecki, T. Ewerth, A. Ferroglia and P. Gambino *et al.*, Phys. Rev. Lett. **98**, 022002 (2007) [hep-ph/0609232];
- [134] J. P. Lees et al. [BaBar Collaboration], Phys. Rev. Lett. 109, 191801 (2012) [arXiv:1207.2690 [hep-ex]].
- [135] M. Bona et al. [UTfit Collaboration], Phys. Lett. B 687, 61 (2010) [arXiv:0908.3470 [hep-ph]].
- [136] J. Beringer et al. [Particle Data Group Collaboration], Phys. Rev. D 86, 010001 (2012).
- [137] LHC Higgs Cross Section Working Group, twiki.cern.ch/twiki/bin/view/LHCPhysics/ HiggsEuropeanStrategy2012
- [138] U. Baur, R. Brock, J. Parsons, M. Albrow, D. Denisov, T. Han, A. Kotwal and F. Olness *et al.*, eConf C **010630**, E4001 (2001) [hep-ph/0201227].
- [139] K. Seidel, F. Simon, and M. Tesar, CERN LCD-Note-2012-013 (2012).
- [140] R. N. Mohapatra and P. B. Pal, Massive neutrinos in physics and astrophysics. Second edition, World Sci. Lect. Notes Phys. 60, 1 (1998) [World Sci. Lect. Notes Phys. 72, 1 (2004)].
- [141] J. Gluza, On teraelectronvolt Majorana neutrinos, Acta Phys. Polon. B 33, 1735 (2002) [hep-ph/0201002].
- [142] M. Fukugita and T. Yanagida, Physics of neutrinos and applications to astrophysics, Berlin, Germany: Springer (2003) 593 p

- [143] V. Barger, D. Marfatia and K. Whisnant, Progress in the physics of massive neutrinos, Int. J. Mod. Phys. E 12, 569 (2003) [hep-ph/0308123].
- [144] B. Kayser, p. 145 in S. Eidelman et al. [Particle Data Group Collaboration], Review of particle physics. Particle Data Group, Phys. Lett. B 592, 1 (2004).
- [145] R. N. Mohapatra and A. Y. Smirnov, Neutrino Mass and New Physics, Ann. Rev. Nucl. Part. Sci. 56, 569 (2006) [hep-ph/0603118].
- [146] A. Strumia and F. Vissani, Neutrino masses and mixings and..., hep-ph/0606054.
- [147] M. C. Gonzalez-Garcia and M. Maltoni, Phys. Rept. 460, 1 (2008) [arXiv:0704.1800 [hep-ph]].
- [148] P. Minkowski, $\mu \to e$ gamma at a Rate of One Out of 1-Billion Muon Decays?, Phys. Lett. B 67, 421 (1977).
- [149] T. Yanagida, Horizontal Symmetry And Masses Of Neutrinos, Conf. Proc. C 7902131, 95 (1979).
- [150] P. Van Nieuwenhuizen and D. Z. Freedman, Supergravity. Proceedings, Workshop At Stony Brook, 27-29 September 1979, Amsterdam, Netherlands: North-holland (1979) 341p
- [151] P. Ramond, The Family Group in Grand Unified Theories, hep-ph/9809459.
- [152] S. L. Glashow, The Future of Elementary Particle Physics, NATO Sci. Ser. B 59, 687 (1980).
- [153] R. N. Mohapatra and G. Senjanovic, Neutrino Mass and Spontaneous Parity Violation, Phys. Rev. Lett. 44, 912 (1980).
- [154] M. Gell-Mann, P. Ramond and R. Slansky, Complex Spinors and Unified Theories, Conf. Proc. C 790927, 315 (1979) [arXiv:1306.4669 [hep-th]].
- [155] J. Schechter and J. W. F. Valle, Neutrino Masses in SU(2) x U(1) Theories, Phys. Rev. D 22, 2227 (1980).
- [156] R. E. Shrock, General Theory of Weak Leptonic and Semileptonic Decays. 1. Leptonic Pseudoscalar Meson Decays, with Associated Tests For, and Bounds on, Neutrino Masses and Lepton Mixing, Phys. Rev. D 24, 1232 (1981).
- [157] J. Schechter and J. W. F. Valle, Neutrino Decay and Spontaneous Violation of Lepton Number, Phys. Rev. D 25, 774 (1982).
- [158] N. Arkani-Hamed, L. J. Hall, H. Murayama, D. Tucker-Smith and N. Weiner, Small neutrino masses from supersymmetry breaking, Phys. Rev. D 64, 115011 (2001) [hep-ph/0006312].
- [159] F. Borzumati and Y. Nomura, Low scale seesaw mechanisms for light neutrinos, Phys. Rev. D 64, 053005 (2001) [hep-ph/0007018].
- [160] A. de Gouvea, See-saw energy scale and the LSND anomaly, Phys. Rev. D 72, 033005 (2005) [hep-ph/0501039].

- [161] A. de Gouvea, J. Jenkins and N. Vasudevan, Neutrino Phenomenology of Very Low-Energy Seesaws, Phys. Rev. D 75, 013003 (2007) [hep-ph/0608147].
- [162] W.-Y. Keung and G. Senjanovic, Majorana Neutrinos and the Production of the Right-handed Charged Gauge Boson, Phys. Rev. Lett. 50, 1427 (1983).
- [163] D. A. Dicus, D. D. Karatas and P. Roy, Lepton nonconservation at supercollider energies, Phys. Rev. D 44, 2033 (1991).
- [164] A. Pilaftsis, Radiatively induced neutrino masses and large Higgs neutrino couplings in the standard model with Majorana fields, Z. Phys. C 55, 275 (1992) [hep-ph/9901206].
- [165] A. Datta, M. Guchait and A. Pilaftsis, Probing lepton number violation via majorana neutrinos at hadron supercolliders, Phys. Rev. D 50, 3195 (1994) [hep-ph/9311257].
- [166] T. Han and B. Zhang, Signatures for Majorana neutrinos at hadron colliders, Phys. Rev. Lett. 97, 171804 (2006) [hep-ph/0604064].
- [167] F. del Aguila, J. A. Aguilar-Saavedra and R. Pittau, *Heavy neutrino signals at large hadron colliders*, JHEP 0710, 047 (2007) [hep-ph/0703261].
- [168] A. Atre, T. Han, S. Pascoli and B. Zhang, The Search for Heavy Majorana Neutrinos, JHEP 0905, 030 (2009) [arXiv:0901.3589 [hep-ph]].
- [169] W. Chao, Z. -g. Si, Y. -j. Zheng and S. Zhou, Testing the Realistic Seesaw Model with Two Heavy Majorana Neutrinos at the CERN Large Hadron Collider, Phys. Lett. B 683, 26 (2010) [arXiv:0907.0935 [hep-ph]].
- [170] J. A. Aguilar-Saavedra, F. Deppisch, O. Kittel and J. W. F. Valle, *Flavour in heavy neutrino searches at the LHC*, Phys. Rev. D 85, 091301 (2012) [arXiv:1203.5998 [hep-ph]].
- [171] S. P. Das, F. F. Deppisch, O. Kittel and J. W. F. Valle, *Heavy Neutrinos and Lepton Flavour Violation in Left-Right Symmetric Models at the LHC*, Phys. Rev. D 86, 055006 (2012) [arXiv:1206.0256 [hep-ph]].
- [172] J. A. Aguilar-Saavedra and F. R. Joaquim, Phys. Rev. D 86, 073005 (2012) [arXiv:1207.4193 [hep-ph]].
- [173] C. -Y. Chen, P. S. B. Dev and R. N. Mohapatra, Probing Heavy-Light Neutrino Mixing in Left-Right Seesaw Models at the LHC, Phys. Rev. D 88, 033014 (2013) [arXiv:1306.2342 [hep-ph]].
- [174] P. S. B. Dev, A. Pilaftsis and U. -k. Yang, New Production Mechanism for Heavy Neutrinos at the LHC, arXiv:1308.2209 [hep-ph].
- [175] H. Davoudiasl and I. M. Lewis, Right-Handed Neutrinos as the Origin of the Electroweak Scale, arXiv:1404.6260 [hep-ph].
- [176] S. Chatrchyan *et al.* [CMS Collaboration], Search for heavy Majorana neutrinos in $\mu^+\mu^+[\mu^-\mu^-]$ and $e^+e^+[e^-e^-]$ events in pp collisions at $\sqrt{s} = 7$ TeV, Phys. Lett. B **717**, 109 (2012) [arXiv:1207.6079 [hep-ex]].

- [177] [ATLAS Collaboration], Search for Majorana neutrino production in pp collisions at sqrt(s)=7TeV in dimuon final states with the ATLAS detector, ATLAS-CONF-2012-139.
- [178] R. Aaij et al. [LHCb Collaboration], Searches for Majorana neutrinos in B⁻ decays, Phys. Rev. D 85, 112004 (2012) [arXiv:1201.5600 [hep-ex]].
- [179] G. Belanger, F. Boudjema, D. London and H. Nadeau, *Inverse neutrinoless double beta decay revisited*, Phys. Rev. D 53, 6292 (1996) [hep-ph/9508317].
- [180] P. Benes, A. Faessler, F. Simkovic and S. Kovalenko, Sterile neutrinos in neutrinoless double beta decay, Phys. Rev. D 71, 077901 (2005) [hep-ph/0501295].
- [181] E. Nardi, E. Roulet and D. Tommasini, *Limits on neutrino mixing with new heavy particles*, Phys. Lett. B **327**, 319 (1994) [hep-ph/9402224].
- [182] E. Nardi, E. Roulet and D. Tommasini, New neutral gauge bosons and new heavy fermions in the light of the new LEP data, Phys. Lett. B 344, 225 (1995) [hep-ph/9409310].
- [183] F. del Aguila, J. de Blas and M. Perez-Victoria, Effects of new leptons in Electroweak Precision Data, Phys. Rev. D 78, 013010 (2008) [arXiv:0803.4008 [hep-ph]].
- [184] S. Antusch and O. Fischer, Non-unitarity of the leptonic mixing matrix: Present bounds and future sensitivities, JHEP 1410, 94 (2014) [arXiv:1407.6607 [hep-ph]].
- [185] R. Hamberg, W. L. van Neerven and T. Matsuura, A Complete calculation of the order α s² correction to the Drell-Yan K factor, Nucl. Phys. B **359**, 343 (1991) [Erratum-ibid. B **644**, 403 (2002)].
- [186] B. A. Kniehl, Elastic e p scattering and the Weizsacker-Williams approximation, Phys. Lett. B 254, 267 (1991).
- [187] M. M. Block, E. M. Gregores, F. Halzen and G. Pancheri, Photon proton and photonphoton scattering from nucleon-nucleon forward amplitudes, Phys. Rev. D 60, 054024 (1999) [hep-ph/9809403].
- [188] M. Gluck, C. Pisano and E. Reya, The Polarized and unpolarized photon content of the nucleon, Phys. Lett. B 540, 75 (2002) [hep-ph/0206126].
- [189] B. E. Cox, A. De Roeck, V. A. Khoze, T. Pierzchala, M. G. Ryskin, I. Nasteva, W. J. Stirling and M. Tasevsky, *Detecting the standard model Higgs boson in the WW decay channel using* forward proton tagging at the LHC, Eur. Phys. J. C 45, 401 (2006) [hep-ph/0505240].
- [190] D. d'Enterria and G. G. da Silveira, Observing light-by-light scattering at the Large Hadron Collider, Phys. Rev. Lett. 111, 080405 (2013) [arXiv:1305.7142 [hep-ph]].
- [191] V. A. Khoze, A. D. Martin and M. G. Ryskin, Prospects for new physics observations in diffractive processes at the LHC and Tevatron, Eur. Phys. J. C 23, 311 (2002) [hepph/0111078].
- [192] T. Han, B. Mukhopadhyaya, Z. Si and K. Wang, Pair production of doubly-charged scalars: Neutrino mass constraints and signals at the LHC, Phys. Rev. D 76, 075013 (2007) [arXiv:0706.0441 [hep-ph]].
- [193] R. S. Gupta, Probing Quartic Neutral Gauge Boson Couplings using diffractive photon fusion at the LHC, Phys. Rev. D 85, 014006 (2012) [arXiv:1111.3354 [hep-ph]].
- [194] I. Sahin, Electromagnetic properties of the neutrinos in gamma-proton collision at the LHC, Phys. Rev. D 85, 033002 (2012) [arXiv:1201.4364 [hep-ph]].
- [195] I. Sahin and B. Sahin, Anomalous quartic $ZZ\gamma\gamma$ couplings in γp collision at the LHC, Phys. Rev. D 86, 115001 (2012) [arXiv:1211.3100 [hep-ph]].
- [196] P. Abreu et al. [DELPHI Collaboration], First evidence of hard scattering processes in single tagged gamma gamma collisions, Phys. Lett. B 342, 402 (1995).
- [197] S. Chatrchyan *et al.* [CMS Collaboration], Study of exclusive two-photon production of W^+W^- in pp collisions at $\sqrt{s} = 7$ TeV and constraints on anomalous quartic gauge couplings, JHEP **1307**, 116 (2013) [arXiv:1305.5596 [hep-ex]].
- [198] S. Aid et al. [H1 Collaboration], Elastic and inelastic photoproduction of J/ψ mesons at HERA, Nucl. Phys. B 472, 3 (1996) [hep-ex/9603005].
- [199] C. Adloff et al. [H1 Collaboration], Elastic photoproduction of J / psi and Upsilon mesons at HERA, Phys. Lett. B 483, 23 (2000) [hep-ex/0003020].
- [200] M. Drees and D. Zeppenfeld, Production of Supersymmetric Particles in Elastic ep Collisions, Phys. Rev. D 39, 2536 (1989).
- [201] K. Fabricius, I. Schmitt, G. Kramer and G. Schierholz, Z. Phys. C 11, 315 (1981).
- [202] G. Kramer and B. Lampe, Fortsch. Phys. **37**, 161 (1989).
- [203] H. Baer, J. Ohnemus and J. F. Owens, Phys. Rev. D 40, 2844 (1989).
- [204] B. W. Harris and J. F. Owens, Phys. Rev. D 65, 094032 (2002) [hep-ph/0102128].
- [205] H. L. Lai, M. Guzzi, J. Huston, Z. Li, P. M. Nadolsky, J. Pumplin and C.-P. Yuan, Phys. Rev. D 82, 074024 (2010) [arXiv:1007.2241 [hep-ph]].
- [206] C. -H. Lee, P. S. Bhupal Dev and R. N. Mohapatra, Natural TeV-scale left-right seesaw mechanism for neutrinos and experimental tests, Phys. Rev. D 88, no. 9, 093010 (2013) [arXiv:1309.0774 [hep-ph]].
- [207] T. Hahn, CUBA: A Library for multidimensional numerical integration, Comput. Phys. Commun. 168, 78 (2005) [hep-ph/0404043].
- [208] A. Alloul, N. D. Christensen, C. Degrande, C. Duhr and B. Fuks, *FeynRules 2.0 A complete toolbox for tree-level phenomenology*, [arXiv:1310.1921 [hep-ph]].
- [209] N. D. Christensen and C. Duhr, FeynRules Feynman rules made easy, Comput. Phys. Commun. 180, 1614 (2009) [arXiv:0806.4194 [hep-ph]].

- [210] R. Gavin, Y. Li, F. Petriello and S. Quackenbush, FEWZ 2.0: A code for hadronic Z production at next-to-next-to-leading order, Comput. Phys. Commun. 182, 2388 (2011) [arXiv:1011.3540 [hep-ph]].
- [211] R. Gavin, Y. Li, F. Petriello and S. Quackenbush, W Physics at the LHC with FEWZ 2.1, Comput. Phys. Commun. 184, 208 (2013) [arXiv:1201.5896 [hep-ph]].
- [212] M. Nemevsek, F. Nesti, G. Senjanovic and Y. Zhang, First Limits on Left-Right Symmetry Scale from LHC Data, Phys. Rev. D 83, 115014 (2011) [arXiv:1103.1627 [hep-ph]].
- [213] S. Chatrchyan *et al.* [CMS Collaboration], Search for leptonic decays of W ' bosons in pp collisions at $\sqrt{s} = 7$ TeV, JHEP **1208**, 023 (2012) [arXiv:1204.4764 [hep-ex]].
- [214] A. Avetisyan, J. M. Campbell, T. Cohen, N. Dhingra, J. Hirschauer, K. Howe, S. Malik and M. Narain et al., Methods and Results for Standard Model Event Generation at √s = 14 TeV, 33 TeV and 100 TeV Proton Colliders (A Snowmass Whitepaper), arXiv:1308.1636 [hep-ex].
- [215] A. De Rujula, J. R. Ellis, R. Petronzio, G. Preparata and W. Scott, CERN-TH-2778.
- [216] A. Das, P. S. Bhupal Dev and N. Okada, Direct bounds on electroweak scale pseudo-Dirac neutrinos from $\sqrt{s} = 8$ TeV LHC data, Phys. Lett. B **735**, 364 (2014) [arXiv:1405.0177 [hep-ph]].
- [217] G. Aad et al. [ATLAS Collaboration], Pg. 1760 Expected Performance of the ATLAS Experiment - Detector, Trigger and Physics, arXiv:0901.0512 [hep-ex].
- [218] S. Chatrchyan et al. [CMS Collaboration], Identification of b-quark jets with the CMS experiment, JINST 8, P04013 (2013) [arXiv:1211.4462 [hep-ex]].
- [219] J. M. Campbell and R. K. Ellis, $t\bar{t}W^{+-}$ production and decay at NLO, JHEP **1207**, 052 (2012) [arXiv:1204.5678 [hep-ph]].
- [220] G. Aad et al. [ATLAS Collaboration], Inclusive search for same-sign dilepton signatures in pp collisions at $\sqrt{s} = 7$ TeV with the ATLAS detector, JHEP **1110**, 107 (2011) [arXiv:1108.0366 [hep-ex]].
- [221] S. Chatrchyan et al. [CMS Collaboration], Search for new physics with same-sign isolated dilepton events with jets and missing transverse energy at the LHC, JHEP 1106, 077 (2011) [arXiv:1104.3168 [hep-ex]].
- [222] T. Binoth, G. Ossola, C. G. Papadopoulos and R. Pittau, NLO QCD corrections to tri-boson production, JHEP 0806, 082 (2008) [arXiv:0804.0350 [hep-ph]].
- [223] S. Weinberg, Phys. Rev. Lett. 43 1566 (1979).
- [224] R. Foot, H. Lew, X. G. He and G. C. Joshi, Z. Phys. C 44 441 (1989); E. Ma, Phys. Rev. Lett. 81 1171 (1998) [arXiv:hep-ph/9805219]; B. Bajc and G. Senjanović, JHEP 0708, 014 (2007) [arXiv:hep-ph/0612029]; P. Fileviez Pérez, Phys. Lett. B654, 189 (2007) [arXiv:hep-ph/0702287]; P. Fileviez Pérez, Phys. Rev. D76, 071701 (2007) [arXiv:0705.3589 [hep-ph]]; P. Fileviez Perez, [arXiv:0809.1202 [hep-ph]]; J. Garayoa, T. Schwetz, JHEP 0803, 009 (2008)

[arXiv:0712.1453 [hep-ph]]; A. G. Akeroyd, M. Aoki, H. Sugiyama, Phys. Rev. **D77**, 075010 (2008) [arXiv:0712.4019 [hep-ph]]; M. Nemevsek, G. Senjanovic, V. Tello, [arXiv:1211.2837].

- [225] R. N. Mohapatra, Phys. Rev. Lett. 56, 561 (1986); R. N. Mohapatra, J. W. F. Valle, Phys. Rev. D34, 1642 (1986); M. C. Gonzalez-Garcia, J. W. F. Valle, Phys. Lett. B216, 360 (1989).
- [226] P. Minkowski, Phys. Lett. B67, 421 (1977); T. Yanagida, in Proc. of the Workshop on Grand Unified Theory and Baryon Number of the Universe, KEK, Japan, 1979; M. Gell-Mann, P. Ramond and R. Slansky in Sanibel Symposium, February 1979, CALT-68-709 [retroprint arXiv:hep-ph/9809459], and in Supergravity, eds. D. Freedman et al. (North Holland, Amsterdam, 1979); S. L. Glashow in Quarks and Leptons, Cargese, eds. M. Levy et al. (Plenum, 1980, New York), p. 707; R. N. Mohapatra and G. Senjanovic, Phys. Rev. Lett. 44, 912 (1980).
- [227] J. C. Pati and A. Salam, Phys. Rev. D10, 275 (1974); R. N. Mohapatra and J. C. Pati, Phys. Rev. D11, 566, 2558 (1975); G. Senjanovic and R. N. Mohapatra, Phys. Rev. D12, 1502 (1975); R. E. Shrock, Nucl. Phys. B 206, 359 (1982). G. Ecker and W. Grimus, Nucl. Phys. B258, 328 (1985); Y. Zhang, H. An, X. Ji and R. N. Mohapatra, Phys. Rev. D76, 091301 (2007) [arXiv:0704.1662 [hep-ph]]; A. Maiezza, M. Nemevsek, F. Nesti and G. Senjanovic, Phys. Rev. D D82, 055022 (2010) [arXiv:1005.5160 [hep-ph]]; J. Chakrabortty, Phys. Lett. B690, 382 (2010) [arXiv:1005.1377 [hep-ph]]; D. Guadagnoli, R. N. Mohapatra, Phys. Lett. B694, 386 (2011) [arXiv:1008.1074 [hep-ph]].
- [228] J. A. Harvey, P. Ramond and D. B. Reiss, Nucl. Phys. B199, 223 (1982); S. Dimopoulos, L. J. Hall, and S. Raby, Phys. Rev. Lett. 68, 1984 (1992); L. J. Hall and S. Raby, Phys. Rev. D51, 6524 (1995) [arXiv:hep-ph/9501298]; S. Blanchet, P. S. B. Dev and R. N. Mohapatra, Phys. Rev. D 82, 115025 (2010) [arXiv:1010.1471 [hep-ph]]; P. S. B. Dev and R. N. Mohapatra, Phys. Rev. D 81, 013001 (2010) [arXiv:0910.3924 [hep-ph]].
- [229] I. Dorsner and P. Fileviez Pérez, Nucl. Phys. B723, 53 (2005) [arXiv:hep-ph/0504276]; see also: I. Dorsner, P. Fileviez Pérez and R. Gonzalez Felipe, Nucl. Phys. B747 (2006) 312 [arXiv:hep-ph/0512068]; P. Fileviez Pérez, AIP Conf. Proc. 903, 385 (2006) [arXiv:hep-ph/0606279]; I. Dorsner, P. F. Pérez and G. Rodrigo, Phys. Rev. D75, 125007 (2007) [arXiv:hep-ph/0607208].
- [230] C. S. Aulakh and R. N. Mohapatra, Phys. Lett. B119, 136 (1982); L. J. Hall and M. Suzuki, Nucl. Phys. B231, 419 (1984); G. G. Ross and J. W. F. Valle, Phys. Lett. B151, 375 (1985); J. Ellis, G. Gelmini, C. Jarlskog, G. G. Ross and J.W.F. Valle, Phys. Lett. B150, 142 (1985); S. Dawson, Nucl. Phys. B261, 297 (1985); M. Drees, S. Pakvasa, X. Tata and T. terVeldhuis, Phys. Rev. D57, R5335 (1998) [arXiv:hep-ph/9712392]; E. J. Chun, S. K. Kang, C. W. Kim and U.W. Lee, Nucl. Phys. B544, 89 (1999) [arXiv:hep-ph/9807327]; V. Barger, T. Han, S. Hesselbach and D.Marfatia, Phys. Lett. B538, 346 (2002) [arXiv:hepph/0108261]; for a recent review see R. Barbieri *et al.*, Phys. Rept. 420, 1 (2005) [arXiv:hepph/0406039]; V. Barger, P. Fileviez Perez and S. Spinner, Phys. Rev. Lett. 102 181802 (2009) [arXiv:0812.3661].
- [231] N. Arkani-Hamed, S. Dimopoulos, G. Dvali and J. March-Russell, Phys. Rev. D65, 024032 (2001) [arXiv:hep-ph/9811448]; Y. Grossman and M. Neubert, Phys. Lett. B474, 361

(2000) [arXiv:hep-ph/9912408]; K. R. Dienes and I. Sarcevic, Phys. Lett. B500, 133 (2001)
[arXiv:hep-ph/0008144]; S. J. Huber and Q. Shafi, Phys. Lett. B 544, 295 (2002) [arXiv:hep-ph/0205327]; M. C. Chen and H.B. Yu, Phys. Lett. B672, 253 (2009) [arXiv:0804.2503];
G. Perez and L. Randall, JHEP 01, 077 (2009) [arXiv:0805.4652].

- [232] M. -C. Chen, J. Huang, Mod. Phys. Lett. A26, 1147 (2011) [arXiv:1105.3188 [hep-ph]].
- [233] M. Schmaltz and C. Spethmann, JHEP 1107, 046 (2011) [arXiv:1011.5918 [hep-ph]];
 M. Nemevsek, F. Nesti, G. Senjanovic and Y. Zhang, Phys. Rev. D 83, 115014 (2011)
 [arXiv:1103.1627 [hep-ph]]; C. Grojean, E. Salvioni and R. Torre, JHEP 1107, 002 (2011)
 [arXiv:1103.2761 [hep-ph]]; C. -W. Chiang, N. D. Christensen, G. -J. Ding and T. Han, Phys.
 Rev. D 85, 015023 (2012) [arXiv:1107.5830 [hep-ph]]; C. -Y. Chen and P. S. B. Dev, Phys.
 Rev. D 85, 093018 (2012) [arXiv:1112.6419 [hep-ph]]; S. P. Das, F. F. Deppisch, O. Kittel
 and J. W. F. Valle, Phys. Rev. D 86, 055006 (2012) [arXiv:1206.0256 [hep-ph]].
- [234] S. Gopalakrishna, T. Han, I. Lewis, Z. -g. Si and Y. -F. Zhou, Phys. Rev. D 82, 115020 (2010) [arXiv:1008.3508 [hep-ph]].
- [235] J. C. Pati and A. Salam, Phys. Rev. D10, 275 (1974) [Erratum-ibid. D11, 703 (1975)];
 G. Senjanovic and R. N. Mohapatra, Phys. Rev. D12, 1502 (1975).
- [236] N. Arkani-Hamed, A. G. Cohen, E. Katz and A. E. Nelson, JHEP **0207**, 034 (2002) [arXiv:hep-ph/0206021]; D. E. Kaplan and M. Schmaltz, JHEP **0310**, 039 (2003) [arXiv:hepph/0302049]; T. Han, H. E. Logan, B. McElrath, and L. T. Wang, Phys. Rev. **D67**, 095004 (2003) [arXiv:hep-ph/0301040].
- [237] T. Appelquist, H. C. Cheng and B. A. Dobrescu, Phys. Rev. D64, 035002 (2001) [arXiv:hep-ph/0012100]; H. C. Cheng, K. T. Matchev and M. Schmaltz, Phys. Rev. D66, 056006 (2002) [arXiv:hep-ph/0205314].
- [238] C. Csaki, C. Grojean, L. Pilo and J. Terning, Phys. Rev. Lett. **92**, 101802 (2004) [hep-ph/0308038]; R. S. Chivukula, B. Coleppa, S. Di Chiara, E. H. Simmons, H. -J. He, M. Kurachi and M. Tanabashi, Phys. Rev. D **74**, 075011 (2006) [hep-ph/0607124]; H. -J. He, et al., Phys. Rev. D **78**, 031701 (2008) [arXiv:0708.2588 [hep-ph]].
- [239] K. Agashe, A. Delgado, M. J. May and R. Sundrum, JHEP 0308, 050 (2003) [hepph/0308036].
- [240] I. Dorsner and P. Fileviez Perez, JHEP 0706, 029 (2007) [hep-ph/0612216].
- [241] B. Bajc, M. Nemevsek and G. Senjanovic, Phys. Rev. D 76, 055011 (2007) [hep-ph/0703080].
- [242] A. de Gouvea, arXiv:0706.1732 [hep-ph].
- [243] C. Dohmen et al. [SINDRUM II. Collaboration], Phys. Lett. B317, 631 (1993); M. L. Brooks et al. [MEGA Collaboration], Phys. Rev. Lett. 83, 1521 (1999) [hep-ex/9905013]; T. Aaltonen et al. [CDF Collaboration], Phys. Rev. Lett. 103, 041801 (2009) [arXiv:0902.3276]; V. M. Abazov et al. [D0 Collaboration], Phys. Lett. B699, 145 (2011) [arXiv:1101.0806]; S. Chatrchyan et al. [CMS Collaboration], JHEP 1106, 077 (2011) [arXiv:1104.3168]; G. Aad et al. [ATLAS Collaboration], Phys. Lett. B705, 28 (2011), [arXiv:1108.1316].

- [244] S. Chatrchyan et al. [CMS Collaboration] CMS-PAS-EXO-12-017; S. Chatrchyan et al. [CMS Collaboration], CMS-EXO-11-024 [arXiv:1204.4764].
- [245] S. Chatrchyan et al. [CMS Collaboration] CMS PAS EXO-12-010.
- [246] G. Aad et al. [ATLAS Collaboration], Eur. Phys. J. C 72, 2056 (2012) [arXiv:1203.5420 [hep-ex]].
- [247] G. Cacciapaglia, C. Csaki, G. Marandella and A. Strumia, Phys. Rev. D 74, 033011 (2006) [hep-ph/0604111].
- [248] J. Donoghue, B. Holstein, Phys. Lett. 113B, 383 (1982); G. Beall, M. Bander, A. Soni, Phys. Rev. Lett. 48, 848 (1982)
- [249] L. Wolfenstein, Phys. Rev. **D29**, 2130 (1984)
- [250] F. I. Olness and M. E. Ebel, Phys. Rev. D 30, 1034 (1984).
- [251] P. Langacker and S. U. Sankar, Phys. Rev. D 40, 1569 (1989).
- [252] A. Denner, H. Eck, O. Hahn and J. Kublbeck, Nucl. Phys. B 387, 467 (1992). [doi:10.1016/0550-3213(92)90169-C].
- [253] G. L. Bayatian et al. [CMS Collaboration], J. Phys. G34, 995 (2007); G. Aad et al. [The AT-LAS Collaboration], arXiv:0901.0512 [hep-ex]; ATLAS Collaboration, Detector and Physics Performances Technical Design Report, Vol.II, ATLAS TDR 14, CERN/LHCC 99-14; CMS Physics TDR, Volume II: CERN-LHCC-2006-01 (2006); J. Phys. G: Nucl. Part. Phys. 34 (2006) 995-1579
- [254] T. M. P. Tait and C. -P. Yuan, Phys. Rev. D 63, 014018 (2000) [hep-ph/0007298].
- [255] P. Langacker, R. W. Robinett and J. L. Rosner, Phys. Rev. D 30, 1470 (1984); M. Dittmar, Phys. Rev. D 55, 161 (1997) [hep-ex/9606002]; M. Dittmar, A. -S. Nicollerat and A. Djouadi, Phys. Lett. B 583, 111 (2004) [hep-ph/0307020]; M. S. Carena, A. Daleo, B. A. Dobrescu and T. M. P. Tait, Phys. Rev. D 70, 093009 (2004) [hep-ph/0408098]; T. Han, H. E. Logan and L. -T. Wang, JHEP 0601, 099 (2006) [hep-ph/0506313]; F. Petriello and S. Quackenbush, Phys. Rev. D 77, 115004 (2008) [arXiv:0801.4389 [hep-ph]].